UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

	_			
		FORM 10-Q		
Mark One)	_			
QUARTERLY REPORT PURSUA	ANT TO SECTION 13 OF	R 15(d) OF THE SECURITIES EXCHANG	GE ACT OF 1934	
	For the	quarterly period ended September 30, 20	21	
		OR		
☐ TRANSITION REPORT PURSUA	ANT TO SECTION 13 OR	15(d) OF THE SECURITIES EXCHANG	GE ACT OF 1934	
		For the transition period from to Commission File Number: 001-40708		
		RENOVORX, INC.	ter)	
(State or other	elaware er jurisdiction of n or organization)		27-1448452 (I.R.S. Employer Identification No.)	
Los Alto	ino Real, Suite B1 s, California cipal executive offices)		94022 (Zip Code)	
	Registrant's tel	ephone number, including area code: (650	284-4433	
Securities registered pursuant to S	ection 12(b) of the Act:			
Title of each cla	ss	Trading Symbol(s)	Name of each exchange on which registered	
Common Stock, par value \$0.	.0001 per share	RNXT	Nasdaq Capital Market	
			13 or 15(d) of the Securities Exchange Act of 1934 during s been subject to such filing requirements for the past 90 c	
		d electronically every Interactive Data File re ch shorter period that the registrant was requi	equired to be submitted pursuant to Rule 405 of Regulation red to submit such files). Yes \boxtimes No \square	S-7
			accelerated filer, smaller reporting company, or an emergany," and "emerging growth company" in Rule 12b-2 of	
Large accelerated filer	1		Accelerated filer	
Non-accelerated filer	1		Smaller reporting company	\boxtimes
Emerging growth company	1			
If an emerging growth company inancial accounting standards provided p			ended transition period for complying with any new or rev	isec
Indicate by check mark whether	the registrant is a shell con	npany (as defined in Rule 12b-2 of the Excha	nge Act). Yes □No ⊠	

As of November 1, 2021, the registrant had 8,908,150 shares of common stock, \$0.0001 par value per share, outstanding.

PART I.	<u>FINANCIAL INFORMATION</u>	2
Item 1.	Condensed Interim Financial Statements (Unaudited)	2
	Condensed Balance Sheets as of September 30, 2021 and December 31, 2020 (Audited)	2
	Condensed Statements of Operations for the three and nine months ended September 30, 2021 and 2020	3
	Condensed Statements of Convertible Preferred Stock and Stockholders' Equity (Deficit) for the three and nine months ended September 30, 2021 and 2020	4
	Condensed Statements of Cash Flows for the nine months ended September 30, 2021 and 2020	6
	Notes to the Unaudited Condensed Interim Financial Statements	7
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	30
Item 4.	Controls and Procedures	31
PART II.	OTHER INFORMATION	32
Item 1.	<u>Legal Proceedings</u>	32
Item 1A.	Risk Factors	32
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	59
Item 3.	Defaults Upon Senior Securities	59
Item 4.	Mine Safety Disclosures	59
Item 5.	Other Information	59
Item 6.	Exhibits	62
Signatures		63

Forward-Looking Statements

This Quarterly Report on Form 10-Q, or Quarterly Report, contains forward-looking statements that are based on our beliefs and assumptions and on information currently available to us. All statements other than statements of historical facts contained in this Quarterly Report are forward-looking statements. These forward-looking statements include, but are not limited to, information regarding our expectations on the timing of clinical study initiation and results and the timing and success of future development of our products; our possible or assumed future results of operations and expenses, including research and development expenses; business strategies and plans; trends; market sizing; competitive position; industry environment; potential growth opportunities; reliance on third parties, including third-party manufacturers; the timing of product revenues, if any; financing needs; liquidity, cash flows and operating losses; the effects of the COVID-19 pandemic, including on our preclinical and clinical studies; and impact of the Affordable Care Act and other legislation, among other things. In some cases, investors can identify forward-looking statements by terms such as "anticipates," "believes," "could," "estimates," "expects," "intends," "may," "plans," "potential," "predicts," "seeks," "should," "will," "would" or similar expressions and the negatives of those terms.

We have based these forward-looking statements largely on our current expectations and projections about our business, the industry in which we operate and financial trends that we believe may affect our business, financial condition, results of operations and prospects, and these forward-looking statements are not guarantees of future performance or development. These forward-looking statements speak only as of the date of this Quarterly Report and are subject to a number of risks, uncertainties and assumptions described in the section titled "Risk Factors" and elsewhere in this Quarterly Report. Because forward-looking statements are inherently subject to risks and uncertainties, some of which cannot be predicted or quantified, investors should not rely on these forward-looking statements as predictions of future events. The events and circumstances reflected in our forward-looking statements may not be achieved or occur and actual results could differ materially from those projected in the forward-looking statements. Except as required by applicable law, we do not plan to publicly update or revise any forward-looking statements contained herein, whether as a result of any new information, future events or otherwise.

In addition, statements that "we believe" and similar statements reflect our beliefs and opinions on the relevant subject. These statements are based upon information available to us as of the date of this Quarterly Report, and while we believe such information forms a reasonable basis for such statements, such information may be limited or incomplete, and our statements should not be read to indicate that we have conducted an exhaustive inquiry into, or review of, all potentially available relevant information. These statements are inherently uncertain and investors are cautioned not to unduly rely upon these statements.

ii

PART I—FINANCIAL INFORMATION

Item 1. Condensed Interim Financial Statements (Unaudited)

RenovoRx, Inc. Condensed Balance Sheets (Unaudited)

(in thousands, except share and per share amounts)

	-	ember 30, 2021	December 31, 2020 (audited)
Assets			
Current assets:			
Cash and cash equivalents	\$	17,725	\$ 1,795
Prepaid expenses and other current assets		439	115
Total current assets		18,164	1,910
Leasehold improvements, net		10	-
Other assets		4	4
Total assets	\$	18,178	\$ 1,914
Liabilities, convertible preferred stock and stockholders' equity (deficit)			
Current liabilities:			
Accounts payable	\$	133	\$ 162
Accrued expenses		492	311
Promissory note, current		-	117
Convertible note		-	2,650
Derivative liability		<u>-</u>	856
Total current liabilities		625	4,096

Promissory note, net of current portion	-	23
Total liabilities	\$ 625	4,119
Commitments and contingencies (Note 8)	 	
Convertible preferred stock, \$0.0001 par value; 15,000,000 and 22,360,455 shares authorized; 0 and		
3,535,469 shares issued and outstanding at September 30, 2021 and December 31, 2020, respectively		
(aggregate liquidation preference of \$0 at September 30, 2021 and \$12,782 as of December 31, 2020)	-	12,451
Stockholders' equity (deficit):		
Common stock, \$0.0001 par value, 250,000,000 and 42,000,000 shares authorized; 8,908,150 and		
2,291,497 shares issued and outstanding as of September 30, 2021 and December 31, 2020, respectively	1	1
Additional paid-in capital	36,481	303
Accumulated deficit	(18,929)	(14,960)
Total stockholders' equity (deficit)	17,553	(14,656)
Total liabilities, convertible preferred stock and stockholders' equity (deficit)	\$ 18,178	\$ 1,914

The accompanying notes are an integral part of these condensed interim financial statements.

2

RenovoRx, Inc. Condensed Statements of Operations (Unaudited)

(in thousands, except share and per share amounts)

	Three Months Ended September 30,			Nine Months Ended September 30,			ed	
		2021		2020		2021		2020
Operating expenses:								
Research and development	\$	767	\$	727	\$	1,938	\$	1,934
General and administrative		628		183		1,377		631
Total operating expenses		1,395		910		3,315		2,565
Loss from operations		(1,395)		(910)		(3,315)		(2,565)
Interest expense, net	<u> </u>	(208)		(186)		(835)		(355)
Other income, net		170		-		119		-
Loss (gain) on loan extinguishment		(78)		-		62		-
Total other expenses, net		(116)		(186)		(654)		(355)
Net loss	\$	(1,511)	\$	(1,096)	\$	(3,969)	\$	(2,920)
Net loss per share, basic and diluted	\$	(0.27)	\$	(0.48)	\$	(1.09)	\$	(1.31)
Weighted-average shares of common stock outstanding, basic and diluted		5,620,135		2,263,589		3,640,988		2,233,645

The accompanying notes are an integral part of these condensed interim financial statements.

3

RenovoRx, Inc. Condensed Statements of Convertible Preferred Stock and Stockholders' Equity (Deficit) (Unaudited)

(in thousands, except share amounts)

	Convertible Pro		l Stock Amount		Commo	n Stock Amo	unt	P	ditional aid-In apital		cumulated Deficit	 Total ockholders' nity (Deficit)
Balance—December 31, 2020 (audited)	3,535,469	s	12,451		2,233,139	\$	1	s	303	\$	(14,960)	\$ (14,656)
Issuance of common stock upon exercise of stock options	-		-		50,058		_		34		-	34
Stock-based compensation expense Net loss			-		-		-		8		- (1.140)	8
Net loss Balance—March 31, 2021	3,535,469	\$	12,451		2,283,197		1	\$	345	\$	(1,148)	\$ (1,148)
Issuance of common stock upon exercise of stock options	· ·		_		8,300		_		5		-	5
Stock-based compensation expense	-		-		-		_		7		-	7
Net loss		_					_	_		_	(1,310)	(1,310)
Balance—June 30, 2021 Conversion of convertible preferred stock to common stock upon initial public offering	3,535,469	\$	12,451		2,291,497		1	\$	357	\$	(17,418)	\$ (17,060)
Conversion of convertible notes and accrued interest to units upon	(3,535,469)		(12,451)	3,535,469		-		12,451		-	12,451
initial public offering	-		-		708,820		-		5,279		-	5,279

Reclassification of derivative liability upon							
conversion of convertible							
notes	_	-	_	-	1,101	_	1,101
Proceeds from initial public					,		, .
offering, net of							
underwriters' commissions,							
discounts and issuance							
costs of \$2,090	-	-	1,850,000	-	14,561	-	14,561
Issuance of common stock							
upon exercise of warrants							
issued upon initial public			247.700		2.675		2.675
offering	-	-	247,700	-	2,675	-	2,675
Issuance of common stock upon exercise of stock							
options			274,574		51	_	51
Reverse stock split			2/7,5/7		51		31
adjustment	_	_	90	_	_	_	_
Stock-based compensation							
expense	-	-	-	-	6	-	6
Net loss		 _				(1,511)	(1,511)
Balance—September 30, 2021		\$ 	8,908,150	\$ 1	\$ 36,481	\$ (18,929)	\$ 17,553

The accompanying notes are an integral part of these condensed interim financial statements.

4

RenovoRx, Inc. Condensed Statements of Convertible Preferred Stock and Stockholders' Deficit (Unaudited)

(in thousands, except share amounts)

	Convertible Pre	ferred Stock Amount	Commo Shares	n Stock Amount	Additional Paid-In Capital	Accumulated Deficit	Total Stockholders'
Balance—December 31, 2019	3,508,631	\$ 12,391	2,177,187	\$ -	\$ 235	\$ (11,162)	\$ (10,926)
Issuance of restricted stock award to Nonemployee for						, ,	` ` `
service	-	-	24,478	-	17	-	17
Issuance of common stock							
upon exercise of stock options	-	-	16,666	-	11	-	11
Issuance of Series A-1							
convertible preferred stock							
upon exercise of warrant	26,838	25	-	-	-	-	-
Warrant liability transferred to							
mezzanine equity							
upon exercise of warrant	-	35	-	-	-	-	-
Stock-based compensation							
expense	-	-	-	-	10	-	10
Net loss						(1,051)	(1,051)
Balance—March 31, 2020	3,535,469	12,451	2,218,331	1	273	(12,213)	(11,939)
Stock-based compensation expense	-	-	-	-	8	-	8
Net loss						(773)	(773)
Balance—June 30, 2020	3,535,469	12,451	2,218,331	1	281	(12,986)	(12,704)
Stock-based compensation expense	-	-	-	-	7	-	7
Net loss	-	-	-	-	-	(1,096)	(1,096)
Balance—September 30, 2020	3,535,469	12,451	2,218,331	1	288	(14,082)	(13,793)

The accompanying notes are an integral part of these condensed interim financial statements.

5

RenovoRx, Inc. Condensed Statements of Cash Flows (Unaudited) (in thousands)

Nine Months Ended September 30

	 September 30,					
	 2021	2020				
Operating activities						
Net loss	\$ (3,969) \$	(2,920)				
Adjustments to reconcile net loss to net cash used in operating activities:						
Stock-based compensation expense	21	25				
Amortization on leasehold improvement	5	-				
Loss on change in fair value of derivative liability	(118)	_				
Gain on loan extinguishment from PPP loan	(140)	-				
Loss on loan extinguishment from convertible notes	78	_				
Amortization of debt discount	697	341				
Changes in operating assets and liabilities:						
Prepaid expenses and other current assets	(324)	26				
Accounts payable	(29)	(287)				
Accrued expenses	 421	257				
Net cash used in operating activities	(3,358)	(2,558)				
Investing activities						

Expenditures for leasehold improvements	(15)	-
Net cash used in investing activities	(15)	-
Financing activities		
Net proceeds from issuance of common stock upon initial public offering	14,561	-
Proceeds from exercise of warrants	2,675	-
Proceeds from convertible notes	1,977	2,570
Proceeds from promissory note	-	140
Proceeds from exercise of Series A-1 warrant	-	25
Proceeds from exercise of common stock options	90	11
Net cash provided by financing activities	19,303	2,746
Increase in cash and cash equivalents	 15,930	188
Cash, cash equivalents, beginning of period	1,795	2,124
Cash, cash equivalents, end of period	\$ 17,725	\$ 2,312
Supplemental disclosure of non-cash investing and financing activities:		
Derivative liability	\$ 738	\$ 743
Conversion of convertible preferred stock upon initial public offering	\$ 12,451	\$ -
Conversion of convertible notes upon initial public offering	\$ 5,279	\$ -

The accompanying notes are an integral part of these condensed interim financial statements.

6

RenovoRx, Inc. Notes to the Unaudited Condensed Interim Financial Statements

1. Business and Principal Activities

Description of Business

RenovoRx, Inc. (the "Company") was incorporated in the state of Delaware in December 2012 and operates from its headquarters in Los Altos, California. The Company is a clinical-stage biopharmaceutical company focused on developing therapies for the local treatment of solid tumors and conducting a phase 3 pancreatic cancer clinical trial for its lead product candidate RenovoGemTM. The Company's therapy platform, RenovoRx Trans-Arterial Micro-Perfusion, or RenovoTAMPTM utilizes approved chemotherapeutics with validated mechanisms of action and well-established safety and side effect profiles, with the goal of increasing their efficacy, improving their safety, and widening their therapeutic window.

Initial Public Offering

On August 25, 2021, the Company's Registration Statement on Form S-1 (File No. 333-258071) relating to its initial public offering ("IPO") of units of securities, or units, was declared effective by the U.S. Securities and Exchange Commission, (or "SEC"), and its shares of common stock began trading on the Nasdaq Capital Market on August 26, 2021. The transaction formally closed on August 30, 2021. In connection with the IPO, the Company issued and sold an aggregate of 1,850,000 units at a price of \$9.00 per unit. Each unit consisted of (a) one share of common stock and (b) one warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share, which is exercisable for a period of five years after the issuance date. The Company also granted the underwriters an over-allotment option, exercisable for 45 days after August 25, 2021, to purchase any combination of up to 277,500 shares of its common stock and/or common stock warrants to purchase 277,500 shares of common stock with an exercise price of \$10.80 per share. The underwriters exercised their over-allotment option to purchase 277,500 common stock warrants on August 30, 2021. In connection with the IPO, the underwriters were issued a five-year warrant, exercisable on or after February 25, 2022, to purchase up to 198,875 shares of the Company's common stock at an exercise price of \$10.80.

The Company received aggregate gross proceeds of \$16.7 million from the IPO, paid underwriting discounts and commissions of \$1.3 million and incurred other expenses of \$0.8 million. As a result, the net offering proceeds to the Company, after deducting underwriting discounts and commissions and other offering expenses, were \$14.6 million. Immediately prior to the closing of the IPO, all shares of convertible preferred stock then outstanding were converted into 3,535,469 shares of common stock after giving effect to the reverse stock split. In addition, all of the outstanding Convertible Notes, representing principal and accrued but unpaid interest of \$5.3 million, converted into an aggregate of 708,820 units. Each unit consisted of (a) one shareof common stock and (b) one warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share, which is exercisable for a period of five years after the issuance date. The 2020 Convertible Notes converted at a 20% discount to the IPO price and the 2021 Convertible Notes converted at a 12.5% discount to the IPO price, see Note 5, Convertible Notes.

Reverse Stock Split

The Company filed a certificate of amendment to its Fifth Amended and Restated Certificate of Incorporation to effect a 1-for-5 reverse stock split of its issued and outstanding preferred stock and common stock, which became effective on August 5, 2021. The number of authorized shares and the par values of the common stock and convertible preferred stock were not adjusted as a result of the reverse stock split. Adjustments corresponding to the reverse stock split were made to the ratio at which the Company's convertible preferred stock converted into the Company's common stock. Accordingly, all share and per share amounts related to the common stock, stock options, warrants and restricted stock awards for all periods presented in the accompanying financial statements and notes thereto have been retroactively adjusted, where applicable, to reflect the effect of the reverse stock split.

Liquidity and Capital Resources

The Company raised \$14.6 million in net proceeds from its IPO in August 2021 and from the Company's inception through September 30, 2021, it raised an aggregate of \$5.0 million in net cash proceeds from the sale and issuance of convertible preferred stock, convertible notes, issuance of common stock in connection with its IPO and the exercise of warrants. The Company had cash and cash equivalents of \$17.7 million as of September 30, 2021.

The Company has incurred significant losses and negative cash flows from operations since its inception. For the nine months ending September 30, 2021, the Company reported a net loss of \$4.0 million and an accumulated deficit of \$18.9 million and does not expect to generate positive cash flows from operations in the foreseeable future. The Company expects to incur significant and increasing losses until regulatory approval is granted for its first product candidate, RenovoGemTM. Regulatory approval is not guaranteed and may never be obtained. The Company may seek to raise additional capital through debt financings, private or public equity financings, license agreements, collaborative agreements or other arrangements with other companies, or other sources of financing. There can be no assurance that such financing will be available or will be at terms acceptable to the Company. The inability to raise capital as and when needed would have a negative impact on the Company's financial condition and its ability to pursue its business strategy. The Company will need to generate significant revenue to achieve profitability, and it may never do so.

The Company has reviewed the relevant conditions and events surrounding its ability to continue as a going concern including among others: historical losses, projected future results, including the effects of the novel coronavirus ("COVID-19") pandemic, cash requirements for the upcoming year, funding capacity, net working capital, total stockholders' deficit and future access to capital. Based upon the Company's current operating plan, management believes that its existing cash and cash equivalents as of September 30, 2021 will be sufficient to allow the Company to fund operating, investing and financing cash flow needs for at least twelve months from the date of issuance of these interim condensed financial statements. The accompanying condensed interim financial statements have been prepared assuming that the Company will continue as a going concern, which contemplates the realization of assets and the settlement of liabilities and commitments in the normal course of business. The accompanying condensed interim financial statements do not reflect any adjustments relating to the recoverability and reclassifications of assets and liabilities that might be necessary if the Company is unable to continue as a going concern.

2. Summary of Significant Accounting Policies

Basis of Presentation and Unaudited Condensed Interim Financial Information

The accompanying unaudited interim financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP") and applicable rules and regulations of the SEC for interim reporting. As permitted under those rules and regulations, certain footnotes or other financial information normally included in financial statements prepared in accordance with GAAP have been condensed or omitted. The condensed interim financial statements have been prepared on the same basis as the annual financial statements and, in the opinion of management, reflect all adjustments, which include only normal, recurring adjustments that are necessary to present fairly the Company's results for the interim periods presented. The condensed balance sheet as of December 31, 2020, is derived from the Company's audited financial statements. The results of operations for the three and nine months ended September 30, 2021, are not necessarily indicative of the results to be expected for the year ending December 31, 2021, or for any other future annual or interim period.

The accompanying unaudited condensed interim financial statements should be read in conjunction with the audited financial statements and the related notes thereto for the year ended December 31, 2020, which are included in the Company's prospectus related to the Company's IPO, filed with the SEC on August 25, 2021, pursuant to Rule 424(b) under the Securities Act of 1933.

Use of Estimates

The preparation of unaudited condensed interim financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the unaudited condensed interim financial statements and accompanying notes. The Company bases its estimates on historical experience and market-specific or other relevant assumptions that it believes are reasonable under the circumstances. Assets and liabilities reported in the Company's condensed balance sheets and expenses and income reported are affected by estimates and assumptions, which are used for, but are not limited to, determining the fair value of assets and liabilities, including the accrual of certain liabilities, the valuation of financial instruments, the fair value of the Company's common stock, income tax uncertainties, and measurement of stock-based compensation expense. Actual results could differ from such estimates or assumptions.

Concentration of Credit Risk and Other Risks and Uncertainties

Financial instruments that potentially subject the Company to significant concentrations of credit risk consist primarily of cash and cash equivalents. The Company maintains bank deposits in federally insured financial institutions and these deposits may exceed federally insured limits. The Company is exposed to credit risk in the event of default by the financial institutions holding its cash and cash equivalents to the extent recorded in the condensed balance sheets. The Company has not experienced any losses on its deposits of cash and cash equivalents. The Company is subject to a number of risks similar to other early-stage biopharmaceutical companies, including, but not limited to, the need to obtain adequate additional funding, possible failure of current or future preclinical studies or clinical trials, its reliance on third parties to conduct its clinical trials, the need to obtain regulatory and marketing approvals for its product candidates, competitors developing new technological innovations, the need to successfully commercialize and gain market acceptance of the Company's product candidates, protection of its proprietary technology, and the need to secure and maintain adequate manufacturing arrangements with third parties.

The Company relied, and expects to rely, on a small number of third-party manufacturers to manufacture and supply its RenovoCath devices and its product candidates for clinical trials. These activities could be adversely affected by a significant interruption in supply of these items. If the Company does not successfully commercialize or partner any of its product candidates, it will be unable to generate product revenue or achieve profitability.

8

Deferred Offering Costs

The Company incurred offering costs consisting of legal, accounting and other fees and costs directly attributable to the Company's IPO. For the three and nine months ended September 30, 2021, the Company charged \$774,000 of deferred offering costs to additional paid-in capital upon completion of the IPO in August 2021. There wereno deferred offering costs at September 30, 2021 and December 31, 2020.

Operating Segment

The Company operates and manages its business as one reportable and operating segment, which is the development of a platform technology to deliver de-risked small molecules for localized treatment of solid cancer tumors. The Company's chief executive officer, who is the chief operating decision maker, reviews financial information on an aggregate basis for allocating resources and evaluating financial performance.

Cash and Cash Equivalents

The Company considers all highly liquid investments purchased with original maturities of 90 days or less from the purchase date to be cash equivalents. Cash and cash equivalents are held in accounts at financial institutions. Cash equivalents consist of amounts held in a money market account. Such deposits have and will continue to exceed federally insured limits in the foreseeable future.

Leasehold Improvements, Net

Leasehold improvements are presented at cost, net of accumulated amortization. Amortization expense is recorded using the straight-line method over the shorter of the remaining lease term or the estimated useful life.

Convertible Instruments and Embedded Derivatives

The Company accounts for certain redemption features that are associated with convertible notes as liabilities at fair value and adjusts the instruments to their fair value at the end of each reporting period. For derivative financial instruments that are accounted for as liabilities, the derivative instrument is initially recorded at its fair value and is then re-valued at each reporting date, with changes in the fair value reported in other income (expense), net in the condensed statements of operations. Derivative instrument liabilities are classified in the condensed balance sheets as current or non-current based on whether or not net-cash settlement of the derivative instrument could be required within 12 months of the balance sheet date. As of December 31, 2020, the Company's only derivative financial instrument was related to the 2020 Convertible Notes, which contained certain redemptive features. The Company completed its IPO on August 30, 2021, which triggered the automatic conversion of all outstanding Convertible Notes,

plus accrued interest, into units, consisting of (a) one share of common stock and (b) one five-year warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share. Upon the conversion of the Convertible Notes, the outstanding Convertible Notes, plus accrued interest thereon totaling \$5.3 million, net of unamortized debt discounts, were derecognized into stockholders' equity (Note 5).

Clinical Trial Expenses

The Company makes payments in connection with its ongoing Phase 3 clinical trial under contracts with clinical trial sites and contract research organizations that support conducting and managing clinical trials. The financial terms of these agreements are subject to negotiation and vary from contract to contract and may result in uneven payment flows. Generally, these agreements set forth the scope of work to be performed at a fixed fee, unit price or on a time and materials basis. A portion of the obligation to make payments under these contracts depends on factors such as the successful enrollment or treatment of patients or the completion of other clinical trial milestones.

Expenses related to clinical trials are accrued based on estimates and/or representations from service providers regarding work performed, including actual level of patient enrollment, completion of patient studies and progress of the clinical trials. Other incidental costs related to patient enrollment or treatment are accrued when reasonably certain. If amounts and obligations to pay under clinical trial agreements are modified (for instance, as a result of changes in the clinical trial protocol or scope of work to be performed), the accruals are adjusted accordingly. Revisions to contractual payment obligations are charged to expense in the period in which the facts that give rise to the revision become reasonably certain. In addition, the clinical trial sites involved in our Phase 3 clinical trial of RenovoGem are charged for the RenovoCath delivery devices used in the trial. The payments received from the clinical trial sites for the devices are adequate to cover the direct costs of manufacturing and offset research and development expenses.

9

Research and Development Expenses

Research and development expenses are charged to expense as incurred. Research and development expenses includes personnel costs including salaries, benefits and stock-based compensation for employees related to research and development activities. In addition, expenses for consultants that support clinical trial studies, materials costs, external clinical drug product manufacturing costs, outside services costs, regulatory activities including filing fees, fees for maintaining licenses and other amounts due to third-party agreements, laboratory materials, clinical trial, as noted above, and supplies to support our research activities, as well as allocated facility related costs.

Fair Value Measurement

Fair value is defined as the exchange price that would be received for an asset or paid to transfer a liability, or an exit price, in the principal or most advantageous market for that asset or liability in an orderly transaction between market participants on the measurement date. Fair value measurement establishes a fair value hierarchy that requires an entity to maximize the use of observable inputs, where available, and minimize the use of unobservable inputs when measuring fair value.

The Company determined the fair value of financial assets and liabilities using the fair value hierarchy that describes three levels of inputs that may be used to measure fair value, as follows:

Level 1—Quoted prices in active markets for identical assets and liabilities;

Level 2—Inputs other than Level 1 that are observable, either directly or indirectly, such as quoted prices for similar assets or liabilities, quoted prices in markets that are not active, or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities; and

Level 3—Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

As of September 30, 2021, fair value measurements consisted solely of cash equivalents which comprise money market securities. The carrying amounts of these instruments approximate their fair value. The carrying value of all remaining current assets and current liabilities approximate their fair value.

Stock-Based Compensation Expense

The Company maintains an equity incentive plan as a long-term incentive for selected employees, consultants, and directors. The plan allows for the issuance of incentive stock options, non-statutory stock options, stock appreciation rights, restricted stock grants, and restricted stock units.

The Company accounts for stock-based compensation expense by measuring and recognizing compensation expense for all share-based payments made to employees and non-employees based on estimated grant-date fair values. The Company uses the straight-line method to allocate compensation cost to reporting periods over each recipient's requisite service period, which is generally the vesting period over four years. The Company estimates the fair value of stock options granted to employees using the Black-Scholes option pricing model. The Black-Scholes option pricing model requires the input of subjective assumptions, including expected volatility, expected dividend yield, expected term and the risk-free rate of return.

Prior to the Company's IPO, given the absence of a public trading market, the Company's Board of Directors considered numerous objective and subjective factors to determine the fair value of the common stock at each grant date. These factors included, but were not limited to: (i) contemporaneous third-party valuations of common stock; (ii) the prices for preferred stock sold to outside investors; (iii) the rights and preferences of preferred stock relative to common stock; (iv) the lack of marketability of the Company's common stock; (v) developments in the business; and (vi) the likelihood of achieving a liquidity event, such as an IPO or sale of the business, given prevailing market conditions. The methodology to determine the fair value of our common stock included estimating the fair value of the enterprise using the "backsolve" method, which is a market approach that assigns an implied enterprise value by accounting for all share class rights and preferences based on the latest round of financing. The total equity value implied was then applied in the context of an option pricing model to determine the value of each class of our shares.

Future awards will be based on the closing price of the Company's common stock as reported on the date of grant to determine the fair value of the award.

10

Emerging Growth Company and Smaller Reporting Company Status

The Company is an emerging growth company ("EGC") as defined in the Jumpstart Our Business Startups Act of 2012 ("JOBS Act") and may take advantage of reduced reporting requirements that are otherwise applicable to public companies. Section 107 of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies are required to comply with those standards. The Company has elected to use the extended transition period for complying with new or revised accounting standards.

From time to time, new accounting pronouncements are issued by the Financial Accounting Standards Board ("FASB") or other standard setting bodies and adopted by the Company as of the specified effective date. Unless otherwise discussed, the impact of recently issued standards that are not yet effective will not have a material impact on the Company's financial position or results of operations upon adoption.

Recent Accounting Pronouncements

Recently Adopted Accounting Pronouncements

In June 2018, the FASB issued Accounting Standards Updates ("ASU") No. 2018-07, Improvements to Nonemployee Share-Based Payment Accounting (ASU 2018-07). The standard simplifies the accounting for share-based payments granted to nonemployees for goods and services and aligns most of the guidance on such payments to the nonemployees with the requirements for share-based payments granted to employees. ASU 2018-07 is effective for the Company for annual reporting periods beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2020, with early adoption permitted. The guidance should be applied to new awards granted after the date of adoption. The Company adopted this new standard on January 1, 2020 and the adoption of this standard did not have an impact on its condensed interim financial statements.

In August 2018, the FASB issued ASU 2018-13, Fair Value Measurement (Topic 820), Disclosure Framework — Changes to the Disclosure Requirements for Fair Value Measurement (ASU 2018-13). The standard eliminates, adds and modifies certain disclosure requirements for fair value measurements. Entities will no longer be required to disclose the amount of and reasons for transfers between Level 1 and Level 2 of the fair value hierarchy, but public companies will be required to disclose the range and weighted average of significant unobservable inputs used to develop Level 3 fair value measurements. The standard is effective for annual reporting periods beginning after December 15, 2019, and for interim periods within those periods. The Company adopted this new standard on January 1, 2020, with no material impact on its condensed interim financial statements.

-In November 2016, the FASB ASU 2016-18, Statement of Cash Flows (Topic 230): Restricted Cash (ASU 2016-18), which requires that a statement of cash flows explain the change during the period in the total of cash, cash equivalents and restricted cash. Therefore, amounts described as restricted cash should be included with cash and cash equivalents when reconciling the beginning of period and end of period amounts shown on the statement of cash flows. The Company adopted this guidance on January 1, 2019, and it did not have an impact on its financial results, but it did result in a change in the presentation of restricted cash and cash equivalents within the statements of cash flows.

Recent Accounting Pronouncements Not Yet Adopted

In February 2016, the FASB issued ASU No. 2016-02, Leases (Topic 842) (ASU 2016-02). The guidance requires lessees to recognize assets and liabilities related to long-term leases on the balance sheet and expands disclosure requirements regarding leasing arrangements. In July 2018, the FASB issued additional guidance, which offers a transition option to entities adopting the new lease standards, and a package of practical expedients an entity can elect to utilize to reduce the level of effort required for adoption. Under the transition option, entities can elect to apply the new guidance using a modified retrospective approach at the beginning of the year in which the new lease standard is adopted, rather than to the earliest comparative period presented in their financial statements. In November 2019, the FASB issued ASU 2019-10, Leases (Topic 842) (ASU 2019-10), deferring the effective date for the Company for fiscal years beginning after December 15, 2020 and interim periods within fiscal years beginning after December 15, 2021. In June 2020, the FASB issued ASU 2020-05, Leases (Topic 842) (ASU 2020-05), which further defers the effective date for the Company for fiscal years beginning after December 15, 2021 and interim periods within fiscal years beginning after December 15, 2021. Early adoption is permitted. The Company plans to adopt this new standard effective January 1, 2022. The Company is currently evaluating its contracts to determine whether there will be a significant impact from the adoption of this guidance on its condensed interim financial statements.

In June 2016, the FASB issued ASU No. 2016-13, Financial Instruments – Credit Losses (ASU 2016-13), which requires the measurement of expected credit losses for financial instruments carried at amortized cost, such as accounts receivable, held at the reporting date based on historical experience, current conditions and reasonable forecasts. The main objective of this standard is to provide financial statement users with more decision-useful information about the expected credit losses on financial instruments and other commitments to extend credit held by a reporting entity at each reporting date. In November 2018, the FASB issued ASU No. 2018-19, Codification Improvements to Topic 326, Financing Instruments – Credit Losses (ASU 2018-19), which included an amendment of the effective date. The standard is effective for the Company for annual reporting periods beginning after December 15, 2021, and for interim periods within those periods. Early adoption is permitted. The Company plans to adopt this new standard on January 1, 2022 and does not believe that adoption will have a significant impact on its condensed interim financial statements.

11

In December 2019, the FASB issued ASU No. 2019-12, *Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes* (ASU 2019-12), which simplifies the accounting for income taxes, and is effective on a prospective basis for annual reporting periods beginning after December 15, 2021 and for interim periods within fiscal years beginning after December 15, 2022, with early adoption permitted. The Company plans to adopt this new standard on January 1, 2022 and does not believe that adoption will have a significant impact on its condensed interim financial statements.

In August 2020, the FASB issued ASU No. 2020-06, *Debt – Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging – Contracts in Entity's Own Equity (Subtopic 815-40)* (ASU 2020-06): *Accounting for Convertible Instruments and Contracts in an Entity*, which simplifies the accounting for certain financial instruments with characteristics of liabilities and equity, including convertible instruments and contracts on an entity's own equity. The updated guidance is effective on a prospective basis for annual reporting periods beginning after December 15, 2023 and for interim periods within those periods. Early adoption is permitted. The Company has not yet determined the impact that this new standard will have on its financial position and results of operations.

3. Fair Value Measurements

As of September 30, 2021 and December 31, 2020, the Company held \$1.6.1 million and \$1.7 million, respectively, in a money market account.

The following tables summarize the Company's financial assets and liabilities measured at fair value on a recurring basis by level within the fair value hierarchy as of September 30, 2021 and December 31, 2020 (in thousands):

		Fair Value Measurements at September 30, 2021 using:								
Assets:	Le	evel 1	Level 2	Level	3	Total				
Money market account	\$	16,141 \$	-	\$	- \$	16,14				
	\$	16,141 \$	-	\$	- \$	16,14				
		Fair Value	Measurements at D	December 31, 2020	(audited) using:	:				
Assets:	Le	evel 1	Level 2	Level	3	Total				
Money market account	\$	1,703 \$	-	\$	- \$	1,703				
	\$	1,703 \$		\$	- \$	1,700				
Liabilities:										
Derivative liability – 2020										
Convertible Notes	\$	<u>-</u> \$		- \$	856 \$	856				
	\$	- \$		- \$	856 \$	856				
		12								

The change in the fair value of the derivative liability is summarized below (in thousands):

	 Derivative Liability at					
	ember 30, 2021	December 31, 2020 (audited)				
Fair value at beginning of the period	\$ 856 \$		-			
Initial fair value of instruments issued	363		856			
Change in fair value of instruments	(118)		-			
Conversion upon IPO	(1,101)		-			
Fair value at end of the period	\$ - \$		856			

The derivative liability in the table above relates to the 2020 and 2021 Convertible Notes and represents the fair value of the redemption-like contingent conversion feature. The Company calculated the fair value of the derivative liability using a probability weighted discounted cash flow analysis. The inputs used to determine the estimated fair value of the derivative were based primarily on the probability of an underlying event occurring that would trigger the embedded derivative and the timing of such event. The Company's derivative liability was measured at fair value on a recurring basis and was classified as a Level 3 liability. The Company recorded subsequent adjustments to reflect the increase or decrease in estimated fair value at each reporting date in other income (expense), net in the condensed statements of operations (see Note 5, Convertible Notes).

There were no transfers among Level 1, Level 2 or Level 3 categories during any of the periods presented. The Company had no other financial assets or liabilities that were required to be measured at fair value on a recurring basis.

4. Accrued Expenses

Accrued expenses consisted of the following (in thousands):

	September 30, 2021		December 31, 2020 (audited)	
Clinical trials	\$	377	\$	171
Research and development		14		_
Professional services		80		_
Interest		_		101
Personnel		13		39
Other		8		
Total accrued expenses	\$	492	\$	311

Accrued research and development expenses were primarily related to clinical trials and materials.

5. Convertible Notes

In March 2020, the Company entered into a note purchase agreement for the issuance of up to \$4.0 million of convertible promissory notes, which, if not converted, had an initial maturity date of March 31, 2021. The Company entered into a series of convertible note payable agreements (the "2020 Convertible Notes") for aggregate borrowings of \$3.0 million. The 2020 Notes bore interest at the rate of 5% per annum and could not be prepaid prior to the maturity date unless approved in writing by the Company and requisite holders.

The terms of the 2020 Convertible Notes provided for automatic conversion into equity shares in the next equity financing round with total proceeds of not less than \$0.0 million (a "Qualified Financing"), at a conversion price per share equal to 80% of the price per share paid by investors purchasing such equity securities in a Qualified Financing. For purposes of the 2020 Convertible Notes, equity securities meant the Company's common stock, preferred stock or any securities providing for rights to purchase the Company's common stock, preferred stock or securities convertible into or exchangeable for the Company's common stock or preferred stock issued in the Qualified Financing. If the Company consummated a Change of Control prior to a Qualified Financing, the Company would repay each holder in cash an amount equal to the greater of (a) two times (2x) the entire outstanding principal balance of the 2020 Convertible Notes or (b) the amount the holder would receive if the 2020 Convertible Notes had been converted into shares of the Company's Series B convertible preferred stock immediately prior to the consummation of the Change in Control, at a conversion price equal to the Series B convertible preferred stock Original Issue Price.

13

On March 1, 2021, the Company entered into an amendment to the 2020 Convertible Notes which extended the maturity date of the 2020 Convertible Notes from March 31, 2021 to October 30, 2021 and provided for the conversion of the 2020 Convertible Notes into shares of the Company's common stock upon a Qualified Financing that is an IPO. No other terms of the 2020 Convertible Notes were amended. This amendment was accounted for as a troubled debt restructuring pursuant to FASB ASC Topic 470-60, "Troubled Debt Restructurings by Debtors." As the future undiscounted cash flows of the 2020 Convertible Notes were greater than their carrying amount, the carrying amount was not adjusted and no gain was recognized as a result of the modification of terms.

The Company determined that the redemption features contained rights and obligations for conversion were contingent upon a potential future financing event or a change in control. Thus, the embedded redemption features were bifurcated from the face value of the notes and accounted for as a derivative liability to be remeasured at the end of each reporting period. The fair value of the derivative liability at September 30, 2021 and December 31, 2020 was \$0 and \$856,000, respectively. Debt issuance costs were \$22,000 at December 31, 2020. There were no debt issuance costs as of September 30, 2021. The derivative liability was subject to fair value remeasurement at the end of each reporting period. The debt discount and debt issuance costs were being amortized to interest expense using the effective interest method over the expected term of the 2020 Convertible Notes. For the three and nine months ended September 30, 2021, the Company recognized \$18,000 and \$379,000 for amortization of the debt discount and debt issuance costs, respectively. For the three and nine months ended September 30, 2020, the Company recognized \$153,000 and \$294,000 for the amortization of the debt discount and debt issuance costs, respectively. This amortization expense is recognized as interest expense in the condensed statements of operations. The effective interest rate of the 2020 Convertible Notes was 0% at September 30, 2021 and 30.8% at December 31, 2020, compared to the stated rate of 5% per annum. The effective interest rate immediately prior to the conversion of the Convertible Notes resulting from the Company's IPO was 8.6% per annum. As a result, the Company's reported interest expense was significantly higher than the contractual cash interest payments. During the three and nine months ended September 30, 2021, the Company recognized interest expense in the condensed statements of operations of \$25,000 and \$101,000, respectively, related to the 2020 Convertible Notes. During the three and nine months ended September 30, 2020,

In April 2021, the Company entered into a note purchase agreement and a series of convertible note payable agreements (the "2021 Convertible Notes," together with the 2020 Convertible Notes, the "2020 and 2021 Convertible Notes") for aggregate borrowings of \$2.0 million. Outstanding borrowings under the 2021 Convertible Notes and accrued interest were due in April 2022, if not previously converted. The 2021 Notes bore interest at the rate of 5% per annum. Pursuant to the 2021 Convertible Notes, the outstanding

principal and accrued interest are automatically convertible into equity shares in a Qualified Financing at a conversion price per share equal to 87.5% of the price per share paid by investors purchasing such equity securities in a Qualified Financing.

The Company determined that these redemption features in the 2021 Convertible Notes contained rights and obligations for conversion that were contingent upon a potential future financing event or a change in control. Thus, the embedded redemption features were bifurcated from the face value of the note and accounted for as a derivative liability to be remeasured at the end of each reporting period. Upon issuance of the notes, the Company recorded the fair value of the derivative liability of \$363,000 and debt issuance costs of \$23,000, with the offsetting amount being recorded as a debt discount. The discount and debt issuance costs were amortized to interest expense using the effective interest method over the expected term of the 2021 Convertible Notes. For the three and nine months ended September 30, 2021, the Company recognized \$141,000 and \$318,000, respectively, for the amortization of the debt discount and debt issuance costs as interest expense in the condensed statements of operations. The effective interest rate immediately prior to the conversion of the 2021 Convertible Notes resulting from the Company's IPO was 46.5% per annum compared to the stated rate of 55% per annum. During the three and nine months ended September 30, 2021, the Company recognized interest expense in the condensed statements of operations of \$17,000 and \$38,000, respectively, relating to the 2021 Convertible Notes.

The Company completed an IPO on August 30, 2021, which triggered the automatic conversion of the outstanding Convertible Notes plus accrued interest into an aggregate of 708,820 units. Each unit consisted of (a) one share of common stock and (b) one five-year warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share (Note 7). Upon conversion of the 2020 and 2021 Convertible Notes, the outstanding principal, including debt discount and debt issuance costs for those Convertible Notes of \$5.3 million, was derecognized into stockholders' equity. The unamortized debt discount totaling \$78,000 was recognized as a loss on extinguishment of debt and is included in loss (gain) on loan extinguishment in the Company's condensed statements of operations.

6. Promissory Note

On April 22, 2020, the Company entered into a promissory note with Silicon Valley Bank that provided for the receipt by the Company of loan proceeds of \$40,000 (the "PPP Loan"), with an interest rate of 1.0% per annum, pursuant to the Paycheck Protection Program under the Coronavirus Aid, Relief and Economic Security Act (the "CARES Act"). Under certain conditions, the loan and accrued interest were forgivable, including if the loan proceeds were used for eligible purposes, including payroll, benefits, rent and utilities, and maintaining payroll levels. In October 2020, the Paycheck Protection Program Flexibility Act of 2020 extended the deferral period for borrower payments of principal, interest, and fees on all PPP loans from 6 months to 10 months. As of December 31, 2020, payments were deferred for 10 months. If not forgiven earlier, the PPP Loan was to mature on April 22, 2022. The PPP Loan contains events of default and other provisions customary for a loan of this type. The Company recorded the PPP Loan as a promissory note in the December 31, 2020 balance sheet as both a current and non-current liability.

14

On February 6, 2021, the Company received notification and confirmation from Silicon Valley Bank that its PPP loan and related accrued interest were forgiven in their entirety by the U.S. Small Business Administration and automatically cancelled. During the nine-months ended September 30, 2021, the \$140,000 was recorded to loss (gain) on loan extinguishment in the condensed statements of operations.

7. Capital Stock

Common Stock

On August 25, 2021, the Company's Registration Statement on Amendment No. 4 to the Form S-1 relating to its IPO was declared effective by the SEC. In connection with the IPO, the Company issued and sold an aggregate of 1,850,000 units at a price of \$9.00 per unit. Each unit consisted of (a) one share of common stock and (b) one warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share, which is exercisable for a period of five years after the issuance date. The Company received net proceeds of \$14.6 million from the IPO, after deducting underwriting discounts and commissions of \$1.3 million and other costs incurred with the offering of \$0.8 million. Upon the closing of the IPO, all of the 3,535,469 outstanding shares of the Company's convertible preferred stock automatically converted into 3,535,469 shares of common stock and the outstanding 2020 and 2021 Convertible Notes, including accrued but unpaid interest, representing \$5.3 million, converted to 708,820 units, which consisted of (a) one share of common stock and (b) one five-year warrant to purchase one share of common stock at an exercise price equal to \$0.80 per share. Upon completion of the offering on August 30, 2021, the Company was authorized to issue 250,000,000 shares of common stock, par value of \$0.0001 per share and 15,000,000 shares of preferred stock, par value of \$0.0001 per share and 15,000,000 shares of preferred

Convertible Preferred Stock

Issued and outstanding convertible preferred stock and its principal terms as of December 31, 2020 (audited) were as follows (in thousands, except share and per share amounts):

		Shares Issued and	1	Aggregate	Net Carrying
Preferred Series	Shares Authorized	Outstanding		idation Value	 Value
Series A-1	3,542,669	708,533	\$	660	\$ 639
Series A-2	3,546,095	709,219		1,150	1,099
Series A-3	2,660,230	532,046		2,227	2,166
Series B	12,611,461	1,585,671		8,745	8,547
Total	22,360,455	3,535,469	\$	12,782	\$ 12,451

The Company classified its convertible preferred stock as mezzanine equity on the condensed balance sheets as the shares were contingently redeemable upon deemed liquidation events, such as a change of control.

In August 2021, immediately prior to the completion of the IPO and after giving effect to the 1-5 reverse stock split, all outstanding shares of the Company's convertible preferred stock were automatically converted into 3,535,469 shares of common stock. In August 2021, the Company filed its Sixth Amended and Restated Articles of Incorporation with the Secretary of State of the State of Delaware, authorizing the issuance of 15,000,000 shares of preferred stock. There were no shares of preferred stock outstanding as of September 30, 2021.

15

8. Commitments and Contingencies

Contingencies

From time to time, the Company may become involved in legal proceedings arising in the ordinary course of business. The Company was not subject to any material legal proceedings during the nine months ended September 30, 2021 and no material legal proceedings are currently pending or threatened.

In the ordinary course of business, the Company enters into agreements that may include indemnification provisions. As permitted under Delaware law and in accordance with its bylaws, the Company indemnifies its officers and directors for certain events or occurrences while the officer or director is or was serving in such capacity. The Company is also party to indemnification agreements with its officers and directors. In some cases, the indemnification will continue after the termination of the agreement. The maximum potential amount of future payments that the Company could be required to make under these provisions is not determinable. The Company has never incurred material costs to defend lawsuits or settle claims related to these indemnification provisions. The Company is not currently aware of any indemnification claims. Accordingly, the Company has not recorded any liabilities for these indemnification rights and agreements as of September 30, 2021 and December 31, 2020.

Operating Leases

The Company leases its headquarters in Los Altos, California under a one-year operating lease agreement which expires on May 31, 2022. Rent expense was \$18,000 and \$11,000 for the three months ended September 30, 2021 and 2020, respectively. Rent expense was \$42,000 and \$33,000 for the nine months ended September 30, 2021 and 2020, respectively.

Coronavirus Pandemic

In December 2019, a novel strain of coronavirus, which causes the disease known as COVID-19, was reported to have surfaced in Wuhan, China. Since then, COVID-19 has spread globally. In March 2020, the World Health Organization declared the COVID-19 outbreak a pandemic. The ongoing COVID-19 global and national health emergency has caused significant disruption in the international and U.S. economies and financial markets. The spread of COVID-19 has caused illness, quarantines, cancellation of events and travel, business and school shutdowns, reduction in business activity and financial transactions, labor shortages, supply chain interruptions and overall economic and financial market instability.

In response to public health directives and orders and to help minimize the risk of the virus to employees, the Company has taken precautionary measures, including implementing work-from home and/or hybrid work policies for employees. The COVID-19 global pandemic has also negatively affected, and the Company expects it will continue to negatively affect, our clinical studies. For example, we have faced challenges in conducting our Phase 3 clinical trial, including recruiting subjects and accommodating patient visits. Additionally, the Company's service providers and their operations may be disrupted, temporarily closed or experience worker or supply shortages, which could result in additional disruptions or delays in shipments of purchased materials or the continued development of our product candidates. To date, the Company has not suffered material supply chain disruptions.

The Company is not able to estimate the duration of the pandemic and the potential impact on its business. As the global pandemic of COVID-19 continues to evolve, it could result in significant long-term disruption of global financial markets, reducing the Company's ability to raise additional capital when needed and on acceptable terms, if at all, which could negatively affect liquidity. The extent to which the COVID-19 pandemic impacts the Company's clinical development and regulatory efforts will depend on future developments that are highly uncertain and cannot be predicted with confidence, such as the ultimate geographic spread of the disease, the duration of the outbreak, travel restrictions, quarantines and social distancing requirements in the United States and other countries, business closures or business disruptions and the effectiveness of actions taken in the United States and other countries to contain and treat the virus. The Company continues to monitor the COVID-19 situation closely.

9. Equity Incentive Plan-Stock-Based Compensation Expense and Common Stock Warrants

2021 Omnibus Equity Incentive Plan

On July 19, 2021, the Company's Board of Directors adopted the RenovoRx, Inc. 2021 Omnibus Equity Incentive Plan (the "2021 Plan"). The 2021 Plan, which became effective immediately prior to the closing of the IPO, provides for the grant of incentive stock options ("ISO"), non-statutory stock options ("NSO"), restricted stock, restricted stock units, stock appreciation rights, and other stock-based awards to selected employees, directors, and consultants. The Company initially reserved 2,175,000 shares of common stock, including the addition of 20,401 common shares available pursuant to the Amended and Restated 2013 Equity Incentive Plan (the "2013 Plan"). The Company's 2013 Plan was terminated immediately prior to the closing of the IPO; however, shares subject to awards granted under the 2013 Plan will continue to be governed by the 2013 Plan. Shares subject to awards granted under the 2013 Plan that are repurchased by, or forfeited to, the Company will also be reserved for issuance under the 2021 Plan. The 2021 Plan provides an annual increase on January 1, beginning on January 1, 2022, during the initial ten-year term of the 2021 Plan, equal to the lesser of (A) three percent (3%) of the shares outstanding (on an as-converted basis) on the final day of the immediately preceding calendar year and (B) such lesser number of shares as determined by the Board; provided that shares of common stock issued under the 2021 Plan with respect to an Exempt Award will not count against the share limit. The term "Exempt Award" means (i) an award granted in assumption of, or in substitution for, outstanding awards previously granted by another business entity acquired by the Company or any of its subsidiaries or with which the Company or any of its subsidiaries merge, or (ii) an award that a participant purchases at fair value. Since the date of adoption of the 2013 and 2021 Plans and through September 30, 2021, the Company has issued stock-based awards to its employees, directors, and consultants. In most instances, the o

16

Options under the 2021 Plan may be granted for periods of up to 10 years and at exercise prices no less than 100% of the estimated fair value of the underlying shares of common stock on the date of grant as determined by the Board of Directors provided that the exercise price of an ISO and NSO granted to a 10% stockholder shall not be less than 110% of the estimated fair value of the shares on the date of grant. The 2021 Plan requires that options be exercised no later than 10 years after the grant. Options granted to employees generally vest ratably on a monthly basis over four years, subject to cliff vesting restrictions.

*** * * * *

The following is a summary of the stock option award activity during the nine months ended September 30, 2021:

	Number of Stock Options	eighted- age Exercise Price	Weighted- Average Remaining Contractual Life (in years)	Intri	ggregate insic Value housands)
Outstanding as of December 31, 2020	997,266	\$ 0.42	5.84	\$	276
Granted	238,269	\$ 4.94	-		-
Exercised	(332,932)	\$ 0.27	-		-
Forfeited	(23,625)	\$ 0.67	-		-
Expired	(83)	\$ 0.70	-		-
Outstanding as of September 30, 2021	878,895	\$ 0.48	6.58	\$	3,817
Exercisable as of September 30, 2021	603,086	\$ 0.53	5.28	\$	3,323
Vested and expected to vest as of September 30, 2021	878,895	\$ 1.70	6.58	\$	3,817

As of September 30, 2021, there was \$364,000 of unrecognized stock-based compensation expense related to options granted but not yet amortized, which will be recognized over a weighted-average period of 2.36 years.

For the nine months ended September 30, 2021, the Company utilized the Black-Scholes option-pricing model for estimating the fair value of the stock option granted. The following table presents the assumptions and the Company's methodology for developing each of the assumptions used:

 Volatility
 41.66% – 42.13%

 Expected life (years)
 5.00 – 10.0

 Risk-free interest rate
 0.62% – 1.00%

 Dividend rate
 -%

- Volatility—The Company estimates the expected volatility of its common stock at the date of grant based on the historical volatility of comparable public companies
 over the expected term.
- Expected life—The expected life is estimated as the contractual term.
- Risk-free interest rate—The risk-free rate for periods within the estimated life of the stock award is based on the U.S. Treasury yield curve in effect at the time of grant.
- Dividend rate—The assumed dividend yield is based upon the Company's expectation of not paying dividends in the foreseeable future.

17

The following table summarizes the components of stock-based compensation expense recognized in the Company's condensed statements of operations during the three and nine months ended September 30, 2021 and 2020 (in thousands):

		Three Months Ended September 30,				Nine Months Ended September 30,				
	202	1	2020		-	2021		2020		
Research and development	\$	2	\$	5	\$	5	\$	19		
General and administrative		4		2		16		6		
Total stock-based compensation expense	\$	6	\$	7	\$	21	\$	25		

Common Stock Warrants

In connection with the IPO, the Company issued warrants to purchase 3,035,195 shares of the Company's common stock, of which, warrants to purchase 198,875 shares of the Company's common stock expire on August 25, 2026 and warrants to purchase 2,836,320 shares of the Company's common stock expire on August 31, 2026. See Note 1, *Initial Public Offering*.

The following is a summary of the common stock warrant activity during the nine months ended September 30, 2021:

	Shares Issuable Upon Exercise of Outstanding Warrants	 Weighted- Average Exercise Price	Weighted- Average Remaining Contractual Life (in years)	 Aggregate Intrinsic Value n thousands)
Outstanding as of December 31, 2020	-	\$ _	-	\$ -
Issued	3,035,195	\$ 9.81	-	-
Exercised	(525,200)	\$ 5.10	-	-
Expired	-	\$ -	-	-
Outstanding as of September 30, 2021	2,509,995	\$ 10.80	4.92	\$ 27,108

10. Income Taxes

The Company had no income tax expense for the three and nine months ended September 30, 2021 and 2020. During the nine months ended September 30, 2021 and 2020, the Company had a net operating loss ("NOL") for each period that generated deferred tax assets for NOL carryforwards. Deferred income tax assets and liabilities are recognized for temporary differences between the financial statements and income tax carrying values using tax rates in effect for the years such differences are expected to reverse. Due to uncertainties surrounding our ability to generate future taxable income and consequently realize such deferred income tax assets, the Company has determined that it is more-likely-than-not that these deferred tax assets will not be realized. Accordingly, the Company has established a full valuation allowance against its deferred tax assets as of September 30, 2021.

The Company's policy is to recognize any interest and penalties related to unrecognized tax benefits as a component of income tax expense. As of September 30, 2021 and December 31, 2020, the Company had no accrued interest or penalties related to uncertain tax positions.

11. Net Loss Per Share

The following table sets forth the computation of the basic and diluted net loss per share (in thousands, except share and per share data):

	Three Months Ended September 30,				Nine Months Ended September 30,			
	2021 2020		2021			2020		
Numerator:								
Net loss	\$	(1,511)	\$	(1,096)	\$	(3,969)	\$	(2,920)
Denominator:								
Weighted-average shares of common stock outstanding used in								
the calculation of basic and diluted net loss per share		5,620,135		2,263,589		3,640,988		2,233,645
Net loss per share, basic and diluted	\$	(0.27)	\$	(0.48)	\$	(1.09)	\$	(1.31)

Since the Company had a net loss for all periods presented, basic net loss per share is the same as diluted net loss per share for all periods as the inclusion of all common stock equivalents outstanding would have been anti-dilutive. Potentially dilutive securities that were not included in the diluted per share calculations because they would be anti-dilutive were as follows:

As of	Sei	otem	ber	30.	
-------	-----	------	-----	-----	--

	2021	2020
Options to purchase common stock	878,895	1,012,075
Convertible preferred stock		3,535,469
Total	878,895	4,547,544

18

12. Related Party Transactions

In January 2018, the Company entered into a consulting agreement with one of the Company's co-founders, Dr. Ramtin Agah, pursuant to which Dr. Agah provides consulting services as the Company's Chief Medical Officer by overseeing Company-sponsored clinical trials. The Agreement, which was amended on September 1, 2019, and November 11, 2021, continues in force for as long as Dr. Agah is providing consulting services and may be terminated by either party on thirty (30) days' notice. Dr. Agah was awarded (i) options to purchase 60,000 shares of the Company's common stock in May 2017, which have vested, (ii)options to purchase 40,000 shares of the Company's common stock in July 2018, of which 25% vested after one year and the remainder vests ratably over the 36 month period ending July 2022, (iii) options to purchase 0,000 shares of the Company's common stock in June 2021, which vest ratably over 24 months from the vesting commencement date of May 14, 2023, and (iv) options to purchase of 52,203 shares of the Company's common stock in September 2021, which vest ratably over 48 months from the vesting commencement date of August 26, 2021.In December 2018, Dr. Agah's agreement was amended to provide that he would receive cash compensation of \$4,000 per month for certain proctoring services, and in September 2019, his compensation was increased to \$10,000 per month to compensate for additional services he was providing. Effective upon the completion of the IPO, Dr. Agah's compensation was increased to \$260,000 annually, based on Dr. Agah spending no less than 24 hours per week on Company matters. Consulting fees paid to Dr. Agah for the three months ending September 30, 2021 and 2020, were \$42,000 and \$30,000, respectively. Consulting fees paid to Dr. Agah for the nine months ending September 30, 2021 and 2020, were \$42,000 and \$30,000, respectively.

In July 2019, the Company entered into a consulting agreement with the Company's then Chief Financial Officer, Paul Manners. In February 2020, the Company granted Mr. Manners an option to purchase 28,000 shares of the Company's common stock, of which 25% were vested at the grant date and the remainder vested ratably over the following 18 months. The CFO Agreement was amended in December 2020 to increase Mr. Manners hourly rate to \$150. Consulting fees paid to Mr. Manners for the three months ending September 30, 2021 and 2020, were \$56,000 and \$12,000, respectively. Consulting fees paid to Mr. Manners for the nine months ending September 30, 2021 and 2020, were \$155,000 and \$32,000, respectively. In August 2021, upon completion of the IPO, Mr. Manners stepped from his role and Christopher J. Lehman, was appointed Chief Financial Officer.

Kamran Najmabadi, another co-founder of the Company, has served as our consulting technical engineering advisor on manufacturing and intellectual property matters since January 2020. Mr. Najmabadi served as the Company's Chief Executive Officer from its inception in December 2009 until January 2013; Chief Technical and Operations Officer from January 2013 until January 2019; and Chief Technology Officer from January 2019 to January 2020. He currently receives cash compensation of \$3,000 per quarter.

13. Subsequent Events

The Company has evaluated all events occurring through November 15, 2021, the date on which the unaudited condensed interim financial statements were issued. Since September 30, 2021, the following material subsequent events occurred:

In October 2021, 66,923 non-qualified stock options were granted to members of the Board of Directors.

In October 2021, the Company filed its Form S-8, Registration Statement Under the Securities Act of 1933 with the SEC.

In November 2021, the Company's Board of Directors approved, and we entered into the following agreements: (i) a Confirmatory Employment Letter with Shaun R. Bagai, our Chief Executive Officer, (ii) an Amendment to our Consulting Agreement with Dr. Ramtin Agah, our Chief Medical Officer, (iii) Change in Control and Severance Agreements with Mr. Bagai and Dr. Agah. The Board of Directors also adopted a Key Service Provider Incentive Compensation Plan, which allows our Compensation Committee to determine which employees or service providers may receive awards under the plan, establish target awards and performance goals for those individuals, and determine the appropriate incentive awards to be granted. See Item 5. Other Information.

19

Item 2: Management's Discussion and Analysis of Financial Condition and Results of Operations

Unless the context otherwise requires, all references in this section to the "Company," "we," "us," or "our" refer to the business of RenovoRx, Inc. You should read the following discussion and analysis of our financial condition and results of operations in conjunction with our unaudited interim condensed financial statements and related notes included elsewhere in this Quarterly Report on Form 10-Q and our final prospectus, dated August 25, 2021, filed with the SEC pursuant to Rule 424(b)(4) under the Securities Act of 1933, as amended (the "Securities Act").

This discussion contains forward-looking statements within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended, that reflect our plans, estimates, and beliefs that involve risks and uncertainties, including those described in the section titled "Special Note Regarding Forward Looking Statements." Our actual results and the timing of selected events could differ materially from those discussed below. Factors that could cause or contribute to such differences include, but are not limited to, those identified below and those set forth under the section titled "Risk Factors" included elsewhere in this report.

Overview

We are a late-stage clinical biopharmaceutical company focused on developing therapies for the local treatment of solid tumors. Our therapy platform, RenovoRx Trans-Arterial Micro-Perfusion, or RenovoTAMP™ utilizes approved chemotherapeutics with validated mechanisms of action and well-established safety and side effect profiles with the goal of increasing their efficacy, improving their safety, and widening their therapeutic window. RenovoTAMP combines our patented FDA cleared delivery system, RenovoCath®, with small molecule chemotherapeutic agents that can be forced across the vessel wall using pressure, targeting these anti-cancer drugs locally to the solid tumors. Our first product candidate, RenovoGemTM, is a drug and device combination consisting of intra-arterial gemcitabine and RenovoCath. FDA has determined that RenovoGem will be regulated as, and if approved we expect will be reimbursed as, a new oncology drug product. We have secured FDA Orphan Drug Designation for RenovoGem in our first two indications: pancreatic cancer and cholangiocarcinoma, or CCA. We have completed Phase 1/2 and observational registry studies in locally advanced pancreatic cancer, or LAPC, demonstrating safety and a median overall survival rate of 27.9 months in patients treated with RenovoGem and radiation versus expected survival rate (historical control) of 12-15 months in patients only receiving intravenous (IV) systemic chemotherapy dosed at 1,000mg/m². RenovoGem is currently being evaluated in a Phase 3 registration Investigational New Drug, or IND, clinical trial and we expect to report data from a planned interim data readout in the second half of 2022. As we prepared the FDA Pre-Investigational New Drug, or Pre-IND, application for our second indication, hilar cholangiocarcinoma (cancer that occurs in the bile ducts that lead out of the liver and join with the gallbladder, also called hilar cholangiocarcinoma, or HCCA), we discovered an alternative approach to treat this patient population with our therapy platform

treatment pathway for HCCA during the first half of 2022 and launching the appropriate study in 2023. In addition, we may evaluate RenovoGem in other indications, potentially including locally advanced lung cancer, locally advanced uterine tumors, and glioblastoma, and develop other chemotherapeutic agents for intra-arterial delivery via RenovoCath

Since our inception, we have devoted substantially all of our efforts to developing our cancer therapy platform and product candidates, raising capital and organizing and staffing our company. To date, we have financed our operations with proceeds from the issuance of convertible preferred stock and convertible notes. Through July 31, 2021, we have financed our operations primarily through issuance of convertible preferred stock with net proceeds of \$11.8 million, including \$5.0 million through the issuance of convertible notes and a loan of \$140,000 pursuant to the Paycheck Protection Program under the CARES Act, which was forgiven in February 2021. In August 2021, we completed our IPO with aggregate gross proceeds of \$16.7 million. We paid underwriting discounts and commissions of \$1.3 million, and we also incurred expenses of \$0.8 million in connection with the offering. As a result, the net offering proceeds to us, after deducting underwriting discounts and commissions and offering expenses, were \$14.6 million.

We have incurred significant operating losses and generated negative cash flows from operations since our inception. As of September 30, 2021, we had cash and cash equivalents of \$17.7 million. We also had net losses of \$1.5 million and \$4.0 million for the three and nine months ended September 30, 2021, respectively and \$1.1 million and \$2.9 million for the three and nine months ended September 30, 2020, respectively. As of September 30, 2021, we had an accumulated deficit of \$18.9 million. We expect to continue to incur significant expenses, increasing operating losses and negative cash flows from operations in 2021 and for the foreseeable future. We do not expect to generate revenues from product sales unless and until we successfully complete development and obtain regulatory approval for one or more product candidates. We expect that our expenses will increase substantially in connection with our ongoing research and development activities, particularly as we:

- Advance clinical development of RenovoGem and our platform technology by continuing to enroll patients in our ongoing TIGeR-PaC Phase 3 clinical trial, expanding
 the number of clinical trials including our planned clinical trial in HCCA, and advancing RenovoGem through preclinical and clinical development in additional
 indications;
- Hire additional research, development, engineering, and general and administrative personnel;
- Maintain, expand, enforce, defend, and protect our intellectual property portfolio; and
- Expand our operational, financial and management systems and increase personnel, including personnel to support our clinical development, manufacturing and commercialization efforts and our operations as a public company.

20

In addition to the variables described above, if and when any of our product candidates successfully complete development, we will incur substantial additional costs associated with establishing a sales, marketing, medical affairs and distribution infrastructure to commercialize products for which we may obtain marketing approval, regulatory filings, marketing approval, and post-marketing requirements, in addition to other commercial costs. We cannot reasonably estimate these costs at this time.

As a result, we will need significant additional funding to support our continuing operations. Until such time, if ever, as we can generate substantial product revenue, we expect to finance our cash needs through equity issuances, debt financings and collaborations, licenses or other similar arrangements. We currently have no credit facility or committed sources of capital. To the extent that we raise additional capital through the future sale of equity or debt, the ownership interests of our stockholders will be diluted and the terms of these securities may include liquidation or other preferences that adversely affect the rights of our existing common stockholders. If we raise additional funds through the issuance of debt securities, these securities could contain covenants that would restrict our operations. We may require additional capital beyond our currently anticipated amounts and additional capital may not be available on reasonable terms, or at all. If we raise additional funds through collaboration arrangements or other strategic transactions in the future, we may have to relinquish valuable rights to our technologies or future revenue streams or grant licenses on terms that may not be favorable to us. If we are unable to raise additional funds through equity or debt financings when needed, we may be required to delay, limit, reduce or terminate development or future commercialization efforts.

Impact of COVID-19

In December 2019, a novel strain of coronavirus, which causes the disease known as COVID-19, was reported to have surfaced in Wuhan, China. Since then, COVID-19 has spread globally. In March 2020, the World Health Organization declared the COVID-19 outbreak a pandemic. The ongoing COVID-19 pandemic has caused significant disruption in the international and U.S. economies and financial markets. The spread of COVID-19 has caused illness, quarantines, cancellation of events and travel, business and school shutdowns, reduction in business activity and financial transactions, labor shortages, supply chain interruptions and overall economic and financial market instability.

In response to public health directives and orders and to help minimize the risk of the virus to employees, we have taken precautionary measures, including implementing work-from home and/or hybrid work policies for our employees. The COVID-19 pandemic also has negatively affected, and we expect will continue to negatively affect, our clinical studies. For example, we have faced challenges in conducting our Phase 3 clinical trial, including recruiting subjects and accommodating patient visits. Additionally, our service providers and their operations may be disrupted, temporarily closed or experience worker or supply shortages, which could result in additional disruptions or delays in shipments of purchased materials or the continued development of our product candidates. To date, we have not suffered material supply chain disruptions.

We are not able to estimate the duration of the pandemic and the potential impact on our business. As the global pandemic of COVID-19 continues to evolve, it could continue to result in significant long-term disruption of global financial markets, reducing our ability to raise additional capital when needed and on acceptable terms, if at all, which could negatively affect our liquidity. The extent to which the COVID-19 pandemic impacts our clinical development and regulatory efforts will depend on future developments that are highly uncertain and cannot be predicted with confidence, such as the duration of the outbreak, vaccine and infection rates, new travel restrictions, quarantines and social distancing requirements in the United States and other countries, business closures or business disruptions and the effectiveness of actions taken in the United States and other countries to contain and treat the virus. We will continue to monitor the COVID-19 situation closely.

Components of Our Results of Operations

Revenue

We have not generated any revenue from product sales and do not expect to generate any revenue from the sale of products for several years, if at all. If our development efforts for our current or future product candidates are successful and result in marketing approval or collaboration or license agreements with third parties, we may generate revenue in the future from a combination of product sales or payments from collaboration or license agreements.

Research and Development

Research and development expenses consist of costs related to the research and development of our platform technology. Clinical trial costs are a significant component of research and development expenses and include costs associated with third-party contractors. We outsource a substantial portion of our clinical trial activities, utilizing the service of third-party clinical trial sites and contract research organizations to assist us with the execution of our clinical trials. In addition, we have FDA 510(k) clearance for the RenovoCath delivery device, which comprises part of the RenovoGem product. Accordingly, we are able to charge our clinical trial sites for the RenovoCath delivery devices have been adequate to cover our direct manufacturing costs. Any payments we receive from clinical trial sites as consideration for use of RenovoCath delivery devices offset our research and development expenses. We expect our research and development expenses to increase for the foreseeable future as we continue the development of our product candidates and enroll subjects in our ongoing clinical Phase 3 trial, initiate future clinical trials and pursue regulatory approval of our product candidates. It is difficult to predict with any certainty the duration and costs of completing our current or future clinical trials of our product candidates or if, when or to what extent we will achieve regulatory approval and generate revenue from the commercialization and sale of our product candidates. The duration, costs and timing of clinical trials and other development of our product candidates will depend on a variety of factors, including uncertainties in clinical trial enrollment, timing and extent of future clinical trials, development of new product candidates and significant and changing government regulation. We may never succeed in achieving regulatory approval for any of our product candidates.

Our research and development expenses include:

- · expenses incurred under agreements with clinical trial sites, contract research organizations, and consultants that conduct our clinical trials,
- · costs of acquiring and developing clinical trial materials,
- · personnel costs, including salaries, benefits, bonuses, and stock-based compensation for employees engaged in preclinical and clinical research and development,
- costs related to compliance with regulatory requirements,
- travel expenses, and
- facilities, insurance, and other allocated expenses which include direct and allocated expenses for rent, insurance and other general overhead costs.

Research and development costs are expensed as incurred. Costs for certain development activities, such as clinical trials and preclinical studies, are recognized based on evaluation of progress to completion of specific tasks using data such as subject enrollment, clinical site activations or information provided to us by third party vendors.

Due to the impact of the COVID-19 pandemic and work-from-home policies and other operational limitations mandated by federal, state, and local governments as a result of the pandemic, certain of our research and development activities have been delayed and may be further delayed until such operational limitations are lifted.

General and Administrative

General and administrative expenses consist of salaries, benefits, and stock-based compensation for personnel in executive, finance and administrative functions, professional services and associated costs related to accounting, tax, audit, legal, intellectual property and other matters, consulting costs, conferences, travel and allocated expenses for rent, insurance and other general overhead costs. Following the listing of our common stock on Nasdaq, we expect to continue to incur additional expenses as a result of operating as a public company, including costs to comply with the rules and regulations of the SEC and Nasdaq listing standards and increased expenses in the areas of insurance, professional services and investor relations. As a result, we expect our general and administrative expenses to increase for the foreseeable future. General and administrative expenses are expensed as incurred.

22

Other Income (Expenses), Net

Interest Income (Expense) Net

Interest expense consists of charges relating to the amortization of the debt discount and debt issuance costs as well as interest on amounts outstanding on our convertible notes. In March 2020, we completed the offering of \$3.0 million of convertible notes, the 2020 Convertible Notes, that provided for the automatic conversion into shares of our common stock and warrants at the closing of our IPO at a 20% discount to the public offering price of the units. In April 2021, we completed the offering of \$2.0 million of convertible notes, the 2021 Convertible Notes, that provided for the automatic conversion into shares of our common stock and warrants at the closing of our IPO at a 12.5% discount to the public offering price of the units.

Interest income is earned from cash deposited in our money market account.

Other Income, Net

Other income, net primarily represents the mark-to-market adjustment on the derivative liability resulting from the 2020 and 2021 Convertible Notes. Upon the completion of our IPO in August 2021, the 2020 and 2021 Convertible Notes were converted into units consisting of (a) one share of common stock and (b) one five-year warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share.

Gain (Loss) on Loan Extinguishment

The gain (loss) on loan extinguishment represents the loss from the conversion and settlement of our 2020 and 2021 Convertible Notes as well as the gain on loan extinguishment from the forgiveness and cancellation of our PPP loan.

Income Tax Expense

We account for income taxes using the asset and liability method. Under this method, deferred income tax assets and liabilities are recorded based on the estimated future tax effects of differences between the financial statement and income tax basis of existing assets and liabilities. Deferred income tax assets and liabilities are recorded net and classified as noncurrent on the balance sheets. A valuation allowance is provided against our deferred income tax assets when their realization is not reasonably assured.

We are subject to income taxes in the federal and state jurisdictions. Tax regulations within each jurisdiction are subject to the interpretation of the related tax laws and regulations and require significant judgment to apply. In accordance with the authoritative guidance on accounting for uncertainty in income taxes, we recognize tax liabilities for uncertain tax positions when it is more likely than not that a tax position will not be sustained upon examination and settlement with various taxing authorities. Liabilities for uncertain tax positions are measured based upon the largest amount of benefit that is more-likely-than-not (greater than 50%) of being realized upon settlement. Our policy is to recognize interest and/or penalties related to income tax matters in income tax expense.

On March 27, 2020, the CARES Act was enacted. The CARES Act includes several significant provisions for corporations, including the usage of net operating losses, interest deductions and payroll benefits. Corporate taxpayers may carryback net operating losses, or NOLs, originating during 2018 through 2020 for up to five years.

Results of Operations

The following table summarizes the significant components of our results of operations for the periods presented (in thousands):

	Three Months Ended September 30,			Nine Months Ended September 30,				
		2021		2020		2021		2020
Statements of Operations Data:								
Operating expenses:								
Research and development	\$	767	\$	727	\$	1,938	\$	1,934
General and administrative		628		183		1,377		631
Total operating expenses		1,395		910		3,315		2,565
Loss from operations	·	(1,395)		(910)		(3,315)		(2,565)
Other income (expense), net								
Interest expense, net		(208)		(186)		(835)		(355)
Other income, net		170		-		119		-
Gain (loss) on loan extinguishment		(78)		-		62		-
Total other expense, net		(116)		(186)		(654)		(355)
Net loss	\$	(1,511)	\$	(1,096)	\$	(3,969)	\$	(2,920)

Comparison of the Three Months Ended September 30, 2021 and 2020

The following table summarizes our results of operations for the three months ended September 30, 2021 and 2020 (in thousands, except percentages):

	Three Months Ended September 30,				Change			
	2021			2020	\$		%	
		(unau	dited)					
Operating expenses:								
Research and development	\$	767	\$	727	\$	40	6%	
General and administrative		628		183		445	243%	
Total operating expenses		1,395		910		485	53%	
Loss from operations		(1,395)		(910)		(485)	53%	
Other income (expense), net								
Interest expense, net		(208)		(186)		(22)	12%	
Other income, net		170		<u>-</u>		170	100%	
Loss on loan extinguishment		(78)		-		(78)	100%	
Total other expense, net		(116)		(186)		70	(38)%	
Net loss	\$	(1,511)	\$	(1,096)	\$	(415)	38%	

Research and Development

Research and development expenses were \$0.8 million for the three months ended September 30, 2021, an increase of 6%, compared to \$0.7 million for the prior year quarter. This increase was due to an increase in clinical development personnel costs, including an allocation of executive-level clinical support from general and administrative of \$0.1 million. The increase was partially offset by reduced spending in leased software costs.

General and Administrative Expenses

General and administrative expenses were \$0.6 million for the three months ended September 30, 2021, an increase of 243%, or \$0.4 million, compared to \$0.2 million for the prior year quarter. This increase was primarily due to higher professional and consulting costs related to preparing for our IPO in August 2021 of \$0.2 million, including salaries and related benefits of \$0.1 million, and insurance costs primarily related to the purchase of Directors and Officers Liability Insurance of \$0.1 million.

24

Interest Expense, Net

Interest expense, net for the three months ended September 30, 2021, includes both the amortization of the discount and debt issuance costs, including the interest associated with the 2020 and 2021 Convertible Notes of \$0.2 million. Interest expense, net for the three months ended September 30, 2020, includes both the amortization of the discount and debt issuance costs, including the interest associated with the 2020 Convertible Notes of \$0.2 million.

Other Income, Ne

Other income, net for the three months ended September 30, 2021 was \$0.2 million and primarily includes the mark-to-market adjustment on the derivative liability resulting from the 2020 and 2021 Convertible Notes. There was no other income or expense, net for the three months ended September 30, 2020.

Loss on Loan Extinguishment

The loss on loan extinguishment, representing the unamortized debt discount on our 2020 and 2021 Convertible Notes, of \$0.1 million during the three months ended September 30, 2021 resulted from the conversion of the 2020 and 2021 Convertible Notes into units (consisting of (a) one share of common stock and (b) one five-year warrant to purchase one share of common stock) upon the completion of our IPO in August 2021.

Comparison of the Nine Months Ended September 30, 2020 and 2021

The following table summarizes our results of operations for the nine months ended September 30, 2021 and 2020 (in thousands, except percentages):

Nine Months Ended

	Septen	nber 30,	C	hange
	2021	2020	\$	%
	(unau	ıdited)		
Operating expenses:				
Research and development	\$ 1,938	\$ 1,934	\$ 4	-%
General and administrative	1,377	631	746	118%
Total operating expenses	3,315	2,565	750	29%
Loss from operations	(3,315)	(2,565)	(750)	29%
Other income (expense), net				
Interest expense, net	(835)	(355)	(480)	135%
Other income, net	119	-	119	100%
Gain on loan extinguishment	62	<u>-</u> _	62	100%
Total other expense, net	(654)	(355)	(299)	84%
Net loss	\$ (3,969)	\$ (2,920)	\$ (1,049)	36%

Research and Development

Research and development expenses were \$1.9 million for the nine months ended September 30, 2021 and 2020. Increases in research and development expenses consisted of clinical development personnel costs and related benefits, as well as the allocation of executive-level clinical support from general and administrative of \$0.1 million. These increases were offset by lower leased software costs and payments received from clinical sites for the use of RenovoCath delivery devices in our Phase 3 clinical trial of \$0.1 million.

General and Administrative Expenses

General and administrative expenses were \$1.4 million for the nine months ended September 30, 2021, an increase of 118%, or \$0.7 million, compared to \$0.6 million for the prior year period. This increase was due to higher professional and consulting services costs of \$0.6 million related to preparing for our IPO in August 2021, including an increase in spending in personnel costs of \$0.2 million and insurance costs of \$0.1 million related to the purchase of Directors and Officers Liability Insurance. These increases were partially offset by a \$0.1 million decrease in expense as a result of allocating executive-level clinical support costs to research and development expense.

25

Interest Expense, Net

Interest expense, net for the nine months ended September 30, 2021 includes both the stated interest on the 2020 and 2021 Convertible Notes of 5% per annum, or \$0.1 million, as well as the amortization of the discount and debt issuance costs associated with the 2020 and 2021 Convertible Notes of \$0.7 million. Interest expense for the nine months ended September 30, 2020 includes both the stated interest on the 2020 Convertible Notes of 5% per annum, or \$0.1 million, as well as the amortization of the debt discount and debt issuance costs of \$0.3 million.

Other Income. Net

Other income, net for the nine months ended September 30, 2021 was \$0.1 million and represents the mark-to-market adjustment on the derivative liabilities resulting from the 2020 and 2021 Convertible Notes. There was no other income, net for the nine months ended September 30, 2020.

On March 1, 2021, the Company entered into an amendment to the 2020 Convertible Notes which extended the maturity date of the 2020 Convertible Notes from March 31, 2021 to October 30, 2021 and provided for the conversion of the 2020 Convertible Notes into shares of the Company's common stock upon a Qualified Financing that is an IPO. No other terms to the 2020 Convertible Notes were amended. This amendment was accounted for as a troubled debt restructuring pursuant to FASB ASC Topic 470-60, "Troubled Debt Restructurings by Debtors." As the future undiscounted cash flows of the 2020 Convertible Notes were greater than their carrying amount, the carrying amount was not adjusted and no gain was recognized as a result of the modification of terms.

Gain on Loan Extinguishment

The gain on loan extinguishment of \$0.1 million during the nine months ended September 30, 2021 represents a loss of \$0.1 million on the automatic conversion of the 2020 and 2021 Convertible Notes upon completion of our IPO offset by the forgiveness and cancellation of our PPP loan of \$0.1 million.

Liquidity and Capital Resources

For the three and nine months ended September 30, 2021 we incurred net losses of \$1.5 million and \$4.0 million, respectively. As of September 30, 2021, we had an accumulated deficit of \$18.9 million. We expect to incur additional losses and increased operating expenses in future periods. Since our inception, our primary sources of liquidity have been the sale and issuance of convertible preferred stock, convertible notes and common stock, including in our IPO, and from the exercise of warrants.

As of September 30, 2021 and December 31, 2020, we had \$17.7 million and \$1.8 million in cash and cash equivalents, respectively. During the nine months ended September 30, 2021, we used \$3.4 million of cash in operations. Our primary requirements for liquidity have been to fund our clinical trial activity and general corporate and working capital needs. In August 2021, we completed our IPO for aggregate gross proceeds of \$16.7 million. We paid underwriting discounts and commissions of \$1.3 million, and we also incurred expenses of \$0.8 million in connection with the offering. As a result, the net offering proceeds to us, after deducting underwriting discounts and commissions and offering expenses, were \$14.6 million. In February 2021, we received notification and confirmation from Silicon Valley Bank that our PPP loan of \$0.1 million, has been forgiven in its entirety and automatically cancelled by the U.S. Small Business Administration.

Based on our current operating plan, we expect that our current cash and cash equivalents as of September 30, 2021 will be sufficient to fund our operating, investing and financing cash flow needs for at least 12 months from the issuance date of these interim financial statements. We intend to raise additional capital through equity offerings and/or debt financings. If we are unable to raise capital when needed or on attractive terms, we would be forced to delay, reduce or eliminate our clinical trials or other operations. If any of these events occur, our ability to achieve our operational goals would be adversely affected. Our future capital requirements and the adequacy of available funds will depend on many factors, including those described in "Risk Factors." Depending on the severity and direct impact of these factors on us, we may be unable to secure additional financing to meet our operating requirements on terms favorable to us, or at all.

Sources of Liquidity

Since our inception, we have not generated any revenue from product sales and we have incurred significant operating losses and negative cash flows from our operations. We do not have any products that have achieved regulatory marketing approval and we do not expect to generate revenue from sales of any product candidates for several years, if ever.

We have financed our operations primarily through the issuance and sale of convertible preferred stock and convertible debt. Through the date of this report, we have raised an aggregate of \$17.0 million of gross proceeds from private placements of our equity and convertible debt securities, net proceeds of \$14.6 million from our IPO in August 2021 and \$2.7 million in proceeds from the exercise of warrants and of the underwriter's exercise of its over-allotment option to purchase common stock warrants. We also received \$0.1 million from a loan under the PPP which was forgiven in February 2021. As of September 30, 2021, we had cash and cash equivalents of \$17.7 million and an accumulated deficit of \$18.9 million.

Cash Flows

Our primary uses of cash are to fund our operations including research and development and general and administrative expenses. We will continue to incur operating losses in the future and expect that our research and development and general and administrative expenses will continue to increase as we continue our research and development efforts with respect to clinical development of our product candidates and further develop our platform. We expect that we will use a substantial portion of the net proceeds of this offering, in combination with our existing cash and cash equivalents, for these purposes and for the increased expenses associated with being a public company. Cash used to fund operating expenses is impacted by the timing of when we pay expenses, as reflected in the change in our outstanding accounts payable and accrued expenses.

The following table summarizes our cash flows for the periods indicated (in thousands):

		Nine Months Ended September 30,						
	2021			2020				
		(unaud	dited)					
Net cash provided by (used in):								
Operating activities	\$	(3,358)	\$	(2,558)				
Investing activities		(15)		-				
Financing activities		19,303		2,746				
Increase in cash and cash equivalents	\$	15,930	\$	188				

Net Cash Used in Operating Activities

Cash used in operating activities for the nine months ended September 30, 2021 reflected a net loss of \$4.0 million and a net change in our operating assets and liabilities of \$0.1 million, offset by non-cash charges of \$0.5 million consisting primarily of amortization of a debt discount and gain/loss on loan/convertible debt extinguishments. Net cash used in operating activities for the nine months ended September 30, 2020 reflected a net loss of \$2.9 million and non-cash charges of \$0.4 million representing amortization of debt discount and stock-based compensation expense.

Net Cash Used in Investing Activities

Net cash used in investing activities for the nine months ended September 30, 2021 consisted of capital expenditures made for leasehold improvements to our new office space. There were no investment activities in the comparable nine months ended September 30, 2020.

Cash Provided by Financing Activities

Net cash provided by financing in the nine months ended September 30, 2021 was \$19.3 million, consisting of net proceeds of \$14.6 million from the issuance of common stock in our IPO, \$2.0 million from the issuance of convertible notes and \$2.8 million from exercise of warrants and stock options. Net cash provided by financing activities in the nine months ended September 30, 2020 was \$2.7 million, consisting of \$2.6 million in proceeds from the issuance of convertible notes and \$0.2 million in proceeds from the PPP loan and the exercise of the Series A-1 warrants and stock options.

Contractual Obligations and Other Commitments

As of the date of this report, we have no contractual obligations or other commitments. In August 2021, the 2020 and 2021 Convertible Notes, including accrued interest, of \$5.3 million were converted to common shares upon the completion of our IPO. In February 2021, the Company received notification and confirmation from Silicon Valley Bank that its PPP loan of \$0.1 million, had been forgiven in its entirety and automatically cancelled by the U.S. Small Business Administration. There have been no other significant changes in our contractual obligations or other commitments as of September 30, 2021.

27

Critical Accounting Policies and Significant Judgments and Estimates

The accompanying management's discussion and analysis of our financial condition and results of operations are based upon our unaudited condensed interim financial statements and the related disclosures, which have been prepared in accordance with accounting principles generally accepted in the United States or GAAP. The preparation of these unaudited condensed interim financial statements requires us to make estimates, assumptions and judgments that affect the reported amounts in our unaudited condensed interim financial statements and accompanying notes. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions. To the extent that there are material differences between these estimates and actual results, our future financial statement presentation, financial condition, results of operations and cash flows will be affected. While our significant accounting policies are described in the notes to our financial statements included elsewhere in this report, we believe that the following critical accounting policies are most important to understanding and evaluating our reported financial results because they require us to make estimates, assumptions and judgments about matters that are inherently uncertain.

A critical accounting policy is defined as one that is both material to the presentation of our financial statements and requires management to make difficult, subjective, or complex judgments that could have a material effect on our financial condition and results of operations. Specifically, critical accounting estimates have the following attributes: (i) we are required to make assumptions about matters that are highly uncertain at the time of the estimate; and (ii) different estimates we could reasonably have used, or changes in the estimate that are reasonably likely to occur, would have a material effect on our financial condition or results of operations.

Clinical Trial Expenses

We make payments in connection with our Phase 3 clinical trial under contracts with clinical trial sites and contract research organizations that support conducting and managing clinical trials. The financial terms of these agreements are subject to negotiation and vary from contract to contract and may result in uneven payment flows. Generally, these agreements set forth the scope of work to be performed at a fixed fee, unit price or on a time and materials basis. A portion of the obligation to make payments under these contracts depends on factors such as the successful enrollment or treatment of patients or the completion of other clinical trial milestones.

Expenses related to clinical trials are accrued based on estimates and/or representations from service providers regarding work performed, including actual level of patient enrollment, completion of patient studies and progress of the clinical trials. Other incidental costs related to patient enrollment or treatment are accrued when reasonably certain. If the amounts we are obligated to pay under clinical trial agreements are modified (for instance, as a result of changes in the clinical trial protocol or scope of work to be performed), the accruals are adjusted accordingly. Revisions to contractual payment obligations are charged to expense in the period in which the facts that give rise to the revision become reasonably certain.

Stock-Based Compensation

We calculate the fair value of stock options using the Black-Scholes option pricing model, which incorporates various assumptions including the fair value of our common stock, volatility, expected life, and risk-free interest rate. Compensation related to service-based awards is recognized starting on the grant date on a straight-line basis over the vesting period, which is generally four years.

Determining the grant date fair value of options using the Black-Scholes option pricing model requires management to make assumptions and judgments. If any of the assumptions used in the Black-Scholes model change significantly, stock-based compensation for future awards may differ materially compared with the awards granted previously. The assumptions and estimates are as follows:

Fair Value of Common Stock—Given the absence of a public trading market prior to the Company's IPO, our Board of Directors considered numerous objective and subjective factors to determine the fair value of our common stock at each grant date. These factors included but were not limited to: (i) contemporaneous third-party valuations of common stock; (ii) the prices for preferred stock sold to outside investors; (iii) the rights and preferences of preferred stock relative to common stock; (iv) the lack of marketability of our common stock; (v) developments in the business; and (vi) the likelihood of achieving a liquidity event, such as an IPO or sale of the business, given prevailing market conditions. The methodology to determine the fair value of our common stock included estimating the fair value of the enterprise using the "backsolve" method, which is a market approach that assigns an implied enterprise value by accounting for all share class rights and preferences based on the latest round of financing. The total equity value implied was then applied in the context of an option pricing model to determine the value of each class of our shares.

28

Expected Term—The expected term represents the period that the stock-based awards are expected to be outstanding. We determine the expected term using the simplified method. The simplified method deems the term to be the average of the time-to-vesting and the contractual life of the options. For stock options granted to non-employees, the expected term equals the remaining contractual term of the option from the vesting date.

Expected Volatility—Given the absence of a public trading market, the expected volatility was estimated by taking the average historic price volatility for industry peers, consisting of several public companies in our industry that are either similar in size, stage, or financial leverage, over a period equivalent to the expected term of the awards.

Risk-Free Interest Rate—The risk-free interest rate is calculated using the average of the published interest rates of U.S. Treasury zero-coupon issues with maturities that are commensurate with the expected term.

Dividend Rate—The dividend yield assumption is zero as we have no plans to make dividend payments.

The determination of the fair value of our common stock after our IPO on August 30, 2021 is determined by the closing price of our common stock on the date of grant.

Convertible Instruments and Embedded Derivatives

We evaluate all our agreements to determine whether such instruments have derivatives or contain features that qualify as embedded derivatives. We account for certain redemption features that are associated with the terms of convertible notes as liabilities at fair value and adjust the instruments to their fair value at the end of each reporting period. For derivative financial instruments that are accounted for as liabilities, the derivative instrument is initially recorded at its fair value and is then re-valued at each reporting date, with changes in the fair value reported in other income (expense), net in the statements of operations. Derivative instrument liabilities are classified in the balance sheets as current or non-current based on whether net-cash settlement of the derivative instrument could be required within 12 months of the balance sheet date. As of December 31, 2020, our derivative financial instruments were related to the 2020 and 2021 Convertible Notes, which contained certain redemptive features. On August 30, 2021, we completed our IPO which triggered the automatic conversion of all outstanding Convertible Notes and accrued interest into shares of common stock.

Emerging Growth Company and Smaller Reporting Company Status

We are an "emerging growth company" as defined in the JOBS Act. Under the JOBS Act, companies have extended transition periods available for complying with new or revised accounting standards. We have elected this exemption to delay adopting new or revised accounting standards. We will remain an emerging growth company until the earlier of (1) December 31, 2026, (2) the last day of the fiscal year in which we have total annual gross revenues of at least \$1.07 billion, (3) the date on which we are deemed to be a "large accelerated filer" as defined in Rule 12b-2 under the Exchange Act, or (4) the date on which we have issued more than \$1.0 billion in non-convertible debt securities during the prior three-year period. An emerging growth company may take advantage of specified reduced reporting requirements and is relieved of certain other significant requirements that are otherwise generally applicable to public companies. As an emerging growth company,

- we may present only two years of audited financial statements, plus unaudited interim condensed financial statements for any interim period, and related Management's Discussion and Analysis of Financial Condition and Results of Operations;
- we may avail ourselves of the exemption from the requirement to obtain an attestation and report from our auditors on the assessment of our internal control over financial reporting pursuant to the Sarbanes-Oxley Act;
- we may provide reduced disclosure about our executive compensation arrangements; and
- we do not require stockholder non-binding advisory votes on executive compensation or golden parachute arrangements.

We have elected to take advantage of certain reduced disclosure obligations in this Quarterly Report on Form 10-Q and may elect to take advantage of other reduced reporting requirements in future filings. As a result, the information that we provide to our stockholders may be different than you might receive from other public reporting companies in which you hold equity interests.

We are also a "smaller reporting company," meaning that the market value of our stock held by non-affiliates plus the proposed aggregate amount of gross proceeds to us as a result of this offering is less than \$700.0 million and our annual revenue is less than \$100.0 million during the most recently completed fiscal year. We may continue to be a smaller reporting company if either (1) the market value of our stock held by non-affiliates is less than \$250.0 million or (2) our annual revenue is less than \$100.0 million during the most recently completed fiscal year and the market value of our stock held by non-affiliates is less than \$700.0 million. If we are a smaller reporting company at the time we cease to be an emerging growth company, we may continue to rely on exemptions from certain disclosure requirements that are available to smaller reporting companies. Specifically, as a smaller reporting company we may choose to present only the two most recent fiscal years of audited financial statements in our Annual Report on Form 10-K and, like emerging growth companies, smaller reporting companies have reduced disclosure obligations regarding executive compensation

A description of recently issued and adopted accounting pronouncements that may potentially impact our financial position and results of operations is disclosed in Note 2, Summary of Significant Accounting Policies to our interim financial statements included elsewhere in this report.

Off-Balance Sheet Arrangements

During the periods presented, we did not have, nor do we currently have, any off-balance sheet arrangements, as defined in the rules and regulations of the SEC.

20

Item 3: Quantitative and Qualitative Disclosures about Market Risk

We are exposed to market risks in the ordinary course of our business. These risks primarily include interest rate risks, foreign currency exchange, risks and inflation risks. Periodically, we maintain deposits in accredited financial institutions in excess of federally insured limits. We deposit our cash in financial institutions that we believe have high credit quality and have not experienced any losses on such accounts and do not believe we are exposed to any unusual credit risk beyond the normal credit risk associated with commercial banking relationships.

Interest Rate Risk

Our cash and cash equivalents consisted primarily of cash on hand and marketable securities at September 30, 2021 and December 31, 2020. The fair value of our cash and cash equivalents would not be significantly affected by either an increase or decrease in interest rates.

Our exposure to risks related to interest rates is minimal. The interest rates for our 2020 and 2021 Convertible Notes were fixed rates.

Foreign Currency Exchange Risk

We are not currently exposed to significant market risk related to changes in foreign currency exchange rates. However, we have contracted with and may continue to contract with vendors such as contract research organizations and clinical trial sites that are in Europe. We may be subject to fluctuations in foreign currency rates in connection with certain of these agreements. Transactions denominated in currencies other than the U.S. dollar are recorded based on exchange rates at the time such transactions arise. While we have not engaged in hedging our foreign currency transactions to date, we may evaluate the costs and benefits of initiating such a program and may in the future, hedge selected significant transactions denominated in currencies other than the U.S. dollar as we expand our clinical trial sites globally.

Inflation Risk

Inflation generally affects us by increasing our labor and clinical trial costs. We do not believe that inflation had a material effect on our business, financial condition or results of operations during the nine months ended September 30, 2021 and the year ended December 31, 2020.

30

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial and Accounting Officer, we conducted an evaluation of the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act) as of the end of the fiscal quarter ended September 30, 2021. Based on this evaluation, our Chief Executive Officer and Chief Financial and Accounting Officer have concluded that, during the period covered by this Quarterly Report, our disclosure controls and procedures were not effective due to our previously identified material weakness in internal control over financial reporting. Notwithstanding the identified material weaknesses, management, including our Chief Executive Officer and Chief Financial and Accounting Officer, believes the financial statements included in this Quarterly Report on Form 10-Q are fairly presented, in all material respects, in accordance with U.S. GAAP.

Disclosure controls and procedures are designed to ensure that information required to be disclosed by us in our Exchange Act reports is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our principal executive officer and principal financial and accounting officer or persons performing similar functions, as appropriate, to allow timely decisions.

Previously Identified Material Weakness and Plans to Remediate

In preparation for our IPO, we identified a material weakness in our internal control over financial reporting related to our control environment. Specifically, we have determined that we have not maintained adequate formal accounting policies, processes and controls related to complex transactions as a result of a lack of finance and accounting staff with the appropriate GAAP technical expertise needed to identify, evaluate and account for complex and non-routine transactions. We also determined that we have not maintained sufficient staffing or written policies and procedures for accounting and financial reporting, which contributed to the lack of a formalized process or controls for management's timely review and approval of financial information. More specifically, we have determined that our financial statement close process includes significant control gaps mainly driven by the small size of our accounting and finance staff and, as a result, a significant lack of appropriate segregation of duties. A material weakness is a deficiency, or combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected and corrected on a timely basis.

We are in the process of implementing a number of measures to address the material weakness that has been identified including: (i) engaging additional accounting and financial reporting personnel with US GAAP, and SEC reporting experience, (ii) developing, communicating and implementing an accounting policy manual for our accounting and financial reporting personnel for recurring transactions and period-end closing processes, and (iii) establishing effective monitoring and oversight controls for non-recurring and complex transactions to ensure the accuracy and completeness of our consolidated financial statements and related disclosures.

These additional resources and procedures are designed to enable us to broaden the scope and quality of our internal review of underlying information related to financial reporting and to formalize and enhance our internal control procedures. With the oversight of senior management and our audit committee, we have begun taking steps and plan to take additional measures to remediate the underlying causes of the material weaknesses.

We intend to complete the implementation of our remediation plan during 2022. Although we believe that our remediation plan will improve our internal control over financial reporting, additional time may be required to fully implement it and to make conclusions regarding the effectiveness of our internal control over financial reporting. Our management will closely monitor and modify, as appropriate, the remediation plan to eliminate the identified material weakness.

Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the quarter ended September 30, 2021 that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

Limitations on the Effectiveness of Controls

In designing and evaluating our disclosure controls and procedures, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives. In addition, the design of disclosure controls and procedures must reflect the fact that there are resource constraints, and that management is required to apply judgment in evaluating the benefits of possible controls and procedures relative to their costs.

31

PART II—OTHER INFORMATION

Item 1. Legal Proceedings

From time to time, we may become involved in legal proceedings arising in the ordinary course of our business. We are not currently a party to any material legal proceedings. Regardless of outcome, litigation can have an adverse impact on us due to defense and settlement costs, diversion of management resources, negative publicity, reputational harm and other factors.

Item 1A. Risk Factors

An investment in our securities involve a high degree of risk. You should carefully consider the risks described below, as well as the other information in this Quarterly Report on Form 10-Q, including our unaudited interim condensed financial statements and the related notes and the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations," and in our other public filings in evaluating our business. The occurrence of any of the events or developments described below could harm our business, financial condition, results of operations, growth prospects or stock price. In such an event, the market price of our common stock could decline, and you may lose all or part of your investment. Additional risks and uncertainties not presently known to us or that we currently deem immaterial may also impair our business operations and the market price of our common stock.

Risk Factors Summary

Investing in shares of our common stock involves a high degree of risk because our business is subject to numerous risks and uncertainties, including those outside of our control, that could cause our actual results to be harmed. The principal factors and uncertainties that make investing in shares of our common stock risky and impact our ability to execute on our business strategy include risks regarding the following, among others:

- We are a clinical stage biopharmaceutical company, have a limited operating history and have no products approved for commercial sale, which makes it difficult to
 evaluate our current business and predict our future success and viability.
- We have incurred significant net losses in each period since inception, and we expect to continue to incur net losses for the foreseeable future.
- We will need to raise substantial additional capital to develop and commercialize RenovoGem, and our failure to obtain funding when needed may force us to delay, reduce or eliminate our product development programs or collaboration efforts.
- Our product candidates' commercial viability remains subject to current and future preclinical studies, clinical trials, regulatory approvals, and the risks generally
 inherent in the development of a pharmaceutical product candidate. If we are unable to successfully advance or develop our product candidates, our business will be
 materially harmed
- If we do not achieve our projected development goals in the timeframes we announce and expect, our stock price may decline.
- Our product candidates may exhibit undesirable side effects when used alone or in combination with other approved pharmaceutical products or investigational new
 drugs, which may delay or preclude further development or regulatory approval or limit their use if approved.
- If the results of preclinical studies or clinical trials for our product candidates are negative, we could be delayed or precluded from the further development or commercialization of our product candidates, which could materially harm our business.
- If we are unable to satisfy regulatory requirements, we may not be able to commercialize our product candidates.
- If our product candidates are unable to compete effectively with marketed drugs targeting similar indications as our product candidates, our commercial opportunity will be reduced or eliminated.
- We may delay or terminate the development of our product candidates at any time if we believe the perceived market or commercial opportunity does not justify further investment, which could materially harm our business.
- · Our future success depends on our ability to retain our key personnel and to attract, retain, and motivate qualified personnel.
- If we are unable to protect our intellectual property effectively, we may be unable to prevent third parties from using our technologies, which would impair our competitive advantage.
- The patents issued to us may not be broad enough to provide any meaningful protection, one or more of our competitors may develop more effective technologies, designs, or methods without infringing our intellectual property rights and one or more of our competitors may design around our proprietary technologies.
- The market price of our common stock may be volatile and fluctuate substantially, which could result in substantial losses for our investors.

32

In addition, we face other risks and uncertainties that may materially affect our business prospects, financial condition, and results of operations. You should consider the risks discussed in "Risk Factors" and in our other public filings before investing in our securities.

Risks Related to Our Business, Financial Condition and Capital Requirements

We are a clinical stage biopharmaceutical company, have a limited operating history and have no products approved for commercial sale, which makes it difficult to evaluate our current business and predict our future success and viability.

We are a clinical stage biopharmaceutical company with a limited operating history upon which you can evaluate our business and prospects. We have no products approved for commercial sale and have not generated any revenue from product sales. We are developing a novel therapy platform, which is an unproven and highly uncertain undertaking and involves a substantial degree of risk. To date, we have not obtained marketing approval for any product candidates, manufactured a commercial scale product or arranged for a third-party to do so on our behalf, or conducted sales and marketing activities necessary for successful product commercialization. Our limited operating history as a company makes any assessment of our future success and viability subject to significant uncertainty. As a result, it may be more difficult for investors to accurately predict our likelihood of success and viability than it could be if we had a longer operating history.

We will encounter expenses, difficulties, complications, delays, and other known and unknown factors and risks frequently experienced by clinical stage companies in rapidly evolving fields, and we have not yet demonstrated an ability to successfully overcome such risks and difficulties. We also may need to transition from a company with a research and clinical development focus to a company capable of supporting commercial activities. We have not yet demonstrated an ability to successfully overcome such risks and difficulties, or to make such a transition. If we do not adequately address these risks and difficulties or successfully make such a transition, our business will suffer.

We have incurred significant net losses in each period since inception, and we expect to continue to incur net losses for the foreseeable future.

We are a clinical stage company and have incurred significant losses since our formation. As of September 30, 2021, we have an accumulated total deficit of approximately \$18.9 million. For the nine months ended September 30, 2021 and 2020, we had net losses of approximately \$4.0 million and \$2.9 million, respectively. To date, we have experienced negative cash flow from development of our product candidate, RenovoGem, our platform technology, Renovo Trans-Arterial Micro-Perfusion, or RenovoTAMP, and our RenovoCath delivery system. We have not generated any revenue from operations, and we expect to incur substantial net losses for the foreseeable future as we seek to further develop and commercialize RenovoGem. Because of the numerous risks and uncertainties associated with developing and commercializing RenovoGem, we are unable to predict the extent of any future losses or when we will attain profitability, if ever. Investors in our common stock must carefully consider the substantial challenges, risks and uncertainties inherent in the attempted development and commercialization of RenovoGem. We may never successfully commercialize RenovoGem, and our business may not be successful.

Our product candidates will require substantial additional development time and resources before we will be able to receive regulatory approvals, if any, and, if approved, to begin generating revenue from product sales. As a result, we expect that will be several years, if ever, before we receive approval to commercialize a product and generate revenue from product sales. Even if we succeed in receiving marketing approval for and commercializing one or more of our product candidates, we expect that we will continue to incur substantial expenses and increasing operating losses for the foreseeable future. The amount of our future net losses will depend, in part, on the level of our future expenditures and revenue. Moreover, our net losses may fluctuate significantly from quarter to quarter and year to year, such that a period-to-period comparison of our results of operations may not be a good indication of our future performance. If we are unable to generate significant revenue from RenovoGem or attain profitability, we will not be able to sustain operations.

33

We anticipate that our expenses will increase substantially if and as we:

- continue our research and discovery activities;
- continue the development of our proprietary technology platform;
- progress our current and any future product candidates through preclinical and clinical development;
- initiate and conduct additional preclinical, clinical, or other studies for our product candidates;
- work with our contract manufacturing organizations to manufacture our product candidates for our clinical trials;
- change or add additional contract manufacturers or suppliers;
- seek regulatory approvals and marketing authorizations for our product candidates;
- establish sales, marketing, and distribution infrastructure to commercialize any products for which we obtain approval;
- take steps to seek protection of our intellectual property and defend our intellectual property against challenges from third parties;
- obtain, expand, maintain, protect, and enforce our intellectual property portfolio;
- pursue any licensing or collaboration opportunities;
- attract, hire, and retain key and qualified personnel including clinical, scientific, management, and administrative personnel;
- provide additional internal infrastructure to support our continued research and development operations and any planned commercialization efforts in the future;
- experience any delays or encounter other issues related to our operations;
- implement operations, financial, and management information systems;
- meet the requirements and demands of being a public company; and
- defend against any product liability claims or other lawsuits related to our products.

Our prior operating losses and expected future net losses have had and will continue to have an adverse effect on our stockholders' equity, working capital, and our ability to fund our development efforts and achieve and maintain profitability. In any particular period, our operating results could be below the expectations of securities analysts or investors, or such analysts or investors could perceive these results to be negative, which could have a substantial adverse effect on the price of our common stock.

We will need to raise substantial additional capital to develop and commercialize RenovoGem, and our failure to obtain funding when needed may force us to delay, reduce or eliminate our product development programs or collaboration efforts.

As of September 30, 2021, our cash and cash equivalents and our working capital were approximately \$17.7 million. Due to our recurring operating losses and the expectation that we will continue to incur net losses in the future, we will be required to raise additional capital to complete the development and commercialization of our product candidates. We have historically financed our operations primarily through private sales of our equity, debt financings and the sale of common stock and warrants in our initial public offering. to fund our operations. To raise additional capital, we may seek to sell additional equity and/or debt securities, obtain a credit facility or other loan or enter into collaborations, licenses or other similar arrangements, which we may not be able to do on favorable terms, or at all. Our ability to obtain additional financing will be subject to a number of factors, including market conditions, fluctuations in interest rates, our operating performance and investor sentiment. If we are unable to raise additional

Risks Related to the Discovery, Development, and Commercialization of Our Product Candidates

Our product candidates' commercial viability remains subject to current and future preclinical studies, clinical trials, regulatory approvals, and the risks generally inherent in the development of a pharmaceutical product candidate. If we are unable to successfully advance or develop our product candidate, our business will be materially harmed.

In the near-term, failure to successfully advance the development of any of our product candidates may have a material adverse effect on us. To date, we have not successfully developed or commercially marketed, distributed, or sold any product candidate. The success of our business depends primarily upon our ability to successfully advance the development of our current and future product candidates through preclinical studies and clinical trials, have the product candidates approved for sale by the FDA or regulatory authorities in other countries, and ultimately have the product candidates successfully commercialized by us or a commercial partner. We cannot assure you that the results of our ongoing preclinical studies or clinical trials will support or justify the continued development of our product candidate, or that we will receive regulatory approval from the FDA, or similar regulatory authorities in other countries, to advance the development of our product candidates.

Our product candidates must satisfy rigorous regulatory standards of safety and efficacy before we can advance or complete their clinical development, or before they can be approved for sale. To satisfy these standards, we must engage in expensive and lengthy preclinical studies and clinical trials, develop acceptable manufacturing processes, and obtain regulatory approval. Despite these efforts, our product candidates may not:

- offer therapeutic or other medical benefits over existing drugs or other product candidates in development to treat the same patient population;
- be proven to be safe and effective in current and future preclinical studies or clinical trials;
- have the desired effects;
- be free from undesirable, unexpected or serious adverse effects;
- meet applicable regulatory standards;
- be capable of being formulated and manufactured in commercially suitable quantities and at an acceptable cost; or
- be successfully commercialized by us or by collaborators.

We cannot assure you that the results of late-stage clinical trials will be favorable enough to support the continued development of our product candidates. A number of companies in the pharmaceutical and biopharmaceutical industries have experienced significant delays, setbacks and failures in all stages of development, including late-stage clinical trials, even after achieving promising results in preclinical testing or early-stage clinical trials. Accordingly, results from completed preclinical studies and early-stage clinical trials of our product candidates may not be predictive of the results we may obtain in later-stage trials. Furthermore, even if the data collected from preclinical studies and clinical trials involving our product candidates demonstrate a favorable safety and efficacy profile, such results may not be sufficient to support the submission of a New Drug Application to obtain regulatory approval from the FDA in the U.S., or other similar regulatory agencies in other jurisdictions, which is required to market and sell the product. Even if we are successful in obtaining approval in one jurisdiction, we may not be successful in obtaining approval in any other jurisdictions. If we are unable to obtain approval for our product candidates in multiple jurisdictions, our business, financial condition, results of operations and our growth prospects could be negatively affected.

Our product candidates will require significant additional research and development efforts, the commitment of substantial financial resources, and regulatory approvals prior to advancing into further clinical development or being commercialized by us or collaborators. Additionally, changes in regulations as part of a response to the COVID-19 pandemic may require us to change the ways in which our clinical trials are conducted, or to discontinue clinical trials altogether, or which may result in unexpected costs. We cannot assure you that our product candidates will successfully progress through the drug development process or will result in commercially viable products. We do not expect our product candidates to be commercialized by us or collaborators for at least several years.

If we do not achieve our projected development goals in the timeframes we announce and expect, our stock price may decline.

From time to time, we estimate the timing of the anticipated accomplishment of various scientific, clinical, regulatory and other product development goals, which we sometimes refer to as milestones. These milestones may include the commencement or completion of scientific studies and clinical trials and the submission of regulatory filings. From time to time, we may publicly announce the expected timing of some of these milestones. All of these milestones are and will be based on numerous assumptions. The actual timing of these milestones can vary dramatically compared to our estimates, in some cases for reasons beyond our control. If we do not meet these milestones as publicly announced, our stock price may decline.

35

Our product candidates may exhibit undesirable side effects when used alone or in combination with other approved pharmaceutical products or investigational new drugs, which may delay or preclude further development or regulatory approval or limit their use if approved.

Throughout the drug development process, we must continually demonstrate the efficacy, safety and tolerability of our product candidates to obtain regulatory approval to further advance clinical development or to market them. Even if our product candidates demonstrate clinical efficacy, any unacceptable, adverse side effects or toxicities, when administered alone or in the presence of other pharmaceutical products, which can arise at any stage of development, may outweigh the potential benefits. In preclinical studies and clinical trials we have conducted to date, each of our product candidate's tolerability profile is based on studies and trials that have involved a small number of subjects or patients over a limited period of time. We may observe adverse or significant adverse events or drug-drug interactions in future preclinical studies or clinical trial candidates, which could result in the delay or termination of development, prevent regulatory approval, or limit market acceptance if ultimately approved.

If the results of preclinical studies or clinical trials for our product candidates, including those that are subject to existing or future license or collaboration agreements, are unfavorable or delayed, we could be delayed or precluded from the further development or commercialization of our product candidates, which could materially harm our business.

To further advance the development of, and ultimately receive regulatory approval to sell, our product candidates, we must conduct extensive preclinical studies and clinical trials to demonstrate their safety and efficacy to the satisfaction of the FDA or similar regulatory authorities in other countries, as the case may be. Preclinical studies and clinical trials are expensive, complex, can take many years to complete, and have highly uncertain outcomes. Delays, setbacks, or failures can occur at any time, or in any phase of preclinical or clinical testing, and can result from concerns about safety or toxicity, a lack of demonstrated efficacy or superior efficacy over other similar products that

have been approved for sale or are in more advanced stages of development, poor study or trial design, and issues related to the formulation or manufacturing process of the materials used to conduct the trials. The results of prior preclinical studies or clinical trials are not necessarily predictive of the results we may observe in later stage clinical trials. In many cases, product candidates in clinical development may fail to show desired safety, efficacy or tolerability characteristics despite having favorably demonstrated such characteristics in preclinical studies or earlier stage clinical trials.

In addition, we may experience numerous unforeseen events during, or as a result of, preclinical studies and the clinical trial process, which could delay or impede our ability to advance the development of, receive regulatory approval for, or commercialize our product candidate, including, but not limited to:

- communications with the FDA, or similar regulatory authorities in different countries, regarding the scope or design of a trial or trials;
- regulatory authorities, including an Institutional Review Board ("IRB") or Ethical Committee ("EC"), not authorizing us to commence or conduct a clinical trial at a prospective trial site;
- enrollment in our clinical trials being delayed, or proceeding at a slower pace than we expected, because we have difficulty recruiting patients or participants dropping out of our clinical trials at a higher rate than we anticipated;
- our third-party contractors, upon whom we rely for conducting preclinical studies, clinical trials and manufacturing of our trial materials, may fail to comply with regulatory requirements or meet their contractual obligations to us in a timely manner;
- having to suspend or ultimately terminate our clinical trials if participants are being exposed to unacceptable health or safety risks;
- IRBs, ECs, or regulators requiring that we hold, suspend or terminate our preclinical studies and clinical trials for various reasons, including non-compliance with regulatory requirements or due to the effects of the COVID-19 pandemic; and
- the supply or quality of drug material or the supply of our RenovoCath device necessary to conduct our preclinical studies or clinical trials being insufficient, inadequate, or unavailable, especially in light of the supply chain issues caused by the effects of the COVID-19 pandemic.

Even if the data collected from preclinical studies or clinical trials involving our product candidates demonstrate a favorable safety and efficacy profile, such results may not be sufficient to support the submission of an NDA to obtain regulatory approval from the FDA in the U.S., or other similar foreign regulatory authorities in foreign jurisdictions, which is required to market and sell the product.

36

If third party vendors upon whom we intend to rely on to conduct our preclinical studies or clinical trials do not perform or fail to comply with strict regulations, these studies or trials of our product candidate may be delayed, terminated, or fail, or we could incur significant additional expenses, which could materially harm our business.

We have limited resources dedicated to designing, conducting, and managing preclinical studies and clinical trials. We intend to rely on third parties, including clinical research organizations, consultants, and principal investigators, to assist us in designing, managing, monitoring and conducting our preclinical studies and clinical trials. We intend to rely on these vendors and individuals to perform many facets of the drug development process, including certain preclinical studies, the recruitment of sites and patients for participation in our clinical trials, maintenance of good relations with the clinical sites, and ensuring that these sites are conducting our trials in compliance with the trial protocol, including safety monitoring and applicable regulations. If these third parties fail to perform satisfactorily, or do not adequately fulfill their obligations under the terms of our agreements with them, we may not be able to enter into alternative arrangements without undue delay or additional expenditures, and therefore the preclinical studies and clinical trials of our product candidate may be delayed or prove unsuccessful. Further, the FDA, or other similar foreign regulatory authorities, may inspect some of the clinical sites participating in our clinical trials in the U.S., or our third-party vendors' sites, to determine if our clinical trials are being conducted according to Good Clinical Practices. If we or the FDA determine that our third-party vendors are not in compliance with, or have not conducted our clinical trials according to, applicable regulations we may be forced to delay, repeat, or terminate such clinical trials. Additionally, certain third parties with whom we engage or may engage for preclinical studies or clinical studies may have to adjust their operations or limit their capacity in light of the effects of the COVID-19 pandemic.

We have limited capacity for recruiting and managing clinical trials, which could impair our timing to initiate or complete clinical trials of our product candidate and materially harm our business.

We have limited capacity to recruit and manage the clinical trials necessary to obtain FDA approval or approval by other regulatory authorities. By contrast, larger pharmaceutical and biopharmaceutical companies often have substantial staff with extensive experience in conducting clinical trials with multiple product candidates across multiple indications. In addition, they may have greater financial resources to compete for the same clinical investigators and patients that we are attempting to recruit for our clinical trials. If potential competitors are successful in completing drug development for their product candidates and obtain approval from the FDA, they could limit the demand to participate in clinical trials of our product candidates. As a result, we may be at a competitive disadvantage that could delay the initiation, recruitment, timing, and completion of our clinical trials, as well as obtaining regulatory approvals, if at all, for our product candidates.

We, and our collaborators, if any, must comply with extensive government regulations in order to advance our product candidates through the development process and ultimately obtain and maintain marketing approval for our products in the U.S. and abroad.

The product candidates that we, or our collaborators, are developing or may develop require regulatory approval to advance through clinical development and to ultimately be marketed and sold and are subject to extensive and rigorous domestic and foreign government regulation. In the U.S., the FDA regulates, among other things, the development, testing, manufacture, safety, efficacy, record-keeping, labeling, storage, approval, advertising, promotion, sale, and distribution of pharmaceutical and biopharmaceutical products. Our product candidates are also subject to similar regulation by foreign governments to the extent we seek to develop or market them in those countries. We, or our collaborators, must provide the FDA and foreign regulatory authorities, if applicable, with preclinical and clinical data, as well as data supporting an acceptable manufacturing process, that appropriately demonstrate each of our product candidate's safety and efficacy before it can be approved for the targeted indications. Our product candidates have not been approved for sale in the U.S. or any foreign market, and we cannot predict whether we or our collaborators will obtain regulatory approval for any product candidates we are developing or plan to develop. The regulatory review and approval process can take many years, is dependent upon the type, complexity, novelty of, and medical need for the product candidate, requires the expenditure of substantial resources, and involves post-marketing surveillance and vigilance and potentially post-marketing studies or Phase 4 clinical trials. In addition, we or our collaborators may encounter delays in, or fail to gain, regulatory approval for any of our product candidates based upon additional governmental regulation resulting from future legislative, administrative action or changes in FDA's or other similar foreign regulatory authorities' policy or interpretation during the period of product development. Delays or failures in obtaining regulatory approval to advance any of our product

- adversely impact our ability to raise sufficient capital, if at all, to fund the development of our product candidates;
- adversely affect our ability to further develop or commercialize our product candidates;
- diminish any competitive advantages that we or our collaborators may have or attain; or
- adversely affect the receipt of potential milestone payments and royalties from collaborators, if any, from the sale of our products or product revenues in the future.

Furthermore, any regulatory approvals, if granted, may later be withdrawn. If we or our collaborators fail to comply with applicable regulatory requirements at any time, or if post-approval safety concerns arise, we or our collaborators may be subject to restrictions or a number of actions, including:

- delays, suspension, or termination of clinical trials related to our product candidates;
- refusal by regulatory authorities to review pending applications or supplements to approved applications;
- product recalls or seizures;
- suspension of manufacturing;
- · withdrawals of previously approved marketing applications; or
- fines, civil penalties, and criminal prosecutions.

Additionally, at any time we or our collaborators may voluntarily suspend or terminate the preclinical or clinical development of a product candidate, or withdraw any approved product from the market if we believe that it may pose an unacceptable safety risk to patients, or if the product candidate or approved product no longer meets our business objectives. The ability to develop or market a pharmaceutical product outside of the U.S. is contingent upon receiving appropriate authorization from the respective foreign regulatory authorities. Foreign regulatory approval processes typically include many, if not all, of the risks and requirements associated with the FDA regulatory process for drug development and may include additional risks. Additionally, results acceptable to support approval in one jurisdiction may be deemed inadequate by another regulatory authority to support regulatory approval in that other jurisdiction.

Clinical trials involve a lengthy and expensive process with an uncertain outcome, and results of earlier studies and trials may not be predictive of future trial results.

Our product candidates may not prove to be safe and efficacious in clinical trials and may not meet all the applicable regulatory requirements needed to receive regulatory approval. To receive regulatory approval for the commercialization of our product candidates, we must conduct, at our own expense, extensive preclinical testing and clinical trials to demonstrate safety and efficacy of our product candidate for the intended indication of use. Clinical testing is expensive, can take many years to complete, if at all, and its outcome is uncertain. Failure can occur at any time during the clinical trial process.

The results of preclinical studies and early clinical trials of new drugs do not necessarily predict the results of later-stage clinical trials. The design of our clinical trials is based on many assumptions about the expected effects of our product candidate, and if those assumptions are incorrect, they may not produce statistically significant results. Preliminary results may not be confirmed on full analysis of the detailed results of a clinical trial. Product candidates in later stages of clinical development may fail to show safety and efficacy sufficient to support intended use claims despite having progressed through earlier clinical testing. The data collected from clinical trials of our product candidates may not be sufficient to support the filing of an NDA or to obtain regulatory approval in the United States or elsewhere. Because of the uncertainties associated with drug development and regulatory approvals, we cannot determine if or when we will have an approved product for commercialization or achieve sales or profits.

Delays in clinical testing could result in increased costs to us and delay our ability to generate revenue.

We may experience delays in clinical testing of our product candidates. We do not know whether planned clinical trials will begin on time, will need to be redesigned or will be completed on schedule, if at all, especially in light of the effects of the COVID-19 pandemic. Clinical trials can be delayed for a variety of reasons, including pandemics, delays in obtaining regulatory approval to commence a clinical trial, in securing clinical trial agreements with prospective sites with acceptable terms, in obtaining IRB approval to conduct a clinical trial at a prospective site, in recruiting patients to participate in a clinical trial or in obtaining sufficient supplies of clinical trial materials, including RenovoCath. Many factors affect patient enrollment, including the size of the patient population, the proximity of patients to clinical sites, the eligibility criteria for the clinical trial, the existing body of safety and efficacy data with respect to the study drug, competing clinical trials, new drugs approved for the conditions we are investigating, clinicians' and patients' perceptions of the potential advantages and side effects of the product candidates being studied in relation to other available therapies and product candidates, and health epidemics such as the COVID-19 pandemic. Clinical investigators will need to decide whether to offer their patients enrollment in clinical trials of our product candidate versus treating these patients with commercially available drugs that have established safety and efficacy profiles. Any delays in completing our clinical trials will increase our costs, slow down our product development, timeliness and approval process, and delay our ability to generate revenue.

38

The regulatory approval processes of the FDA and comparable foreign authorities are lengthy, time consuming and inherently unpredictable, and if we are ultimately unable to obtain regulatory approval for our product candidates, our business will be substantially harmed.

The time required to obtain approval by the FDA and comparable foreign authorities is unpredictable but it typically takes many years following the commencement of clinical trials and depends upon numerous factors, including the substantial discretion of the regulatory authorities. In addition, approval policies, regulations, or the type and amount of clinical data necessary to gain approval may change during the course of a product candidate's clinical development and may vary among jurisdictions. We have not obtained regulatory approval for any product candidate and it is possible that any of our existing product candidates, or any product candidate we may seek to develop in the future, may never obtain regulatory approval.

Our product candidates could fail to receive regulatory approval for many reasons, including the following:

- the FDA or comparable foreign regulatory authorities may disagree with the design or implementation of our clinical trials;
- we may be unable to demonstrate to the satisfaction of the FDA or comparable foreign regulatory authorities that a product candidate is safe and effective for its proposed indication;
- the results of clinical trials may not meet the level of statistical significance required for approval by the FDA or comparable foreign regulatory authorities;
- the FDA or comparable foreign regulatory authorities may disagree with our interpretation of data from preclinical studies or clinical trials;
- the data collected from clinical trials of our product candidates may not be sufficient to support the submission of an NDA or other submission or to obtain regulatory approval in the United States or elsewhere;
- the FDA or comparable foreign regulatory authorities may fail to approve the manufacturing processes or facilities of third-party manufacturers with which we contract for clinical and commercial supplies; or
- the approval policies or regulations of the FDA or comparable foreign regulatory authorities may significantly change in a manner rendering our clinical data insufficient for approval.

This lengthy approval process as well as the unpredictability of future clinical trial results may result in our failing to obtain regulatory approval to market our product candidates, which would significantly harm our business, results of operations, prospects and our underlying stock price.

In addition, even if we were to obtain approval, regulatory authorities may approve our product candidates for fewer or more limited indications than we request, may grant approval contingent on the performance of costly post-marketing clinical trials, or may approve a product candidate with a label that does not include the labeling claims necessary or desirable for the successful commercialization of that product candidate. Any of the foregoing scenarios could materially harm the commercial prospects for any of our product candidates.

We have not previously submitted an NDA to the FDA, nor similar drug approval filings to comparable foreign authorities, for our product candidates, and we cannot be certain that our product candidates will be successful in clinical trials or receive regulatory approval. Further, our product candidates may not receive regulatory approval even if they are successful in clinical trials. If we do not receive regulatory approvals for our product candidates, we may not be able to continue our operations. Even if we successfully obtain regulatory approvals to market one or more of our product candidates, our revenues will be dependent on many factors including the size of the markets in the territories for which we gain regulatory approval and have commercial rights. If the markets for patients that we are targeting for our product candidates are not as significant as we estimate, we may not generate significant revenues from sales of such products, if approved.

We plan to seek regulatory approval and to commercialize our product candidates, directly or with collaborators in the United States, the European Union, and other foreign countries which we have not yet identified. While the scope of regulatory approval is similar in other countries, to obtain separate regulatory approval in many other countries we must comply with numerous and varying regulatory requirements of such countries regarding the safety and efficacy, among other things, of clinical trials and commercial sales, pricing, and distribution of our product candidates, and we cannot predict success in these jurisdictions.

39

We may be required to suspend or discontinue clinical trials due to unexpected side effects or other safety risks that could preclude approval of our product candidates.

Our clinical trials may be suspended at any time for a number of reasons. For example, we may voluntarily suspend or terminate our clinical trials if at any time we believe that they present an unacceptable risk to the clinical trial patients. In addition, the FDA or other regulatory agencies may order the temporary or permanent discontinuation of our clinical trials at any time if they believe that the clinical trials are not being conducted in accordance with applicable regulatory requirements or that they present an unacceptable safety risk to the clinical trial patients.

Administering our product candidates to humans may produce undesirable side effects. These adverse side effects could interrupt, delay or halt clinical trials of our product candidates and could result in the FDA or other regulatory authorities denying further development or approval of any of our product candidates for any or all targeted indications. Ultimately, our product candidates may prove to be unsafe for human use. Moreover, we could be subject to significant liability if any volunteer or patient suffers, or appears to suffer, adverse health effects as a result of participating in our clinical trials. Prosecution, enforcement actions, damages or adverse media coverage related to such events, if any, will likely result in a materially significant diversion of management's attention and resources and significant defense costs and other professional fees. As a general matter, such events could damage our reputation, brand, international activities, business, prospects, operating results and financial condition.

If we fail to comply with healthcare regulations, we could face substantial enforcement actions, including civil and criminal penalties and our business, operations and financial condition could be adversely affected.

As a developer of pharmaceuticals, certain federal and state healthcare laws and regulations pertaining to fraud and abuse, false claims and patients' privacy rights are and will be applicable to our business. We could be subject to healthcare fraud and abuse laws and patient privacy laws of both the federal government and the states in which we conduct our business. The laws include:

- the federal healthcare program anti-kickback law, which prohibits, among other things, persons from soliciting, receiving or providing remuneration, directly or
 indirectly, to induce either the referral of an individual, for an item or service or the purchasing or ordering of a good or service, for which payment may be made
 under federal healthcare programs such as the Medicare and Medicaid programs;
- federal false claims laws which prohibit, among other things, individuals, or entities from knowingly presenting, or causing to be presented, claims for payment from Medicare, Medicaid, or other third-party payors that are false or fraudulent, and which may apply to entities like us which provide coding and billing information to customers;
- the federal Health Insurance Portability and Accountability Act of 1996, which prohibits executing a scheme to defraud any healthcare benefit program or making false statements relating to healthcare matters and which also imposes certain requirements relating to the privacy, security and transmission of individually identifiable health information;
- the Federal Food, Drug, and Cosmetic Act, which among other things, strictly regulates drug manufacturing and product marketing, prohibits manufacturers from
 marketing drug products for off-label use and regulates the distribution of drug samples; and
- state law equivalents of each of the above federal laws, such as anti-kickback and false claims laws which may apply to items or services reimbursed by any third-party payor, including commercial insurers, and state laws governing the privacy and security of health information in certain circumstances, many of which differ from each other in significant ways and often are not preempted by federal laws, thus complicating compliance efforts.

If our operations are found to be in violation of any of the laws described above or any governmental regulations that apply to us, we may be subject to penalties, including civil and criminal penalties, damages, fines, and the curtailment or restructuring of our operations. Any penalties, damages, fines, curtailment or restructuring of our operations could adversely affect our ability to operate our business and our financial results. Although compliance programs can mitigate the risk of investigation and prosecution for violations of these laws, the risks cannot be entirely eliminated. Any action against us for violation of these laws, even if we successfully defend against it, could cause us to incur significant legal expenses and divert management's attention from the operation of our business. Moreover, achieving and sustaining compliance with applicable federal and state privacy, security and fraud laws may prove costly.

40

If we are unable to satisfy regulatory requirements, we may not be able to commercialize our product candidates.

We need FDA approval prior to marketing our product candidates in the United States. This approval process is lengthy and subject to extensive governmental regulations, and given the unpredictability of the results of clinical trials, our failure to obtain regulatory approval from the FDA to market any of our product candidates would significantly harm our business, results of operations and prospects. Any delay or failure in seeking or obtaining required approvals from the FDA to market any of our product candidates would have a material and adverse effect on our ability to sell our product candidates in the United States and to generate revenue from any such candidates we are developing and for which we are seeking approval.

The FDA's review and approval process, including among other things, evaluation of preclinical studies and clinical trials of a product candidate as well as the

manufacturing process and facility, is lengthy, expensive, and uncertain. To receive approval, we must, among other things, demonstrate with substantial evidence from well-designed and well-controlled pre-clinical testing and clinical trials that the product candidates are both safe and effective for each indication for which approval is sought. Satisfaction of these requirements typically takes several years, and the time needed to satisfy them may vary substantially, based on the type, complexity and novelty of the pharmaceutical product. We cannot predict if or when we will submit an NDA for approval for any of our product candidates currently under development. Any approvals we may obtain may not cover all of the clinical indications for which we are seeking approval or may contain significant limitations on the conditions of use.

The FDA has substantial discretion in the NDA review process and may either refuse to file our NDA for substantive review or may decide that our data is insufficient to support approval of our product candidates for the claimed intended uses. Following any regulatory approval of our product candidates, we will be subject to continuing regulatory obligations such as safety reporting, required and additional post marketing obligations, and regulatory oversight of promotion and marketing. Even if we receive regulatory approvals for any of our product candidates, the FDA may subsequently seek to withdraw approval of our NDA if we determine that new data or a reevaluation of existing data show the product is unsafe for use under the conditions of use upon the basis of which the NDA was approved, or based on new evidence of adverse effects or adverse clinical experience, or upon other new information. If the FDA does not file or approve our NDA or withdraws approval of our NDA, the FDA may require that we conduct additional clinical trials, preclinical or manufacturing studies and submit that data before it will reconsider our application. Depending on the extent of these or any other requested studies, approval of any applications that we submit may be delayed by several years, may require us to expend more resources than we have available, or may never be obtained at all. In addition, we have obtained FDA clearance for our RenovoCath delivery system. In the event adverse events arise with respect to the RenovoCath delivery system, the FDA could revoke its clearance which would have a material adverse effect on our business.

We will also be subject to a wide variety of foreign regulations governing the development, manufacture, and marketing of our products to the extent we seek regulatory approval to develop and market any of our product candidates in a foreign jurisdiction. As of the date hereof we have not identified any foreign jurisdictions which we intend to seek approval from. Whether or not FDA approval has been obtained, approval of a product candidate by the comparable regulatory authorities of foreign countries must still be obtained prior to marketing the product candidate in those countries. The approval process varies, and the time needed to secure approval in any region such as the European Union or in a country with an independent review procedure may be longer or shorter than that required for FDA approval. We cannot assure you that clinical trials conducted in one country will be accepted by other countries or that an approval in one country or region will result in approval elsewhere.

If our product candidates are unable to compete effectively with marketed drugs targeting similar indications as our product candidates, our commercial opportunity will be reduced or eliminated.

We face competition generally from established pharmaceutical and biotechnology companies, as well as from academic institutions, government agencies and private and public research institutions. Many of our competitors have significantly greater financial resources and expertise in research and development, manufacturing, preclinical testing, conducting clinical trials, obtaining regulatory approvals and marketing approved products than we do. Small or early-stage companies may also prove to be significant competitors, particularly through collaborative arrangements with large, established companies. We are aware of a number of companies in Phase 3 clinical trials for the treatment of LAPC including Angiodynamics, Bausch Health, FibroGen, Novocure, and SynCore Biotechnology. In addition, we are aware of a number of companies in Phase 2 clinical trials for the treatment of LAPC including one interventional company, TriSalus Lifesciences. Our commercial opportunity will be reduced or eliminated if our competitors develop and commercialize any products that are safer, more effective, have fewer side effects or are less expensive than our product candidates. These potential competitors compete with us in recruiting and retaining key and qualified scientific and management personnel, establishing clinical trial sites, and patient enrollment for clinical trials, as well as in acquiring technologies and technology licenses complementary to our programs or advantageous to our business.

If approved and commercialized, RenovoGem would compete with several currently approved prescription therapies for the treatment of LAPC and hilar cholangiocarcinoma. To our knowledge, other potential competitors are in earlier stages of development. If potential competitors are successful in completing drug development for their product candidates and obtain approval from the FDA, they could limit the demand for RenovoGem.

4

We expect that our ability to compete effectively will depend upon our ability to:

- successfully identify and develop key points of product differentiation from currently available therapies;
- successfully and timely complete clinical trials and submit for and obtain all requisite regulatory approvals in a cost-effective manner;
- maintain a proprietary position for our products and manufacturing processes and other related product technology;
- attract and retain key and qualified personnel;
- develop relationships with physicians prescribing these products; and
- build an adequate sales and marketing infrastructure for our products, if approved.

Because we will be competing against significantly larger companies with established track records, we will have to demonstrate that, based on experience, clinical data, side-effect profiles and other factors, our products, if approved, are competitive with other products. If we are unable to compete effectively and differentiate our products from other marketed drugs, we may never generate meaningful revenue.

We may expend our limited resources to pursue one or more product candidates or indications within our product development strategy, which has and may continue to change over time, and fail to capitalize on product candidates or indications that may be more profitable or for which there is a greater likelihood of success.

Because we have limited financial and managerial resources, we intend to focus on developing product candidates for specific indications that we identify as most likely to succeed, in terms of their potential both to gain regulatory approval and to achieve commercialization. As a result, we may forego or delay pursuit of opportunities with other product candidates or in other indications with greater commercial potential. Such resource allocation and strategic decisions may cause us to fail to capitalize on viable commercial products or profitable market opportunities. Our spending on current and future research and development programs and product candidates for specific indications may not yield any commercially viable product candidates. If we do not accurately evaluate the commercial potential or target market for a particular product candidate, we may relinquish valuable rights to that product candidate through collaboration, licensing, or other royalty arrangements in cases in which it would have been more advantageous for us to retain sole development and commercialization rights to the product candidate.

If the manufacturers upon whom we rely fail to produce our product candidates, in the volumes that we require on a timely basis or fail to comply with stringent regulations applicable to pharmaceutical drug manufacturers, we may face delays in the development and commercialization of our product candidates.

We do not currently possess internal manufacturing capacity. We plan to utilize the services of cGMP manufacturers, FDA inspected contract manufacturers to manufacture our clinical supplies. Any curtailment in the availability of gemcitabine, however, could result in production or other delays with consequent adverse effects on us. In addition, because regulatory authorities must generally approve raw material sources for pharmaceutical products, changes in raw material suppliers may result in production delays or higher raw material costs.

We obtain our RenovoCath delivery system from a single source. Gemcitabine is supplied from our clinical sites' pharmacies and used off-label for intra-arterial use within our clinical study. We continue to pursue supply agreements for gemcitabine and our RenovoCath delivery system. We may be required to agree to minimum volume requirements, exclusivity arrangements or other restrictions with the contract manufacturers. We may not be able to enter into long-term agreements on commercially

reasonable terms, or at all. If we change or add manufacturers, the FDA and comparable foreign regulators may require approval of the changes. Approval of these changes could require new testing by the manufacturer and compliance inspections to ensure the manufacturer is conforming to all applicable laws and regulations and cGMP.

The manufacture of pharmaceutical products requires significant expertise and capital investment, including the development of advanced manufacturing techniques and process controls. Manufacturers of pharmaceutical products may encounter difficulties in production, particularly in scaling up production. These problems include difficulties with production costs and yields, quality control, including stability of the product and quality assurance testing, shortages of qualified personnel, as well as compliance with federal, state, and foreign regulations. In addition, any delay or interruption in the supply of clinical trial supplies, due to the effects of the COVID-19 pandemic or otherwise, could delay the completion of our clinical trials, increase the costs associated with conducting our clinical trials and, depending upon the period of delay, require us to commence new clinical trials at significant additional expense or to terminate a clinical trial.

42

We will be responsible for ensuring that our future contract manufacturers comply with the cGMP requirements of the FDA and other regulatory authorities from which we seek to obtain product approval. These requirements include, among other things, quality control, quality assurance and the maintenance of records and documentation. The approval process for NDAs includes an inspection of the manufacturer's compliance with cGMP requirements. We will be responsible for regularly assessing a contract manufacturer's compliance with cGMP requirements through record reviews and periodic audits and for ensuring that the contract manufacturer takes responsibility and corrective action for any identified deviations. Manufacturers of our product candidates may be unable to comply with these cGMP requirements and with other FDA and foreign regulatory requirements, if any.

While we will oversee compliance of our contract manufacturers, ultimately, we will not have control over our manufacturers' compliance with these regulations and standards. A failure to comply with these requirements may result in fines and civil penalties, suspension of production, suspension or delay in product approval, product seizure or recall, or withdrawal of product approval. If the safety of any of our product candidates is compromised due to a manufacturers' failure to adhere to applicable laws or for other reasons, we may not be able to obtain regulatory approval for or successfully commercialize any of our product candidates, and we may be held liable for any injuries sustained as a result. Any of these factors could cause a delay of clinical trials, regulatory submissions, approvals, or commercialization of RenovoGem or other product candidates, entail higher costs or result in us being unable to effectively commercialize any of our product candidates. Furthermore, if our manufacturers fail to deliver the required commercial quantities on a timely basis and at commercially reasonable prices, we may be unable to meet demand for any approved products and would lose potential revenues.

We may not be able to manufacture our product candidates in commercial quantities, which would prevent us from commercializing our product candidates.

To date, our product candidates have been manufactured in small quantities for preclinical studies and clinical trials. If any of our product candidates are approved by the FDA or comparable regulatory authorities in other countries for commercial sale, we will need to manufacture such product candidates in larger quantities. We may not be able to successfully increase the manufacturing capacity for our product candidates in a timely or economic manner, or at all. Significant scale-up of manufacturing may require additional validation studies, which the FDA must review and approve. If we are unable to successfully increase the manufacturing capacity for a product candidate, the clinical trials as well as the regulatory approval or commercial launch of that product candidate may be delayed or there may be a shortage in supply. Our product candidates require precise, high quality manufacturing in accordance with cGMP. Our failure to achieve and maintain these high-quality manufacturing standards in collaboration with our third-party manufacturers, including the incidence of manufacturing errors, could result in patient injury or death, product recalls or withdrawals, delays or failures in product testing or delivery, cost overruns or other problems that could harm our business, financial condition and results of operations.

Our product candidates, if approved for sale, may not gain acceptance among physicians, patients, and the medical community, thereby limiting our potential to generate revenues.

If our product candidates are approved for commercial sale by the FDA or other regulatory authorities, the degree of market acceptance of any approved product by physicians, healthcare professionals and third-party payors and our profitability and growth will depend on a number of factors, including:

- demonstration of safety and efficacy;
- changes in the practice guidelines and the standard of care for the targeted indication;
- relative convenience and ease of administration;
- the prevalence and severity of any adverse side effects;
- budget impact of adoption of our product on relevant drug formularies and the availability, cost, and potential advantages of alternative treatments, including less expensive generic drugs;
- pricing, reimbursement, and cost effectiveness, which may be subject to regulatory control;
- effectiveness of our or any of our partners' sales and marketing strategies;
- the product labeling or product insert required by the FDA or regulatory authority in other countries; and
- the availability of adequate third-party insurance coverage or reimbursement.

43

If any product candidate that we develop does not provide a treatment regimen that is as beneficial as, or is perceived as being as beneficial as, the current standard of care or otherwise does not provide patient benefit, that product candidate, if approved for commercial sale by the FDA or other regulatory authorities, likely will not achieve market acceptance. Our ability to effectively promote and sell any approved products will also depend on pricing and cost-effectiveness, including our ability to produce a product at a competitive price and our ability to obtain sufficient third-party coverage or reimbursement. If any product candidate is approved but does not achieve an adequate level of acceptance by physicians, patients and third-party payors, our ability to generate revenues from that product would be substantially reduced. In addition, our efforts to educate the medical community and third-party payors on the benefits of our product candidates may require significant resources, may be constrained by FDA rules and policies on product promotion, and may never be successful.

Guidelines and recommendations published by various organizations can impact the use of our products.

Government agencies promulgate regulations and guidelines directly applicable to us and to our product. In addition, professional societies, practice management groups, private health and science foundations and organizations involved in various diseases from time to time may also publish guidelines or recommendations to the healthcare and patient communities. Recommendations of government agencies or these other groups or organizations may relate to such matters as usage, dosage, route of administration and use of concomitant therapies. Recommendations or guidelines suggesting the reduced use of our products or the use of competitive or alternative products that are followed by patients and healthcare providers could result in decreased use of our proposed products.

If third-party contract manufacturers upon whom we rely to formulate and manufacture our product candidates do not perform, fail to manufacture according to our specifications or fail to comply with strict regulations, our preclinical studies or clinical trials could be adversely affected and the development of our product candidate could be delayed or terminated or we could incur significant additional expenses.

We do not own or operate any manufacturing facilities. We intend to rely on cGMP manufacturers, FDA inspected third-party contractors, at least for the foreseeable future to formulate and manufacture these preclinical and clinical materials. Our reliance on third-party contract manufacturers exposes us to a number of risks, any of which could delay or prevent the completion of our preclinical studies or clinical trials, or the regulatory approval or commercialization of any of our product candidates, result in higher costs, or deprive us of potential product revenues. Some of these risks include:

- our third-party contractors failing to develop an acceptable formulation to support later-stage clinical trials for, or the commercialization of, our product candidates;
- our contract manufacturers failing to manufacture our product candidates according to their own standards, our specifications or Current Good Manufacturing Practice ("cGMP"), or otherwise manufacturing material that we or the FDA may deem to be unsuitable in our clinical trials;
- our contract manufacturers being unable to increase the scale of, increase the capacity for, or reformulate the form of any of our product candidates. We may experience a shortage in supply, or the cost to manufacture our products may increase to the point where it adversely affects the cost of our product candidates. We cannot assure you that our contract manufacturers will be able to manufacture our product candidates at a suitable scale, or we will be able to find alternative manufacturers acceptable to us that can do so;
- our contract manufacturers placing a priority on the manufacture of their own products, or other customers' products;
- our contract manufacturers failing to perform as agreed or not remaining in the contract manufacturing business;
- our contract manufacturers experiencing the effects of any strikes or other work stoppages; or
- our contract manufacturers' plants being closed as a result of regulatory sanctions or a natural disaster.

44

In the event that we need to change our third-party contract manufacturers, our preclinical studies, clinical trials or the commercialization of our product candidates could be delayed, adversely affected or terminated, or such a change may result in significantly higher costs.

Due to regulatory restrictions inherent in an IND or NDA, or for economic reasons, various steps in the manufacture of any of our product candidates may need to be sole-sourced. We currently obtain our RenovoCath delivery system from a single supplier. In accordance with cGMP regulations, changing manufacturers may require the revalidation of manufacturing processes and procedures, and may require further preclinical studies or clinical trials to show comparability between the materials produced by different manufacturers. Changing our current or future contract manufacturers may be difficult for us and could be costly, which could result in our inability to manufacture our product candidate for an extended period of time and therefore a delay in the development of any of our product candidates. Further, to maintain our development timelines in the event of a change in our third-party contract manufacturer, we may incur significantly higher costs to manufacture any of our product candidates.

We currently do not have any internal drug discovery capabilities, and therefore we are dependent on identifying drugs that are off patent or on in-licensing or acquiring development programs from third parties in order to obtain additional product candidates.

If in the future we decide to further expand our pipeline of product candidates, we will be dependent on identifying drugs that are off patent or on in-licensing or acquiring product candidates as we do not have significant internal discovery capabilities at this time. We may face substantial competition from other biotechnology and pharmaceutical companies, many of which may have greater resources then we have, in obtaining in-licensing, sponsored research or acquisition opportunities. In-licensing or acquisition opportunities may not be available to us on terms we find acceptable, if at all. In-licensed compounds that appear promising in research or in preclinical studies may fail to progress into further preclinical studies or clinical trials.

If a product liability claim is successfully brought against us for uninsured liabilities, or such claim exceeds our insurance coverage, we could be forced to pay substantial damage awards that could materially harm our business.

The use of any of our existing or future product candidates in clinical trials and the sale of any approved pharmaceutical products may expose us to significant product liability claims. We have product liability insurance coverage for our proposed clinical trials; however, such insurance coverage may be inadequate and may not protect us against any or all of the product liability claims that may be brought against us now or in the future. We may not be able to acquire or maintain adequate product liability insurance coverage at a commercially reasonable cost or in sufficient amounts or scope to protect us against potential losses. In the event a product liability claim is brought against us, we may be required to pay legal and other expenses to defend the claim, as well as uncovered damage awards resulting from a claim brought successfully against us. In the event that any of our product candidates are approved for sale by the FDA and commercialized, we may need to substantially increase the amount of our product liability coverage. Defending any product liability claim or claims could require us to expend significant financial and managerial resources, which could have a material adverse effect on our business.

We may delay or terminate the development of our product candidates at any time if we believe the perceived market or commercial opportunity does not justify further investment, which could materially harm our business.

Even though the results of preclinical studies and clinical trials that have been conducted or may be conducted in the future may support further development of our product candidates, we may delay, suspend or terminate the future development of a product candidate at any time for strategic, business, financial or other reasons, including the determination or belief that the emerging profile of the product candidate is such that it may not receive FDA approval, gain meaningful market acceptance, generate a significant return to stockholders, or otherwise provide any competitive advantages in its intended indication or market.

Risks Related to Our Operations

Our future success depends on our ability to retain our key personnel and to attract, retain and motivate qualified personnel.

We are highly dependent on the development, regulatory, commercialization, and business development expertise of Shaun Bagai, our Chief Executive Officer, as well as the other principal members of our management, scientific and clinical teams. Although we have employment agreements, offer letters or consulting agreements with our executive officers, these agreements do not prevent them from terminating their services at any time.

breadth of skills and experience required to successfully develop product candidates, gain regulatory approval, and commercialize new products. Competition to hire from this limited pool is intense, and we may be unable to hire, train, retain or motivate these additional key personnel on acceptable terms given the competition among numerous pharmaceutical and biotechnology companies for similar personnel. We also experience competition for the hiring of scientific and clinical personnel from universities and research institutions. In addition, we rely on consultants and advisors, including scientific and clinical advisors, to assist us in formulating our research and development and commercialization strategy. Our consultants and advisors may be engaged by entities other than us and may have commitments under consulting or advisory contracts with other entities that may limit their availability to us. One such key consultant is Dr. Ramtin Agah, our Chief Medical Officer. If we are unable to continue to attract and retain highly qualified personnel, our ability to develop and commercialize product candidates will be limited.

We will need to increase the size of our organization, and we may experience difficulties in managing growth.

We are a small company with less than 10 employees. Future growth of our company will impose significant additional responsibilities on members of management, including the need to identify, attract, retain, motivate and integrate highly skilled personnel. We may increase the number of employees in the future depending on the progress of our development and commercialization of our product candidates. Our future financial performance and our ability to commercialize our product candidates and to compete effectively will depend, in part, on our ability to manage any future growth effectively. To that end, we must be able to:

- manage our clinical studies effectively;
- integrate additional and future management, administrative, manufacturing, sales and marketing, and regulatory personnel;
- · maintain sufficient administrative, accounting and management information systems and controls; and
- hire and train additional qualified personnel.

There is no guarantee that we will be able to accomplish these tasks, and our failure to accomplish any of them could materially adversely affect our business, prospects, and financial condition.

Business disruptions could seriously harm future revenue and financial condition and increase our costs and expenses.

Our operations, and those of our third-party manufacturers, contract research organizations ("CROs"), and other contractors and consultants, could be subject to earthquakes, power shortages, telecommunications failures, water shortages, floods, hurricanes, typhoons, fires, extreme weather conditions, medical epidemics and other natural or man-made disasters or business interruptions beyond our control, for which we are predominantly self-insured. The occurrence of any of these business disruptions could seriously harm our operations and financial condition and increase our costs and expenses.

Our corporate headquarters are located in Silicon Valley, California, an area prone to wildfires and earthquakes. These and other natural disasters could severely disrupt our operations, and have a material adverse effect on our business, results of operations, financial condition and prospects. If a natural disaster, power outage or other event occurred that prevented us from using all or a significant portion of our headquarters, that damaged critical infrastructure, such as the manufacturing facilities of our third-party contract manufacturers, or that otherwise disrupted operations, it may be difficult or, in certain cases, impossible for us to continue our business for a substantial period of time. Any disaster recovery and business continuity plans we have in place may prove inadequate in the event of a serious disaster or similar event. We may incur substantial expenses as a result of the limited nature of our disaster recovery and business continuity plans, which could have a material adverse effect on our business.

Security threats to our information technology infrastructure and/or our physical buildings could expose us to liability and damage our reputation and business.

It is essential to our business strategy that our and our vendors, partners, clinical trial sites, and third-party providers' technology and network infrastructure and physical buildings remain secure and are perceived by our customers and corporate partners to be secure. Despite security measures, however, any network infrastructure may be vulnerable to cyber-attacks by hackers and other security threats, including distributed denial of service (DDoS) attacks, phishing attacks or ransomware attacks. We may face cyber-attacks that attempt to penetrate our network security, sabotage, or otherwise disable our research and development activities, products and services, misappropriate our or our customers' and partners' proprietary and confidential information, which may include personally identifiable information, or cause interruptions of our internal systems and services. During the COVID-19 pandemic, with many of our employees working from home from time to time and accessing our corporate network via remote devices, the potential for such events to occur is even greater. Despite security measures, we also cannot guarantee security of our physical buildings. Physical building penetration or any cyber-attacks could negatively affect our reputation, damage our network infrastructure and our ability to deploy our products and services, harm our relationship with customers and partners that are affected, and expose us to financial liability, including the possibility of consequential damages resulting from cyber-attacks and other security threats.

46

Additionally, there are a number of state, federal, and international laws protecting the privacy and security of health information and personal data. For example, the Health Insurance Portability and Accountability Act of 1996 ("HIPAA") imposes limitations on the use and disclosure of an individual's healthcare information by healthcare providers, healthcare clearinghouses, and health insurance plans, or, collectively, covered entities, and also grants individuals rights with respect to their health information. HIPAA also imposes compliance obligations and corresponding penalties for non-compliance on individuals and entities that provide services to healthcare providers and other covered entities. As part of the American Recovery and Reinvestment Act of 2009 ("ARRA"), the privacy and security provisions of HIPAA were amended. ARRA also made significant increases in the penalties for improper use or disclosure of an individual's health information under HIPAA and extended enforcement authority to state attorneys general. As amended by ARRA and subsequently by the final omnibus rule adopted in 2013, HIPAA also imposes notification requirements on covered entities in the event that certain health information has been inappropriately accessed or disclosed, including notification requirements to individuals, federal regulators, and in some cases, notification to local and national media. Notification is not required under HIPAA if the health information that is improperly used or disclosed is deemed secured in accordance with encryption or other standards developed by the U.S. Department of Health and Human Services. Most states have laws requiring notification of affected individuals and/or state regulators in the event of a breach of personal information, which is a broader class of information than the health information protected by HIPAA. Many state laws impose significant data security requirements, such as encryption or mandatory contractual terms, to ensure ongoing protection of personal information. Act

We and our third-party contract manufacturers must comply with environmental, health and safety laws and regulations, and failure to comply with these laws and regulations could expose us to significant costs or liabilities.

We and our third-party manufactures are subject to numerous environmental, health and safety laws and regulations, including those governing laboratory procedures and the use, generation, manufacture, distribution, storage, handling, treatment, remediation and disposal of hazardous materials and wastes. Hazardous chemicals, including flammable and biological materials, are involved in certain aspects of our business, and we cannot eliminate the risk of injury or contamination from the use, generation, manufacture, distribution, storage, handling, treatment or disposal of hazardous materials and wastes. In the event of contamination or injury, or failure to comply with environmental, health and safety laws and regulations, we could be held liable for any resulting damages and any such liability could exceed our assets and resources. We could also incur significant costs associated with civil or criminal fines and penalties for failure to comply with such laws and regulations.

Although we maintain workers' compensation insurance to cover us for costs and expenses we may incur due to injuries to our employees resulting from the use of hazardous materials, this insurance may not provide adequate coverage against potential liabilities. We do not maintain insurance for environmental liability or toxic tort claims

that may be asserted against us in connection with our storage or disposal of biological, hazardous or radioactive materials.

Environmental, health and safety laws and regulations are becoming increasingly more stringent. We may incur substantial costs in order to comply with current or future environmental, health and safety laws and regulations. These current or future laws and regulations may impair our research, development or production efforts. Our failure to comply with these laws and regulations also may result in substantial fines, penalties or other sanctions.

Further, with respect to the operations of our third-party contract manufacturers, it is possible that if they fail to operate in compliance with applicable environmental, health and safety laws and regulations or properly dispose of wastes associated with our products, we could be held liable for any resulting damages, suffer reputational harm or experience a disruption in the manufacture and supply of our product candidates or products.

47

A variety of risks associated with operating internationally could materially adversely affect our business.

Doing business internationally involves a number of risks, including but not limited to:

- multiple, conflicting and changing laws and regulations, such as privacy regulations, tax laws, export and import restrictions, employment laws, regulatory requirements and other governmental approvals, permits and licenses;
- failure by us to obtain and maintain regulatory approvals for the use of our products in various countries;
- additional potentially relevant third-party patent rights;
- complexities and difficulties in obtaining protection and enforcing our intellectual property;
- · difficulties in staffing and managing foreign operations;
- complexities associated with managing multiple payor reimbursement regimes, government payors or patient self-pay systems;
- limits in our ability to penetrate international markets;
- financial risks, such as longer payment cycles, difficulty collecting accounts receivable, the impact of local and regional financial crises on demand and payment for our products and exposure to foreign currency exchange rate fluctuations;
- natural disasters, political and economic instability, including wars, terrorism and political unrest, outbreak of disease, boycotts, curtailment of trade and other business restrictions;
- certain expenses including, among others, expenses for travel, translation, and insurance; and
- regulatory and compliance risks that relate to maintaining accurate information and control over sales and activities that may fall within the purview of the U.S.
 Foreign Corrupt Practices Act, its books and records provisions, or its anti-bribery provisions.

Any of these factors could significantly harm any current or future international operations and, consequently, our results of operations.

General economic or business conditions may have a negative impact on our business.

Continuing concerns over U.S. healthcare reform legislation and energy costs, geopolitical issues, fluctuations in inflation rates, the availability and cost of credit and government stimulus programs in the U.S. and other countries, including those related to the COVID-19 pandemic, have contributed to increased volatility. If the economic climate deteriorates or is poor, our business, as well as the financial condition of our suppliers and our third-party payors, could be negatively impacted, which could materially adversely affect our business, prospects and financial condition.

Healthcare reform measures could adversely affect our business.

In the United States and foreign jurisdictions, there have been, and continue to be, a number of legislative and regulatory changes and proposed changes to the healthcare system that could affect our future results of operations. In particular, there have been and continue to be a number of initiatives at the federal and state levels in the United States that seek to reduce healthcare costs. In 2010, the Patient Protection and Affordable Care Act (the "PPACA") was enacted, which includes measures to significantly change the way healthcare is financed by both governmental and private insurers. Among the provisions of the PPACA of greatest importance to the pharmaceutical and biotechnology industry are the following:

- an annual, nondeductible fee on any entity that manufactures or imports certain branded prescription drugs and biologic agents, apportioned among these entities according to their market share in certain government healthcare programs;
- implementation of the federal physician payment transparency requirements, sometimes referred to as the "Physician Payments Sunshine Act";
- a licensure framework for follow-on biologic products;
- a new Patient-Centered Outcomes Research Institute to oversee, identify priorities in, and conduct comparative clinical effectiveness research, along with funding for such research;
- establishment of a Center for Medicare Innovation at the Centers for Medicare & Medicaid Services to test innovative payment and service delivery models to lower Medicare and Medicaid spending, potentially including prescription drug spending;

18

- an increase in the statutory minimum rebates a manufacturer must pay under the Medicaid Drug Rebate Program, to 23.1% and 13% of the average manufacturer price for most branded and generic drugs, respectively and capped the total rebate amount for innovator drugs at 100% of the AMP;
- a new methodology by which rebates owed by manufacturers under the Medicaid Drug Rebate Program are calculated for certain drugs and biologics, including our
 product candidates, that are inhaled, infused, instilled, implanted or injected;
- extension of manufacturers' Medicaid rebate liability to covered drugs dispensed to individuals who are enrolled in Medicaid managed care organizations;

- expansion of eligibility criteria for Medicaid programs by, among other things, allowing states to offer Medicaid coverage to additional individuals and by adding
 new mandatory eligibility categories for individuals with income at or below 133% of the federal poverty level, thereby potentially increasing manufacturers'
 Medicaid rebate liability;
- a new Medicare Part D coverage gap discount program, in which manufacturers must agree to offer 50% point-of-sale discounts off negotiated prices of applicable
 brand drugs to eligible beneficiaries during their coverage gap period, as a condition for the manufacturer's outpatient drugs to be covered under Medicare Part D;
 and
- expansion of the entities eligible for discounts under the Public Health program.

Some of the provisions of the PPACA have yet to be implemented, and there have been legal and political challenges to certain aspects of the PPACA. During President Trump's administration, he signed two executive orders and other directives designed to delay, circumvent, or loosen certain requirements mandated by the PPACA. Concurrently, Congress has considered legislation that would repeal or repeal and replace all or part of the PPACA. While Congress has not passed repeal legislation, Congress passed the Tax Cuts and Jobs Act in 2017 which eliminated, effective January 1, 2019, the tax-based shared responsibility payment imposed by the PPACA on certain individuals who fail to maintain qualifying health coverage for all or part of a year that is commonly referred to as the "individual mandate". Congress may consider other legislation to repeal or replace elements of the PPACA.

Many of the details regarding the implementation of the PPACA are yet to be determined, and at this time, the full effect that the PPACA would have on our business remains unclear.

Individual states have become increasingly aggressive in passing legislation and implementing regulations designed to control pharmaceutical product pricing, including price or patient reimbursement constraints, discounts, restrictions on certain product access, and marketing cost disclosure and transparency measures, and to encourage importation from other countries and bulk purchasing. Legally mandated price controls on payment amounts by third-party payors or other restrictions could harm our business, results of operations, financial condition and prospects. In addition, regional healthcare authorities and individual hospitals are increasingly using bidding procedures to determine what pharmaceutical products and which suppliers will be included in their prescription drug and other healthcare programs. This could reduce ultimate demand for our products or put pressure on our product pricing, which could negatively affect our business, results of operations, financial condition and prospects.

In addition, given recent federal and state government initiatives directed at lowering the total cost of healthcare, Congress and state legislatures will likely continue to focus on healthcare reform, the cost of prescription drugs and biologics and the reform of the Medicare and Medicaid programs. While we cannot predict the full outcome of any such legislation, it may result in decreased reimbursement for drugs and biologics, which may further exacerbate industry-wide pressure to reduce prescription drug prices. This could harm our ability to generate revenues. Increases in importation or re-importation of pharmaceutical products from foreign countries into the United States could put competitive pressure on our ability to profitably price our products, which, in turn, could adversely affect our business, results of operations, financial condition and prospects. We might elect not to seek approval for or market our products in foreign jurisdictions in order to minimize the risk of re-importation, which could also reduce the revenue we generate from product sales. It is also possible that other legislative proposals having similar effects will be adopted.

Furthermore, regulatory authorities' assessment of the data and results required to demonstrate safety and efficacy can change over time and can be affected by many factors, such as the emergence of new information, including on other products, changing policies and agency funding, staffing and leadership. We cannot be sure whether future changes to the regulatory environment will be favorable or unfavorable to our business prospects. For example, average review times at the FDA for marketing approval applications can be affected by a variety of factors, including budget and funding levels and statutory, regulatory and policy changes.

49

The outbreak of the novel coronavirus disease, COVID-19, could materially adversely impact our business, results of operations and financial condition, including our clinical trials.

In January 2020, the World Health Organization declared the outbreak of COVID-19 as a "Public Health Emergency of International Concern," which continues to spread throughout the world and has adversely impacted global commercial activity and contributed to significant volatility in financial markets. The COVID-19 outbreak and government responses are creating disruption in global supply chains and adversely impacting many industries. The outbreak could have a continued material adverse impact on economic and market conditions. We continue to monitor the impact of the COVID-19 outbreak closely. The extent to which the COVID-19 outbreak will impact our operations or financial results is uncertain.

The outbreak and government measures taken in response have also had a significant impact, both direct and indirect, on businesses and commerce, as worker shortages have occurred; supply chains have been disrupted; activity at facilities and production have been suspended; and demand for certain goods and services, such as medical services and supplies, has spiked, while demand for other goods and services, such as travel, has fallen. While the extent of the impact of the COVID-19 pandemic on our business and financial results is uncertain, a continued and prolonged public health crisis such as the COVID-19 pandemic could have a material adverse effect on our business, financial condition and results of operations. As a result of the COVID-19 pandemic, we may experience disruptions that could severely impact our business and clinical trials, including:

- delays or difficulties in enrolling and retaining patients in our clinical trials;
- delays or difficulties in clinical site initiation, including difficulties in recruiting clinical site investigators and clinical site staff;
- diversion of healthcare resources away from the conduct of clinical trials, including the diversion of hospitals serving as our clinical trial sites and hospital staff supporting the conduct of our clinical trials;
- interruption of key clinical trial activities, such as clinical trial site data monitoring, due to limitations on travel imposed or recommended by federal or state governments, employers and others or interruption of clinical trial subject visits and study procedures (such as endoscopies that are deemed non-essential), which may impact the integrity of subject data and clinical study endpoints;
- interruption or delays in the operations of the FDA or other regulatory authorities, which may impact review and approval timelines;
- interruption of, or delays in receiving, supplies of our product candidates from our contract manufacturing organizations due to staffing shortages, production slowdowns or stoppages and disruptions in delivery systems;
- limitations on employee and consulting resources that would otherwise be focused on the conduct of our clinical trials, including because of sickness of employees, consultants or their families or the desire of employees or consultants to avoid contact with large groups of people;
- interruption or delays to our outsourced clinical activities; or
- changes in clinical site procedures and requirements as well as regulatory requirements for conducting clinical trials during the pandemic.

We may be required to develop and implement additional clinical trial policies and procedures designed to help protect subjects from the COVID-19 virus. For example, in March 2020, the FDA issued guidance, which FDA subsequently updated, on conducting clinical trials during the pandemic, which describes a number of considerations for sponsors of clinical trials impacted by the pandemic, including the requirement to include in the clinical trial report contingency measures implemented to manage the trial, and any disruption of the trial as a result of the COVID-19 pandemic; a list of all subjects affected by the COVID-19 pandemic related study disruption by unique subject identifier and by investigational site and a description of how the individual's participation was altered; and analyses and corresponding discussions that address the impact of implemented contingency measures (e.g., participant discontinuation from investigational product and/or study, alternative procedures used to collect critical safety and/or efficacy data) on the safety and efficacy results reported for the trial.

The COVID-19 pandemic continues to evolve rapidly, with the status of operations and government restrictions evolving frequently. The extent to which the outbreak impacts our business and clinical trials will depend on future developments, which are highly uncertain and cannot be predicted with confidence, such as the duration of the pandemic, the continued effectiveness of the vaccines, vaccination and infection rates, new travel restrictions or social distancing in the United States and other countries, business closures or business disruptions and the effectiveness of actions taken in the United States and other countries to contain and treat the disease.

50

The ultimate impact of the COVID-19 pandemic on our business operations is highly uncertain and subject to change and will depend on future developments, which cannot be accurately predicted, including the duration of the pandemic, the ultimate geographic spread of the disease, additional or modified government actions, new information that will emerge concerning the severity and impact of COVID-19 and the actions taken to contain COVID-19, including vaccination efforts, or address its impact in the short and long term, among others. We do not yet know the full extent of potential delays or impacts on our business, our clinical trials, our research programs, healthcare systems or the global economy. We will continue to monitor the situation closely.

In addition, our business could be materially adversely affected by other business disruptions to us or our third-party providers that could materially adversely affect our potential future revenue and financial condition and increase our costs and expenses. Our operations, and those of our CROs, third party manufacturers, and other contractors, consultants and third parties could be subject to other global pandemics, earthquakes, power shortages, telecommunications failures, water shortages, floods, hurricanes, typhoons, fires, extreme weather conditions, medical epidemics and other natural or man-made disasters or business interruptions, for which we are predominantly self-insured. The occurrence of any of these business disruptions could materially adversely affect our operations and financial condition and increase our costs and expenses. We rely on third-party manufacturers to produce and process our product candidate. Our ability to obtain clinical supplies of our product candidates could be disrupted if the operations of these suppliers are affected by a man-made or natural disaster or other business interruption.

Any legal proceedings or claims against us could be costly and time-consuming to defend and could harm our reputation regardless of the outcome.

We may in the future become subject to legal proceedings and claims that arise in the ordinary course of business, including intellectual property, product liability, employment, wage and hour, class action, derivative, whistleblower and other litigation claims, and governmental and other regulatory investigations and proceedings. Such matters can be time-consuming, divert management's attention and resources, cause us to incur significant expenses or liability, or require us to change our business practices. In addition, the expense of litigation, for which we are either not insured or only partially insured depending on the claim, and the timing of this expense from period to period will be difficult to estimate, subject to change, and could adversely affect our financial condition and results of operations. Because of the potential risks, expenses, and uncertainties of litigation, we may, from time to time, settle disputes, even where we have meritorious claims or defenses, by agreeing to settlement agreements. Any of the foregoing could adversely affect our business, financial condition, and results of operations.

Risks Related to Intellectual Property

If we are unable to protect our intellectual property effectively, we may be unable to prevent third parties from using our technologies, which would impair our competitive advantage.

We rely on patent protection as well as a combination of trademark, copyright and trade secret protection, and other contractual restrictions, to protect our proprietary technologies, all of which provide limited protection and may not adequately protect our rights or permit us to gain or keep any competitive advantage. We may not be successful in defending challenges made in connection with our patents and patent applications. If we fail to protect our intellectual property, we will be unable to prevent third parties from using our technologies and they will be able to compete more effectively against us.

In addition to our patents, we rely on contractual restrictions to protect our proprietary technology. We require our employees and third parties to sign confidentiality agreements and our employees are also required to sign agreements assigning to us all intellectual property arising from their work for us. Nevertheless, we cannot guarantee that these measures will be effective in protecting our intellectual property rights. Any failure to protect our intellectual property rights could materially adversely affect our business, prospects and financial condition.

Our currently pending or future patent applications may not result in issued patents and any patents issued to us may be challenged, invalidated, or held unenforceable. Furthermore, we cannot be certain that we were the first to make the invention claimed in our issued patents or pending patent applications in the U.S., or that we were the first to file for protection of the inventions claimed in our foreign issued patents or pending patent applications. In addition, there are numerous recent changes to the patent laws and proposed changes to the rules of the United States Patent and Trademark Office ("USPTO"), which may have a significant impact on our ability to protect our technology and enforce our intellectual property rights. For example, in September 2011, the U.S. enacted sweeping changes to the U.S. patent system under the Leahy-Smith America Invents Act, including changes that transitioned the U.S. from a "first-to-invent" system to a "first-to-file" system and alter the processes for challenging issued patents. These changes could increase the uncertainties and costs surrounding the prosecution of our patent applications and the enforcement or defense of our issued patents. In addition, we may become subject to interference proceedings conducted in the patent and trademark offices of various countries to determine our entitlement to patents, and these proceedings may conclude that other patents or patent applications have priority over our patents or patent applications. It is also possible that a competitor may successfully challenge our patents through various proceedings and those challenges may result in the elimination or narrowing of our patents, and therefore reduce our patent protection. Accordingly, rights under any of our issued patents, patent applications or future patents may not provide us with commercially meaningful protection for our products or afford us a commercial advantage against our competitors or their competitive products or processes.

51

If our trademarks and trade names are not adequately protected, then we may not be able to build name recognition in our markets of interest and our business may be adversely affected.

Our unregistered trademarks or trade names may be challenged, infringed, circumvented, or declared generic or determined to be infringing on other marks. We may not be able to protect our rights to these trademarks and trade names, which we need to build name recognition among potential collaborators or customers in our markets of interest. At times, competitors may adopt trade names or trademarks similar to ours, thereby impeding our ability to build brand identity and possibly leading to market confusion. In addition, there could be potential trade name or trademarks infringement claims brought by owners of other registered trademarks or trademarks that incorporate variations of our unregistered trademarks or trade names. Over the long term, if we are unable to successfully register our trademarks and trade names and establish name recognition based on our trademarks and trade names, then we may not be able to compete effectively, and our business may be adversely affected. Our efforts to enforce or protect our proprietary rights related to trademarks, trade secrets, domain names, copyrights or other intellectual property may be ineffective and could result in substantial costs and diversion of resources and could adversely impact our financial condition or results of operations.

The patents issued to us may not be broad enough to provide any meaningful protection, one or more of our competitors may develop more effective technologies, designs or methods without infringing our intellectual property rights and one or more of our competitors may design around our proprietary technologies.

If we are not able to protect our proprietary technology, trade secrets and know-how, our competitors may use our inventions to develop competing products. Our patents may not protect us against our competitors, and patent litigation is very expensive. We may not have sufficient cash available to pursue any patent litigation to its conclusion because we currently do not generate revenues other than licensing, milestone and royalty income.

We cannot rely solely on our current patents to be successful. The standards that the USPTO and foreign patent offices use to grant patents, and the standards that U.S. and foreign courts use to interpret patents, are not the same, are not always applied predictably or uniformly and can change, particularly as new technologies develop. As such, the degree of patent protection obtained in the U.S. may differ substantially from that obtained in various foreign countries.

We cannot be certain of the level of protection, if any, that will be provided by our patents if they are challenged in court, where our competitors may raise defenses such as invalidity, unenforceability, or possession of a valid license. In addition, the type and extent of any patent claims that may be issued to us in the future are uncertain. Any patents that are issued may not contain claims that will permit us to stop competitors from using similar technology.

We may incur substantial costs as a result of litigation or other proceedings relating to patent and other intellectual property rights.

Third parties may challenge the validity, inventorship or ownership of our patents and other intellectual property rights, resulting in costly litigation or other time-consuming and expensive proceedings, which could deprive us of valuable rights. If we become involved in any intellectual property litigation, interference or other judicial or administrative proceedings, we will incur substantial expenses and the attention of our technical and management personnel will be diverted. An adverse determination may subject us to significant liabilities or require us to seek licenses that may not be available from third parties on commercially favorable terms, if at all. Further, if such claims are proven valid, through litigation or otherwise, we may be required under applicable law to pay substantial monetary damages, which can be tripled if the infringement is deemed willful, or be required to discontinue or significantly delay development, marketing, selling and licensing of the affected products and intellectual property rights.

Our competitors may have filed, and may in the future file, patent applications covering technology similar to ours. Any such patent application may have priority over our patent applications and could further require us to obtain rights to issued patents covering such technologies. There may be third-party patents, patent applications and other intellectual property relevant to our potential products that may block or compete with our potential products or processes. If another party has filed a U.S. patent application on inventions similar to ours, we may have to participate in an interference proceeding declared by the USPTO to determine priority of invention in the U.S. The costs of these proceedings could be substantial, and it is possible that such efforts would be unsuccessful, resulting in a loss of our U.S. patent position with respect to such inventions. In addition, we cannot assure you that we would prevail in any of these suits or that the damages or other remedies that we are ordered to pay, if any, would not be substantial. Claims of intellectual property infringement, misappropriation or other violations against us may require us to enter into royalty or license agreements with third parties that may not be available on acceptable terms, if at all. We may also be subject to injunctions against the further development and use of our technology, which could materially adversely affect our business, prospects and financial condition.

52

Some of our competitors may be able to sustain the costs of complex patent litigation more effectively than we can because they have substantially greater resources. In addition, any uncertainties resulting from the initiation and continuation of any litigation could materially adversely affect our ability to raise the funds necessary to continue our operations.

Obtaining and maintaining patent protection depends on compliance with various procedural, document submission, fee payment and other requirements imposed by governmental patent agencies, and our patent protection could be reduced or eliminated for non-compliance with these requirements.

Periodic maintenance fees on any issued patent are due to be paid to the USPTO and foreign patent agencies in several stages over the lifetime of the patent. In the United States, patents have a limited lifespan, and if all maintenance fees are timely paid, the natural expiration of a patent is generally 20-years from its earliest U.S. non-provisional filing date. The USPTO and various foreign governmental patent agencies require compliance with a number of procedural, documentary, fee payment and other provisions during the patent application process and following the issuance of a patent. While an inadvertent lapse can in many cases be cured by payment of a late fee or by other means in accordance with the applicable rules, there are situations in which noncompliance can result in abandonment or lapse of the patent or patent application, resulting in partial or complete loss of patent rights in the relevant jurisdiction. Noncompliance events that could result in abandonment or lapse of a patent or patent application include, but are not limited to, failure to respond to official actions within prescribed time limits, non-payment of fees and failure to properly legalize and submit formal documents. In certain circumstances, even inadvertent noncompliance events may permanently and irrevocably jeopardize patent rights. In such an event, our competitors might be able to enter the market, which would have a material adverse effect on our business.

We rely on confidentiality agreements to protect our trade secrets. If these agreements are breached by our employees or other parties, our trade secrets may become known to our competitors.

We rely on trade secrets that we seek to protect through confidentiality agreements with our employees and other parties. If these agreements are breached, our competitors may obtain and use our trade secrets to gain a competitive advantage over us. We may not have any remedies against our competitors and any remedies that may be available to us may not be adequate to protect our business or compensate us for the damaging disclosure. In addition, we may have to expend resources to protect our interests from possible infringement by others.

Risks Related to Our Common Stock

An active trading market for our common stock may not be sustained.

Prior to the closing of our initial public offering in August 2021, there was no public trading market for our common stock. Although our common stock is listed on the Nasdaq Capital Market, the market for our shares has demonstrated varying levels of trading activity. Our ability to raise capital to continue to fund operations by selling shares of our common stock and our ability to acquire other companies or technologies by using shares of our common stock as consideration may be impaired if an active trading market for our common stock is not sustained.

The market price of our common stock may be volatile and fluctuate substantially, which could result in substantial losses for investors.

The market price of our common stock is likely to be highly volatile and may be subject to wide fluctuations in response to a variety of factors, some of which are beyond our control. These factors include the following:

- any delay in the commencement, enrollment and ultimate completion of our clinical trials;
- any delay in submitting an NDA and any adverse development or perceived adverse development with respect to the FDA's review of that NDA;
- failure to successfully develop and commercialize RenovoGem;
- inability to obtain additional funding;

- adverse regulatory decisions;
- changes in the structure of healthcare payment systems;
- inability to obtain adequate product supply for RenovoGem, RenovoCath or any other product candidate, or the inability to do so at acceptable prices;
- introduction of new products, services or technologies by our competitors;
- failure to meet or exceed financial projections we provide to the public;
- failure to meet or exceed the estimates and projections of the investment community;
- changes in the market valuations of companies similar to ours;
- market conditions in the pharmaceutical and biotechnology sectors, and the issuance of new or changed securities analysts' reports or recommendations;
- announcements of significant acquisitions, strategic collaborations, joint ventures or capital commitments by us or our competitors;
- significant lawsuits, including patent or stockholder litigation, and disputes or other developments relating to our proprietary rights, including patents, litigation matters
 and our ability to obtain patent protection for our technologies;
- additions or departures of key scientific or management personnel;
- sales of our common stock by us, our insiders or our other stockholders;
- · expiration of market standoff or lock-up agreements;
- trading volume of our common stock;
- fluctuations in interest rates and inflation rates;
- · general economic, industry and market conditions;
- health epidemics and outbreaks, including the COVID-19 pandemic, or other natural or manmade disasters which could significantly disrupt our preclinical studies and clinical trials, and therefore our receipt of necessary regulatory approvals could be delayed or prevented; and
- the other factors described in this "Risk Factors" section.

In addition, the stock markets have experienced extreme price and volume fluctuations that have affected and continue to affect the market prices of equity securities of many companies. These fluctuations have often been unrelated or disproportionate to the operating performance of those companies. Broad market and industry factors, as well as general economic, political, regulatory and market conditions, may negatively affect the market price of our common stock, regardless of our actual operating performance. In particular, stock markets have experienced extreme volatility due to the ongoing COVID-19 pandemic and investor concerns and uncertainty related to the impact of the pandemic on the economies of countries worldwide.

Raising additional capital may cause dilution to our existing stockholders, restrict our operations or require us to relinquish rights to our product candidates on unfavorable terms to us.

We may seek additional capital through a variety of means, including through public or private equity, debt financings or other sources, including up-front payments and milestone payments from strategic collaborations. To the extent that we raise additional capital through the sale of equity or convertible debt or equity securities, your ownership interest will be diluted, and the terms may include liquidation or other preferences that adversely affect your rights as a stockholder. Such financing may result in dilution to stockholders, imposition of debt covenants, increased fixed payment obligations or other restrictions that may affect our business. If we raise additional funds through up-front payments or milestone payments pursuant to strategic collaborations with third parties, we may have to relinquish valuable rights to our product candidates or grant licenses on terms that are not favorable to us. In addition, we may seek additional capital due to favorable market conditions or strategic considerations even if we believe we have sufficient funds for our current or future operating plans.

54

The Nasdaq Stock Market may delist our securities from its exchange, which could limit investors' ability to make transactions in our securities and subject us to additional trading restrictions.

Our common stock is listed on the Nasdaq Capital Market. We cannot assure you that, in the future, our securities will meet the continued listing requirements to be listed on the Nasdaq Capital Market. If the Nasdaq Stock Market delists our common stock, we could face significant material adverse consequences, including:

- a limited availability of market quotations for our securities;
- a determination that our common stock is a "penny stock" which will require brokers trading in our common stock to adhere to more stringent rules and possibly resulting in a reduced level of trading activity in the secondary trading market for our common stock;
- a limited amount of news and analyst coverage for our company; and
- a decreased ability to issue additional securities or obtain additional financing in the future.

We could be subject to securities class action litigation.

In the past, securities class action and derivative litigation has often been brought against companies following a decline in the market price of their securities or upon the occurrence of other corporate events. This risk is especially relevant for us because biotechnology companies have experienced significant share price volatility in recent

years. If we face such litigation, it could result in substantial costs, for which we are not insured, and a diversion of management's attention and resources, which could harm our business.

If securities or industry analysts do not publish research or publish inaccurate or unfavorable research about our business, the market price for the shares and trading volume could decline.

The trading market for our common stock depends in part on the research and reports that securities or industry analysts publish about us or our business. If our revenue or operating results fall below the expectations of analysts or investors or below any forecasts we may provide to the market, or if the forecasts we provide to the market are below the expectations of analysts or investors, the price of our common stock could decline substantially. If one or more of the analysts who covers us downgrades our common stock or publishes inaccurate or unfavorable research about our business, the market price for our common stock would likely decline. If one or more of these analysts cease coverage of our company or fail to publish reports on us regularly, we could lose visibility in the financial markets, which, in turn, could cause the market price or trading volume for our common stock to decline.

We do not expect to pay dividends in the foreseeable future, and you must rely on price appreciation of your shares for return on your investment.

We have paid no cash dividends on any class of our stock to date and we do not anticipate paying cash dividends in the near term. For the foreseeable future, we intend to retain any earnings to finance the development and expansion of our business, and we do not anticipate paying any cash dividends on our stock. Accordingly, investors must be prepared to rely on sales of their shares of common stock after price appreciation to earn an investment return, which may never occur. Investors seeking cash dividends should not purchase our shares of common stock. Any determination to pay dividends in the future will be made at the discretion of our board of directors and will depend on our results of operations, financial condition, contractual restrictions, restrictions imposed by applicable law and other factors our board deems relevant.

55

We have incurred and will continue to incur increased costs as a result of operating as a public company, and our management has devoted and will continue to devote substantial time to new compliance initiatives and corporate governance practices.

As a public company, we have incurred and particularly after we no longer qualify as an emerging growth company, we will continue to incur significant legal, accounting and other expenses that we did not incur as a private company. The Sarbanes-Oxley Act of 2002 ("SOX"), the Dodd-Frank Wall Street Reform and Consumer Protection Act, the listing requirements of Nasdaq, and other applicable securities rules and regulations impose various requirements on U.S. reporting public companies, including the establishment and maintenance of effective disclosure and financial controls and corporate governance practices. Our management and other personnel will need to continue to devote a substantial amount of time to these compliance initiatives. Moreover, these rules and regulations have increased our legal and financial compliance costs and have made some activities more time-consuming and costly. For example, we expect that these rules and regulations may make it more expensive for us to obtain director and officer liability insurance, which in turn could make it more difficult for us to attract and retain qualified senior management personnel or members for our board of directors. In addition, these rules and regulations are often subject to varying interpretations, and, as a result, their application in practice may evolve over time as new guidance is provided by regulatory and governing bodies. This could result in continuing uncertainty regarding compliance matters and higher costs necessitated by ongoing revisions to disclosure and governance practices.

Pursuant to Section 404 of SOX ("Section 404"), we will be required to furnish a report by our senior management on our internal control over financial reporting beginning with our second filing of an Annual Report on Form 10-K with the SEC.

However, while we remain an emerging growth company, we will not be required to include an attestation report on internal control over financial reporting issued by our independent registered public accounting firm. To prepare for eventual compliance with Section 404, once we no longer qualify as an emerging growth company, we will be engaged in a process to document and evaluate our internal control over financial reporting, which is both costly and challenging. In this regard, we will need to continue to dedicate internal resources, potentially engage outside consultants and adopt a detailed work plan to assess and document the adequacy of internal control over financial reporting, continue steps to improve control processes as appropriate, validate through testing that controls are functioning as documented and implement a continuous reporting and improvement process for internal control over financial reporting. Despite our efforts, there is a risk that we will not be able to conclude, within the prescribed timeframe or at all, that our internal control over financial reporting is effective as required by Section 404.

We have identified material weaknesses in our internal control over financial reporting. Failure to maintain effective internal controls could cause our investors to lose confidence in us and adversely affect the market price of our common stock. If our internal controls are not effective, we may not be able to accurately report our financial results or prevent fraud.

Effective internal control over financial reporting is necessary for us to provide reliable financial reports in a timely manner. In connection with the audit of our financial statements as of and for the years ended December 31, 2019 and 2020 and the review of our financial statements for the six months ended June 30, 2020 and 2021, we identified material weaknesses in our internal control over financial reporting. A material weaknesse is a deficiency, or combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of the annual or interim financial statements will not be prevented or detected on a timely basis. Specifically, we determined that we lacked a sufficient number of qualified accounting and financial reporting personnel with an appropriate level of knowledge, training and experience to address complex accounting issues, sufficient written policies and procedures for accounting and financial reporting in accordance with U.S. Generally Accepted Accounting Principles ("GAAP"), and adequate management review controls. In addition, we determined that our financial statement close process includes significant control gaps mainly driven by the small size of our accounting and finance staff and, as a result, a significant lack of appropriate segregation of duties.

The above material weaknesses could result in a misstatement of our account balances or disclosures that would result in a material misstatement of our annual or interim financial statements that would not be prevented or detected. To address the material weaknesses, we have implemented, and are continuing to implement, measures designed to improve internal control over financial reporting, including expanding our accounting and finance team to add additional qualified accounting and finance resources, which may include third party consultants, and new financial processes. We intend to continue to take steps to remediate the material weaknesses through the hiring or engagement of additional experienced accounting and financial reporting personnel, formalizing documentation of policies and procedures and further evolving the accounting processes, including implementing appropriate segregation of duties. We expect to incur additional costs to remediate these weaknesses, including personnel, consulting and other costs.

We may not be successful in implementing these changes or in developing other internal controls, which may undermine our ability to provide accurate, timely and reliable reports on our financial and operating results. Further, we will not be able to fully assess whether the steps we are taking will remediate the material weakness in our internal control over financial reporting until we have completed our implementation efforts and sufficient time passes in order to evaluate their effectiveness. In addition, until we remediate these weaknesses, or if we identify additional material weaknesses in our internal control over financial reporting, we may not detect errors on a timely basis and our financial statements may be materially misstated. Moreover, in the future we may engage in business transactions, such as acquisitions, reorganizations or implementation of new information systems that could negatively affect our internal control over financial reporting and result in material weaknesses.

result of such failures, we could also become subject to investigations by the stock exchange on which our securities are listed, the SEC, or other regulatory authorities, and become subject to litigation from investors and stockholders, which could harm our reputation, financial condition or divert financial and management resources from our core business

Our disclosure controls and procedures may not prevent or detect all errors or acts of fraud.

As a public company, we are subject to the periodic reporting requirements of the Exchange Act. We designed our disclosure controls and procedures to reasonably assure that information we must disclose in reports we file or submit under the Exchange Act is accumulated and communicated to management, and recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC. We believe that any disclosure controls and procedures or internal controls and procedures, no matter how well-conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met.

These inherent limitations include the fact that judgments in decision-making can be faulty and that breakdowns can occur because of simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people or by an unauthorized override of the controls. Accordingly, because of the inherent limitations in our control system, misstatements due to error or fraud may occur and not be detected.

We are an "emerging growth company," and the reduced reporting requirements applicable to emerging growth companies may make our common stock less attractive to investors.

We are an "emerging growth company," as defined in the JOBS Act. For as long as we continue to be an emerging growth company, we may take advantage of exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies, including exemption from compliance with the auditor attestation requirements of Section 404; exemption from any requirement that may be adopted by the Public Company Accounting Oversight Board regarding mandatory audit firm rotation or a supplement to the auditor's report providing additional information about the audit and the financial statements; reduced disclosure obligations regarding executive compensation; and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. As a result, the information we provide stockholders will be different than the information that is available with respect to other public companies. We will remain an emerging growth company until the earlier of (1) the last day of the fiscal year (a) following the fifth anniversary of the closing of our initial public offering, (b) in which we have total annual gross revenue of at least \$1.07 billion or (c) in which we are deemed to be a large accelerated filer, which means the market value of our common stock held by non-affiliates exceeds \$700 million as of the end of our prior second fiscal quarter, or (2) the date on which we have issued more than \$1.0 billion in non-convertible debt during the prior three-year period.

In addition, under the JOBS Act, emerging growth companies may delay adopting new or revised accounting standards until such time as those standards apply to private companies. We may elect not to avail ourselves of this exemption from new or revised accounting standards and, therefore, may be subject to the same new or revised accounting standards as other public companies that are not emerging growth companies.

We cannot predict if investors will find our common stock less attractive because we may rely on these exemptions. If some investors find our common stock less attractive as a result of these exemptions, there may be a less active trading market for our common stock and our share price may be more volatile.

Anti-takeover provisions contained in our certificate of incorporation and bylaws, as well as provisions of Delaware law, could impair a takeover attempt.

Our certificate of incorporation, bylaws and Delaware law contain provisions that could have the effect of rendering more difficult, delaying or preventing an acquisition deemed undesirable by our board of directors. Our corporate governance documents include provisions:

- authorizing "blank check" preferred stock, which could be issued by our board of directors without stockholder approval and may contain voting, liquidation, dividend, and other rights superior to our common stock;
- limiting the liability of, and providing indemnification to, our directors and officers;
- limiting the ability of our stockholders to call and bring business before special meetings;

5

- requiring advance notice of stockholder proposals for business to be conducted at meetings of our stockholders and for nominations of candidates for election to our board of directors:
- controlling the procedures for the conduct and scheduling of board of directors and stockholder meetings; and
- providing our board of directors with the express power to postpone previously scheduled annual meetings and to cancel previously scheduled special meetings.

These provisions, alone or together, could delay or prevent hostile takeovers and changes in control or changes in our management.

As a Delaware corporation, we are also subject to provisions of Delaware law, including Section 203 of the Delaware General Corporation law, which prevents some stockholders holding more than 15% of our outstanding common stock from engaging in certain business combinations without approval of the holders of substantially all of our outstanding common stock.

Any provision of our certificate of incorporation, bylaws or Delaware law that has the effect of delaying or deterring a change in control could limit the opportunity for our stockholders to receive a premium for their shares of our common stock, and could also affect the price that some investors are willing to pay for our common stock.

Our certificate of incorporation, as amended, designates the Court of Chancery of the State of Delaware as the sole and exclusive forum for certain types of actions and proceedings that may be initiated by our stockholders, which could limit our stockholders' ability to obtain a favorable judicial forum for disputes with us or our directors, officers or other employees.

Our certificate of incorporation requires that, unless we consent in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware will, to the fullest extent permitted by law, be the sole and exclusive forum for each of the following:

- any derivative action or proceeding brought on our behalf;
- any action asserting a claim for breach of any fiduciary duty owed by any director, officer or other employee of ours to the Company or our stockholders, creditors or other constituents:
- any action asserting a claim against us or any director or officer of ours arising pursuant to, or a claim against us or any of our directors or officers, with respect to the interpretation or application of any provision of, the DGCL, our certificate of incorporation or bylaws; or
- any action asserting a claim governed by the internal affairs doctrine;

provided, that, if and only if the Court of Chancery of the State of Delaware dismisses any of the foregoing actions for lack of subject matter jurisdiction, any such action or actions may be brought in another state court sitting in the State of Delaware.

The exclusive forum provision is limited to the extent permitted by law, and it will not apply to claims arising under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or for any other federal securities laws which provide for exclusive federal jurisdiction.

Furthermore, Section 22 of the Securities Act creates concurrent jurisdiction for federal and state courts over all such Securities Act actions. Accordingly, both state and federal courts have jurisdiction to entertain such claims. To prevent having to litigate claims in multiple jurisdictions and the threat of inconsistent or contrary rulings by different courts, among other considerations, our second amended and restated certificate of incorporation provides that the federal district courts of the United States of America will be the exclusive forum for resolving any complaint asserting a cause of action arising under the Securities Act. While the Delaware courts have determined that such choice of forum provisions are facially valid, a stockholder may nevertheless seek to bring such a claim arising under the Securities Act against us, our directors, officers, or other employees in a venue other than in the federal district courts of the United States of America. In such instance, we would expect to vigorously assert the validity and enforceability of the exclusive forum provisions of our second amended and restated certificate of incorporation.

58

Although we believe this provision benefits us by providing increased consistency in the application of Delaware law in the types of lawsuits to which it applies, this provision may limit or discourage a stockholder's ability to bring a claim in a judicial forum that it finds favorable for disputes with us or our directors, officers or other employees, which may discourage such lawsuits against us and our directors, officers and other employees. Alternatively, if a court were to find the choice of forum provision contained in our certificate of incorporation to be inapplicable or unenforceable in an action, we may incur additional costs associated with resolving such action in other jurisdictions, which could adversely affect our business and financial condition.

We note that there is uncertainty as to whether a court would enforce the provision and that investors cannot waive compliance with the federal securities laws and the rules and regulations thereunder. Although we believe this provision benefits us by providing increased consistency in the application of Delaware law in the types of lawsuits to which it applies, the provision may have the effect of discouraging lawsuits against our directors and officers.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Unregistered Sales of Equity Securities

None

Use of Proceeds from Public Offering of Common Stock

On August 25, 2021, our Registration Statement on Form S-1 was declared effective by the SEC for our initial public offering of common stock. We began trading on the Nasdaq Capital Market on August 26, 2021, and the transaction formally closed on August 30, 2021. In connection with our IPO, we issued and sold an aggregate of 1,850,000 units of securities at a price of \$9.00 per unit. Each unit consisted of (a) one share of common stock and (b) one warrant to purchase one share of common stock at an exercise price equal to \$10.80 per share. The aggregate offering price for shares sold in our IPO was \$16.7 million. The sole book-running manager for the initial public offering was Roth Capital Partners. Upon the closing of the IPO, we issued an underwriter's warrant to Roth Capital Partners, LLC entitling it to purchase 277,500 shares of common stock at an exercise price of \$10.80. After deducting underwriting discounts and commissions of \$1.3 million and other offering costs paid or payable by us of \$0.8 million, the net proceeds from the offering were \$14.6 million. No payments were made by us to directors, officers or persons owning ten percent or more of our common stock or to their associates, or to our affiliates, other than payments in the ordinary course of business to officers for salaries and to non-employee directors pursuant to our director compensation policy. The offer and sale of our common stock units in our initial public offering was effected through a Registration Statement on Form S-1 (File No. 333-258071).

There has been no material change in the planned use of proceeds from our IPO as described in our final prospectus filed with the SEC on August 27, 2021 pursuant to Rule 424(b)(4). We invested the funds received in interest-bearing U.S. Treasury securities.

Item 3. Defaults Upon Senior Securities

None.

Item 4. Mine Safety Disclosures

None.

Item 5. Other Information

On November 11, 2021, the Company's Board of Directors approved the following, which are described in detail below: (i) a Confirmatory Employment Letter with Shaun R. Bagai, our Chief Executive Officer, (ii) an Amendment to Consulting Agreement with Dr. Ramtin Agah, our Chief Medical Officer, (iii) Change in Control and Severance Agreements with Mr. Bagai and Dr. Agah, and (iv) a Key Service Provider Incentive Compensation Plan.

Confirmatory Employment Letter with Shaun R. Bagai

On November 11, 2021, we entered into a Confirmatory Employment Letter with Mr. Bagai. The Confirmatory Employment Letter has no specific term and provides that Mr. Bagai is an at-will employee and is eligible to enter into a Severance Agreement (as defined below) with the Company. The Confirmatory Employment Letter supersedes all existing agreements and understandings that Mr. Bagai may have entered into concerning his employment relationship with us. Mr. Bagai's current annual base salary is \$363,000, and he is eligible for an annual target cash incentive bonus equal to 50% of his annual base salary. He will also be entitled to receive other employee benefits generally available to all employees of the Company. The description of the Confirmatory Employment Letter is qualified in its entirety by reference to the full text of letter filed herewith as Exhibit 10.4 and incorporated herein by reference.

59

Amendment to Consulting Agreement with Dr. Ramtin Agah

On November 11, 2021, we entered into a third amendment to the Consulting Agreement with Dr. Agah (the "Amendment"), which continues in force for as long as Dr. Agah is providing consulting services and may be terminated by either party on thirty (30) days' notice. The Amendment provides for a monthly consulting fee of \$21,667.67, based on Dr. Agah spending no less than 24 hours per week on Company matters. The Company may, in its discretion, proportionally adjust the monthly consulting fee if Dr. Agah's time commitment decreases. The Amendment also provides for Dr. Agah's eligibility for an annual target cash incentive bonus equal to 35% of his annualized base consulting fee. The description of the Amendment is qualified in its entirety by reference to the full text of Amendment filed herewith as Exhibit 10.6 and incorporated herein by reference.

Change In Control and Severance Agreement

On November 11, 2021, we entered into a Change in Control and Severance Agreement, or the Severance Agreement, with each of Mr. Bagai and Dr. Agah (each an "Executive" and collectively, the "Executives"). Capitalized terms used herein and not otherwise defined shall have the meaning assigned to such term in the Severance Agreement. The description of each Severance Agreement is qualified in its entirety by reference to the full text of Mr. Bagai's and Dr. Agah's Severance Agreements filed herewith as Exhibits 10.7 and 10.8, respectively, and incorporated herein by reference.

Each Severance Agreement will continue indefinitely until terminated by written consent of the parties to the Severance Agreement.

Termination Outside of Change in Control Period

Under the Severance Agreement, if Mr. Bagai or Dr. Agah are terminated outside a period beginning on the date of a Change in Control and ending on (and inclusive of) the date that is the one-year anniversary of a Change in Control (the "Change in Control Period"), either by us without Cause (other than due to death or Disability) or by the Executive for Good Reason, they will receive:

- Annual Base Compensation Severance: A single, lump sum payment equal to the specified percent of the Executive Officer's Annual Base Compensation (which is base salary or, for Dr. Agah, annual base consulting fee, if applicable), which was in effect immediately before such termination (or, if the termination is due to a resignation for Good Reason based on a material reduction in the Executive Officer's annual base salary (or, for Dr. Agah, annual base consulting fee, if applicable), then Executive's annual base salary (or, for Dr. Agah, annual base consulting fee, if applicable) in effect immediately prior to the reduction), or if greater, the base salary (or, for Dr. Agah, annual base consulting fee, if applicable) in effect immediately prior to the Change in Control):
 - (i) Mr. Bagai: 100% of Annual Base Compensation
 - (ii) Dr. Agah: 50% of Annual Base Compensation; and
- Bonus Severance: A single, lump sum payment of the pro rata portion (based on period of employment) of the Executive Officer's target bonus in effect immediately before such termination or if greater, the target bonus in effect immediately prior to the Change in Control; and
- COBRA Severance: payment or reimbursement of COBRA continuation coverage premiums for group health, dental and vision coverage for the Executive Officer and his eligible dependents, for:
 - (i) Mr. Bagai: up to 12 months
 - (ii) Dr. Agah (if he was an employee immediately prior to the termination): up to 6 months

Or, if providing such payment would violate applicable law, a taxable payment for an equivalent amount in lieu thereof.

Termination During Change in Control Period

Under the Severance Agreement, if Mr. Bagai or Dr. Agah are terminated during the Change in Control Period, either by us without Cause (other than due to death or Disability), or by the Executive for Good Reason, they will receive:

- Base Compensation Severance: A single, lump sum payment equal to the specified percent of the Executive Officer's Annual Base Compensation:
 - (i) Mr. Bagai: 150% of Annual Base Compensation
 - (ii) Dr. Agah: 100% of Annual Base Compensation; and
- COBRA Severance: payment or reimbursement of COBRA continuation coverage premiums for group health, dental and vision coverage for the executive officer and his eligible dependents, for:
 - (i) Mr. Bagai: up to 18 months
 - (ii) Dr. Agah (if he was an employee immediately prior to the termination): up to 12 months

Or, if providing such payment would violate applicable law, a taxable payment for an equivalent amount in lieu thereof; and

Vesting Acceleration of Service-based Equity Awards: Full vesting of the outstanding and unvested equity awards (other than equity award subject to performance-based vesting criteria).

60

The Severance Agreement provides that if any payments or benefits received by Mr. Bagai or Dr. Agah under the Severance Agreement or otherwise would constitute "parachute payments" within the meaning of Section 280G of the Internal Revenue Code (the "Code") and be subject to excise taxes imposed by Section 4999 of the Code, such amount will either be delivered in full or reduced so as not to be subject to excise taxation, whichever amount is higher. The Severance Agreement does not require us to provide any tax gross-ups.

To receive the severance described above, the Executive Officers must sign and not revoke our standard separation agreement and release of claims within the timeframe that is set forth in the Severance Policy.

Key Service Provider Incentive Compensation Plan

The Key Service Provider Incentive Compensation Plan, or our Bonus Plan, allows our Compensation Committee to (i) determine which employees or other service providers may receive incentive awards under the Bonus Plan, and (ii) provide incentive awards to selected employees, including our named executive officers and other service providers, which may be based upon performance goals established by our Compensation Committee. Our Compensation Committee, in its sole discretion, may establish a target award for each participant under the Bonus Plan, which may be expressed as a percentage of the participant's average annual base salary for the applicable performance period or a fixed dollar amount or such other amount or based on such other formula or factors as the Compensation Committee determines.

The Bonus Plan is administered by our Board or a committee appointed by our Board. Unless and until our Board determines otherwise, our Compensation Committee will be the administrator of the Bonus Plan.

Under the Bonus Plan, our Compensation Committee will determine the performance goals, if any, applicable to awards and such performance goals may differ from participant to participant and from award to award. Performance goals may be based on any factors our Compensation Committee determines relevant, including, without limitation, on an individual, divisional, portfolio, project, business unit, segment, or Company-wide basis, and may include criteria related to research and development, regulatory, business development, financial and operational performance or other subjective or objective criteria. Any criteria used may be measured on such basis as our Compensation Committee determines. As determined by our Compensation Committee, the performance goals may be based on GAAP or non-GAAP results and any actual

results may be adjusted by our Compensation Committee for one-time items or unbudgeted or unexpected items and/or payment of actual awards when determining whether the performance goals have been met.

Our Compensation Committee, at any time prior to payment of an actual award, may increase, reduce or eliminate a participant's actual award, and/or increase, reduce or eliminate the amount allocated to the bonus pool. The actual award may be below, at or above a participant's target award, in our Compensation Committee's discretion. Our Compensation Committee may determine the amount of any increase, reduction or elimination of an actual award based on such factors as it deems relevant, and it will not be required to establish any allocation or weighting with respect to the factors it considers.

Actual awards generally will be paid in cash (or its equivalent) in a single lump sum. The Compensation Committee reserves the right to settle an actual award with a grant of an equity award with such terms and conditions, including any vesting requirements, as determined by the Compensation Committee. Unless otherwise determined by our Compensation Committee, to earn an actual award, a participant must be employed by us (or an affiliate of us, as applicable) on the date the bonus is paid. Payment of bonuses occurs as soon as practicable after the end of the applicable performance period, but no later than the dates set forth in the Bonus Plan.

All awards under the Bonus Plan will be subject to reduction, cancellation, forfeiture, or recoupment in accordance with any clawback policy that we are required to adopt pursuant to any rule, regulation or law. Our Compensation Committee may also impose such other clawback, recovery or recoupment provisions with respect to an award under the Bonus Plan as it may determine is necessary or appropriate.

If we are required to prepare an accounting restatement due to our material noncompliance, as a result of misconduct, with any financial reporting requirement under the securities laws, then any participant who knowingly or through gross negligence engaged in, or failed to prevent, the misconduct, will reimburse us for the amount of any payment with respect to an award earned or accrued under the Bonus Plan during the twelve month period following the first to occur of the public issuance or filing with the SEC, of the financial document embodying such financial reporting requirement.

Our Board or its Compensation Committee will have the authority to amend or terminate the Bonus Plan provided such action does not alter or impair the existing rights of any participant with respect to any earned bonus without the participant's consent. The Bonus Plan will remain in effect until terminated in accordance with the terms of the Bonus Plan.

The description of the Bonus Plan is qualified in its entirety by reference to the full text of the Bonus Plan filed herewith as Exhibit 10.9 and incorporated herein by reference.

61

Item 6. Exhibits

		Incorporated by Reference			
Exhibit Number	Exhibit Description	Form	File No.	Exhibit	Filing Date
3.1	Sixth Amended and Restated Certificate of Incorporation of RenovoRx, Inc.	8-K	001-40738	3.1	August 31, 2021
3.2	Amended and Restated Bylaws of RenovoRx, Inc.	8-K	001-40738	3.2	August 31, 2021
4.1±	Form of Private Common Stock Warrant (related to the 2020 Convertible Notes and 2021 Convertible Notes)	Filed herewith			
10.1	Amended and Restated Investor Rights Agreement, dated as of April 18, 2018	Filed herewith			
10.2	2021 Omnibus Equity Incentive Plan and Forms of Stock Option Grant Notice and Option Agreement	Filed herewith			
10.3*	Outside Director Compensation Policy	Filed herewith			
10.4*	Confirmatory Offer Letter, by and between RenovoRx, Inc. and Shaun Bagai, dated November 11, 2021	Filed herewith			
10.5*	Consulting Agreement, by and between RenovoRx, Inc. and Ramtin Agah, MD, dated January 1, 2018 (Replaces <i>incorrect exhibit</i> previously filed.)	Filed herewith			
10.6*	Amendment to Consulting Agreement, by and between RenovoRx, Inc. and Ramtin Agah, MD, dated November 11, 2021	Filed herewith			
10.7*	Change in Control and Severance Agreement, by and between RenovoRx, Inc. and Shaun Bagai, effective as of November 11, 2021	Filed herewith			
10.8*	Change in Control and Severance Agreement, by and between RenovoRx, Inc. and Ramtin Agah, effective as of November 11, 2021	Filed herewith			
10.9*	Key Service Provider Incentive Compensation Plan	Filed herewith			
31.1	Certification of Principal Executive Officer pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	Filed herewith			
31.2	Certification of Principal Financial Officer pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	Filed herewith			
32.1†	Certification of Principal Executive Officer pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.	Furnished herewith			

32.2†	Certification of Principal Financial Officer pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.	Furnished herewith
101.INS	XBRL Instance Document - the instance document does not appear in the Interactive Data File because XBRL tags are embedded within the Inline XBRL document	Filed herewith
101.SCH	XBRL Taxonomy Extension Schema Document	Filed herewith
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document	Filed herewith
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document	Filed herewith
101.LAB	XBRL Taxonomy Extension Label Linkbase Document	Filed herewith
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document (embedded within the Inline XBRL document)	Filed herewith

[±] Certain attachments to this document have been omitted pursuant to Item 601(a)(5) of Regulation S-K but will be provided to the SEC or its staff upon request.

62

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: November 15, 2021

By: /s/ Shaun R. Bagai
Shaun R. Bagai
Chief Executive Officer

Date: November 15, 2021

By: /s/ Christopher J. Lehman
Chief Financial Officer

53

^{*} Indicates management contract or compensatory plan.

[†] The certifications attached as Exhibits 32.1 and 32.2 that accompany this Quarterly Report on Form 10-Q are deemed furnished and not filed with the Securities and Exchange Commission and are not to be incorporated by reference into any filing of the Registrant under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, whether made before or after the date of this Quarterly Report on Form 10-Q, irrespective of any general incorporation language contained in such filing.

THIS WARRANT AND THE SECURITIES ISSUABLE UPON EXERCISE OF THIS WARRANT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR QUALIFIED UNDER ANY STATE OR FOREIGN SECURITIES LAWS AND MAY NOT BE OFFERED FOR SALE, SOLD, PLEDGED, HYPOTHECATED OR OTHERWISE TRANSFERRED OR ASSIGNED UNLESS (I) A REGISTRATION STATEMENT COVERING SUCH SHARES IS EFFECTIVE UNDER THE ACT AND IS QUALIFIED UNDER APPLICABLE STATE AND FOREIGN LAW OR (II) THE TRANSACTION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS UNDER THE ACT AND THE QUALIFICATION REQUIREMENTS UNDER APPLICABLE STATE AND FOREIGN LAW AND, IF THE CORPORATION REQUESTS, AN OPINION SATISFACTORY TO THE CORPORATION TO SUCH EFFECT HAS BEEN RENDERED BY COUNSEL. THIS WARRANT MUST BE SURRENDERED TO THE COMPANY OR ITS TRANSFER AGENT AS A CONDITION PRECEDENT TO THE SALE, TRANSFER, PLEDGE OR HYPOTHECATION OR ANY INTEREST IN ANY OF THE SECURITIES REPRESENTED HEREBY.

COMMON STOCK PURCHASE WARRANT

RENOVORX, INC.

Warrant Shares: [-]

Initial Exercise Date: August 30, 2021

THIS CERTIFIES THAT, for value received, each holder listed on Exhibit A, or its registered assigns (the "Holder"), is entitled, subject to the provisions and upon the terms and conditions set forth herein, at any time on or after the date hereof (the "Initial Exercise Date") and on or prior to 5:00 p.m. (New York City time) on August 31, 2026 (the "Termination Date") but not thereafter, to purchase from RenovoRx, Inc., a Delaware corporation (the "Company") up to that number of shares listed opposite the Holder on Exhibit A (as subject to adjustment hereunder, the "Warrant Shares") of Common Stock. The term "Warrant" as used herein shall include this Warrant and any warrants delivered in substitution or exchange therefor as provided herein. The purchase price of one share of Common Stock under this Warrant shall be equal to the Exercise Price, as defined in Section 2(b).

Section 1. Definitions. In addition to the terms defined elsewhere in this Warrant, the following terms have the meanings indicated in this Section 1:

"Affiliate" means any Person that, directly or indirectly through one or more intermediaries, controls or is controlled by or is under common control with a Person, as such terms are used in and construed under Rule 405 under the Securities Act.

1

"Bid Price" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the bid price of the Common Stock for the time in question (or the nearest preceding date) on the Trading Market on which the Common Stock is then listed or quoted as reported by Bloomberg L.P. (based on a Trading Day from 9:30 a.m. (New York City time) to 4:02 p.m. (New York City time)), (b) if OTCQB or OTCQX is not a Trading Market, the volume weighted average price of the Common Stock for such date (or the nearest preceding date) on OTCQB or OTCQX as applicable, (c) if the Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Common Stock are then reported on the Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Common Stock so reported, or (d) in all other cases, the fair market value of a share of Common Stock as determined by an independent appraiser selected in good faith by the Holders of a majority in interest of the Warrants then outstanding and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

"Business Day" means any day other than Saturday, Sunday or other day on which commercial banks in The City of New York are authorized or required by law to remain closed; provided, however, for clarification, commercial banks shall not be deemed to be authorized or required by law to remain closed due to "stay at home", "shelter-in-place", "non-essential employee" or any other similar orders or restrictions or the closure of any physical branch locations at the direction of any governmental authority so long as the electronic funds transfer systems (including for wire transfers) of commercial banks in The City of New York generally are open for use by customers on such day.

"Commission" means the United States Securities and Exchange Commission.

"Common Stock" means the common stock of the Company, par value \$0.0001 per share, and any other class of securities into which such securities may hereafter be reclassified or changed.

"Common Stock Equivalents" means any securities of the Company or the Subsidiaries which would entitle the holder thereof to acquire at any time Common Stock, including, without limitation, any debt, preferred stock, right, option, warrant or other instrument that is at any time convertible into or exercisable or exchangeable for, or otherwise entitles the holder thereof to receive, Common Stock.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Person" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Subsidiary" means any subsidiary of the Company and shall, where applicable, also include any direct or indirect subsidiary of the Company formed or acquired after the date hereof.

2

"Trading Day" means a day on which the Common Stock is traded on a Trading Market.

"Trading Market" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE American, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market or the New York Stock Exchange (or any successors to any of the foregoing).

"Transfer Agent" means Philadelphia Stock Transfer, Inc., the current transfer agent of the Company, with a mailing address of 2320 Haverford Rd., Suite 230, Ardmore, PA 19003 and any successor transfer agent of the Company.

"<u>VWAP</u>" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the daily volume weighted average price of the Common Stock for such date (or the nearest preceding date) on the Trading Market on which the Common Stock is then listed or quoted as reported by Bloomberg L.P. (based on a Trading Day from 9:30 a.m. (New York City time) to 4:02 p.m. (New York City

time)), (b) if OTCQB or OTCQX is not a Trading Market, the volume weighted average price of the Common Stock for such date (or the nearest preceding date) on OTCQB or OTCQX as applicable, (c) if the Common Stock is not then listed or quoted for trading on OTCQB or OTCQX and if prices for the Common Stock are then reported on the Pink Open Market (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Common Stock so reported, or (d) in all other cases, the fair market value of a share of Common Stock as determined by an independent appraiser selected in good faith by the holders of a majority in interest of the Warrants then outstanding and reasonably acceptable to the Company, the fees and expenses of which shall be paid by the Company.

"Warrant Agent" means the Transfer Agent and any successor warrant agent of the Company.

"Warrants" means this Warrant.

Section 2. Exercise.

a) Exercise of Warrant. Exercise of the purchase rights represented by this Warrant may be made, in whole or in part, at any time or times on or after the Initial Exercise Date and on or before the Termination Date by delivery to the Company of a duly executed facsimile copy or PDF copy submitted by e-mail (or e-mail attachment) of the Notice of Exercise in the form annexed hereto (the "Notice of Exercise"). Within the earlier of (i) two (2) Trading Days and (ii) the number of Trading Days comprising the Standard Settlement Period (as defined in Section 2(d)(i) herein) following the date of exercise as aforesaid, the Holder shall deliver the aggregate Exercise Price for the shares specified in the applicable Notice of Exercise by wire transfer or cashier's check drawn on a United States bank unless the cashless exercise procedure specified in Section 2(c) below is specified in the applicable Notice of Exercise. No ink-original Notice of Exercise shall be required, nor shall any medallion guarantee (or other type of guarantee or notarization) of any Notice of Exercise be required. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company until the Holder has purchased all of the Warrant Shares available hereunder and the Warrant has been exercised in full, in which case, the Holder shall surrender this Warrant to the Company for cancellation within three (3) Trading Days of the date on which the final Notice of Exercise is delivered to the Company. Partial exercises of this Warrant resulting in purchases of a portion of the total number of Warrant Shares available hereunder shall have the effect of lowering the outstanding number of Warrant Shares purchased and the date of such purchases. The Company shall maintain records showing the number of Warrant Shares purchased and the date of such purchases. The Company shall deliver any objection to any Notice of Exercise within one (1) Trading Day of receipt of such notice. The Holder and any

3

- b) Exercise Price. The exercise price per share of Common Stock under this Warrant shall be \$10.80, subject to adjustment hereunder (the 'Exercise Price'').
- c) <u>Cashless Exercise</u>. If at the time of exercise hereof there is no effective registration statement registering, or the prospectus contained therein is not available for the issuance of the Warrant Shares to the Holder, then this Warrant may also be exercised, in whole or in part, at such time by means of a "cashless exercise" in which the Holder shall be entitled to receive a number of Warrant Shares equal to the quotient obtained by dividing [(A-B) (X)] by (A), where:
 - (A) = as applicable: (i) the VWAP on the Trading Day immediately preceding the date of the applicable Notice of Exercise if such Notice of Exercise is (1) both executed and delivered pursuant to Section 2(a) hereof on a day that is not a Trading Day or (2) both executed and delivered pursuant to Section 2(a) hereof on a Trading Day prior to the opening of "regular trading hours" (as defined in Rule 600(b) of Regulation NMS promulgated under the federal securities laws) on such Trading Day, (ii) at the option of the Holder, either (y) the VWAP on the Trading Day immediately preceding the date of the applicable Notice of Exercise or (z) the Bid Price of the Common Stock on the principal Trading Market as reported by Bloomberg L.P. as of the time of the Holder's execution of the applicable Notice of Exercise if such Notice of Exercise is executed during "regular trading hours" on a Trading Day and is delivered within two (2) hours thereafter (including until two (2) hours after the close of "regular trading hours" on a Trading Day) pursuant to Section 2(a) hereof or (iii) the VWAP on the date of the applicable Notice of Exercise if the date of such Notice of Exercise is a Trading Day;
 - (B) = the Exercise Price of this Warrant, as adjusted hereunder; and
 - (X) = the number of Warrant Shares that would be issuable upon exercise of this Warrant in accordance with the terms of this Warrant if such exercise were by means of a cash exercise rather than a cashless exercise.
- If Warrant Shares are issued in such a cashless exercise, the parties acknowledge and agree that in accordance with Section 3(a)(9) of the Securities Act, the Warrant Shares shall take on the characteristics of the Warrants being exercised, and the holding period of the Warrant Shares being issued may be tacked on to the holding period of this Warrant. The Company agrees not to take any position contrary to this Section 2(c).

Notwithstanding anything herein to the contrary, on the Termination Date, this Warrant shall be automatically exercised via cashless exercise pursuant to this Section 2(c).

i. Delivery of Warrant Shares Upon Exercise. The Company shall cause the Warrant Shares purchased hereunder to be transmitted by the Transfer Agent to the Holder by crediting the account of the Holder's or its designee's balance account with The Depository Trust Company through its Deposit or Withdrawal at Custodian system ("DWAC") if the Company is then a participant in such system and either (A) there is an effective registration statement permitting the issuance of the Warrant Shares to or resale of the Warrant Shares by Holder or (B) this Warrant is being exercised via cashless exercise and the Holder of this Warrant has satisfied the applicable holding period and the issuance of the Warrant Shares is otherwise compliant with Rule 144 or another applicable exception to the registration requirements under the securities laws, and otherwise by physical delivery of a certificate, registered in the Company's share register in the name of the Holder or its designee, for the number of Warrant Shares to which the Holder is entitled pursuant to such exercise to the address specified by the Holder in the Notice of Exercise by the date that is the earliest of (i) two (2) Trading Days after the delivery to the Company of the Notice of Exercise, (ii) one (1) Trading Day after delivery of the aggregate Exercise Price to the Company and (iii) the number of Trading Days comprising the Standard Settlement Period after the delivery to the Company of the Notice of Exercise (such date, the "Warrant Share Delivery Date"). Upon delivery of the Notice of Exercise, the Holder shall be deemed for all corporate purposes to have become the holder of record of the Warrant Shares with respect to which this Warrant has been exercised, irrespective of the date of delivery of the Warrant Shares, provided that payment of the aggregate Exercise Price (other than in the case of a cashless exercise) is received within the earlier of (i) two (2) Trading Days and (ii) the number of Trading Days comprising the Standard Settlement Period following delivery of the Notice of Exercise. If the Company fails for any reason to deliver to the Holder the Warrant Shares subject to a Notice of Exercise by the Warrant Share Delivery Date, the Company shall pay to the Holder, in cash, as liquidated damages and not as a penalty, for each \$1,000 of Warrant Shares subject to such exercise (based on the VWAP of the Common Stock on the date of the applicable Notice of Exercise), \$10 per Trading Day (increasing to \$20 per Trading Day on the fifth Trading Day after such liquidated damages begin to accrue) for each Trading Day after such Warrant Share Delivery Date until such Warrant Shares are delivered or Holder rescinds such exercise. The Company agrees to maintain a transfer agent that is a participant in the FAST program so long as this Warrant remains outstanding and exercisable. As used herein, "Standard Settlement Period" means the standard settlement period, expressed in a number of Trading Days, on the Company's primary Trading Market with respect to the Common Stock as in effect on the date of delivery of the Notice of Exercise.

ii. <u>Delivery of New Warrants Upon Exercise</u>. If this Warrant shall have been exercised in part, the Company shall, at the request of a Holder and upon surrender of this Warrant certificate, at the time of delivery of the Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Warrant Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant.

iii. <u>Rescission Rights</u>. If the Company fails to cause the Transfer Agent to transmit to the Holder the Warrant Shares pursuant to Section 2(d) (i) by the Warrant Share Delivery Date, then the Holder will have the right to rescind such exercise.

5

iv. Compensation for Buy-In on Failure to Timely Deliver Warrant Shares Upon Exercise. In addition to any other rights available to the Holder, if the Company fails to cause the Transfer Agent to transmit to the Holder the Warrant Shares in accordance with the provisions of Section 2(d)(i) above pursuant to an exercise on or before the Warrant Share Delivery Date, and if after such date the Holder is required by its broker to purchase (in an open market transaction or otherwise) or the Holder's brokerage firm otherwise purchases, shares of Common Stock to deliver in satisfaction of a sale by the Holder of the Warrant Shares which the Holder anticipated receiving upon such exercise (a "Buy-In"), then the Company shall (A) pay in cash to the Holder the amount, if any, by which (x) the Holder's total purchase price (including brokerage commissions, if any) for the shares of Common Stock so purchased exceeds (y) the amount obtained by multiplying (1) the number of Warrant Shares that the Company was required to deliver to the Holder in connection with the exercise at issue times (2) the price at which the sell order giving rise to such purchase obligation was executed, and (B) at the option of the Holder, either reinstate the portion of the Warrant and equivalent number of Warrant Shares for which such exercise was not honored (in which case such exercise shall be deemed rescinded) or deliver to the Holder the number of shares of Common Stock that would have been issued had the Company timely complied with its exercise and delivery obligations hereunder. For example, if the Holder purchases Common Stock having a total purchase price of \$11,000 to cover a Buy-In with respect to an attempted exercise of shares of Common Stock with an aggregate sale price giving rise to such purchase obligation of \$10,000, under clause (A) of the immediately preceding sentence the Company shall be required to pay the Holder \$1,000. The Holder shall provide the Company written notice indicating the amounts payable to the Holder in respect of the Buy-In and, upon request of the Company, evidence of the amount of such loss. Nothing herein shall limit a Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Company's failure to timely deliver shares of Common Stock upon exercise of the Warrant as required pursuant to the terms hereof.

v. No Fractional Shares or Scrip. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant. As to any fraction of a share which the Holder would otherwise be entitled to purchase upon such exercise, the Company shall, at its election, either pay a cash adjustment in respect of such final fraction in an amount equal to such fraction multiplied by the Exercise Price or round up to the next whole share.

vi. <u>Charges, Taxes and Expenses</u>. Issuance of Warrant Shares shall be made without charge to the Holder for any issue or transfer tax or other incidental expense in respect of the issuance of such Warrant Shares, all of which taxes and expenses shall be paid by the Company, and such Warrant Shares shall be issued in the name of the Holder or in such name or names as may be directed by the Holder; <u>provided, however</u>, that, in the event that Warrant Shares are to be issued in a name other than the name of the Holder, this Warrant when surrendered for exercise shall be accompanied by the Assignment Form attached hereto duly executed by the Holder and the Company may require, as a condition thereto, the payment of a sum sufficient to reimburse it for any transfer tax incidental thereto. The Company shall pay all Transfer Agent fees required for same-day processing of any Notice of Exercise and all fees to the Depository Trust Company (or another established clearing corporation performing similar functions) required for same-day electronic delivery of the Warrant Shares.

6

vii. <u>Closing of Books</u>. The Company will not close its stockholder books or records in any manner which prevents the timely exercise of this Warrant, pursuant to the terms hereof.

e) <u>Holder's Exercise Limitations</u>. The Company shall not effect any exercise of this Warrant, and a Holder shall not have the right to exercise any portion of this Warrant, pursuant to Section 2 or otherwise, to the extent that after giving effect to such issuance after exercise as set forth on the applicable Notice of Exercise, the Holder (together with the Holder's Affiliates, and any other Persons acting as a group together with the Holder or any of the Holder's Affiliates (such Persons, "<u>Attribution Parties</u>")), would beneficially own in excess of the Beneficial Ownership Limitation (as defined below). For purposes of the foregoing sentence, the number of shares of Common Stock beneficially owned by the Holder and its Affiliates and Attribution Parties shall include the number of shares of Common Stock issuable upon exercise of this Warrant with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which would be issuable upon (i) exercise of the remaining, nonexercised portion of this Warrant beneficially owned by the Holder or any of its Affiliates or Attribution Parties of the Company (including, without limitation, any other Common Stock Equivalents) subject to a limitation on conversion or exercise analogous to the limitation contained herein beneficially owned by the Holder or any of its Affiliates or Attribution Parties. Except as set forth in the preceding sentence, for purposes of this Section 2(e), beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder, it being acknowledged by the Holder that the Company is not representing to the Holder

that such calculation is in compliance with Section 13(d) of the Exchange Act and the Holder is solely responsible for any schedules required to be filed in accordance therewith. To the extent that the limitation contained in this Section 2(e) applies, the determination of whether this Warrant is exercisable (in relation to other securities owned by the Holder together with any Affiliates and Attribution Parties) and of which portion of this Warrant is exercisable shall be in the sole discretion of the Holder, and the submission of a Notice of Exercise shall be deemed to be the Holder's determination of whether this Warrant is exercisable (in relation to other securities owned by the Holder together with any Affiliates and Attribution Parties) and of which portion of this Warrant is exercisable, in each case subject to the Beneficial Ownership Limitation, and the Company shall have no obligation to verify or confirm the accuracy of such determination. In addition, a determination as to any group status as contemplated above shall be determined in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder. For purposes of this Section 2(e), in determining the number of outstanding shares of Common Stock, a Holder may rely on the number of outstanding shares of Common Stock as reflected in (A) the Company's most recent periodic or annual report filed with the Commission, as the case may be, (B) a more recent public announcement by the Company or (C) a more recent written notice by the Company or the Transfer Agent setting forth the number of shares of Common Stock outstanding. Upon the written or oral request of a Holder, the Company shall within one Trading Day confirm orally and in writing to the Holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of securities of the Company, including this Warrant, by the Holder or its Affiliates or Attribution Parties since the date as of which such number of outstanding shares of Common Stock was reported. The "Beneficial Ownership Limitation" shall be 4.99% (or, upon election by a Holder prior to the issuance of any Warrants, 9.99%) of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock issuable upon exercise of this Warrant; provided, however, that the Beneficial Ownership Limitation does not apply to any Holder that is an Affiliate of the Company as of Initial Exercise Date. The Holder, upon notice to the Company, may increase or decrease the Beneficial Ownership Limitation provisions of this Section 2(e), provided that the Beneficial Ownership Limitation in no event exceeds 9.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock upon exercise of this Warrant held by the Holder and the provisions of this Section 2(e) shall continue to apply. Any increase in the Beneficial Ownership Limitation will not be effective until the 61st day after such notice is delivered to the Company. The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this Section 2(e) to correct this paragraph (or any portion hereof) which may be defective or inconsistent with the intended Beneficial Ownership Limitation herein contained or to make changes or supplements necessary or desirable to properly give effect to such limitation. The limitations contained in this paragraph shall apply to a successor holder of this Warrant.

7

Section 3. Certain Adjustments.

a) Stock Dividends and Splits. If the Company, at any time while this Warrant is outstanding: (i) pays a stock dividend or otherwise makes a distribution or distributions on shares of its Common Stock or any other equity or equity equivalent securities payable in shares of Common Stock (which, for avoidance of doubt, shall not include any shares of Common Stock issued by the Company upon exercise of this Warrant), (ii) subdivides outstanding shares of Common Stock into a larger number of shares, (iii) combines (including by way of reverse stock split) outstanding shares of Common Stock into a smaller number of shares, or (iv) issues by reclassification of shares of the Common Stock any shares of capital stock of the Company, then in each case the Exercise Price shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock (excluding treasury shares, if any) outstanding immediately before such event and of which the denominator shall be the number of shares of Common Stock outstanding immediately after such event, and the number of shares issuable upon exercise of this Warrant shall be proportionately adjusted such that the aggregate Exercise Price of this Warrant shall remain unchanged. Any adjustment made pursuant to this Section 3(a) shall become effective immediately after the record date for the determination of stockholders entitled to receive such dividend or distribution and shall become effective immediately after the effective date in the case of a subdivision, combination or re-classification.

b) <u>Subsequent Rights Offerings</u>. In addition to any adjustments pursuant to Section 3(a) above, if at any time the Company grants, issues or sells any Common Stock Equivalents or rights to purchase stock, warrants, securities or other property pro rata to the record holders of any class of shares of Common Stock (the "<u>Purchase Rights</u>"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the grant, issue or sale of such Purchase Rights (provided, however, that, to the extent that the Holder's right to participate in any such Purchase Right would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Purchase Right to such extent (or beneficial ownership of such shares of Common Stock as a result of such Purchase Right to such extent) and such Purchase Right to such extent shall be held in abeyance for the Holder until such time, if ever, as its right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

c) Pro Rata Distributions. During such time as this Warrant is outstanding, if the Company shall declare or make any dividend or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property or options by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "Distribution"), at any time after the issuance of this Warrant, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date of which a record is taken for such Distribution, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the participation in such Distribution (provided, however, that, to the extent that the Holder's right to participate in any such Distribution would result in the Holder exceeding the Beneficial Ownership Limitation, then the beneficial ownership of any shares of Common Stock as a result of such Distribution to such extent) and the portion of such Distribution shall be held in abeyance for the benefit of the Holder until such time, if ever, as its right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

8

d) Fundamental Transaction. If, at any time while this Warrant is outstanding, (i) the Company, directly or indirectly, in one or more related transactions effects any merger or consolidation of the Company with or into another Person, (ii) the Company or any Subsidiary, directly or indirectly, effects any sale, lease, license, assignment, transfer, conveyance or other disposition of all or substantially all of its assets in one or a series of related transactions, (iii) any, direct or indirect, purchase offer, tender offer or exchange offer (whether by the Company or another Person) is completed pursuant to which holders of Common Stock are permitted to sell, tender or exchange their shares for other securities, cash or property and has been accepted by the holders of 50% or more of the outstanding Common Stock, (iv) the Company, directly or indirectly, in one or more related transactions effects any reclassification, reorganization or recapitalization of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for other securities, cash or property, or (v) the Company, directly or indirectly, in one or more related transactions consummates a stock or share purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off, merger or scheme of arrangement) with another Person or group of Persons whereby such other Person or group acquires more than 50% of the outstanding shares of Common Stock (not including any shares of Common Stock held by the other Person or other Persons making or party to, or associated or affiliated with the other Persons making or party to, such stock or share purchase agreement or other business combination) (each a "Fundamental Transaction"), then, upon any subsequent exercise of this Warrant, the Holder shall have the right to receive, for each Warrant Share that would have been issuable upon such exercise immediately prior to the occurrence of such Fundamental Transaction, at the option of the Holder (without regard to any limitation in Section 2(e) on the exercise of this Warrant), the number of shares of Common Stock of the successor or acquiring corporation or of the Company, if it is the surviving corporation, and any additional consideration (the "Alternate Consideration") receivable as a result of such Fundamental Transaction by a holder of the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such Fundamental Transaction (without regard to any limitation in Section 2(e) on the exercise of this Warrant). For purposes of any such exercise, the determination of the Exercise Price shall be appropriately adjusted to apply to such Alternate Consideration based on

the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Company shall apportion the Exercise Price among the Alternate Consideration in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holder shall be given the same choice as to the Alternate Consideration it receives upon any exercise of this Warrant following such Fundamental Transaction. The Company shall cause any successor entity in a Fundamental Transaction in which the Company is not the survivor (the "Successor Entity") to assume in writing all of the obligations of the Company under this Warrant in accordance with the provisions of this Section 3(d) pursuant to written agreements in form and substance reasonably satisfactory to the Holder and approved by the Holder (without unreasonable delay) prior to such Fundamental Transaction and shall, at the option of the Holder, deliver to the Holder in exchange for this Warrant a security of the Successor Entity evidenced by a written instrument substantially similar in form and substance to this Warrant which is exercisable for a corresponding number of shares of capital stock of such Successor Entity (or its parent entity) equivalent to the shares of Common Stock acquirable and receivable upon exercise of this Warrant (without regard to any limitations on the exercise of this Warrant) prior to such Fundamental Transaction, and with an exercise price which applies the exercise price hereunder to such shares of capital stock (but taking into account the relative value of the shares of Common Stock pursuant to such Fundamental Transaction and the value of such shares of capital stock, such number of shares of capital stock and such exercise price being for the purpose of protecting the economic value of this Warrant immediately prior to the consummation of such Fundamental Transaction), and which is reasonably satisfactory in form and substance to the Holder. Upon the occurrence of any such Fundamental Transaction, the Successor Entity shall succeed to, and be substituted for (so that from and after the date of such Fundamental Transaction, the provisions of this Warrant referring to the "Company" shall refer instead to the Successor Entity), and may exercise every right and power of the Company and shall assume all of the obligations of the Company under this Warrant with the same effect as if such Successor Entity had been named as the Company herein.

9

e) <u>Calculations</u>. All calculations under this Section 3 shall be made to the nearest cent or the nearest 1/100th of a share, as the case may be. For purposes of this Section 3, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding treasury shares, if any) issued and outstanding.

f) Notice to Holder.

- i. <u>Adjustment to Exercise Price</u>. Whenever the Exercise Price is adjusted pursuant to any provision of this Section 3, the Company shall promptly deliver to the Holder by facsimile or email a notice setting forth the Exercise Price after such adjustment and any resulting adjustment to the number of Warrant Shares and setting forth a brief statement of the facts requiring such adjustment.
- ii. Notice to Allow Exercise by Holder. If (A) the Company shall declare a dividend (or any other distribution in whatever form) on the Common Stock, (B) the Company shall declare a special nonrecurring cash dividend on or a redemption of the Common Stock, (C) the Company shall authorize the granting to all holders of the Common Stock rights or warrants to subscribe for or purchase any shares of capital stock of any class or of any rights, (D) the approval of any stockholders of the Company shall be required in connection with any reclassification of the Common Stock, any consolidation or merger to which the Company (or any of its Subsidiaries) is a party, any sale or transfer of all or substantially all of its assets, or any compulsory share exchange whereby the Common Stock is converted into other securities, cash or property, or (E) the Company shall authorize the voluntary or involuntary dissolution, liquidation or winding up of the affairs of the Company, then, in each case, the Company shall cause to be delivered by facsimile or email to the Holder at its last facsimile number or email address as it shall appear upon the Warrant Register of the Company, at least 20 calendar days prior to the applicable record or effective date hereinafter specified, a notice stating (x) the date on which a record is to be taken for the purpose of such dividend, distribution, redemption, rights or warrants, or if a record is not to be taken, the date as of which the holders of the Common Stock of record to be entitled to such dividend, distributions, redemption, rights or warrants are to be determined or (y) the date on which such reclassification, consolidation, merger, sale, transfer or share exchange is expected to become effective or close, and the date as of which it is expected that holders of the Common Stock of record shall be entitled to exchange their shares of the Common Stock for securities, cash or other property deliverable upon such reclassification, consolidation, merger, sale, transfer or share exchange; provided that the failure to deliver such notice or any defect therein or in the delivery thereof shall not affect the validity of the corporate action required to be specified in such notice. To the extent that any notice provided in this Warrant constitutes, or contains, material, non-public information regarding the Company or any of the Subsidiaries, the Company shall simultaneously file such notice with the Commission pursuant to a Current Report on Form 8-K. The Holder shall remain entitled to exercise this Warrant during the period commencing on the date of such notice to the effective date of the event triggering such notice except as may otherwise be expressly set forth herein.

10

g) <u>Voluntary Adjustment By Company</u>. Subject to the rules and regulations of the Trading Market, the Company may at any time during the term of this Warrant, subject to the prior written consent of the Holder, reduce the then current Exercise Price to any amount and for any period of time deemed appropriate by the board of directors of the Company.

Section 4. Transfer of Warrant.

- a) Transferability. This Warrant and all rights hereunder (including, without limitation, any registration rights) are transferable, in whole or in part, upon surrender of this Warrant at the principal office of the Company or its designated agent, together with a written assignment of this Warrant substantially in the form attached hereto duly executed by the Holder or its agent or attorney and funds sufficient to pay any transfer taxes payable upon the making of such transfer. Upon such surrender and, if required, such payment, the Company shall execute and deliver a new Warrant or Warrants in the name of the assignee or assignees, as applicable, and in the denomination or denominations specified in such instrument of assignment, and shall issue to the assignor a new Warrant evidencing the portion of this Warrant not so assigned, and this Warrant shall promptly be cancelled. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company unless the Holder has assigned this Warrant in full, in which case, the Holder shall surrender this Warrant to the Company within three (3) Trading Days of the date on which the Holder delivers an assignment form to the Company assigning this Warrant in full. The Warrant, if properly assigned in accordance herewith, may be exercised by a new holder for the purchase of Warrant Shares without having a new Warrant issued.
- b) New Warrants. If this Warrant is not held in global form through DTC (or any successor depositary), this Warrant may be divided or combined with other Warrants upon presentation hereof at the aforesaid office of the Company, together with a written notice specifying the names and denominations in which new Warrants are to be issued, signed by the Holder or its agent or attorney. Subject to compliance with Section 4(a), as to any transfer which may be involved in such division or combination, the Company shall execute and deliver a new Warrant or Warrants in exchange for the Warrant or Warrants to be divided or combined in accordance with such notice. All Warrants issued on transfers or exchanges shall be dated the initial issuance date of this Warrant and shall be identical with this Warrant except as to the number of Warrant Shares issuable pursuant thereto.
- c) <u>Warrant Register</u>. The Warrant Agent shall register this Warrant, upon records to be maintained by the Warrant Agent for that purpose (the "<u>Warrant Register</u>"), in the name of the record Holder hereof from time to time. The Company and the Warrant Agent may deem and treat the registered Holder of this Warrant as the absolute owner hereof for the purpose of any exercise hereof or any distribution to the Holder, and for all other purposes, absent actual notice to the contrary.

Section 5. Compliance with the Securities Act.

(a) Agreement to Comply with the Securities Act; Legend The Holder, by acceptance of this Warrant, agrees to comply in all respects with the provisions of this Section 5 and the restrictive legend requirements set forth on the face of this Warrant and further agrees that such Holder shall not offer, sell or otherwise dispose of this Warrant or any Warrant Shares to be issued upon exercise hereof except under circumstances that will not result in a violation of the Securities Act. This Warrant and all Warrant Shares issued upon exercise of this Warrant (unless registered under the Securities Act) shall be stamped or imprinted with a legend in substantially the following form:

"THIS WARRANT AND THE SECURITIES ISSUABLE UPON EXERCISE OF THIS WARRANT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR QUALIFIED UNDER ANY STATE OR FOREIGN SECURITIES LAWS AND MAY NOT BE OFFERED FOR SALE, SOLD, PLEDGED, HYPOTHECATED OR OTHERWISE TRANSFERRED OR ASSIGNED UNLESS (I) A REGISTRATION STATEMENT COVERING SUCH SHARES IS EFFECTIVE UNDER THE ACT AND IS QUALIFIED UNDER APPLICABLE STATE AND FOREIGN LAW OR (II) THE TRANSACTION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS UNDER THE ACT AND THE QUALIFICATION REQUIREMENTS UNDER APPLICABLE STATE AND FOREIGN LAW AND, IF THE CORPORATION REQUESTS, AN OPINION SATISFACTORY TO THE CORPORATION TO SUCH EFFECT HAS BEEN RENDERED BY COUNSEL. THIS WARRANT MUST BE SURRENDERED TO THE COMPANY OR ITS TRANSFER AGENT AS A CONDITION PRECEDENT TO THE SALE, TRANSFER, PLEDGE OR HYPOTHECATION OR ANY INTEREST IN ANY OF THE SECURITIES REPRESENTED HEREBY.

THE SECURITIES REPRESENTED HEREBY ARE SUBJECT TO RESTRICTIONS ON TRANSFERABILITY AND RESALE, WHICH MAY INCLUDE A LOCK-UP PERIOD IN THE EVENT OF A PUBLIC OFFERING, AS SET FORTH IN THE WARRANT PURSUANT TO WHICH THESE SHARES WERE ISSUED, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE COMPANY."

The Company may also include any legend required by the Blue Sky laws of any state to the extent those laws are applicable to the shares represented by the certificate so legended.

- (b) <u>Representations of the Holder</u>. In connection with the issuance of this Warrant, the Holder specifically represents, as of the date hereof, to the Company by acceptance of this Warrant as follows:
- (i) The Holder is an "accredited investor" as defined in Rule 501(a) of Regulation D promulgated under the Securities Act. The Holder is acquiring this Warrant and the Warrant Shares to be issued upon exercise hereof for investment for its own account and not with a view towards, or for resale in connection with, the public sale or distribution of this Warrant or the Warrant Shares, except pursuant to sales registered or exempted under the Securities Act.
- (ii) The Holder understands and acknowledges that this Warrant and the Warrant Shares to be issued upon exercise hereof are "restricted securities" under the federal securities laws inasmuch as they are being acquired from the Company in a transaction not involving a public offering and that, under such laws and applicable regulations, such securities may be resold without registration under the Securities Act only in certain limited circumstances. In addition, the Holder represents that it is familiar with Rule 144 under the Securities Act, as presently in effect, and understands the resale limitations imposed thereby and by the Securities Act.
- (iii) The Holder acknowledges that it can bear the economic and financial risk of its investment for an indefinite period, and has such knowledge and experience in financial or business matters that it is capable of evaluating the merits and risks of the investment in the Warrant and the Warrant Shares. The Holder has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of the Warrant and the business, properties, prospects and financial condition of the Company.

12

Section 6. Miscellaneous.

- a) No Rights as Stockholder Until Exercise; No Settlement in Cash. This Warrant does not entitle the Holder to any voting rights, dividends or other rights as a stockholder of the Company prior to the exercise hereof as set forth in Section 2(d)(i), except as expressly set forth in Section 3. Without limiting any rights of a Holder to receive Warrant Shares on a "cashless exercise" pursuant to Section 2(c) or to receive cash payments pursuant to Section 2(d)(i) and Section 2(d)(iv) herein, in no event shall the Company be required to net cash settle an exercise of this Warrant.
- b) Loss, Theft, Destruction or Mutilation of Warrant. The Company covenants that upon receipt by the Company of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Warrant or any stock certificate relating to the Warrant Shares, and in case of loss, theft or destruction, of indemnity or security reasonably satisfactory to it (which, in the case of the Warrant, shall not include the posting of any bond), and upon surrender and cancellation of such Warrant or stock certificate, if mutilated, the Company will make and deliver a new Warrant or stock certificate of like tenor and dated as of such cancellation, in lieu of such Warrant or stock certificate.
- c) <u>Saturdays, Sundays, Holidays, etc.</u> If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall not be a Trading Day, then such action may be taken or such right may be exercised on the next succeeding Trading Day.

d) Authorized Shares.

The Company covenants that, during the period the Warrant is outstanding, it will reserve from its authorized and unissued Common Stock a sufficient number of shares to provide for the issuance of the Warrant Shares upon the exercise of any purchase rights under this Warrant. The Company further covenants that its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of issuing the necessary Warrant Shares upon the exercise of the purchase rights under this Warrant. The Company will take all such reasonable action as may be necessary to assure that such Warrant Shares may be issued as provided herein without violation of any applicable law or regulation, or of any requirements of the Trading Market upon which the Common Stock may be listed. The Company covenants that all Warrant Shares which may be issued upon the exercise of the purchase rights represented by this Warrant will, upon exercise of the purchase rights represented by this Warrant and payment for such Warrant Shares in accordance herewith, be duly authorized, validly issued, fully paid and nonassessable and free from all taxes, liens and charges created by the Company in respect of the issue thereof (other than taxes in respect of any transfer occurring contemporaneously with such issue).

13

Except and to the extent as waived or consented to by the Holder, the Company shall not by any action, including, without limitation, amending its certificate of incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of Holder as set forth in this Warrant against impairment. Without limiting the generality of the foregoing, the Company will (i) not increase the par value of any Warrant Shares above the amount payable therefor upon such exercise immediately prior to such increase in par value, (ii) take all such action as may be necessary or appropriate in order that the

Company may validly and legally issue fully paid and nonassessable Warrant Shares upon the exercise of this Warrant and (iii) use commercially reasonable efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof, as may be, necessary to enable the Company to perform its obligations under this Warrant.

Before taking any action which would result in an adjustment in the number of Warrant Shares for which this Warrant is exercisable or in the Exercise Price, the Company shall obtain all such authorizations or exemptions thereof, or consents thereto, as may be necessary from any public regulatory body or bodies having jurisdiction thereof.

e) Governing Law. All questions concerning the construction, validity, enforcement and interpretation of this Warrant shall be governed by and construed and enforced in accordance with the internal laws of the State of New York, without regard to the principles of conflicts of law thereof. Each party agrees that all legal proceedings concerning the interpretations, enforcement and defense of the transactions contemplated by this Warrant (whether brought against a party hereto or their respective affiliates, directors, officers, shareholders, partners, members, employees or agents) shall be commenced exclusively in the state and federal courts sitting in the City of New York. Each party hereby irrevocably submits to the exclusive jurisdiction of the state and federal courts sitting in the City of New York, Borough of Manhattan for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is improper or is an inconvenient venue for such proceeding. Each party hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to such party at the address in effect for notices to it under this Warrant and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any other manner permitted by law. If either party shall commence an action, suit or proceeding to enforce any provisions of this Warrant, the prevailing party in such action, suit or proceeding shall be reimbursed by the other party for their reasonable attorneys' fees and ot

f) <u>Restrictions</u>. The Holder acknowledges that the Warrant Shares acquired upon the exercise of this Warrant, if not registered, will have restrictions upon resale imposed by state and federal securities laws.

14

- g) Nonwaiver and Expenses. No course of dealing or any delay or failure to exercise any right hereunder on the part of Holder shall operate as a waiver of such right or otherwise prejudice the Holder's rights, powers or remedies. Without limiting any other provision of this Warrant, if the Company willfully and knowingly fails to comply with any provision of this Warrant, which results in any material damages to the Holder, the Company shall pay to the Holder such amounts as shall be sufficient to cover any costs and expenses including, but not limited to, reasonable attorneys' fees, including those of appellate proceedings, incurred by the Holder in collecting any amounts due pursuant hereto or in otherwise enforcing any of its rights, powers or remedies hereunder.
- h) Notices. Any and all notices or other communications or deliveries to be provided by the Holders hereunder including, without limitation, any Notice of Exercise, shall be in writing and delivered personally, by facsimile or e-mail, or sent by a nationally recognized overnight courier service, addressed to the Company, at 4546 El Camino Real, Suite B1, Los Altos, CA 94022, Attention: Shaun Bagai, CEO, facsimile number: 650-397-4433, email address: sbagai@renovorx.com, or such other facsimile number, email address or address as the Company may specify for such purposes by notice to the Holders. Any and all notices or other communications or deliveries to be provided by the Company hereunder shall be in writing and delivered personally, by facsimile or e-mail, or sent by a nationally recognized overnight courier service addressed to each Holder at the facsimile number, e-mail address or address of such Holder appearing on the books of the Company. Any notice or other communication or deliveries hereunder shall be deemed given and effective on the earliest of (i) the time of transmission, if such notice or communication is delivered via facsimile at the facsimile number or via e-mail at the e-mail address set forth in this Section prior to 5:30 p.m. (New York City time) on any date, (ii) the next Trading Day after the time of transmission, if such notice or communication is delivered via facsimile at the facsimile number or via e-mail at the e-mail address set forth in this Section on a day that is not a Trading Day or later than 5:30 p.m. (New York City time) on any Trading Day, (iii) the second Trading Day following the date of mailing, if sent by U.S. nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given. To the extent that any notice provided hereunder constitutes, or contains, material, non-public information regarding the Company or any Subsidiaries, the Company shall simultaneously file such notice with the Commission pursuant
- i) <u>Limitation of Liability</u>. No provision hereof, in the absence of any affirmative action by the Holder to exercise this Warrant to purchase Warrant Shares, and no enumeration herein of the rights or privileges of the Holder, shall give rise to any liability of the Holder for the purchase price of any Common Stock or as a stockholder of the Company, whether such liability is asserted by the Company or by creditors of the Company.
- j) Remedies. The Holder, in addition to being entitled to exercise all rights granted by law, including recovery of damages, will be entitled to specific performance of its rights under this Warrant. The Company agrees that monetary damages would not be adequate compensation for any loss incurred by reason of a breach by it of the provisions of this Warrant and hereby agrees to waive and not to assert the defense in any action for specific performance that a remedy at law would be adequate.

15

- k) <u>Successors and Assigns</u>. Subject to applicable securities laws, this Warrant and the rights and obligations evidenced hereby shall inure to the benefit of and be binding upon the successors and permitted assigns of the Company and the successors and permitted assigns of Holder. The provisions of this Warrant are intended to be for the benefit of any Holder from time to time of this Warrant and shall be enforceable by the Holder or holder of Warrant Shares.
- l) Amendment. Except as expressly provided herein, neither this Warrant nor any term hereof may be amended, waived, discharged or terminated other than by a written instrument referencing this Warrant and signed by the Company and the Holder. Notwithstanding the foregoing, without the consent of any Holder, the Company at any time may amend this Warrant (i) to cure any ambiguity, defect, error or inconsistency, or (ii) to comply with applicable law.
- m) <u>Severability</u>. Wherever possible, each provision of this Warrant shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Warrant shall be prohibited by or invalid under applicable law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provisions or the remaining provisions of this Warrant.
 - n) Headings. The headings used in this Warrant are for the convenience of reference only and shall not, for any purpose, be deemed a part of this Warrant.
- o) Entire Agreement. This Warrant (including the exhibit and forms attached hereto) constitutes the sole and entire agreement of the parties to this Warrant with respect to the subject matter contained herein, and supersedes all prior and contemporaneous understandings and agreements, both written and oral, with respect to such subject matter.

IN WITNESS WHEREOF, the Com	many has assessed this Warrant to	be executed by its officer thereunt	to duly outhorized or of t	ha data firat abayya indigatar
IN WITHESS WHEREOF, HE COIL	idaliv has caused this wallant to	De executed by its officer increum	lo durv admorized as of t	ne date inst above mulcated

RENOVORX, INC.

By:			
By: Name: Title:			
17			

RENOVORX, INC.

AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT

RENOVORX, INC.

AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT

THIS AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT (the "Agreement") is entered into as of the 18 day of April, 2018, by and among RenovoRx, Inc., a Delaware corporation (the "Company") and the investors listed on Exhibit A hereto, referred to hereinafter as the "Investors" and each individually as an "Investor."

RECITALS

WHEREAS, certain of the Investors are purchasing shares of the Company's Series D Preferred Stock (the "Series D Stock") pursuant to that certain Series D Preferred Stock Purchase Agreement (the "Purchase Agreement") of even date herewith (the "Financing");

WHEREAS, the obligations in the Purchase Agreement are conditioned upon the execution and delivery of this Agreement;

WHEREAS, certain of the Investors (the "Prior Investors") are holders of the Company's Series A Preferred Stock (the "Series A Stock"), the Company's Series B Preferred Stock (the "Series B Stock") and the Company's Series C Preferred Stock (the "Series C Stock" and together with the Series A Stock, the Series B Stock and the Series D Stock, the "Preferred Stock");

WHEREAS, the Prior Investors and the Company are parties to an Investor Rights Agreement dated December 7, 2015 (the "Prior Agreement");

WHEREAS, the parties to the Prior Agreement desire to amend and restate the Prior Agreement and accept the rights and covenants hereof in lieu of their rights and covenants under the Prior Agreement; and

WHEREAS, in connection with the consummation of the Financing, the Company and the Investors have agreed to the registration rights, information rights, and other rights as set forth below.

NOW, THEREFORE, in consideration of these premises and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

SECTION 1. GENERAL.

1.1 Amendment and Restatement of Prior Agreement. The Prior Agreement is hereby amended in its entirety and restated herein. Such amendment and restatement is effective upon the execution of this Agreement by the Company and the holders of a majority of the Series A Stock, Series B Stock and Series C Stock held by the Prior Investors outstanding as of the date of this Agreement. Upon such execution, all provisions of, rights granted and covenants made in the Prior Agreement are hereby waived, released and superseded in their entirety and shall have no further force or effect, including, without limitation, all rights of first refusal and any notice period associated therewith otherwise applicable to the transactions contemplated by the Purchase Agreement.

1

- 1.2 Definitions. As used in this Agreement the following terms shall have the following respective meanings:
 - (a) "BSC" means Boston Scientific Corporation, a Delaware corporation.
 - (b) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (c) "Form S-3" means such form under the Securities Act as in effect on the date hereof or any successor or similar registration form under the Securities Act subsequently adopted by the SEC which permits inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.
- (d) "Holder" means any person owning of record Registrable Securities that have not been sold to the public or any assignee of record of such Registrable Securities in accordance with Section 2.9 hereof.
 - (e) "Initial Offering" means the Company's first firm commitment underwritten public offering of its Common Stock registered under the Securities Act.
- (f) "Major Investor" means each of (i) Golden Seeds RenovoRx, LLC (and any of its affiliates, including any members of Golden Seeds that individually own Shares), (ii) The Angels' Forum 103, LLC, (iii) the Halo Fund III, L.P., (iv) Astia Angels RenovoRx, LLC, (v) Amidi, LLC, (vi) BSC, and (vii) each Investor under the Purchase Agreement, that certain Series C Preferred Stock Purchase Agreement dated December 7, 2015, that certain Series B Preferred Stock Purchase Agreement dated December 20, 2013 or that certain Series A Preferred Stock and Warrant Purchase Agreement, dated as of January 23, 2013 who (together with their affiliates) holds at least (A) 536,768 shares of Series A Stock, (B) 308,356 shares of Series B Stock, (C) 119,474 shares of Series C Stock or (D) 453,309 shares of Series D Stock (or, in each case, Common Stock issued upon conversion of such Shares, and as adjusted for any stock dividends, combinations, splits, recapitalizations and the like after the date hereof).
- (g) "Register," "registered," and "registration" refer to a registration effected by preparing and filing a registration statement in compliance with the Securities Act, and the declaration or ordering of effectiveness of such registration statement or document.
- (h) "Registrable Securities" means (a) Common Stock of the Company issuable or issued upon conversion of the Shares and (b) any Common Stock of the Company issued as (or issuable upon the conversion or exercise of any warrant, right or other security which is issued as) a dividend or other distribution with respect to, or in exchange for or in replacement of, such above-described securities. Notwithstanding the foregoing, Registrable Securities shall not include any securities (i) sold by a person to the public either pursuant to a registration statement or Rule 144 or (ii) sold in a private transaction in which the transferor's rights under Section 2 of this Agreement are not assigned.

- (i) "Registrable Securities then outstanding" shall be the number of shares of the Company's Common Stock that are Registrable Securities and either (a) are then issued and outstanding or (b) are issuable pursuant to then exercisable or convertible securities.
- (j) "Registration Expenses" shall mean all expenses incurred by the Company in complying with Sections 2.2, 2.3 and 2.4 hereof, including, without limitation, all registration and filing fees, printing expenses, fees and disbursements of counsel for the Company, reasonable fees and disbursements not to exceed thirty thousand dollars (\$30,000) of a single special counsel for the Holders, blue sky fees and expenses and the expense of any special audits incident to or required by any such registration (but excluding the compensation of regular employees of the Company which shall be paid in any event by the Company).
- (k) "Required Preferred Approval" means, after the Milestone Closing, the approval of at least one of (i) the Series D Director (as defined in the Voting Agreement) or (ii) the Series A/B Director (as defined in the Voting Agreement).
 - (1) "SEC" or "Commission" means the Securities and Exchange Commission.
 - (m) "Securities Act" shall mean the Securities Act of 1933, as amended.
 - (n) "Selling Expenses" shall mean all underwriting discounts and selling commissions applicable to the sale.
- (o) "Shares" shall mean the Series A Stock, the Series B Stock, the Series C Stock and the Series D Stock held from time to time by the Investors listed on Exhibit A hereto and their permitted assigns.
- (p) "Special Registration Statement" shall mean (i) a registration statement relating to any employee benefit plan or (ii) with respect to any corporate reorganization or transaction under Rule 145 of the Securities Act, any registration statements related to the issuance or resale of securities issued in such a transaction or (iii) a registration related to stock issued upon conversion of debt securities.
 - (q) "Voting Agreement" means the Amended and Restated Voting Agreement by and between the Company and its stockholders, of even date herewith.

SECTION 2. REGISTRATION; RESTRICTIONS ON TRANSFER.

2.1 Restrictions on Transfer.

- (a) Each Holder agrees not to make any disposition of all or any portion of the Shares or Registrable Securities unless and until:
- (i) there is then in effect a registration statement under the Securities Act covering such proposed disposition and such disposition is made in accordance with such registration statement; or

3

- (ii) (A) The transferee has agreed in writing to be bound by the terms of this Agreement, (B) such Holder shall have notified the Company of the proposed disposition and shall have furnished the Company with a detailed statement of the circumstances surrounding the proposed disposition, and (C) if reasonably requested by the Company, such Holder shall have furnished the Company with an opinion of counsel, reasonably satisfactory to the Company, that such disposition will not require registration of such shares under the Securities Act. It is agreed that the Company will not require opinions of counsel for transactions made pursuant to Rule 144, except in unusual circumstances. After the Initial Offering, the Company will not require any transferee pursuant to Rule 144 to be bound by the terms of this Agreement if the shares so transferred do not remain Registrable Securities hereunder following such transfer.
- (b) Notwithstanding the provisions of subsection (a) above, no such restriction shall apply to a transfer by a Holder that is (A) a partnership transferring to its partners or former partners in accordance with partnership interests, (B) a corporation transferring to a wholly-owned subsidiary or a parent corporation that owns all of the capital stock of the Holder, (C) a limited liability company transferring to its members or former members in accordance with their interest in the limited liability company, or (D) an individual transferring to the Holder's family member or trust for the benefit of an individual Holder or the Holder's family members; provided that in each case the transferee agrees in writing to be subject to the terms of this Agreement to the same extent as if the transferee were an original Holder hereunder.
- (c) Each certificate representing Shares or Registrable Securities shall be stamped or otherwise imprinted with legends substantially similar to the following (in addition to any legend required under applicable state securities laws):

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "ACT") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, ASSIGNED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR PURSUANT TO AN EXEMPTION THEREFROM.

THE SALE, PLEDGE, HYPOTHECATION OR TRANSFER OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO THE TERMS AND CONDITIONS OF A CERTAIN INVESTOR RIGHTS AGREEMENT BY AND BETWEEN THE STOCKHOLDER AND THE COMPANY. COPIES OF SUCH AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF THE COMPANY.

(d) The Company shall be obligated to reissue promptly unlegended certificates at the request of any Holder thereof if (i) the securities are registered under the Act or (ii) the Holder shall have obtained an opinion of counsel (which counsel may be counsel to the Company) reasonably acceptable to the Company to the effect that the securities proposed to be disposed of may lawfully be so disposed of without registration or qualification, *provided that* the second legend listed above shall be removed only at such time as the Holder of such certificate is no longer subject to any restrictions hereunder.

4

(e) In addition to the foregoing, any legend endorsed on an instrument pursuant to applicable state securities laws and the stop-transfer instructions with respect to such securities shall be removed upon receipt by the Company of an order of the appropriate blue sky authority authorizing such removal.

2.2 Demand Registration.

(a) Subject to the conditions of this Section 2.2, if the Company shall receive a written request from Holders of Registrable Securities (the 'Initiating Holders') that the Company file a registration statement under the Securities Act covering the registration of at least 40% of the Registrable Securities then outstanding (or such lesser number of Registrable Securities if the anticipated aggregate offering price would exceed \$30,000,000), then the Company shall, within thirty (30) days of the receipt thereof, give written notice of such request to all Holders, and subject to the limitations of this Section 2.2, effect, as expeditiously as reasonably possible, the registration under

the Securities Act of all Registrable Securities that all Holders request to be registered.

(b) If the Initiating Holders intend to distribute the Registrable Securities covered by their request by means of an underwriting, they shall so advise the Company as a part of their request made pursuant to this Section 2.2 or any request pursuant to Section 2.4 and the Company shall include such information in the written notice referred to in Section 2.2(a) or Section 2.4(a), as applicable. In such event, the right of any Holder to include its Registrable Securities in such registration shall be conditioned upon such Holder's participation in such underwriting and the inclusion of such Holder's Registrable Securities in the underwriting to the extent provided herein. All Holders proposing to distribute their securities through such underwriting shall enter into an underwriting agreement in customary form with the underwriters selected for such underwriting by the Holders of a majority of the Registrable Securities held by all Initiating Holders (which underwriter or underwriters shall be reasonably acceptable to the Company). Notwithstanding any other provision of this Section 2.2 or Section 2.4, if the underwriter advises the Company that marketing factors require a limitation of the number of securities to be underwritten (including Registrable Securities) then the Company shall so advise all Holders of Registrable Securities that would otherwise be underwritten pursuant hereto, and the number of shares that may be included in the underwriting shall be allocated to the Holders of such Registrable Securities on a *pro rata* basis based on the number of Registrable Securities held by all such Holders (including the Initiating Holders), provided that, for the avoidance of doubt, all securities held by persons other than Holders shall first be excluded. Any Registrable Securities excluded or withdrawn from such underwriting shall be withdrawn from the registration.

(c) The Company shall not be required to effect a registration pursuant to this Section 2.2:

(i) prior to the earlier of (A) the fourth anniversary of the date of this Agreement or (B) six months following the effective date of the Initial

Offering;

5

(ii) after the Company has effected two (2) registrations pursuant to this Section 2.2, and such registrations have been declared or ordered effective;

(iii) during the period starting with the date of filing of, and ending on the date one hundred eighty (180) days following the effective date of the registration statement pertaining to the Initial Offering (or such longer period as may be determined pursuant to Section 2.11 hereof); provided that the Company (A) delivers written notice to the Holders of the Company's intention to file a registration statement for its Initial Offering within thirty (30) days after the Initiating Holders' written request under Section 2.2(a) and (B) makes reasonable good faith efforts to cause such registration statement to become effective;

(iv) if within thirty (30) days of receipt of a written request from Initiating Holders pursuant to Section 2.2(a), the Company gives notice to the Holders of the Company's intention to file a registration statement for its Initial Offering within ninety (90) days and the Company makes reasonable good faith efforts to cause such registration statement to become effective;

(v) if the Company shall furnish to Holders requesting a registration statement pursuant to this Section 2.2 a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board of Directors of the Company, it would be seriously detrimental to the Company and its stockholders for such registration statement to be effected at such time, in which event the Company shall have the right to defer such filing for a period of not more than sixty (60) days after receipt of the request of the Initiating Holders; provided that such right to delay a request shall be exercised by the Company not more than twice in any twelve (12) month period;

(vi) if the Initiating Holders propose to dispose of shares of Registrable Securities that may be immediately registered on Form S-3 pursuant to a request made pursuant to Section 2.4 below; or

(vii) in any particular jurisdiction in which the Company would be required to qualify to do business or to execute a general consent to service of process in effecting such registration, qualification or compliance, unless the Company is already subject to service in such jurisdiction and except as may be required by the Securities Act.

2.3 Piggyback Registrations. The Company shall notify all Holders of Registrable Securities in writing at least thirty (30) days prior to the filing of any registration statement under the Securities Act for purposes of a public offering of securities of the Company (including, but not limited to, registration statements relating to secondary offerings of securities of the Company, but excluding Special Registration Statements) and will afford each such Holder an opportunity to include in such registration statement all or part of such Registrable Securities held by such Holder. Each Holder desiring to include in any such registration statement all or any part of the Registrable Securities held by it shall, within fifteen (15) days after the above-described notice from the Company, so notify the Company in writing. Such notice shall state the intended method of disposition of the Registrable Securities by such Holder. If a Holder decides not to include all of its Registrable Securities in any registration statement thereafter filed by the Company, such Holder shall nevertheless continue to have the right to include any Registrable Securities in any subsequent registration statement or registration statements as may be filed by the Company with respect to offerings of its securities, all upon the terms and conditions set forth herein.

6

(a) Underwriting. If the registration statement of which the Company gives notice under this Section 2.3 is for an underwritten offering, the Company shall so advise the Holders of Registrable Securities. In such event, the right of any such Holder to include Registrable Securities in a registration pursuant to this Section 2.3 shall be conditioned upon such Holder's participation in such underwriting and the inclusion of such Holder's Registrable Securities in the underwriting to the extent provided herein. All Holders proposing to distribute their Registrable Securities through such underwriting shall enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting by the Company. Notwithstanding any other provision of this Agreement, if the managing underwriter determines in good faith that marketing factors require a limitation of the number of shares to be underwritten, the number of shares that may be included in the underwriting shall be allocated, first, to the Company; second, to the Holders on a pro rata basis based on the total number of Registrable Securities held by the Holders; and third, to any stockholder of the Company (other than a Holder) on a pro rata basis; provided that in no event shall the number of Registrable Securities be reduced below 30% of the offering except for the Initial Offering in which case Registrable Securities may be reduced to zero. In no event will shares of any other selling stockholder be included in such registration that would reduce the number of shares which may be included by Holders without the written consent of Holders of not less than a majority of the Registrable Securities proposed to be sold in the offering. If any Holder disapproves of the terms of any such underwriting, such Holder may elect to withdraw therefrom by written notice to the Company and the underwriter, delivered at least ten (10) business days prior to the effective date of the registration statement. Any Registrable Securities excluded or withdrawn from such underwriting shall be excluded and withdrawn from the registration. For any Holder which is a partnership, limited liability company or corporation, the partners, retired partners, members, retired members, stockholders and other affiliates of such Holder, or the estates and family members of any such partners, retired partners, members and retired members and any trusts for the benefit of any of the foregoing person shall be deemed to be a single "Holder," and any pro rata reduction with respect to such "Holder" shall be based upon the aggregate amount of shares carrying registration rights owned by all entities and individuals included in such "Holder," as defined in this sentence.

(b) Right to Terminate Registration. The Company shall have the right to terminate or withdraw any registration initiated by it under this Section 2.3 whether or not any Holder has elected to include securities in such registration, and shall promptly notify any Holder that has elected to include shares in such registration of such termination or withdrawal. The Registration Expenses of such withdrawn registration shall be borne by the Company in accordance with Section 2.5 hereof.

2.4 Form S-3 Registration. In case the Company shall receive from any Holder or Holders of Registrable Securities a written request or requests that the Company effect a registration on Form S-3 (or any successor to Form S-3) or any similar short-form registration statement and any related qualification or compliance with respect to all or a part of the Registrable Securities owned by such Holder or Holders, the Company will:

- (a) promptly give written notice of the proposed registration, and any related qualification or compliance, to all other Holders of Registrable Securities; and
- **(b)** as soon as practicable, effect such registration and all such qualifications and compliances as may be so requested and as would permit or facilitate the sale and distribution of all or such portion of such Holder's or Holders' Registrable Securities as are specified in such request, together with all or such portion of the Registrable Securities of any other Holder or Holders joining in such request as are specified in a written request given within fifteen (15) days after receipt of such written notice from the Company; *provided, however*, that the Company shall not be obligated to effect any such registration, qualification or compliance pursuant to this Section 2.4:
 - (i) if Form S-3 is not available for such offering by the Holders, or
- (ii) if the Holders, together with the holders of any other securities of the Company entitled to inclusion in such registration, propose to sell Registrable Securities and such other securities (if any) at an aggregate price to the public of less than one million dollars (\$1,000,000), or
- (iii) if within thirty (30) days of receipt of a written request from any Holder or Holders pursuant to this Section 2.4, the Company gives notice to such Holder or Holders of the Company's intention to file a registration statement for a public offering within ninety (90) days, other than pursuant to a Special Registration Statement, and the Company makes reasonable good faith efforts to cause such registration statement to become effective, or
- (iv) if the Company shall furnish to the Holders a certificate signed by the Chairman of the Board of Directors of the Company stating that in the good faith judgment of the Board of Directors of the Company, it would be seriously detrimental to the Company and its stockholders for such Form S-3 registration to be effected at such time, in which event the Company shall have the right to defer the filing of the Form S-3 registration statement for a period of not more than one hundred twenty (120) days after receipt of the request of the Holder or Holders under this Section 2.4; provided, that such right to delay a request shall be exercised by the Company not more than twice in any twelve (12) month period, or
- (v) if the Company has, within the twelve (12) month period preceding the date of such request, already effected two (2) registrations on Form S-3 for the Holders pursuant to this Section 2.4, or
- (vi) in any particular jurisdiction in which the Company would be required to qualify to do business or to execute a general consent to service of process in effecting such registration, qualification or compliance, unless the Company is already subject to service in such jurisdiction and except as may be required by the Securities Act.
- (c) Subject to the foregoing, the Company shall file a Form S-3 registration statement covering the Registrable Securities and other securities so requested to be registered as soon as practicable after receipt of the requests of the Holders. Registrations effected pursuant to this Section 2.4 shall not be counted as demands for registration or registrations effected pursuant to Section 2.2.

8

- 2.5 Expenses of Registration. Except as specifically provided herein, all Registration Expenses incurred in connection with any registration, qualification or compliance pursuant to Section 2.2, 2.3 or 2.4 herein, including the fees and disbursements of one counsel for the selling Holders not to exceed \$30,000, shall be borne by the Company. All Selling Expenses incurred in connection with any registrations hereunder, shall be borne by the holders of the securities so registered *pro rata* on the basis of the number of shares so registered. The Company shall not, however, be required to pay for expenses of any registration proceeding begun pursuant to Section 2.2 or 2.4, the request of which has been subsequently withdrawn at the request of the Holders holding a majority of the Registrable Securities to be registered or because of sufficient number of Holders have withdrawn so that the minimum offering conditions set forth in Sections 2.2(a) and 2.4(b)(ii) are no longer satisfied unless (a) the withdrawal is based upon material adverse information concerning the Company of which the Holders were not aware at the time of such request or (b) the Holders of a majority of Registrable Securities to be registered agree to deem such registration to have been effected as of the date of such withdrawal for purposes of determining whether the Company shall be obligated pursuant to Section 2.2(c)(ii) or 2.4(b)(iii), as applicable, to undertake any subsequent registration Expenses of a withdrawn offering pursuant to clause (a) above, then such registration shall not be deemed to have been effected for purposes of determining whether the Company shall be obligated pursuant to Section 2.2(c)(iii) or 2.2(c)(iii) or 2.2(c)(iiii) or 2.2(c)(iiiii) or 2.2(c)(iiii) or 2.2(c)(iiii) or 2.2(c)(iiii) or 2.2(c)(iiii
- 2.6 Obligations of the Company. Whenever required to effect the registration of any Registrable Securities, the Company shall, as expeditiously as reasonably possible:
- (a) prepare and file with the SEC a registration statement with respect to such Registrable Securities and use all commercially reasonable best efforts to cause such registration statement to become effective, and shall keep such registration statement effective for one hundred twenty (120) days or, if earlier, until the Holder or Holders have completed the distribution related thereto; provided, however, such one hundred twenty (120) day period shall be extended for the period of time, not to exceed sixty (60) days, that a Holder refrains, at the reasonable written request of the Company based on the existence of material nonpublic information involving the Company, the failure of which to be disclosed in the prospectus included in the registration statement could result in a Violation (as defined below), from selling any securities included in such registration, provided the Company is taking commercially reasonable best efforts to correct the prospectus. Notwithstanding the foregoing, the Company shall not be required to file, cause to become effective or maintain the effectiveness of any registration statement other than a registration statement on Form S-3 that contemplates a distribution of securities on a delayed or continuous basis pursuant to Rule 415 under the Securities Act.
- (b) Prepare and file with the SEC such amendments and supplements to such registration statement and the prospectus used in connection with such registration statement as may be necessary to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement for the period set forth in subsection (a) above.

9

- (c) Furnish to the Holders such number of copies of a prospectus, including a preliminary prospectus, in conformity with the requirements of the Securities Act, and such other documents as they may reasonably request in order to facilitate the disposition of Registrable Securities owned by them.
- (d) Use its commercially reasonable best efforts to register and qualify the securities covered by such registration statement under such other securities or Blue Sky laws of such jurisdictions as shall be reasonably requested by the Holders; provided that the Company shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdictions, unless the Company is already subject to service in such jurisdiction and except as may be required by the Securities Act.
- (e) In the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing underwriter(s) of such offering.
 - (f) Use its commercially reasonable efforts to cause all such Registrable Securities covered by such registration statement to be listed on a national securities

exchange or trading system and each securities exchange and trading system (if any) on which similar securities issued by the Company are then listed.

- (g) Provide a transfer agent and registrar for all Registrable Securities registered pursuant to this Agreement and provide a CUSIP number for all such Registrable Securities, in each case not later than the effective date of such registration.
- (h) Promptly make available for inspection by the selling Holders, any managing underwriter(s) participating in any disposition pursuant to such registration statement, and any attorney or accountant or other agent retained by any such underwriter or selected by the selling Holders, all financial and other records, pertinent corporate documents, and properties of the Company, and cause the Company's officers, directors, employees, and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant, or agent, in each case, as necessary or advisable to verify the accuracy of the information in such registration statement and to conduct appropriate due diligence in connection therewith.
- (i) Notify each Holder of Registrable Securities covered by such registration statement at any time when a prospectus relating thereto is required to be delivered under the Securities Act of the happening of any event as a result of which the prospectus included in such registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing. The Company will use commercially reasonable best efforts to amend or supplement such prospectus in order to cause such prospectus not to include any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing.

10

(j) Use its commercially reasonable best efforts to furnish, on the date that such Registrable Securities are delivered to the underwriters for sale, if such securities are being sold through underwriters, (i) an opinion, dated as of such date, of the counsel representing the Company for the purposes of such registration, in form and substance as is customarily given to underwriters in an underwritten public offering, addressed to the underwriters, if any, and (ii) a letter, dated as of such date, from the independent certified public accountants of the Company, in form and substance as is customarily given by independent certified public accountants to underwriters in an underwritten public offering addressed to the underwriters.

2.7 Delay of Registration; Furnishing Information.

- (a) No Holder shall have any right to obtain or seek an injunction restraining or otherwise delaying any such registration as the result of any controversy that might arise with respect to the interpretation or implementation of this Section 2.
- (b) Each selling Holder shall furnish to the Company such information regarding themselves, the Registrable Securities held by them and the intended method of disposition of such securities as shall be reasonably required to effect the registration of their Registrable Securities.
- (c) The Company shall have no obligation to complete a registration with respect to any registration requested pursuant to Section 2.2 or Section 2.4 if the number of shares or the anticipated aggregate offering price of the Registrable Securities to be included in the registration does not equal or exceed the number of shares or the anticipated aggregate offering price required to originally trigger the Company's obligation to initiate such registration as specified in Section 2.2(a) or Section 2.4(b)(ii), whichever is applicable.
 - 2.8 Indemnification. In the event any Registrable Securities are included in a registration statement under Sections 2.2, 2.3 or 2.4:
- (a) To the extent permitted by law, the Company will indemnify and hold harmless each Holder, the partners, members, legal counsel, accountants, officers and directors of each Holder, any underwriter (as defined in the Securities Act) for such Holder and each person, if any, who controls such Holder or underwriter within the meaning of the Securities Act or the Exchange Act, against any losses, claims, damages, or liabilities (joint or several) (or actions, proceedings or settlements in respect thereof) to which they may become subject under the Securities Act, the Exchange Act or other federal or state law, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively a "Violation") by the Company: (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement or incorporated reference therein, including any preliminary prospectus or final prospectus contained therein or any amendments or supplements thereto, (ii) the omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading, or (iii) any violation or alleged violation by the Company of the Securities Act, the Exchange Act, any state securities law in connection with the offering covered by such registration statement; and the Company will reimburse each such Holder, partner, member, legal counsel, accountant, officer, director, underwriter or controlling person for any legal or other expenses reasonably incurred by them in connection with investigating or defending any such loss, claim, damage, liability or action; provided however, that the indemnity agreement contained in this Section 2.8(a) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Company, which consent shall not be unreasonab

11

(b) To the extent permitted by law, each Holder will, severally and not jointly, if Registrable Securities held by such Holder are included in the securities as to which such registration qualifications or compliance is being effected, indemnify and hold harmless the Company, each of its directors, its officers and each person, if any, who controls the Company within the meaning of the Securities Act, any underwriter and any other Holder selling securities under such registration statement or any of such other Holder's partners, directors or officers or any person who controls such Holder, against any losses, claims, damages or liabilities (joint or several) to which the Company or any such director, officer, controlling person, underwriter or other such Holder, or partner, director, officer or controlling person of such other Holder may become subject under the Securities Act, the Exchange Act or other federal or state law, insofar as such losses, claims, damages or liabilities (or actions in respect thereto) arise out of or are based upon any of the following statements: (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement or incorporated reference therein, including any preliminary prospectus or final prospectus contained therein or any amendments or supplements thereto, (ii) the omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading, or (iii) any violation or alleged violation by the Company of the Securities Act (collectively, a "Holder Violation"), in each case to the extent (and only to the extent) that such Holder Violation occurs in reliance upon and in conformity with written information furnished by such Holder under an instrument duly executed by such Holder and stated to be specifically for use in connection with such registration; and each such Holder will reimburse any legal or other expenses reasonably incurred by the Company or any such director, officer, controlling person, underwriter or other Holder, or partner, officer, director or controlling person of such other Holder in connection with investigating or defending any such loss, claim, damage, liability or action if it is judicially determined that there was such a Holder Violation; provided, however, that the indemnity agreement contained in this Section 2.8(b) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Holder, which consent shall not be unreasonably withheld; provided further, that in no event shall any indemnity under this Section 2.8 exceed the net proceeds from the offering received by such Holder.

(c) Promptly after receipt by an indemnified party under this Section 2.8 of notice of the commencement of any action (including any governmental action), such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section 2.8, deliver to the indemnifying party a written notice of the commencement thereof and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense thereof with counsel mutually satisfactory to the parties; provided, however, that an indemnified party shall have the right to retain its own counsel, with the fees and expenses thereof to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the

(d) If the indemnification provided for in this Section 2.8 is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any losses, claims, damages or liabilities referred to herein, the indemnifying party, in lieu of indemnifying such indemnified party thereunder, shall to the extent permitted by applicable law contribute to the amount paid or payable by such indemnified party as a result of such loss, claim, damage or liability in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other in connection with the Violation(s) or Holder Violation(s) that resulted in such loss, claim, damage or liability, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by a court of law by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission; *provided*, *that* in no event shall any contribution by a Holder hereunder exceed the net proceeds from the offering received by such Holder.

(e) The obligations of the Company and Holders under this Section 2.8 shall survive completion of any offering of Registrable Securities in a registration statement and, with respect to liability arising from an offering to which this Section 2.8 would apply that is covered by a registration filed before termination of this Agreement, such termination. No indemnifying party, in the defense of any such claim or litigation, shall, except with the consent of each indemnified party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Party of a release from all liability in respect to such claim or litigation.

2.9 Assignment of Registration Rights. The rights to cause the Company to register Registrable Securities pursuant to this Section 2 may be assigned by a Holder to a transferee or assignee of Registrable Securities (for so long as such shares remain Registrable Securities) that (a) is a subsidiary, parent, partner, retired partner, member or retired member, or stockholder of a Holder that is a corporation, partnership or limited liability company, (b) is a Holder's family member or trust for the benefit of an individual Holder, or (c) acquires at least twenty percent (20%) of a Holder's shares of Registrable Securities; or (d) is an entity affiliated by common control (or other related entity) with such Holder; provided, however, (i) the transferor shall, within ten (10) days after such transfer, furnish to the Company written notice of the name and address of such transferee or assignee and the securities with respect to which such registration rights are being assigned and (ii) such transferee shall agree to be subject to all restrictions set forth in this Agreement.

13

2.10 Limitation on Subsequent Registration Rights. Other than as provided in Section 5.10, after the date of this Agreement, the Company shall not enter into any agreement with any holder or prospective holder of any securities of the Company that would grant such holder rights to demand the registration of shares of the Company's capital stock, or to include such shares in a registration statement.

2.11 Market Stand-Off Agreement. Each Holder hereby agrees that such Holder shall not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale, any shares of Common Stock (or other securities) of the Company held by such Holder (other than those included in the registration) during (i) the 180-day period following the effective date of the Initial Offering (or such longer period, not to exceed 34 days after the expiration of the 180-day period, as the underwriters or the Company shall request in order to facilitate compliance with FINRA Rule 2711 or NYSE Member Rule 472 or any successor or similar rule or regulation), and (ii) the 90-day period following the effective date of a registration statement of the Company filed under the Securities Act (or such longer period, not to exceed 34 days after the expiration of the 180-day period, as the underwriters or the Company shall request in order to facilitate compliance with FINRA Rule 2711 or NYSE Member Rule 472 or any successor or similar rule or regulation); provided, that (i) all officers and directors of the Company and holders of at least one percent (1%) of the Company's voting securities are bound by and have entered into similar agreements and (ii) if the Company or managing underwriters permit any Company stockholder to sell securities in the Initial Offering, all Holders shall receive the same right. The Company shall require all future holders of any capital stock of the Company, or any securities convertible into such capital stock, to agree to a market stand-off provision no less restrictive than the foregoing.

2.12 Stop Transfer Instructions. In connection with the Initial Offering, each Holder agrees to execute and deliver such other agreements as may be reasonably requested by the Company or the managing underwriters that are consistent with the Holder's obligations under Section 2.11 or that are necessary to give further effect thereto. The obligations described in Section 2.11 and this Section 2.12 shall not apply to a Special Registration Statement. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to such shares of Common Stock (or other securities) until the end of such period. Each Holder agrees that any transferee of any shares of Registrable Securities shall be bound by Sections 2.11 and 2.12. The underwriters of the Company's stock are intended third party beneficiaries of Sections 2.11 and 2.12 and shall have the right, power and authority to enforce the provisions hereof as though they were a party hereto.

2.13 Agreement to Furnish Information. If reasonably requested by the Company or the representative of the underwriters of Common Stock (or other securities) of the Company, each Holder shall provide such information regarding the Holder and the distribution proposed by the Holder as may be required by the Company or such representative in connection with the completion of any public offering of the Company's securities in which the Holder is participating pursuant to a registration statement filed under the Securities Act referred to in this Section 2.

14

2.14 Rule 144 Reporting. With a view to making available to the Holders the benefits of certain rules and regulations of the SEC which may permit the sale of the Registrable Securities to the public without registration, the Company agrees to use its best efforts to:

(a) Make and keep public information available, as those terms are understood and defined in SEC Rule 144 or any similar or analogous rule promulgated under the Securities Act, at all times after the effective date of the first registration filed by the Company for an offering of its securities to the general public;

(b) File with the SEC, in a timely manner, all reports and other documents required of the Company under the Exchange Act; and

(c) So long as a Holder owns any Registrable Securities, furnish to such Holder forthwith upon request: a written statement by the Company as to its compliance with the reporting requirements of said Rule 144 of the Securities Act, and of the Exchange Act (at any time after it has become subject to such reporting requirements); a copy of the most recent annual or quarterly report of the Company filed with the Commission; and such other reports and documents as a Holder may reasonably request in connection with availing itself of any rule or regulation of the SEC allowing it to sell any such securities without registration.

2.15 Termination of Registration Rights. The right of any Holder to request registration or inclusion of Registrable Securities in any registration pursuant to Section 2.2, Section 2.3, or Section 2.4 hereof shall terminate upon the earlier of: (i) the date six (6) years following an initial firm underwritten public offering that results in the automatic conversion of all outstanding shares of the Company's Preferred Stock; or (ii) such time as such Holder, as reflected on the Company's list of stockholders, holds less than 1% of the Company's outstanding Common Stock (treating all shares of Preferred Stock on an as converted basis), the Company has completed the Initial Offering and all Registrable Securities of the Company issuable or issued upon conversion of the Shares held by and issuable to such Holder (and its affiliates) may be sold pursuant to Rule 144 during any ninety (90) day period. Upon such termination, such shares shall cease to be "Registrable Securities" hereunder for all purposes.

SECTION 3. COVENANTS OF THE COMPANY.

3.1 Basic Financial Information and Reporting.

(a) The Company will maintain true books and records of account in which full and correct entries will be made of all its business transactions pursuant to a system of accounting established and administered in accordance with generally accepted accounting principles consistently applied, and will set aside on its books all such proper accruals and reserves as shall be required under generally accepted accounting principles consistently applied.

(b) The Company will furnish to each Major Investor, as soon as practicable, but in any event within ninety (90) days after the end of each fiscal year of the Company (i) a balance sheet as of the end of such year, (ii) statements of income and of cash flows for such year, and a comparison between (x) the actual amounts as of and for such fiscal year and (y) the comparable amounts for the prior year and as included in the Budget (as defined in Section 3.1(e)) for such year, with an explanation of any material differences between such amounts and a schedule as to the sources and applications of funds for such year, and (iii) a statement of stockholders' equity as of the end of such year, all such financial statements audited and certified by independent public accountants of regionally recognized standing selected by the Company;

15

(c) The Company will furnish to each Major Investor, as soon as practicable after the end of the first, second and third quarterly accounting periods in each fiscal year of the Company, and in any event within thirty (30) days thereafter, a balance sheet of the Company and a statement of stockholders' equity as of the end of each such quarterly period, and a statement of income and a statement of cash flows of the Company for such period and for the current fiscal year to date, prepared in accordance with generally accepted accounting principles consistently applied, with the exception that no notes need be attached to such statements and year-end audit adjustments may not have been made.

(d) The Company will furnish to each Major Investor, as soon as practicable, but in any event within forty-five (45) days after the end of each quarter of each fiscal year of the Company, a statement showing the number of shares of each class and series of capital stock and securities convertible into or exercisable for shares of capital stock outstanding at the end of the period, the Common Stock issuable upon conversion or exercise of any outstanding securities convertible or exercisable for Common Stock and the exchange ratio or exercise price applicable thereto, and the number of shares of issued stock options and stock options not yet issued but reserved for issuance, if any, all in sufficient detail as to permit the Major Investors to calculate their respective percentage equity ownership in the Company.

(e) Additionally, the Company will furnish to each Major Investor (i) by December 31 of each year, an annual budget and business plan for the next fiscal year (the "Budget") as approved by the Company's Board of Directors, including the Series D Director (as defined in the Voting Agreement) after the Milestone Closing, prepared on a quarterly basis, including balance sheets, income statements, and statements of cash flow for such months; (ii) a capitalization table upon the closing of the Financing and after the closing of any subsequent rounds of preferred stock financings; (iii) promptly after prepared, any other budgets or revised budgets prepared by the Company.

(f) Additionally, the Company will furnish to each Major Investor such other information relating to the financial condition, business, prospects, or corporate affairs of the Company as any Major Investor may from time to time reasonably request; <u>provided, however</u>, that the Company shall not be obligated under this Section 3.1 to provide information (i) that the Company reasonably determines in good faith to be a trade secret or confidential information (unless covered by an enforceable confidentiality agreement, in a form acceptable to the Company); or (ii) the disclosure of which would adversely affect the attorney-client privilege between the Company and its counsel.

3.2 Inspection Rights. Each Major Investor shall have the right to visit and inspect any of the properties of the Company or any of its subsidiaries, and to discuss the affairs, finances and accounts of the Company or any of its subsidiaries with its officers, and to review such information as is reasonably requested all at such reasonable times and as often as may be reasonably requested; provided, however, that the Company shall not be obligated under this Section 3.2 with respect to a competitor of the Company or with respect to information which the Board of Directors determines in good faith is a trade secret or attorney-client privileged and should not, therefore, be disclosed; provided, however that BSC shall not be deemed a competitor of the Company.

16

3.3 Confidentiality of Records. Each Investor agrees to use the same degree of care as such Investor uses to protect its own confidential information to keep confidential any information furnished to such Investor hereunder that the Company marks as being confidential or proprietary (so long as such information is not in the public domain), except that such Investor may disclose such proprietary or confidential information: (i) to any of such Investor's attorneys, accountants and other advisors, to the extent necessary to obtain their services; (ii) to any partner, member, affiliate, subsidiary or parent of such Investor as long as such partner, member, affiliate, subsidiary or parent is obligated to Investor maintain the confidentiality of the same; (iii) at such time as it enters the public domain through no fault of such Investor; (iv) that is communicated to it free of any obligation of confidentiality; (v) that is developed by Investor or its agents independently of and without reference to any confidential information communicated by the Company; or (vi) as required by applicable law.

3.4 Reservation of Common Stock. The Company will at all times reserve and keep available, solely for issuance and delivery upon the conversion of the Preferred Stock, all Common Stock issuable from time to time upon such conversion.

3.5 Stock Vesting. Unless otherwise approved by the Board of Directors, including the Required Preferred Approval following the Milestone Closing, all employees, directors, consultants and other service providers who purchase, receive options to purchase or receive awards of shares of the Company's capital stock after the date hereof (including any future option grants to Kamran Najmabadi, Ramtin Agah and Shaun Bagai) shall be required to execute a restricted stock or option agreement, as applicable, providing for (a) vesting of shares over a four (4) year period, with the first twenty-five percent (25%) of such stock vesting following one (1) year of continued employment or service, and the remaining seventy-five percent (75%) of such stock vesting in equal monthly installments over the remaining three (3) years, subject to such person's continuous service and (b) a market stand-off provision substantially similar to that contained in Section 2.11 hereof. In addition, unless otherwise approved by the Board of Directors, including the Required Preferred Approval following the Milestone Closing, the Company shall retain a "right of first refusal" on employee stock transfers until the Company's IPO and shall have the right to repurchase unvested shares at cost upon termination of employment of a holder of restricted stock.

3.6 Director and Officer Insurance and Indemnification. The Company will use its best efforts to obtain and maintain in full force and effect director and officer liability insurance in an amount satisfactory to the Company's Board of Directors and shall ensure that any successors of the Company shall assume the Company's obligations (whether through contractual agreement and/or the Delaware General Corporation Law) with respect to indemnification of members of the Company's Board of Directors. The Company hereby acknowledges that one (1) or more of the directors nominated to serve on the Board of Directors by the Investors (each an "Investor Director") may have certain rights to indemnification, advancement of expenses and/or insurance provided by one or more of the Investors and certain of their affiliates (collectively, the "Investor Indemnitors"). The Company hereby agrees (a) that it is the indemnitor of first resort (i.e., its obligations to any such Investor Directors are primary and any obligation of the Investor Indemnitors to advance expenses or to provide indemnification for the same expenses or liabilities incurred by such Investor Director are secondary), (b) that it shall be required to advance the full amount of expenses incurred by such Investor Director and shall be liable for the full amount of all expenses, judgments, penalties, fines and amounts paid in settlement by or on behalf of any such Investor Director to the extent legally permitted and as required by the Company's Certificate of Incorporation or Bylaws of the Company (or any agreement between the Company and such Investor Director), without regard to any rights such Investor Director may have against the Investor Indemnitors, and, (c) that it irrevocably waives, relinquishes and releases the Investor Indemnitors from any and all claims against the Investor Indemnitors on behalf of any such Investor Director with respect to any claim for which such Investor Director has sought indemnification from the Company shall affect the foregoi

17

3.7 Board Observer Rights. The Company shall allow (a) an individual designated by Golden Seeds Fund 2 LP (so long as it holds any shares of Registrable Securities (as adjusted for any stock dividends, combinations, splits, recapitalizations and the like after the date hereof)) (the "Golden Seeds Observer"), (b) an individual designated by Astia Angels RenovoRx, LLC (so long as it holds any shares of Registrable Securities (as adjusted for any stock dividends, combinations, splits, recapitalizations and the like after the date hereof)) (the "Astia Angels Observer"), (c) either of Kamran Najmabadi or Ramtin Agah (each, a "Founder"), as designated in writing signed by the Founders, so long as the Founder so designated is (i) providing services to the Company as an employee or pursuant to written agreements for consulting or advisory services and (ii) not serving as a director on the Company's Board of Directors (the "Founder Observer" and collectively with the Golden Seeds Observer and the Astia Angels Observer, the "Board Observers") (initially Kamran Najmabadi) and (d) two individuals designated by BSC (so long as it holds any Shares of Registrable Securities (as adjusted for any stock dividends, combinations, splits, recapitalizations and the like after the date hereof)) (the "BSC Observers"), provided, however, that upon BSC's appointment of the Series D Director pursuant to the terms of the Voting Agreement, the number of BSC Observers shall be reduced from two (2) to one (1), to attend all meetings of the Company's Board of Directors as an observer, in a nonvoting capacity, and in connection therewith, the Company shall give such representative copies of all notices, minutes, consents and other materials, financial or otherwise, which the Company provides to its Board of Directors; provided, however, that the Company reserves the right to exclude any Board Observer from access to any material or meeting or portion thereof if the Company believes upon advice of counsel that such exclusion is reasonably neces

3.8 Proprietary Information and Inventions Agreement. The Company shall require all employees and consultants to execute and deliver a Proprietary Information and Inventions Agreement substantially in a form approved by the Company's counsel or Board of Directors.

18

3.9 Board Meetings and Committees. The Board of Directors shall have regular meetings, the frequency of which shall be determined by the Board of Directors before the Milestone Closing (as defined in the Purchase Agreement), and shall be no less than quarterly after the Milestone Closing. Following the Milestone Closing, the Series D Director shall have the right to sit on each committee. No member of any committee of the Board of Directors shall have the right to participate in any compensation committee discussions pertaining to his or her own compensation. The Company hereby agrees to establish an Audit Committee and a Compensation Committee within 30 days after the First Closing (as defined in the Stock Purchase Agreement), with customary charters, policies and provisions.

3.10 Right to Conduct Activities. The Company and each Investor hereby agrees and acknowledges that the Major Investors (and their affiliates) invest in numerous portfolio companies and conduct certain business activities, some of which are or may be deemed competitive with the Company's business. The Company and each Investor hereby agrees that, to the extent permitted under applicable law, the Major Investors (and their affiliates) shall not be liable to the Company or any other Investor for any claim arising out of, or based upon, (i) the investment by the Major Investors (or their affiliates) in any entity competitive with the Company, or (ii) actions taken by any partner, officer or other representative of the Major Investors (or their affiliates) to assist any such competitive company, whether or not such action was taken as a member of the board of directors of such competitive company or otherwise, and whether or not such action has a detrimental effect on the Company; provided, however, that the foregoing shall not relieve (x) the Major Investors (and their affiliates) from liability associated with the unauthorized disclosure of the Company's confidential information obtained pursuant to this Agreement, or (y) any director or officer of the Company from any liability associated with his or her fiduciary duties to the Company.

3.11 Directors' Expenses. The Company shall reimburse all reasonable out-of-pocket and travel expenses incurred by any non-employee director in connection with (i) attendance at Board of Directors meetings (including any meetings of committees thereof) and (ii) the performance of his or her duties as a director of the Company. Any travel reimbursements shall be consistent with the Company's travel policies for its officers and shall be prorated based on expenses incurred by such director on behalf of companies other than the Company.

3.12 Additional Board of Director Approvals. For so long as at least 3,000,000 shares of Registrable Securities are outstanding (as adjusted for any stock dividends, combinations, splits, recapitalizations and the like after the date hereof), the Company shall obtain approval of the Board of Directors, including the Required Preferred Approval following the Milestone Closing, for any of the following actions that:

(1) Results in any spinout, sale or exclusive license of material assets or intellectual property rights of the Company outside the ordinary course of business, including by means of transfer to a subsidiary or another entity;

(2) Causes the Company to incur or guarantee any aggregate indebtedness in excess of \$25,000 that is not already included in the Budget, other than trade credit incurred in the ordinary course of business;

19

(3) Causes the Company to make any investment inconsistent with an investment policy approved by the Board of Directors;

(4) Causes the Company to hire, terminate, or change the compensation of the executive officers, including approving any option grants or stock awards to executive officers;

(5) Results in a fundamental change in the nature of the business of the Company, including changing the principal business of the Company, entering new lines of business, or exiting the current line of business;

(6) Adopts any Company equity incentive plan or authorizes any grants of equity securities under such Company equity incentive plans;

(7) Results in the Company becoming party to any transaction with any director, officer, or employee of the Company or any "associate" (as defined in Rule 12b- 2 promulgated under the Exchange Act) of any such person, except for transactions contemplated by this Agreement, the Purchase Agreement, and the Related Agreements (as defined in the Purchase Agreement), or transactions made upon fair and reasonable terms as determined and approved by a majority of the Board of Directors, including the Required Preferred Approval following the Milestone Closing;

(8) Increases the number of shares reserved under the Company's equity incentive plan(s);

(9) Results in a compensation package to any Company employee or consultant in excess of \$200,000 per year in cash that is not included

in the Budget; or

(10) Results in the grant or issuance of any capital stock, options, warrants or other equity securities of the Company that exceeds one percent (1%) of the fully-diluted capitalization of the Company on the date of such grant or issuance.

3.13 Option Issuances. Effective as of the date hereof, of the total number of 1,677,193 shares available for issuance under the Company's 2013 Equity Incentive Plan

as of the date hereof, the Company may issue such shares (or options for such shares) only as follows: (a) an aggregate of 419,298 of such shares shall be reserved for issuance, and may be issued, only to Mutual Directors (as defined in the Voting Agreement); (b) an aggregate of 419,298 of such shares shall be reserved for issuance, and may be issued, only to employees, consultants or other services providers hired after the date hereof; and (c) no more than an aggregate of 838,597 of such shares (or options for such shares) may be issued in such instances and to such individuals or entities as approved by the Board of Directors. Notwithstanding the foregoing, (x) beginning on July 1, 2019, upon unanimous approval of any directors designated solely by the holders of one or more series of Preferred Stock as a single class (to the extent there are any), the restrictions set forth in clauses (a) and (b) of the preceding sentence shall no longer apply; and (y) on July 1, 2020 all restrictions set forth in this Section 3.13 shall automatically terminate unless prior to July 1, 2020 all directors designated solely by the holders of one or more series of Preferred Stock as a single class (to the extent there are any) unanimously elect to keep any such restrictions (except for clause (x) which shall operate pursuant to its terms).

20

3.14 Termination of Covenants. All covenants of the Company contained in Section 3 of this Agreement (other than the provisions of Section 3.3 and 3.6 shall expire and terminate as to each Investor upon the earlier of (i) the effective date of the registration statement pertaining to an Initial Offering or (ii) upon an "Acquisition" as defined in the Company's Amended and Restated Certificate of Incorporation as in effect as of the date hereof (the "Restated Certificate")).

SECTION 4. RIGHTS OF FIRST REFUSAL.

- 4.1 Subsequent Offerings. Each Major Investor shall have a right of first refusal to purchase its pro rata share of all Equity Securities, as defined below, that the Company may, from time to time, propose to sell and issue after the date of this Agreement, other than the Equity Securities excluded by Section 4.6 hereof. Each Major Investor's pro rata share is equal to the ratio of (a) the number of shares of the Company's Common Stock (including all shares of Common Stock issuable or issued upon conversion of the Shares or upon the exercise of outstanding warrants or options or other convertible securities) of which such Investor is or is deemed to be a holder immediately prior to the issuance of such Equity Securities to (b) the total number of shares of the Company's outstanding Common Stock (including all shares of Common Stock issued or issuable upon conversion of the Shares or upon the exercise of any outstanding warrants, options or other convertible securities) held by all Major Investors immediately prior to the issuance of the Equity Securities. The term "Equity Securities" shall mean (i) any Common Stock, Preferred Stock or other security of the Company, (ii) any security convertible into or exercisable or exchangeable for, with or without consideration, any Common Stock, Preferred Stock or other security (including without limitation any option to purchase such a convertible security and any convertible promissory note), (iii) any security carrying any warrant or right to subscribe to or purchase any Common Stock, Preferred Stock or other security or (iv) any such warrant or right.
- 4.2 Exercise of Rights. If the Company proposes to issue any Equity Securities, it shall give each Major Investor written notice of its intention, describing the Equity Securities, the price and the terms and conditions upon which the Company proposes to issue the same. Each Major Investor shall have twenty (20) days after the receipt of such notice to agree to purchase its *pro rata* share (or any portion thereof) of the Equity Securities for the price and upon the terms and conditions specified in the notice by giving written notice to the Company and stating therein the quantity of Equity Securities to be purchased. Notwithstanding the foregoing, the Company shall not be required to offer or sell such Equity Securities to any Major Investor who would cause the Company to be in violation of applicable federal securities laws by virtue of such offer or sale.
- 4.3 Overallotment; Issuance of Equity Securities to Other Persons If not all of the Major Investors elect to purchase their full pro rata share of the Equity Securities, then the Company shall promptly notify in writing the Major Investors who do so elect and shall offer such Major Investors the right to acquire such unsubscribed Equity Securities on a relative pro rata basis. The Major Investors shall have ten (10) days after receipt of such overallotment notice to notify the Company of its election to purchase all or a portion thereof of the unsubscribed Equity Securities. The Company shall have until the date that is one hundred twenty (120) days after the date of the notice provided pursuant to Section 4.2 to sell the remaining unsubscribed Equity Securities in respect of which the Major Investor's rights were not exercised, at a price not lower and upon general terms and conditions not more favorable to the purchasers thereof than specified in the Company's notice to the Major Investors pursuant to Section 4.2 hereof. If the Company has not sold such Equity Securities within one hundred twenty (120) days of the notice provided pursuant to Section 4.2, the Company shall not thereafter issue or sell any Equity Securities, without first offering such securities to the Major Investors in the manner provided above.

21

- **4.4 Termination and Waiver of Rights of First Refusal.** The rights of first refusal established by this Section 4 shall not apply to, and shall terminate upon the earlier of (i) the effective date of the registration statement pertaining to the Company's Initial Offering that results in the automatic conversion of all outstanding shares of Preferred Stock or (ii) an Acquisition as defined in the Restated Certificate.
- 4.5 Assignment of Rights of First Refusal. The rights of first refusal of each Major Investor under this Section 4 may be assigned to the same parties, subject to the same restrictions as any transfer of registration rights pursuant to Section 2.9.
- **4.6 Excluded Securities.** The rights of first refusal established by this Section 4 shall have no application to any Equity Securities sold pursuant to the Purchase Agreement or that are "Additional Shares of Common Stock" as defined in the Restated Certificate.

SECTION 5. MISCELLANEOUS.

- **5.1 Governing Law; Venue.** This Agreement shall be governed by and construed under the laws of the State of Delaware in all respects as such laws are applied to agreements among Delaware residents entered into and to be performed entirely within Delaware, without reference to conflicts of laws or principles thereof. The parties agree that any action brought by either party under or in relation to this Agreement, including without limitation to interpret or enforce any provision of this Agreement, shall be brought exclusively in, and each party agrees to and does hereby submit to the exclusive jurisdiction and venue of, any state or federal court located in Delaware.
- **5.2 Successors and Assigns.** Except as otherwise expressly provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the parties hereto and their respective successors, assigns, heirs, executors, and administrators and shall inure to the benefit of and be enforceable by each person who shall be a holder of Registrable Securities from time to time.
- **5.3 Entire Agreement.** This Agreement, the Exhibits and Schedules hereto, the Purchase Agreement and the other documents delivered pursuant thereto constitute the full and entire understanding and agreement between the parties with regard to the subjects hereof and no party shall be liable or bound to any other in any manner by any oral or written representations, warranties, covenants and agreements except as specifically set forth herein and therein. Each party expressly represents and warrants that it is not relying on any oral or written representations, warranties, covenants or agreements outside of this Agreement.

22

5.4 Severability. In the event one or more of the provisions of this Agreement should, for any reason, be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provisions of this Agreement, and this Agreement shall be construed as if such invalid, illegal or unenforceable provision had never been contained herein.

5.5 Amendment and Waiver.

under this Agreement may be waived, only upon the written consent of the Company and the holders of at least a majority of the then-outstanding Registrable Securities; provided that any waiver or amendment of Sections 3.1, 3.2, 3.7 (with respect to the particular Major Investor entitled to appoint an observer or observers) or Section 4 shall require the further written consent of Major Investors holding a majority of the Registrable Securities; provided further that this Agreement may not be amended or terminated and the observance of any term hereof may not be waived with respect to any Investor without the written consent of such Investor, unless such amendment, termination, or waiver applies to all Investors in the same fashion; and provided further that Section 1.2(f), Section 1.2(k), Section 3.5, and this clause of this Section 5.5 may not be amended without BSC's consent.

- (b) For the purposes of determining the number of Holders or Investors entitled to vote or exercise any rights hereunder, the Company shall be entitled to reasonably rely on the list of record holders of its stock as maintained by or on behalf of the Company.
- 5.6 Delays or Omissions. It is agreed that no delay or omission to exercise any right, power, or remedy accruing to any party, upon any breach, default or noncompliance by another party under this Agreement shall impair any such right, power, or remedy, nor shall it be construed to be a waiver of any such breach, default or noncompliance, or any acquiescence therein, or of any similar breach, default or noncompliance thereafter occurring. It is further agreed that any waiver, permit, consent, or approval of any kind or character on any party's part of any breach, default or noncompliance under the Agreement or any waiver on such party's part of any provisions or conditions of this Agreement must be in writing and shall be effective only to the extent specifically set forth in such writing. All remedies, either under this Agreement, by law, or otherwise afforded to any party, shall be cumulative and not alternative.
- 5.7 Notices. All notices required or permitted hereunder shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) upon confirmation of receipt when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient; if not, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent to the party to be notified at the address as set forth on the signature pages hereof or Exhibit A hereto or at such other address or electronic mail address as such party may designate by ten (10) days advance written notice to the other parties hereto.

23

- 5.8 Attorneys' Fees. In the event that any suit or action is instituted under or in relation to this Agreement, including without limitation to enforce any provision in this Agreement, the prevailing party in such dispute shall be entitled to recover from the losing party all fees, costs and expenses of enforcing any right of such prevailing party under or with respect to this Agreement, including without limitation, such reasonable fees and expenses of attorneys and accountants, which shall include, without limitation, all fees, costs and expenses of appeals.
- 5.9 Titles and Subtitles. The titles of the sections and subsections of this Agreement are for convenience of reference only and are not to be considered in construing this Agreement.
- 5.10 Additional Investors. Notwithstanding anything to the contrary contained herein, if the Company shall issue additional shares of its Preferred Stock pursuant to the Purchase Agreement, any purchaser of such shares of Preferred Stock shall become a party to this Agreement by executing and delivering an additional counterpart signature page to this Agreement and shall be deemed an "Investor," a "Holder" and a party hereunder.
- 5.11 Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be an original, but all of which together shall constitute one instrument.
- 5.12 Aggregation of Stock. All shares of Registrable Securities held or acquired by affiliated entities or persons or persons or entities under common management or control shall be aggregated together for the purpose of determining the availability of any rights under this Agreement.
- 5.13 Pronouns. All pronouns contained herein, and any variations thereof, shall be deemed to refer to the masculine, feminine or neutral, singular or plural, as to the identity of the parties hereto may require.

[THIS SPACE INTENTIONALLY LEFT BLANK]

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

COMPANY:

RenovoRx, Inc.

Signature: /s/ Shaun R. Bagai Print Name: Shaun R. Bagai

Title: Chief Executive Officer

> RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Print Name:

Boston Scientific Corporation

Signature: /s/ Daniel J. Brennan

Daniel J. Brennan Title:

Executive Vice President and Chief Financial Officer

(if applicable)

AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

		ners		

 Signature:
 /s/ Christian Schuetz

 Print Name:
 Christian Schuetz

 Title:
 Managing Director

 (if applicable)

B-TO-V Partners S.AR.L

Signature: /s/ Florian Schweitzer

Print Name: Florian Schweitzer

Title: Partner (if applicable)

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Acorn Campus Taiwan II, L.P.

Signature: /s/ chu, chih-Hoa

Print Name: chu, chih-Hoa

Title: General Partner

(if applicable)

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

ASTIA Angels RenovoRX, LLC

 Signature:
 /s/ Sharon J. Vosmek

 Print Name:
 Sharon J. Vosmek

 Title:
 CEO

(if applicable)

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Family Ventures, LLC

 Signature:
 /s/ Chandrakant Patel

 Print Name:
 Chandrakant Patel

 Title:
 President

(if applicable)

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Sattva Group LLC

 Signature:
 /s/ Viveka Boddipalli

 Print Name:
 Viveka Boddipalli

 Title:
 Manager

 (if applicable)

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Manish Menda

Print Name:

Signature: /s/ Manish Menda
Print Name: Manish Menda

Suman Menda

Signature: /s/ Suman Menda

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

The Angels' Forum 103, LLC

Suman Menda

 Signature:
 /s/ Carol M. Sands

 Print Name:
 Carol M. Sands

 Title:
 Managing Member

(if applicable)

RENOVORX, INC.
AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof. INVESTOR(S):

William Samuel Silva Revocable Trust

Signature: /s/ William Silva
Print Name: William Silva
Title: Trustee

(if applicable)

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

SV Tech Fund II, L.P.

Signature:	/s/ P.R. Yu		
Print Name:	P.R. Yu		-
Title:	Managing Partner		-
		(if applicable)	-
		RENOVO AMENDED AND RESTATED INV SIGNATU	VESTOR RIGHTS AGREEMENT
		s hereto have executed this Amended and Rest	ated Investor Rights Agreement as of the date set forth in the first paragraph hereof.
INVESTOR(S):		
Amit Patel			
Signature:	/s/ Amit Patel		-
Print Name: Title:	Amit Patel		-
		(if applicable)	-
		RENOVO AMENDED AND RESTATED INV SIGNATU	VESTOR RIGHTS AGREEMENT
INVESTOR(S RMP Partners): Limited		ated Investor Rights Agreement as of the date set forth in the first paragraph hereof.
Signature: Print Name:	/s/ Sandra Padmini Red Sandra Padmini Reddy	'dy	<u>.</u>
Title:	Director	(if applicable)	-
		RENOVO AMENDED AND RESTATED INV SIGNATU	VESTOR RIGHTS AGREEMENT
INVESTOR(S	_	s hereto have executed this Amended and Rest	ated Investor Rights Agreement as of the date set forth in the first paragraph hereof.
Signature:	/s/ Peggy Wallace		
Print Name:	Peggy Wallace		-
Title:	Managing Director	/·f!:11.\	- -
		(if applicable)	
		RENOVO AMENDED AND RESTATED INV SIGNATU	VESTOR RIGHTS AGREEMENT
In Wi INVESTOR(S Jyotsna Gai	_	s hereto have executed this Amended and Rest	rated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

RENOVORX, INC. AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

Signature: Print Name:

Title:

/s/ Jyotsna Gai Jyotsna Gai

(if applicable)

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):
Brush Hill Ventures

Signature: /s/ Michael O'Toole
Print Name: Michael O'Toole
Title: Managing Partner

(if applicable)

RENOVORX, INC.
AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT

SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Kanwar Bagai

Signature: /s/ Kanwar Bagai
Print Name: Kanwar Bagai
Title:

(if applicable)

RENOVORX, INC.
AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Fumiaki Ikeno

Signature: /s/ Fumiaki Ikeno
Print Name: Fumiaki Ikeno
Title:

(if applicable)

RENOVORX, INC.
AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Halo Fund III, LP

Signature: /s/ Carol M. Sands
Print Name: Carol M. Sands
Title: Managing Member

(if applicable)

RENOVORX, INC.
AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

In Witness Whereof, the parties hereto have executed this Amended and Restated Investor Rights Agreement as of the date set forth in the first paragraph hereof.

INVESTOR(S):

Barry & Lynda Family 2003 Revocable Trust

Signature: /s/ Barry R. Keller
Print Name: Barry R Keller

Title: Trustee

(if applicable)

RENOVORX, INC.
AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT SIGNATURE PAGE

RENOVORX, INC. 2021 OMNIBUS EQUITY INCENTIVE PLAN

Section 1. Purpose of Plan.

The name of the Plan is the RenovoRx, Inc. 2021 Omnibus Equity Incentive Plan (the 'Plan'"). The purposes of the Plan are to (i) provide an additional incentive to selected employees, directors, and independent contractors of the Company or its Affiliates whose contributions are essential to the growth and success of the Company, (ii) strengthen the commitment of such individuals to the Company and its Affiliates, (iii) motivate those individuals to faithfully and diligently perform their responsibilities and (iv) attract and retain competent and dedicated individuals whose efforts will result in the long-term growth and profitability of the Company. To accomplish these purposes, the Plan provides that the Company may grant Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Other Stock-Based Awards or any combination of the foregoing.

Section 2. Definitions.

For purposes of the Plan, the following terms shall be defined as set forth below:

- (a) "Administrator" means the Board, or, if and to the extent the Board does not administer the Plan, the Committee in accordance with Section 3 hereof.
- (b) "Affiliate" means a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, the Person specified as of any date of determination.
- (c) "Applicable Laws" means the applicable requirements under U.S. federal and state corporate laws, U.S. federal and state securities laws, including the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any other country or jurisdiction where Awards are granted under the Plan, as are in effect from time to time.
 - (d) "Award" means any Option, Stock Appreciation Right, Restricted Stock, Restricted Stock Unit or Other Stock-Based Award granted under the Plan.
- (e) "Award Agreement" means any written notice, agreement, contract or other instrument or document evidencing an Award, including through electronic medium, which shall contain such terms and conditions with respect to an Award as the Administrator shall determine, consistent with the Plan.
 - (f) "Beneficial Owner" (or any variant thereof) has the meaning defined in Rule 13d-3 under the Exchange Act.
 - (g) "Board" means the Board of Directors of the Company.
 - (h) "Bylaws" mean the bylaws of the Company, as may be amended and/or restated from time to time.
- (i) "Cause" has the meaning assigned to such term in any individual service, employment or severance agreement or Award Agreement with the Participant or, if no such agreement exists or if such agreement does not define "Cause," then "Cause" means a Participant's (i) conviction of a felony or a crime involving fraud or moral turpitude; (ii) theft, material act of dishonesty or fraud, intentional falsification of any employment or Company records, or commission of any criminal act which impairs Participant's ability to perform appropriate employment duties for the Company; (iii) intentional or reckless conduct or gross negligence materially harmful to the Company or the successor to the Company after a Change in Control, including violation of a non-competition or confidentiality agreement; (iv) willful failure to follow lawful instructions of the person or body to which Participant reports; or (v) gross negligence or willful misconduct in the performance of Participant's assigned duties. Cause shall not include mere unsatisfactory performance in the achievement of a Participant's job objectives. Any voluntary termination of employment or service by the Participant in anticipation of an involuntary termination of the Participant's employment or service, as applicable, for Cause shall be deemed to be a termination for Cause.
- (j) "Change in Capitalization" means any (i) merger, consolidation, reclassification, recapitalization, spin-off, spin-out, repurchase or other reorganization or corporate transaction or event, (ii) special or extraordinary dividend or other extraordinary distribution (whether in the form of cash, Common Stock or other property), stock split, reverse stock split, share subdivision or consolidation, (iii) combination or exchange of shares or (iv) other change in corporate structure, which, in any such case, the Administrator determines, in its sole discretion, affects the Shares such that an adjustment pursuant to Section 5 hereof is appropriate.
 - (k) "Change in Control" means the first occurrence of an event set forth in any one of the following paragraphs following the Effective Date:
- (1) any Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities Beneficially Owned by such Person which were acquired directly from the Company or any Affiliate thereof) representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding securities, excluding any Person who becomes such a Beneficial Owner in connection with a transaction described in clause (i) of paragraph (3) below; or
- (2) the date on which individuals who constitute the Board as of the Effective Date and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest, including, but not limited to, a consent solicitation, relating to the election of directors of the Company) whose appointment or election by the Board or nomination for election by the Company's stockholders was approved or recommended by a vote of at least two-thirds (2/3) of the directors then still in office who either were directors on the Effective Date or whose appointment, election or nomination for election was previously so approved or recommended cease for any reason to constitute a majority of the number of directors serving on the Board; or
- (3) there is consummated a merger or consolidation of the Company or any direct or indirect Subsidiary with any other corporation or other entity, other than (i) a merger or consolidation (A) which results in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity or any parent thereof), in combination with the ownership of any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary, fifty percent (50%) or more of the combined voting power of the securities of the Company or such surviving entity or any parent thereof outstanding immediately after such merger or consolidation and (B) following which the individuals who comprise the Board immediately prior thereto constitute at least a majority of the board of directors of the Company, the entity surviving such merger or consolidation or, if the Company or the entity surviving such merger or consolidation is then a Subsidiary, the ultimate parent thereof, or (ii) a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities Beneficially Owned by such Person any securities acquired directly from the Company or its Affiliates) representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding securities; or
- (4) the stockholders of the Company approve a plan of complete liquidation or dissolution of the Company or there is consummated an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets, other than (A) a sale or disposition by the Company of all or substantially all of the Company's assets to an entity, more than fifty percent (50%) of the combined voting power of the voting securities of which are owned by stockholders of the Company following the completion of such transaction in substantially the same proportions as their ownership of the Company immediately prior to such sale or (B) a sale or disposition

of all or substantially all of the Company's assets immediately following which the individuals who comprise the Board immediately prior thereto constitute at least a majority of the board of directors of the entity to which such assets are sold or disposed or, if such entity is a subsidiary, the ultimate parent thereof.

Notwithstanding the foregoing, (i) a Change in Control shall not be deemed to have occurred by virtue of the consummation of any transaction or series of integrated transactions immediately following which the holders of Common Stock immediately prior to such transaction or series of transactions continue to have substantially the same proportionate ownership in an entity which owns all or substantially all of the assets of the Company immediately following such transaction or series of transactions and (ii) to the extent required to avoid accelerated taxation and/or tax penalties under Section 409A of the Code, a Change in Control shall be deemed to have occurred under the Plan with respect to any Award that constitutes deferred compensation under Section 409A of the Code only if a change in the ownership or effective control of the Company or a change in ownership of a substantial portion of the assets of the Company shall also be deemed to have occurred under Section 409A of the Code. For purposes of this definition of Change in Control, the term "Person" shall not include (i) the Company or any Subsidiary thereof, (ii) a trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary thereof, (iii) an underwriter temporarily holding securities pursuant to an offering of such securities, or (iv) a corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of shares of the Company.

- (1) "Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor thereto.
- (m) "Committee" means any committee or subcommittee the Board may appoint to administer the Plan. Subject to the discretion of the Board, the Committee shall be composed entirely of individuals who meet the qualifications of a "non-employee director" within the meaning of Rule 16b-3 under the Exchange Act and any other qualifications required by the applicable stock exchange on which the Common Stock is traded.
 - (n) "Common Stock" means the common stock of the Company, par value \$0.0001.
- (o) "Company" means RenovoRx, Inc., a Delaware corporation (or any successor company, except as the term "Company" is used in the definition of "Change in Control" above).
- (p) "Disability" has the meaning assigned to such term in any individual service, employment or severance agreement or Award Agreement with the Participant or, if no such agreement exists or if such agreement does not define "Disability," then "Disability" means that a Participant, as determined by the Administrator in its sole discretion, (i) is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to last for a continuous period of not less than twelve (12) months, or (ii) is, by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than twelve (12) months, receiving income replacement benefits for a period of not less than three (3) months under an accident and health plan covering employees of the Company or an Affiliate thereof.
 - (q) "Effective Date" has the meaning set forth in Section 17 hereof.
- (r) "Eligible Recipient" means an employee, director or independent contractor of the Company or any Affiliate of the Company who has been selected as an eligible participant by the Administrator; provided, however, to the extent required to avoid accelerated taxation and/or tax penalties under Section 409A of the Code, an Eligible Recipient of an Option or a Stock Appreciation Right means an employee, non-employee director or independent contractor of the Company or any Affiliate of the Company with respect to whom the Company is an "eligible issuer of service recipient stock" within the meaning of Section 409A of the Code.
 - (s) "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.
 - (t) "Exempt Award" shall mean the following:
- (1) An Award granted in assumption of, or in substitution for, outstanding awards previously granted by a corporation or other entity acquired by the Company or any of its Subsidiaries or with which the Company or any of its Subsidiaries combines by merger or otherwise. The terms and conditions of any such Awards may vary from the terms and conditions set forth in the Plan to the extent the Administrator at the time of grant may deem appropriate, subject to Applicable Laws.
- (2) An award that an Eligible Recipient purchases at Fair Market Value (including awards that an Eligible Recipient elects to receive in lieu of fully vested compensation that is otherwise due) whether or not the Shares are delivered immediately or on a deferred basis.
- (u) "Exercise Price" means, (i) with respect to any Option, the per share price at which a holder of such Option may purchase Shares issuable upon exercise of such Award, and (ii) with respect to a Stock Appreciation Right, the base price per share of such Stock Appreciation Right.
- (v) "Fair Market Value" of a share of Common Stock or another security as of a particular date shall mean the fair market value as determined by the Administrator in its sole discretion; provided, that, (i) if the Common Stock or other security is admitted to trading on a national securities exchange, the fair market value on any date shall be the closing sale price reported on such date, or if no shares were traded on such date, on the last preceding date for which there was a sale of a share of Common Stock on such exchange, or (ii) if the Common Stock or other security is then traded in an over-the-counter market, the fair market value on any date shall be the average of the closing bid and asked prices for such share in such over-the-counter market for the last preceding date on which there was a sale of such share in such market.
 - (w) "Free Standing Rights" has the meaning set forth in Section 8.
- (x) "Good Reason" has the meaning assigned to such term in any individual service, employment or severance agreement or Award Agreement with the Participant or, if no such agreement exists or if such agreement does not define "Good Reason," "Good Reason" and any provision of this Plan that refers to "Good Reason" shall not be applicable to such Participant.
- (y) "Grandfathered Arrangement" means an Award which is provided pursuant to a written binding contract in effect on November 2, 2017, and which was not modified in any material respect on or after November 2, 2017, within the meaning of Section 13601(e)(2) of P.L. 115.97, as may be amended from time to time (including any rules and regulations promulgated thereunder).
 - (z) "Incentive Compensation" means annual cash bonus and any Award.
 - (aa) "ISO" means an Option intended to be and designated as an "incentive stock option" within the meaning of Section 422 of the Code.
 - (bb) "Nonqualified Stock Option" shall mean an Option that is not designated as an ISO.
- (cc) "Option" means an option to purchase shares of Common Stock granted pursuant to Section 7 hereof. The term "Option" as used in the Plan includes the terms "Nonqualified Stock Option" and "ISO."

- (dd) "Other Stock-Based Award" means a right or other interest granted pursuant to Section 10 hereof that may be denominated or payable in, valued in whole or in part by reference to, or otherwise based on or related to, Common Stock, including, but not limited to, unrestricted Shares, dividend equivalents or performance units, each of which may be subject to the attainment of performance goals or a period of continued provision of service or employment or other terms or conditions as permitted under the Plan.
- (ee) "Participant" means any Eligible Recipient selected by the Administrator, pursuant to the Administrator's authority provided for in Section 3 below, to receive grants of Awards, and, upon his or her death, his or her successors, heirs, executors and administrators, as the case may be.
 - (ff) "Person" shall have the meaning given in Section 3(a)(9) of the Exchange Act, as modified and used in Sections 13(d) and 14(d) thereof.
 - (gg) "Plan" means this 2021 Omnibus Equity Incentive Plan, as amended and/or restated from time to time.
 - (hh) "Prior Plan" means the Company's Amended and Restated 2013 Equity Incentive Plan, as in effect immediately prior to the Effective Date.
 - (ii) "Related Rights" has the meaning set forth in Section 8.
 - (jj) "Restricted Period" has the meaning set forth in Section 9.
- (kk) "Restricted Stock" means a Share granted pursuant to Section 9 below subject to certain restrictions that lapse at the end of a specified period (or periods) of time and/or upon attainment of specified performance objectives.
- (II) "Restricted Stock Unit" means the right granted pursuant to Section 9 hereof to receive a Share at the end of a specified restricted period (or periods) of time and/or upon attainment of specified performance objectives.
 - (mm) "Rule 16b-3" has the meaning set forth in Section 3.
- (nn) "Section 16 Officer" means any officer of the Company whom the Board has determined is subject to the reporting requirements of Section 16 of the Exchange Act, whether or not such individual is a Section 16 Officer at the time the determination to recoup compensation is made.
- (00) "Shares" means Common Stock reserved for issuance under the Plan, as adjusted pursuant to the Plan, and any successor (pursuant to a merger, consolidation or other reorganization) security.
- (pp) "Stock Appreciation Right" means a right granted pursuant to Section 8 hereof to receive an amount equal to the excess, if any, of (i) the aggregate Fair Market Value, as of the date such Award or portion thereof is surrendered, of the Shares covered by such Award or such portion thereof, over (ii) the aggregate Exercise Price of such Award or such portion thereof.
- (qq) "Subsidiary" means, with respect to any Person, as of any date of determination, any other Person as to which such first Person owns or otherwise controls, directly or indirectly, more than 50% of the voting shares or other similar interests or a sole general partner interest or managing member or similar interest of such other Person.
 - (rr) "Transfer" has the meaning set forth in Section 15.

Section 3. Administration.

- (a) The Plan shall be administered by the Administrator and shall be administered, to the extent applicable, in accordance with Rule 16b-3 under the Exchange Act ("Rule 16b-3").
- (b) Pursuant to the terms of the Plan, the Administrator, subject, in the case of any Committee, to any restrictions on the authority delegated to it by the Board, shall have the power and authority, without limitation:
 - (1) to select those Eligible Recipients who shall be Participants;
- (2) to determine whether and to what extent Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Other Stock-Based Awards or a combination of any of the foregoing, are to be granted hereunder to Participants;
 - (3) to determine the number of Shares to be covered by each Award granted hereunder;
- (4) to determine the terms and conditions, not inconsistent with the terms of the Plan, of each Award granted hereunder (including, but not limited to, (i) the restrictions applicable to Restricted Stock or Restricted Stock Units and the conditions under which restrictions applicable to such Restricted Stock or Restricted Stock Units shall lapse, (ii) the performance goals and periods applicable to Awards, (iii) the Exercise Price of each Option and each Stock Appreciation Right or the purchase price of any other Award, (iv) the vesting schedule and terms applicable to each Award; provided, however, that at least ninety-five percent (95%) of the Awards under the Plan shall not vest, in whole or in part, earlier than one (1) year from the date of grant, (v) the number of Shares or amount of cash or other property subject to each Award and (vi) subject to the requirements of Section 409A of the Code (to the extent applicable) any amendments to the terms and conditions of outstanding Awards, including, but not limited to, extending the exercise period of such Awards and accelerating the payment schedules of such Awards and/or accelerating the vesting schedules of such Awards);
 - (5) to determine the terms and conditions, not inconsistent with the terms of the Plan, which shall govern all written instruments evidencing Awards;
 - (6) to determine the Fair Market Value in accordance with the terms of the Plan;
- (7) to determine the duration and purpose of leaves of absence which may be granted to a Participant without constituting termination of the Participant's service or employment for purposes of Awards granted under the Plan;
- (8) to adopt, alter and repeal such administrative rules, regulations, guidelines and practices governing the Plan as it shall from time to time deem advisable;
- (9) to construe and interpret the terms and provisions of, and supply or correct omissions in, the Plan and any Award issued under the Plan (and any Award Agreement relating thereto), and to otherwise supervise the administration of the Plan and to exercise all powers and authorities either specifically granted under the Plan

or necessary and advisable in the administration of the Plan; and

- (10) to prescribe, amend and rescind rules and regulations relating to sub-plans established for the purpose of satisfying applicable non-United States laws or for qualifying for favorable tax treatment under applicable non-United States laws, which rules and regulations may be set forth in an appendix or appendixes to the Plan.
- (c) Subject to Section 5, neither the Board nor the Committee shall have the authority to reprice or cancel and regrant any Award at a lower exercise, base or purchase price in exchange for cash, property or other Awards without first obtaining the approval of the Company's stockholders.
- (d) All decisions made by the Administrator pursuant to the provisions of the Plan shall be final, conclusive and binding on all Persons, including the Company and the Participants.
 - (e) The expenses of administering the Plan shall be borne by the Company and its Affiliates.
- (f) If at any time or to any extent the Board shall not administer the Plan, then the functions of the Administrator specified in the Plan shall be exercised by the Committee. Except as otherwise provided in the Articles of Incorporation or Bylaws of the Company, any action of the Committee with respect to the administration of the Plan shall be taken by a majority vote at a meeting at which a quorum is duly constituted or unanimous written consent of the Committee's members.

Section 4. Shares Reserved for Issuance Under the Plan.

- (a) Subject to Section 5 hereof, the number of shares of Common Stock that are reserved and available for issuance pursuant to Awards granted under the Plan shall be equal to the sum of (i) 2,175,000 shares, plus (ii) the number of shares of Common Stock reserved, but unissued under the Prior Plan; (iii) the number of shares of Common Stock underlying forfeited awards under the Prior Plan; and (iv) an annual increase on the first day of each calendar year beginning with the first January 1 following the Effective Date and ending with the last January 1 during the initial ten-year term of the Plan, equal to the lesser of (A) three percent (3%) of the Shares outstanding (on an as-converted basis) on the final day of the immediately preceding calendar year and (B) such lesser number of Shares as determined by the Board; provided, that, shares of Common Stock issued under the Plan with respect to an Exempt Award shall not count against such share limit. Following the Effective Date, no further awards shall be issued under the Prior Plan, but all awards under the Prior Plan which are outstanding as of the Effective Date (including any Grandfathered Arrangement) shall continue to be governed by the terms, conditions and procedures set forth in the Prior Plan and any applicable Award Agreement.
- (b) Shares issued under the Plan may, in whole or in part, be authorized but unissued Shares or Shares that shall have been or may be reacquired by the Company in the open market, in private transactions or otherwise. If an Award entitles the Participant to receive or purchase Shares, the number of Shares covered by such Award or to which such Award relates shall be counted on the date of grant of such Award against the aggregate number of Shares available for granting Awards under the Plan. If any Shares subject to an Award are forfeited, cancelled, exchanged or surrendered or if an Award otherwise terminates or expires without a distribution of Shares to the Participant, the Shares with respect to such Award shall, to the extent of any such forfeiture, cancellation, exchange, surrender, termination or expiration, again be available for granting Awards under the Plan. Notwithstanding the foregoing, (i) Shares surrendered or withheld as payment of either the Exercise Price of an Award (including Shares otherwise underlying a Stock Appreciation Right that are retained by the Company to account for the Exercise Price of such Stock Appreciation Right) and/or withholding taxes in respect of an Award and (ii) any Shares reacquired by the Company on the open market or otherwise using cash proceeds from the exercise of Options shall no longer be available for grant under the Plan. In addition, (i) to the extent an Award is denominated in shares of Common Stock, but paid or settled in cash, the number of shares of Common Stock with respect to which such payment or settlement is made shall again be available for grants of Awards pursuant to the Plan and (ii) shares of Common Stock underlying Awards that can only be settled in cash shall not be counted against the aggregate number of shares of Common Stock available for Awards under the Plan. Upon the exercise of any Award granted in tandem with any other Awards, such related Awards shall be cancelled to the extent of the number of Shares as to which the Award is exercis
- (c) No more than 2,175,000 Shares (as increased on an annual basis, on the first day of each calendar year beginning with the first January 1 following the Effective Date and ending with the last January 1 during the initial ten-year term of the Plan, by the lesser of (A) three percent (3%) of the Shares outstanding (on an asconverted basis) on the final day of the immediately preceding calendar year; (B) 343,734 Shares; and (C) such lesser number of Shares as determined by the Board) shall be issued pursuant to the exercise of ISOs.

Section 5. Equitable Adjustments.

In the event of any Change in Capitalization, an equitable substitution or proportionate adjustment shall be made in (i) the aggregate number and kind of securities reserved for issuance under the Plan pursuant to Section 4, (ii) the kind, number of securities subject to, and the Exercise Price subject to outstanding Options and Stock Appreciation Rights granted under the Plan, (iii) the kind, number and purchase price of Shares or other securities or the amount of cash or amount or type of other property subject to outstanding Restricted Stock, Restricted Stock Units or Other Stock-Based Awards granted under the Plan; and/or (iv) the terms and conditions of any outstanding Awards (including, without limitation, any applicable performance targets or criteria with respect thereto); provided, however, that any fractional shares resulting from the adjustment shall be eliminated. Such other equitable substitutions or adjustments shall be made as may be determined by the Administrator, in its sole discretion. Without limiting the generality of the foregoing, in connection with a Change in Capitalization, the Administrator may provide, in its sole discretion, but subject in all events to the requirements of Section 409A of the Code, for the cancellation of any outstanding Award granted hereunder in exchange for payment in cash or other property having an aggregate Fair Market Value equal to the Fair Market Value of the Shares, cash or other property covered by such Award, reduced by the aggregate Exercise Price or purchase price of any outstanding Award is equal to or greater than the Fair Market Value of the shares of Common Stock, cash or other property covered by such Award, the Administrator may cancel such Award without the payment of any consideration to the Participant. Further, without limiting the generality of the foregoing, with respect to Awards subject to foreign laws, adjustments made hereunder shall be made in compliance with applicable requirements. Except to the extent determined by the Administrator

Section 6. Eligibility.

The Participants in the Plan shall be selected from time to time by the Administrator, in its sole discretion, from those individuals that qualify as Eligible Recipients. No Participant who is a director, but is not also an employee or consultant, of the Company shall receive Awards and be paid cash compensation during any calendar year that exceed, in the aggregate, \$300,000 in total value (with cash compensation measured for this purpose at its value upon payment and any Awards measured for this purpose at their grant date fair value, as determined for the Company's financial reporting purposes).

Section 7. Options.

(a) <u>General</u>. Options granted under the Plan shall be designated as Nonqualified Stock Options or ISOs. Each Participant who is granted an Option shall enter into an Award Agreement with the Company, containing such terms and conditions as the Administrator shall determine, in its sole discretion, including, among other things,

the Exercise Price of the Option, the term of the Option and provisions regarding exercisability of the Option, and whether the Option is intended to be an ISO or a Nonqualified Stock Option (and in the event the Award Agreement has no such designation, the Option shall be a Nonqualified Stock Option). The provisions of each Option need not be the same with respect to each Participant. More than one Option may be granted to the same Participant and be outstanding concurrently hereunder. Options granted under the Plan shall be subject to the terms and conditions set forth in this Section 7 and shall contain such additional terms and conditions, not inconsistent with the terms of the Plan, as the Administrator shall deem desirable and set forth in the applicable Award Agreement.

- (b) Exercise Price. The Exercise Price of Shares purchasable under an Option shall be determined by the Administrator in its sole discretion at the time of grant, but in no event shall the exercise price of an Option be less than one hundred percent (100%) of the Fair Market Value of a share of Common Stock on the date of grant.
- (c) Option Term. The maximum term of each Option shall be fixed by the Administrator, but no Option shall be exercisable more than ten (10) years after the date such Option is granted. Each Option's term is subject to earlier expiration pursuant to the applicable provisions in the Plan and the Award Agreement. Notwithstanding the foregoing, subject to Section 4(d) of the Plan, the Administrator shall have the authority to accelerate the vesting and/or exercisability of any outstanding Option at such time and under such circumstances as the Administrator, in its sole discretion, deems appropriate.
- (d) Exercisability. Each Option shall be exercisable at such time or times and subject to such terms and conditions, including the attainment of performance goals, as shall be determined by the Administrator in the applicable Award Agreement. The Administrator may also provide that any Option shall be exercisable only in installments, and the Administrator may waive such installment exercise provisions at any time, in whole or in part, based on such factors as the Administrator may determine in its sole discretion.
- (e) Method of Exercise. Options may be exercised in whole or in part by giving written notice of exercise to the Company specifying the number of whole Shares to be purchased, accompanied by payment in full of the aggregate Exercise Price of the Shares so purchased in cash or its equivalent, as determined by the Administrator. As determined by the Administrator, in its sole discretion, with respect to any Option or category of Options, payment in whole or in part may also be made (i) by means of consideration received under any cashless exercise procedure approved by the Administrator (including the withholding of Shares otherwise issuable upon exercise), (ii) in the form of unrestricted Shares already owned by the Participant which have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which such Option shall be exercised, (iii) any other form of consideration approved by the Administrator and permitted by Applicable Laws or (iv) any combination of the foregoing.
- (f) <u>ISOs</u>. The terms and conditions of ISOs granted hereunder shall be subject to the provisions of Section 422 of the Code and the terms, conditions, limitations and administrative procedures established by the Administrator from time to time in accordance with the Plan. At the discretion of the Administrator, ISOs may be granted only to an employee of the Company, its "parent corporation" (as such term is defined in Section 424(e) of the Code) or a Subsidiary of the Company.
- (1) ISO Grants to 10% Stockholders. Notwithstanding anything to the contrary in the Plan, if an ISO is granted to a Participant who owns shares representing more than ten percent (10%) of the voting power of all classes of shares of the Company, its "parent corporation" (as such term is defined in Section 424(e) of the Code) or a Subsidiary of the Company, the term of the ISO shall not exceed five (5) years from the time of grant of such ISO and the Exercise Price shall be at least one hundred and ten percent (110%) of the Fair Market Value of the Shares on the date of grant.
- (2) \$100,000 Per Year Limitation For ISOs. To the extent the aggregate Fair Market Value (determined on the date of grant) of the Shares for which ISOs are exercisable for the first time by any Participant during any calendar year (under all plans of the Company) exceeds \$100,000, such excess ISOs shall be treated as Nonqualified Stock Options.
- (3) Disqualifying Dispositions. Each Participant awarded an ISO under the Plan shall notify the Company in writing immediately after the date the Participant makes a "disqualifying disposition" of any Share acquired pursuant to the exercise of such ISO. A "disqualifying disposition" is any disposition (including any sale) of such Shares before the later of (i) two years after the date of grant of the ISO and (ii) one year after the date the Participant acquired the Shares by exercising the ISO. The Company may, if determined by the Administrator and in accordance with procedures established by it, retain possession of any Shares acquired pursuant to the exercise of an ISO as agent for the applicable Participant until the end of the period described in the preceding sentence, subject to complying with any instructions from such Participant as to the sale of such Shares.
- (g) Rights as Stockholder. A Participant shall have no rights to dividends, dividend equivalents or distributions or any other rights of a stockholder with respect to the Shares subject to an Option until the Participant has given written notice of the exercise thereof, and has paid in full for such Shares and has satisfied the requirements of Section 15 hereof.
- (h) <u>Termination of Employment or Service</u>. Treatment of an Option upon termination of employment of a Participant shall be provided for by the Administrator in the Award Agreement.
- (i) Other Change in Employment or Service Status. An Option shall be affected, both with regard to vesting schedule and termination, by leaves of absence, including unpaid and un-protected leaves of absence, changes from full-time to part-time employment, partial Disability or other changes in the employment status or service status of a Participant, in the discretion of the Administrator.

Section 8. Stock Appreciation Rights.

(a) General. Stock Appreciation Rights may be granted either alone ('Free Standing Rights'') or in conjunction with all or part of any Option granted under the Plan ("Related Rights"). Related Rights may be granted either at or after the time of the grant of such Option. The Administrator shall determine the Eligible Recipients to whom, and the time or times at which, grants of Stock Appreciation Rights shall be made. Each Participant who is granted a Stock Appreciation Right shall enter into an Award Agreement with the Company, containing such terms and conditions as the Administrator shall determine, in its sole discretion, including, among other things, the number of Shares to be awarded, the Exercise Price per Share, and all other conditions of Stock Appreciation Rights. Notwithstanding the foregoing, no Related Right may be granted for more Shares than are subject to the Option to which it relates. The provisions of Stock Appreciation Rights need not be the same with respect to each Participant. Stock Appreciation Rights granted under the Plan shall be subject to the following terms and conditions set forth in this Section 8 and shall contain such additional terms and conditions, not inconsistent with the terms of the Plan, as the Administrator shall deem desirable, as set forth in the applicable Award Agreement.

- (b) Awards; Rights as Stockholder. A Participant shall have no rights to dividends or any other rights of a stockholder with respect to the shares of Common Stock, if any, subject to a Stock Appreciation Right until the Participant has given written notice of the exercise thereof and has satisfied the requirements of Section 15 hereof.
- (c) Exercise Price. The Exercise Price of Shares purchasable under a Stock Appreciation Right shall be determined by the Administrator in its sole discretion at the time of grant, but in no event shall the exercise price of a Stock Appreciation Right be less than one hundred percent (100%) of the Fair Market Value of a share of Common Stock on the date of grant.

(1) Stock Appreciation Rights that are Free Standing Rights shall be exercisable at such time or times and subject to such terms and conditions as shall be determined by the Administrator in the applicable Award Agreement.

(2) Stock Appreciation Rights that are Related Rights shall be exercisable only at such time or times and to the extent that the Options to which they relate shall be exercisable in accordance with the provisions of Section 7 hereof and this Section 8 of the Plan.

(e) Payment Upon Exercise.

- (1) Upon the exercise of a Free Standing Right, the Participant shall be entitled to receive up to, but not more than, that number of Shares equal in value to the excess of the Fair Market Value as of the date of exercise over the Exercise Price per share specified in the Free Standing Right multiplied by the number of Shares in respect of which the Free Standing Right is being exercised.
- (2) A Related Right may be exercised by a Participant by surrendering the applicable portion of the related Option. Upon such exercise and surrender, the Participant shall be entitled to receive up to, but not more than, that number of Shares equal in value to the excess of the Fair Market Value as of the date of exercise over the Exercise Price specified in the related Option multiplied by the number of Shares in respect of which the Related Right is being exercised. Options which have been so surrendered, in whole or in part, shall no longer be exercisable to the extent the Related Rights have been so exercised.
- (3) Notwithstanding the foregoing, the Administrator may determine to settle the exercise of a Stock Appreciation Right in cash (or in any combination of Shares and cash).
- (f) <u>Termination of Employment or Service</u>. Treatment of a Stock Appreciation Right upon termination of employment of a Participant shall be provided for by the Administrator in the Award Agreement.

(g) Term.

- (1) The term of each Free Standing Right shall be fixed by the Administrator, but no Free Standing Right shall be exercisable more than ten (10) years after the date such right is granted.
- (2) The term of each Related Right shall be the term of the Option to which it relates, but no Related Right shall be exercisable more than ten (10) years after the date such right is granted.
- (h) Other Change in Employment or Service Status. Stock Appreciation Rights shall be affected, both with regard to vesting schedule and termination, by leaves of absence, including unpaid and un-protected leaves of absence, changes from full-time to part-time employment, partial Disability or other changes in the employment or service status of a Participant, in the discretion of the Administrator.

Section 9. Restricted Stock and Restricted Stock Units.

- (a) General. Restricted Stock or Restricted Stock Units may be issued under the Plan. The Administrator shall determine the Eligible Recipients to whom, and the time or times at which, Restricted Stock or Restricted Stock Units shall be made. Each Participant who is granted Restricted Stock or Restricted Stock Units shall enter into an Award Agreement with the Company, containing such terms and conditions as the Administrator shall determine, in its sole discretion, including, among other things, the number of Shares to be awarded; the price, if any, to be paid by the Participant for the acquisition of Restricted Stock or Restricted Stock Units; the period of time restrictions, performance goals or other conditions that apply to Transferability, delivery or vesting of such Awards (the "Restricted Period"); and all other conditions applicable to the Restricted Stock and Restricted Stock Units. If the restrictions, performance goals or conditions established by the Administrator are not attained, a Participant shall forfeit his or her Restricted Stock or Restricted Stock Units, in accordance with the terms of the grant. The provisions of the Restricted Stock or Restricted Stock Units need not be the same with respect to each Participant.
- (b) Awards and Certificates. Except as otherwise provided below in Section 9(c), (i) each Participant who is granted an Award of Restricted Stock may, in the Company's sole discretion, be issued a share certificate in respect of such Restricted Stock; and (ii) any such certificate so issued shall be registered in the name of the Participant, and shall bear an appropriate legend referring to the terms, conditions and restrictions applicable to any such Award. The Company may require that the share certificates, if any, evidencing Restricted Stock granted hereunder be held in the custody of the Company until the restrictions thereon shall have lapsed, and that, as a condition of any Award of Restricted Stock, the Participant shall have delivered a share transfer form, endorsed in blank, relating to the Shares covered by such Award. Certificates for shares of unrestricted Common Stock may, in the Company's sole discretion, be delivered to the Participant only after the Restricted Period, share certificates in respect of the shares of Common Stock underlying such Restricted Stock Units to be settled in Shares, at the expiration of the Restricted Period, share certificates in respect of the shares of Common Stock underlying such Restricted Stock Units may, in the Company's sole discretion, be delivered to the Participant, or his legal representative, in a number equal to the number of shares of Common Stock underlying the Restricted Stock Units Award. Notwithstanding anything in the Plan to the contrary, any Restricted Stock or Restricted Stock Units to be settled in Shares (at the expiration of the Restricted Period, and whether before or after any vesting conditions have been satisfied) may, in the Company's sole discretion, be issued in uncertificated form. Further, notwithstanding anything in the Plan to the contrary, with respect to Restricted Stock Units, at the expiration of the Restricted Period, Shares, or cash, as applicable, shall promptly be issued (either in certificated or uncertificated form) to the P
- (c) <u>Restrictions and Conditions</u>. The Restricted Stock or Restricted Stock Units granted pursuant to this Section 9 shall be subject to the following restrictions and conditions and any additional restrictions or conditions as determined by the Administrator at the time of grant or, subject to Section 409A of the Code where applicable, thereafter:
- (1) The Administrator may, in its sole discretion, provide for the lapse of restrictions in installments and may accelerate or waive such restrictions in whole or in part based on such factors and such circumstances as the Administrator may determine, in its sole discretion, including, but not limited to, the attainment of certain performance goals, the Participant's termination of employment or service with the Company or any Affiliate thereof, or the Participant's death or Disability. Notwithstanding the foregoing, upon a Change in Control, the outstanding Awards shall be subject to Section 11 hereof.
- (2) Except as provided in the applicable Award Agreement, the Participant shall generally have the rights of a stockholder of the Company with respect to Restricted Stock during the Restricted Period; provided, however, that dividends declared during the Restricted Period with respect to an Award, shall only become payable if (and to the extent) the underlying Restricted Stock vests. Except as provided in the applicable Award Agreement, the Participant shall generally not have the rights of a stockholder with respect to Shares subject to Restricted Stock Units during the Restricted Period; provided, however, that, subject to Section 409A of the Code, an amount equal to dividends declared during the Restricted Period with respect to the number of Shares covered by Restricted Stock Units shall, unless otherwise set forth in an Award Agreement, be paid to the Participant at the time (and to the extent) Shares in respect of the related Restricted Stock Units are delivered to the Participant. Certificates for Shares of unrestricted Common Stock may, in the Company's sole discretion, be delivered to the Participant only after the Restricted Period has expired without forfeiture in respect of such Restricted Stock Units, except as the Administrator, in its sole discretion, shall otherwise determine.

(3) The rights of Participants granted Restricted Stock or Restricted Stock Units upon termination of employment or service as a director or independent contractor to the Company or to any Affiliate thereof terminates for any reason during the Restricted Period shall be set forth in the Award Agreement.

(d) Form of Settlement. The Administrator reserves the right in its sole discretion to provide (either at or after the grant thereof) that any Restricted Stock Unit represents the right to receive the amount of cash per unit that is determined by the Administrator in connection with the Award.

Section 10. Other Stock-Based Awards.

Other Stock-Based Awards may be issued under the Plan. Subject to the provisions of the Plan, the Administrator shall have sole and complete authority to determine the individuals to whom and the time or times at which such Other Stock-Based Awards shall be granted. Each Participant who is granted an Other Stock-Based Award shall enter into an Award Agreement with the Company, containing such terms and conditions as the Administrator shall determine, in its sole discretion, including, among other things, the number of shares of Common Stock to be granted pursuant to such Other Stock-Based Awards, or the manner in which such Other Stock-Based Awards shall be settled (e.g., in shares of Common Stock, cash or other property), or the conditions to the vesting and/or payment or settlement of such Other Stock-Based Awards (which may include, but not be limited to, achievement of performance criteria) and all other terms and conditions of such Other Stock-Based Awards. In the event that the Administrator grants a bonus in the form of Shares, the Shares constituting such bonus shall, as determined by the Administrator, be evidenced in uncertificated form or by a book entry record or a certificate issued in the name of the Participant to whom such grant was made and delivered to such Participant as soon as practicable after the date on which such bonus is payable. Notwithstanding anything set forth in the Plan to the contrary, any dividend or dividend equivalent Award issued hereunder shall be subject to the same restrictions, conditions and risks of forfeiture as apply to the underlying Award.

Section 11. Change in Control.

Unless otherwise determined by the Administrator and evidenced in an Award Agreement, in the event that a Change in Control occurs, the Administrator, in its sole and absolute discretion, may:

- (a) provide that any unvested or unexercisable portion of any Award carrying a right to exercise become fully vested and exercisable; and
- (b) cause the restrictions, deferral limitations, payment conditions and forfeiture conditions applicable to an Award granted under the Plan to lapse and such Awards shall be deemed fully vested and any performance conditions imposed with respect to such Awards shall be deemed to be fully achieved at target performance levels.

If the Administrator determines in its discretion pursuant to Section 3(b)(4) hereof to accelerate the vesting of Options and/or Share Appreciation Rights in connection with a Change in Control, the Administrator shall also have discretion in connection with such action to provide that all Options and/or Stock Appreciation Rights outstanding immediately prior to such Change in Control shall expire on the effective date of such Change in Control.

Section 12. Amendment and Termination.

The Board may amend, alter or terminate the Plan at any time, but no amendment, alteration or termination shall be made that would impair the rights of a Participant under any Award theretofore granted without such Participant's consent. The Board shall obtain approval of the Company's stockholders for any amendment that would require such approval in order to satisfy the requirements of any rules of the stock exchange on which the Common Stock is traded or other Applicable Law. Subject to Section 3(c), the Administrator may amend the terms of any Award theretofore granted, prospectively or retroactively, but, subject to Section 5 of the Plan and the immediately preceding sentence, no such amendment shall materially impair the rights of any Participant without his or her consent.

Section 13. Unfunded Status of Plan.

The Plan is intended to constitute an "unfunded" plan for incentive compensation. With respect to any payments not yet made to a Participant by the Company, nothing contained herein shall give any such Participant any rights that are greater than those of a general creditor of the Company.

Section 14. Withholding Taxes.

Each Participant shall, no later than the date as of which the value of an Award first becomes includible in the gross income of such Participant for purposes of applicable taxes, pay to the Company, or make arrangements satisfactory to the Administrator regarding payment of an amount up to the maximum statutory tax rates in the Participant's applicable jurisdiction with respect to the Award, as determined by the Company. The obligations of the Company under the Plan shall be conditional on the making of such payments or arrangements, and the Company shall, to the extent permitted by Applicable Laws, have the right to deduct any such taxes from any payment of any kind otherwise due to such Participant. Whenever cash is to be paid pursuant to an Award, the Company shall have the right to deduct therefrom an amount sufficient to satisfy any applicable withholding tax requirements related thereto. Whenever Shares or property other than cash are to be delivered pursuant to an Award, the Company shall have the right to require the Participant to remit to the Company in cash an amount sufficient to satisfy any related taxes to be withheld and applied to the tax obligations; provided, that, with the approval of the Administrator, a Participant may satisfy the foregoing requirement by either (i) electing to have the Company withhold from delivery of Shares or other property, as applicable, or (ii) delivering already owned unrestricted shares of Common Stock, in each case, having a value not exceeding the applicable taxes to be withheld and applied to the tax obligations. Such already owned and unrestricted shares of Common Stock shall be valued at their Fair Market Value on the date on which the amount of tax to be withheld is determined and any fractional share amounts resulting therefrom shall be settled in cash. Such an election may be made with respect to all or any portion of the Shares to be delivered pursuant to an award. The Company may also use any other method of obtaining the necessary payment or proceeds, a

Section 15. Transfer of Awards.

Until such time as the Awards are fully vested and/or exercisable in accordance with the Plan or an Award Agreement, no purported sale, assignment, mortgage, hypothecation, transfer, charge, pledge, encumbrance, gift, transfer in trust (voting or other) or other disposition of, or creation of a security interest in or lien on, any Award or any agreement or commitment to do any of the foregoing (each, a "Transfer") by any holder thereof in violation of the provisions of the Plan or an Award Agreement will be valid, except with the prior written consent of the Administrator, which consent may be granted or withheld in the sole discretion of the Administrator. Any purported Transfer of an Award or any economic benefit or interest therein in violation of the Plan or an Award Agreement shall be null and void *ab initio* and shall not create any obligation or liability of the Company, and any Person purportedly acquiring any Award or any economic benefit or interest therein transferred in violation of the Plan or an Award Agreement shall not be entitled to be recognized as a holder of such Shares or other property underlying such Award. Unless otherwise determined by the Administrator in accordance with the provisions of the immediately preceding sentence, an Option or a Stock Appreciation Right may be exercised, during the lifetime of the Participant, only by the Participant or, during any period during which the Participant is under a legal Disability, by the Participant's guardian or legal representative.

Section 16. Continued Employment or Service.

Neither the adoption of the Plan nor the grant of an Award shall confer upon any Eligible Recipient any right to continued employment or service with the Company or any Affiliate thereof, as the case may be, nor shall it interfere in any way with the right of the Company or any Affiliate thereof to terminate the employment or service of any of its Eligible Recipients at any time.

Section 17. Effective Date.

The Plan was initially approved by the Board on July 19, 2021 and was adopted and became effective on the date that it was first approved by the Company's stockholders (the "Effective Date").

Section 18. Electronic Signature.

Participant's electronic signature of an Award Agreement shall have the same validity and effect as a signature affixed by hand.

Section 19. Term of Plan.

No Award shall be granted pursuant to the Plan on or after the tenth anniversary of the Effective Date, but Awards theretofore granted may extend beyond that date, and no ISO may be granted after the tenth anniversary of the earlier of the initial Board adoption of the Plan or initial shareholder approval of the Plan.

Section 20. Securities Matters and Regulations.

- (a) Notwithstanding anything herein to the contrary, the obligation of the Company to sell or deliver Shares with respect to any Award granted under the Plan shall be subject to all Applicable Laws, rules and regulations, including all applicable federal and state securities laws, and the obtaining of all such approvals by governmental agencies as may be deemed necessary or appropriate by the Administrator. The Administrator may require, as a condition of the issuance and delivery of certificates evidencing shares of Common Stock pursuant to the terms hereof, that the recipient of such shares make such agreements and representations, and that such certificates bear such legends, as the Administrator, in its sole discretion, deems necessary or advisable.
- (b) Each Award is subject to the requirement that, if at any time the Administrator determines that the listing, registration or qualification of Shares is required by any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory body is necessary or desirable as a condition of, or in connection with, the grant of an Award or the issuance of Shares, no such Award shall be granted or payment made or Shares issued, in whole or in part, unless listing, registration, qualification, consent or approval has been effected or obtained free of any conditions not acceptable to the Administrator.
- (c) In the event that the disposition of Shares acquired pursuant to the Plan is not covered by a then current registration statement under the Securities Act and is not otherwise exempt from such registration, such Shares shall be restricted against transfer to the extent required by the Securities Act or regulations thereunder, and the Administrator may require a Participant receiving Common Stock pursuant to the Plan, as a condition precedent to receipt of such Common Stock, to represent to the Company in writing that the Common Stock acquired by such Participant is acquired for investment only and not with a view to distribution.

Section 21. Section 409A of the Code.

The Plan as well as payments and benefits under the Plan are intended to be exempt from, or to the extent subject thereto, to comply with Section 409A of the Code, and, accordingly, to the maximum extent permitted, the Plan shall be interpreted in accordance therewith. Notwithstanding anything contained herein to the contrary, to the extent required in order to avoid accelerated taxation and/or tax penalties under Section 409A of the Code, the Participant shall not be considered to have terminated employment or service with the Company for purposes of the Plan and no payment shall be due to the Participant under the Plan or any Award until the Participant would be considered to have incurred a "separation from service" from the Company and its Affiliates within the meaning of Section 409A of the Code. Any payments described in the Plan that are due within the "short term deferral period" as defined in Section 409A of the Code shall not be treated as deferred compensation unless Applicable Law requires otherwise. Notwithstanding anything to the contrary in the Plan, to the extent that any Awards (or any other amounts payable under any plan, program or arrangement of the Company or any of its Affiliates) are payable upon a separation from service and such payment would result in the imposition of any individual tax and penalty interest charges imposed under Section 409A of the Code, the settlement and payment of such awards (or other amounts) shall instead be made on the first business day after the date that is six (6) months following such separation from service (or death, if earlier). Each amount to be paid or benefit to be provided under this Plan shall be construed as a separate identified payment for purposes of Section 409A of the Code. The Company makes no representation that any or all of the payments or benefits described in this Plan will be exempt from or comply with Section 409A of the Code and makes no undertaking to preclude Section 409A of the Code from applying to any such payment. The Participa

Section 22. Notification of Election Under Section 83(b) of the Code.

If any Participant shall, in connection with the acquisition of shares of Common Stock under the Plan, make the election permitted under Section 83(b) of the Code, such Participant shall notify the Company of such election within ten (10) days after filing notice of the election with the Internal Revenue Service.

Section 23. No Fractional Shares.

No fractional shares of Common Stock shall be issued or delivered pursuant to the Plan. The Administrator shall determine whether cash, other Awards, or other property shall be issued or paid in lieu of such fractional shares or whether such fractional shares or any rights thereto shall be forfeited or otherwise eliminated.

Section 24. Beneficiary.

A Participant may file with the Administrator a written designation of a beneficiary on such form as may be prescribed by the Administrator and may, from time to time, amend or revoke such designation. If no designated beneficiary survives the Participant, the executor or administrator of the Participant's estate shall be deemed to be the Participant's beneficiary.

Section 25. Paperless Administration.

In the event that the Company establishes, for itself or using the services of a third party, an automated system for the documentation, granting or exercise of Awards, such as a system using an internet website or interactive voice response, then the paperless documentation, granting or exercise of Awards by a Participant may be permitted through the use of such an automated system.

Section 26. Severability.

If any provision of the Plan is held to be invalid or unenforceable, the other provisions of the Plan shall not be affected but shall be applied as if the invalid or unenforceable provision had not been included in the Plan.

Section 27. Clawback.

(a) If the Company is required to prepare a financial restatement due to the material non-compliance of the Company with any financial reporting requirement, then the Committee may require any Section 16 Officer to repay or forfeit to the Company, and each Section 16 Officer agrees to so repay or forfeit, that part of the Incentive Compensation received by that Section 16 Officer during the three-year period preceding the publication of the restated financial statement that the Committee determines was in excess of the amount that such Section 16 Officer would have received had such Incentive Compensation been calculated based on the financial results reported in the restated financial statement. The Committee may take into account any factors it deems reasonable in determining whether to seek recoupment of previously paid Incentive Compensation and how much Incentive Compensation to recoup from each Section 16 Officer (which need not be the same amount or proportion for each Section 16 Officer), including any determination by the Committee that a Section 16 Officer engaged in fraud, willful misconduct or committed grossly negligent acts or omissions which materially contributed to the events that led to the financial restatement. The amount and form of the Incentive Compensation to be recouped shall be determined by the Committee in its sole and absolute discretion, and recoupment of Incentive Compensation may be made, in the Committee's sole and absolute discretion, through the cancellation of vested or unvested Awards, cash repayment or both.

(b) Notwithstanding any other provisions in this Plan, any Award which is subject to recovery under any Applicable Laws, government regulation or stock exchange listing requirement, will be subject to such deductions and clawback as may be required to be made pursuant to such Applicable Law, government regulation or stock exchange listing requirement (or any policy adopted by the Company pursuant to any such law, government regulation or stock exchange listing requirement).

Section 28. Governing Law.

The Plan shall be governed by, and construed in accordance with, the laws of the State of Delaware, without giving effect to principles of conflicts of law of such state.

Section 29. Indemnification.

To the extent allowable pursuant to applicable law, each member of the Board and the Administrator and any officer or other employee to whom authority to administer any component of the Plan is designated shall be indemnified and held harmless by the Company from any loss, cost, liability, or expense that may be imposed upon or reasonably incurred by such member in connection with or resulting from any claim, action, suit, or proceeding to which he or she may be a party or in which he or she may be involved by reason of any action or failure to act pursuant to the Plan and against and from any and all amounts paid by him or her in satisfaction of judgment in such action, suit, or proceeding against him or her; provided, however, that he or she gives the Company an opportunity, at its own expenses, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such individuals may be entitled pursuant to the Company's Articles of Incorporation or Bylaws, as a matter of law, or otherwise, or any power that the Company may have to indemnify them or hold them harmless.

Section 30. Titles and Headings, References to Sections of the Code or Exchange Act.

The titles and headings of the sections in the Plan are for convenience of reference only and, in the event of any conflict, the text of the Plan, rather than such titles or headings, shall control. References to sections of the Code or the Exchange Act shall include any amendment or successor thereto.

Section 31. Successors.

The obligations of the Company under the Plan shall be binding upon any successor corporation or organization resulting from the merger, consolidation or other reorganization of the Company, or upon any successor corporation or organization succeeding to substantially all of the assets and business of the Company.

Section 32. Relationship to other Benefits.

No payment pursuant to the Plan shall be taken into account in determining any benefits under any pension, retirement, savings, profit sharing, group insurance, welfare, or other benefit plan of the Company or any Affiliate except to the extent otherwise expressly provided in writing in such other plan or an agreement thereunder.

RENOVORX, INC. STOCK OPTION GRANT NOTICE AND OPTION AGREEMENT (2021 Omnibus Equity Incentive Plan)

As a key leader in our business, you are in a position to have significant influence on the performance and success of RenovoRx, Inc. (the **Company**"). I am pleased to inform you that, in recognition of the role you play in our collective success, you have been granted an option to purchase shares of the Company's Common Stock. This award is subject to the terms and conditions of the RenovoRx, Inc. 2021 Omnibus Equity Incentive Plan, this Grant Notice, and the following Stock Option Agreement. The details of this award are indicated below.

ISO/Nonqualified Stock Option]

THIS STOCK OPTION AGREEMENT (together with the above grant notice (the "Grant Notice"), the "Agreement") is made and entered into as of the date set forth on the Grant Notice by and between RenovoRx, Inc., a Delaware corporation (the "Company"), and the individual (the "Optionee") set forth on the Grant Notice.

- A. Pursuant to the RenovoRx, Inc. 2021 Omnibus Equity Incentive Plan (the "Plan"), the Administrator has determined that it is to the advantage and best interest of the Company to grant to the Optionee an option to purchase the number of Shares (the "Shares") set forth on the Grant Notice, at the exercise price per Share set forth on the Grant Notice, and in all respects subject to the terms, definitions and provisions of the Plan, which is incorporated herein by reference, and this Agreement (the "Option").
- B. Unless otherwise defined herein, capitalized terms used in this Agreement shall have the meanings set forth in the Plan. For purposes of this Agreement, the following definitions shall apply:
- (i) "**Termination**" shall mean the termination of the employment or service of the Optionee with the Company and all Affiliates thereof (including because of the Optionee's employer ceasing to be an affiliate of the Company). For purposes of this Agreement, Termination will not occur when Optionee goes on a military leave, a sick leave or another bona fide leave of absence that was approved by the Company in writing if the terms of the leave provide for continued service crediting, or when continued service crediting is required by Applicable Laws. Notwithstanding the foregoing, an approved leave of absence for six months or less, which does not in fact exceed six months, will not result in Termination for purposes of this Agreement. However, Termination will occur when an approved leave described in this Section A ends, unless Optionee immediately returns to active work.
 - (ii) "Termination Date" shall mean the date of the Optionee's Termination of Service.

NOW, THEREFORE, in consideration of the mutual agreements contained herein, the Optionee and the Company hereby agree as follows:

1. Acceptance of Agreement. Optionee has reviewed all of the provisions of the Plan and this Agreement. Optionee hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator on questions relating to the Plan and this Agreement, and, solely as they relate to this Option, the applicable provisions (if any) contained in a written employment agreement between the Company or an Affiliate and the Optionee. The Optionee's electronic signature of this Agreement shall have the same validity and effect as a signature affixed by hand.

2. Grant and Terms of Stock Option.

2.1 Grant of Option. Pursuant to this Agreement, the Company has granted to the Optionee the right and option to purchase, subject to the terms and conditions set forth in the Plan and this Agreement, all or any part of the number of Shares set forth on the Grant Notice at a purchase price per Share equal to the exercise price per Share set forth on the Grant Notice. An Option granted pursuant to the Grant Notice and this Agreement shall be [an ISO/a Nonqualified Stock Option].

2

- 2.2 Vesting and Term of Option. This Section 2.2 is subject to the provisions of the Plan and the other provisions of this Agreement.
 - 2.2.1 This Option shall vest and become exercisable as described in the Grant Notice.
- 2.2.2 The "**Term**" of this Option shall begin on the Date of Grant set forth in the Grant Notice and end on the expiration of the Term specified in the Grant Notice. No portion of this Option may be exercised after the expiration of the Term.
 - 2.2.3 In the event of Optionee's Termination for any reason other than death, Disability, or Cause:
 - 2.2.3.1 the portion of this Option that is not vested and exercisable as of the Termination Date shall not continue to vest and shall be immediately cancelled and terminated; and
 - 2.2.3.2 the portion of this Option that is vested and exercisable as of the Termination Date shall terminate and be cancelled on the earlier of:
 - (a) the expiration of the Term and
 - (b) ninety (90) days after such Termination Date.
 - 2.2.4 In the event of Termination due to death or Disability:
 - 2.2.4.1 the portion of this Option that is not vested and exercisable as of the Termination Date shall not continue to vest and shall be immediately cancelled and terminated; and
 - 2.2.4.2 the portion of this Option that is vested and exercisable as of the Termination Date shall terminate and be cancelled on the earlier of (a) the expiration of the Term and (b) the date that is twelve (12) months after the Termination Date.
- 2.2.5 In the event of Optionee's Termination for Cause, or if, after the Termination, the Administrator determines that Cause existed before such Termination, this entire Option shall not continue to vest, shall be cancelled and terminated as of the Termination Date, and shall no longer be exercisable as to any Shares, whether or not previously vested.

-3-

3. Method of Exercise.

3.1 Method of Exercise. Each election to exercise the Option shall be subject to the terms and conditions of the Plan and shall be in writing, signed by the Optionee or by his or her executor, administrator, or permitted transferee (subject to any restrictions provided under the Plan), made pursuant to and in accordance with the terms and conditions set forth in the Plan and received by the Company at its principal offices, accompanied by payment in full as provided in the Plan or in this Agreement. Notwithstanding any of the foregoing, the Administrator shall have the right to specify all conditions of the manner of exercise. Upon the Company's determination that the Option has been validly exercised as to any of the Shares, the Company may issue certificates in the Optionee's name for such Shares. However, the Company shall not be liable to the Optionee for damages relating to any reasonable delays in issuing the certificates to the Optionee, any loss of the certificates, or any mistakes or errors in the issuance of the certificates or in the certificates themselves which it promptly undertakes to correct.

3.2 <u>Restrictions on Exercise</u>. No Shares will be issued pursuant to the exercise of this Option unless and until there shall have been full compliance with all applicable requirements of the Securities Act of 1933 ("Securities Act"), as amended (whether by registration or satisfaction of exemption conditions), all applicable listing requirements of any national securities exchange or other market system on which the Common Stock is then listed and all applicable requirements of any Applicable Laws and of any regulatory bodies having jurisdiction over such issuance. As a condition to the exercise of this Option, the Company may require the Optionee to make any representation and warranty to the Company as may be necessary or appropriate, in the judgment of the Administrator, to comply with any Applicable Law. In addition,

Optionee shall not sell any Shares acquired upon exercise of this Option at a time when Applicable Laws, regulations or Company's or underwriter trading policies prohibit such sale. Any other provision of this Agreement notwithstanding, the Company shall have the right to designate one or more periods of time, each of which shall not exceed 180 days in length, during which this Option shall not be exercisable if the Administrator determines (in its sole discretion) that such limitation on exercise could in any way facilitate a lessening of any restriction on transfer pursuant to the Securities Act or any state securities laws with respect to any issuance of securities by the Company, facilitate the registration or qualification of any securities by the Company under the Securities Act or any state securities laws, or facilitate the perfection of any exemption from the registration or qualification requirements of the Securities Act or any applicable state securities laws for the issuance or transfer of any securities. Such limitation on exercise shall not alter the vesting schedule set forth in this Agreement other than to limit the periods during which this Option shall be exercisable.

3.3 <u>Method of Payment</u>. Payment of the exercise price shall be made in full at the time of exercise (a) by the delivery of cash or check acceptable to the Administrator, including an amount to cover the withholding taxes (as provided in Section 7.11) with respect to such exercise, or (b) any other method, if any, approved by the Administrator, including (i) by means of consideration received under any cashless exercise procedure, if any, approved by the Administrator (including the withholding of Shares otherwise issuable upon exercise) or (ii) any other form of consideration approved by the Administrator and permitted by Applicable Laws.

-4-

- 3.4 No Rights as a Shareholder. Until the Shares are issued to the Optionee (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder will exist with respect to the Shares, notwithstanding the exercise of the Option.
- 4. Non-Transferability of Option. Except as provided below, this Option may not be sold, assigned or transferred in any manner, pledged or otherwise encumbered other than by will or by the laws of descent or distribution or to a beneficiary designated pursuant to the Plan, and may be exercised during the lifetime of Optionee only by Optionee or the Optionee's guardian or legal representative. Subject to all of the other terms and conditions of this Agreement, following the death of Optionee, this Option may, to the extent it is vested and exercisable by Optionee in accordance with its terms on the Termination Date, be exercised by Optionee's executor or administrator, or the person or persons to whom the Optionee's rights under this Agreement shall pass by will or by the laws of descent and distribution as the case may be. Any heir or legatee of the Optionee shall take rights herein granted subject to the terms and conditions hereof.
- 5. <u>Restrictions</u>; <u>Restrictive Legends</u>. Ownership and transfer of Shares issued pursuant to the exercise of this Option will be subject to the provisions of, including ownership and transfer restrictions contained in, the Company's Certificate of Incorporation or Bylaws, as amended from time to time, restrictions imposed by Applicable Laws and restrictions set forth or referenced in legends imprinted on certificates representing such Shares.
- 6. <u>Dissolution or Liquidation</u>. In the event of the proposed dissolution or liquidation of the Company, to the extent that this Option had not been previously exercised, it will terminate immediately prior to the consummation of such proposed dissolution or liquidation. In such instance, the Administrator may, in the exercise of its sole discretion, declare that this Option will terminate as of a date fixed by the Administrator and give the Optionee the right to exercise this Option prior to such date as to all or any part of the optioned stock, including Shares as to which this Option would not otherwise be exercisable.

7. General.

- 7.1 Governing Law. This Agreement shall be governed by and construed under the laws of the State of Delaware applicable to agreements made and to be performed entirely in Delaware, without regard to the conflicts of law provisions of Delaware or any other jurisdiction.
- 7.2 Community Property. Without prejudice to the actual rights of the spouses as between each other, for all purposes of this Agreement, the Optionee shall be treated as agent and attorney-in-fact for that interest held or claimed by his or her spouse with respect to this Option and the parties hereto shall act in all matters as if the Optionee was the sole owner of this Option. This appointment is coupled with an interest and is irrevocable.
- 7.3 No Employment Rights. Nothing herein contained shall be construed as an agreement by the Company or any of its Subsidiaries, express or implied, to employ the Optionee or contract for the Optionee's services, to restrict the Company's or such Subsidiary's right to discharge the Optionee or cease contracting for the Optionee's services or to modify, extend or otherwise affect in any manner whatsoever the terms of any employment agreement or contract for services which may exist between the Optionee and the Company or any Affiliate.

-5-

- 7.4 Application to Other Stock. In the event any capital stock of the Company or any other corporation shall be distributed on, with respect to, or in exchange for Shares as a stock dividend, stock split, reclassification or recapitalization in connection with any merger or reorganization or otherwise, all restrictions, rights and obligations set forth in this Agreement shall apply with respect to such other capital stock to the same extent as they are, or would have been applicable, to the Shares on or with respect to which such other capital stock was distributed, and references to "Company" in respect of such distributed stock shall be deemed to refer to the company to which such distributed stock relates.
- 7.5 No Third-Party Benefits. Except as otherwise expressly provided in this Agreement, none of the provisions of this Agreement shall be for the benefit of, or enforceable by, any third-party beneficiary.
- 7.6 <u>Successors and Assigns</u>. Except as provided herein to the contrary, this Agreement shall be binding upon and inure to the benefit of the parties, their respective successors and permitted assigns.
- 7.7 No Assignment. Except as otherwise provided in this Agreement, the Optionee may not assign any of his or her rights under this Agreement without the prior written consent of the Company, which consent may be withheld in its sole discretion. The Company shall be permitted to assign its rights or obligations under this Agreement so long as such assignee agrees to perform all of the Company's obligations hereunder.
- 7.8 <u>Severability</u>. The validity, legality or enforceability of the remainder of this Agreement shall not be affected even if one or more of the provisions of this Agreement shall be held to be invalid, illegal or unenforceable in any respect.
- 7.9 Equitable Relief. The Optionee acknowledges that, in the event of a threatened or actual breach of any of the provisions of this Agreement, damages alone will be an inadequate remedy, and such breach will cause the Company great, immediate and irreparable injury and damage. Accordingly, the Optionee agrees that the Company shall be entitled to injunctive and other equitable relief, and that such relief shall be in addition to, and not in lieu of, any remedies it may have at law or under this Agreement.
- 7.10 <u>Jurisdiction</u>. Any suit, action or proceeding with respect to this Agreement, or any judgment entered by any court in respect of any thereof, shall be brought in any court of competent jurisdiction in the State of Delaware, and the Company and the Optionee hereby submit to the exclusive jurisdiction of such courts for the purpose of any such suit, action, proceeding or judgment. The Optionee and the Company hereby irrevocably waive (i) any objections which it may now or hereafter have to the laying of the venue of any suit, action or proceeding arising out of or relating to this Agreement brought in any court of competent jurisdiction in the State of Delaware, (ii) any claim that any such suit, action or proceeding brought in any such court has been brought in any inconvenient forum and (iii) any right to a jury trial.

- 7.11 Taxes. By agreeing to this Agreement, the Optionee represents that he or she has reviewed with his or her own tax advisors the federal, state, local and foreign tax consequences of the transactions contemplated by this Agreement and that he or she is relying solely on such advisors and not on any statements or representations of the Company or any of its agents. The Company shall be entitled to require a cash payment by or on behalf of the Optionee and/or to deduct from the Shares or cash otherwise issuable hereunder or other compensation payable to the Optionee the minimum amount of any sums required by federal, state or local tax law to be withheld (or other such sums that will not cause adverse accounting consequences for the Company and is permitted under applicable withholding rules promulgated by the Internal Revenue Service or another applicable governmental entity) in respect of the Option, its exercise or any payment or transfer under or with respect to the Option.
- 7.12 <u>Headings</u>. The section headings in this Agreement are inserted only as a matter of convenience, and in no way define, limit, extend or interpret the scope of this Agreement or of any particular section.
- 7.13 <u>Number and Gender</u>. Throughout this Agreement, as the context may require, (a) the masculine gender includes the feminine and the neuter gender includes the masculine and the feminine; (b) the singular tense and number includes the plural, and the plural tense and number includes the singular; (c) the past tense includes the present, and the present tense includes the past; (d) references to parties, sections, paragraphs and exhibits mean the parties, sections, paragraphs and exhibits of and to this Agreement; and (e) periods of days, weeks or months mean calendar days, weeks or months.
- 7.14 <u>Data Privacy</u>. Optionee agrees that all of Optionee's information that is described or referenced in this Agreement and the Plan may be used by the Company, its affiliates and the designated broker and its affiliates to administer and manage Optionee's participation in the Plan.
- 7.15 <u>Acknowledgments of Optionee</u>. Optionee has reviewed the Plan and this Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Agreement, fully understands all provisions of the Plan and this Agreement and, by accepting the Notice of Grant, acknowledges and agrees to all of the provisions of the Grant Notice, the Plan and this Agreement.
- 7.16 Complete Agreement. The Grant Notice, this Stock Option Agreement, the Plan, and the applicable provisions (if any) contained in a written employment agreement between the Company or an Affiliate and the Optionee constitute the parties' entire agreement with respect to the subject matter hereof and supersede all agreements, representations, warranties, statements, promises and understandings, whether oral or written, with respect to the subject matter hereof. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan shall control.

-7-

- 7.17 <u>Waiver</u>. The Optionee acknowledges that a waiver by the Company of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any other provision of this Agreement, or of any subsequent breach by the Optionee.
- 7.18 Signature in Counterparts. This Agreement may be signed in counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.
- 7.19 Amendments and Termination. To the extent permitted by the Plan, this Agreement may be wholly or partially amended, altered or terminated at any time or from time to time by the Administrator or the Board, but no amendment, alteration or termination shall be made that would materially impair the rights of an Optionee under the Option without such Optionee's consent. If it is determined that the terms of this Agreement have been structured in a manner that would result in adverse tax treatment under Section 409A of the Code, the parties agree to cooperate in taking all reasonable measures to restructure the arrangement to minimize or avoid such adverse tax treatment without materially impairing Optionee's economic rights.
- 7.20 Waiver of Jury Trial. TO THE EXTENT EITHER PARTY INITIATES LITIGATION INVOLVING THIS AGREEMENT OR ANY ASPECT OF THE RELATIONSHIP BETWEEN US (EVEN IF OTHER PARTIES OR OTHER CLAIMS ARE INCLUDED IN SUCH LITIGATION), ALL OF THE PARTIES WAIVE THEIR RIGHT TO A TRIAL BY JURY. THIS WAIVER WILL APPLY TO ALL CAUSES OF ACTION THAT ARE OR MIGHT BE INCLUDED IN SUCH ACTION, INCLUDING CLAIMS RELATED TO THE ENFORCEMENT OR INTERPRETATION OF THIS AGREEMENT, ALLEGATIONS OF STATE OR FEDERAL STATUTORY VIOLATIONS, FRAUD, MISREPRESENTATION, OR SIMILAR CAUSES OF ACTION, AND IN CONNECTION WITH ANY LEGAL ACTION INITIATED FOR THE RECOVERY OF DAMAGES BETWEEN OR AMONG US OR BETWEEN OR AMONG ANY OF OUR OWNERS, AFFILIATES, OFFICERS, EMPLOYEES OR AGENTS.
- 7.21 Electronic Delivery and Disclosure. The Company may, in its sole discretion, decide to deliver or disclose, as applicable, any documents related to this Award granted under the Plan, future awards that may be granted under the Plan, the prospectus related to the Plan, the Company's annual reports or proxy statements by electronic means or to request Optionee's consent to participate in the Plan by electronic means, including, but not limited to, the Securities and Exchange Commission's Electronic Data Gathering, Analysis, and Retrieval system or any successor system ("EDGAR"). Optionee hereby consents to receive such documents delivered electronically or to retrieve such documents furnished electronically (including on EDGAR), as applicable, and agrees to participate in the Plan through any online or electronic system established and maintained by the Company or another third party designated by the Company.
- 7.22 Section 409A. The parties intend for the Option to be exempt from Section 409A of the Code or, if not so exempt, to be treated in a manner which complies with the requirements of such section, and intend that this Agreement be construed and administered in accordance with such intention. In the event that the parties determine that the terms of this Agreement or the Option needs to be modified in order to comply with Section 409A of the Code, the parties shall cooperate reasonably to do so in a manner intended to best preserve the economic benefits of this Agreement. Any payments that qualify for the "short-term deferral" exception or another exception under Section 409A of the Code shall be paid under the applicable exception. For purposes of the limitations on nonqualified deferred compensation under Section 409A of the Code, payment of compensation under this Agreement shall be treated as a separate payment of compensation. Notwithstanding anything contained herein to the contrary, to the extent required in order to avoid accelerated taxation and/or tax penalties under Section 409A of the Code, amounts that would otherwise be payable and benefits that would otherwise be provided pursuant to this Agreement during the six-month period immediately following the Participant's separation from service shall instead be paid on the first business day after the date that is six months following the Participant's termination date (or death, if earlier).

RENOVORX, INC.

OUTSIDE DIRECTOR COMPENSATION POLICY

Adopted and Approved by the Board of Directors on September 30, 2021

RenovoRx, Inc. (the "Company") believes that providing cash and equity compensation to its members of the Board of Directors (the "Board," and members of the Board, the "Directors") represents an effective tool to attract, retain and reward Directors who are not employees of the Company (the "Outside Directors"). This Outside Director Compensation Policy (the "Policy") is intended to formalize the Company's policy regarding the compensation to its Outside Directors. Unless otherwise defined herein, capitalized terms used in this Policy will have the meaning given to such terms in the Company's 2021 Omnibus Equity Incentive Plan (the "Plan"), or if the Plan is no longer in place, the meaning given to such terms or any similar terms in the equity plan then in place. Each Outside Director will be solely responsible for any tax obligations incurred by such Outside Director as a result of the equity and cash payments such Outside Director receives under this Policy.

This Policy is adopted and approved as of the date stated above and applies to Outside Director compensation as of the date of the Company's initial public offering on August 26, 2021 (such date, the "Effective Date").

1. Cash Compensation

Annual Cash Retainer

Each Outside Director will be paid an annual cash retainer of \$36,000. There are no per-meeting attendance fees for attending Board meetings. This cash compensation will be paid quarterly in arrears on a prorated basis from the Effective Date.

Committee Annual Cash Retainer

Effective as of the Effective Date, each Outside Director who serves as the chair or a member of a committee of the Board listed below will be eligible to earn additional annual cash fees (paid quarterly in arrears on a prorated basis) as follows:

Chair of Audit Committee:	\$ 15,000
Chair of Compensation Committee:	\$ 10,000
Chair of Corporate Governance and Nominating Committee:	\$ 10,000
Non-Chair Members of Various Committees:	\$ 5,000

For clarity, each Outside Director who serves as the chair of a committee shall receive only the additional annual cash fee as the chair of the committee, and not the additional annual cash fee as a member of the committee.

2. Equity Compensation

Outside Directors will be eligible to receive all types of Awards (except Incentive Stock Options) under the Plan (or the applicable equity plan in place at the time of grant), including discretionary Awards not covered under this Policy. All grants of Awards to Outside Directors pursuant to Section 2 of this Policy will be automatic and nondiscretionary, except as otherwise provided herein, and will be made in accordance with the following provisions:

- (a) No Discretion. No person will have any discretion to select which Outside Directors will be granted any Awards under this Policy or to determine the number of Shares to be covered by such Awards.
- (b) Initial Award. Each individual who first becomes an Outside Director following the Effective Date will be granted Options at a fair market value of \$120,000 in the aggregate (an "Initial Award"), on the date on which such individual first becomes an Outside Director, whether through election by the stockholders of the Company or appointment by the Board to fill a vacancy. Subject to Section 11 of the Plan and Section 3 of this Policy, each Initial Award will vest and become exercisable over three years, with 1/36th of the Initial Award vesting each month on the same day of the month as the commencement of the applicable Outside Director's service as an Outside Director, subject to the Outside Director continuing to be a Participant (as defined in the Plan) through such date.
- (c) Annual Award. On October 1 of each year, commencing October 1, 2022, each Outside Director will be automatically granted Options at a fair market value of \$65,000 in the aggregate (an "Annual Award"). Subject to Section 11 of the Plan and Section 3 of this Policy, 1/12th of each Annual Award will vest monthly after October 1 on the first day of each subsequent month, subject to the applicable Outside Director continuing to be a Participant through such date.
 - (c) Terms. The terms and conditions of each Initial Award or Annual Award will be as follows:
 - (i) Exercise Price. The per Share exercise price for an Option granted under this Policy will be 100% of the Fair Market Value on the grant date.
 - (ii) Term. The maximum term to expiration of an Option granted under this Policy will be 10 years, subject to earlier termination as provided in the Plan.

3. Change in Control

In the event of a Change in Control, each Outside Director will fully vest in his or her outstanding Company equity Awards, including any Initial Award or Annual Award, provided that the Outside Director continues to be an Outside Director through such date.

4. Annual Compensation Limit

In any fiscal year, other than the fiscal year in which he or she joins the Board, no Outside Director may be paid, issued or granted compensation (including in the form of cash or equity compensation, which will be valued based on its grant-date fair value) with an aggregate value greater than \$250,000 (increased to \$300,000 for an Outside Director's first fiscal year of service). Any cash compensation paid or equity compensation award (including any Awards) granted to an individual for his or her services as an employee, or for his or her services as a consultant (other than as an Outside Director), will not count for purposes of the limitation under this Section 4.

5. <u>Travel Expenses</u>

Each Outside Director's reasonable, customary and documented travel expenses to Board or Board committee meetings will be reimbursed by the Company.

6. Additional Provisions

All provisions of the Plan not inconsistent with this Policy will apply to Awards granted to Outside Directors.

7. Adjustments

In the event that any extraordinary dividend or other extraordinary distribution (whether in the form of cash, Shares, other securities or other property), recapitalization, stock split, reverse stock split, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase, or exchange of Shares or other securities of the Company, or other change in the corporate structure of the Company affecting the Shares occurs, the Administrator, in order to prevent diminution or enlargement of the benefits or potential benefits intended to be made available under this Policy, will adjust the number of Shares issuable pursuant to Awards granted under this Policy.

-2-

8. Section 409A

In no event will cash compensation or expense reimbursement payments under this Policy be paid after the later of (i) 15th day of the third month following the end of the Company's fiscal year in which the compensation is earned or expenses are incurred, as applicable, or (ii) 15th day of the third month following the end of the calendar year in which the compensation is earned or expenses are incurred, as applicable, in compliance with the "short-term deferral" exception under Section 409A of the Internal Revenue Code of 1986, as amended, and the final regulations and guidance thereunder, as may be amended from time to time (together, "Section 409A"). It is the intent of this Policy that this Policy and all payments hereunder be exempt from or otherwise comply with the requirements of Section 409A so that none of the compensation to be provided hereunder will be subject to the additional tax imposed under Section 409A, and any ambiguities or ambiguous terms herein will be interpreted to be so exempt or comply. In no event will the Company reimburse an Outside Director for any taxes imposed or other costs incurred as a result of Section 409A.

9. Revisions

The Board may amend, alter, suspend or terminate this Policy at any time and for any reason. No amendment, alteration, suspension or termination of this Policy will materially impair the rights of an Outside Director with respect to compensation that already has been paid or awarded, unless otherwise mutually agreed between the Outside Director and the Company. Termination of this Policy will not affect the Board's or the Compensation Committee's ability to exercise the powers granted to it under the Plan with respect to Awards granted under the Plan pursuant to this Policy prior to the date of such termination.

November 11, 2021

Shaun Bagai Via Email

Re: Confirmatory Employment Letter

Dear Shaun:

This confirmatory employment letter agreement (the "Agreement") is entered into between Shaun Bagai ("you") and RenovoRx, Inc. (the "Company" or "we"), effective as of November 11, 2021 (the "Effective Date"), to confirm the terms and conditions of your employment with the Company as of the Effective Date.

- 1. <u>Title; Position; Location</u>. You will continue to serve as the Company's Chief Executive Officer. You also will continue to report to the Company's Board of Directors (the "Board") and will perform the duties and responsibilities customary for such position and such other related duties as are lawfully assigned by the Company's Board. You will perform your duties from the Company's corporate offices located in Los Altos, California (with the exception of any period during which a shelter-in-place order or similar work-from-home arrangement affecting your employment with the Company remains in effect), subject to customary travel as reasonably required by the Company and necessary to perform your job duties.
- 2. <u>Best Efforts.</u> You will continue to devote your full business time and best efforts to the faithful and loyal performance of your duties to the Company. While you render services to the Company, you will not engage in any other employment, consulting or other business activity (whether full-time or part-time) that would create a conflict of interest with the Company, and any engagement or participation in any outside business activity or business affairs by you will be subject to full disclosure and prior written approval from the Company's Board or its Compensation Committee, as applicable. By signing this Agreement, you confirm that you have no contractual commitments or other legal obligations that would prohibit you from continuing to perform your duties for the Company.
- 3. <u>Base Salary.</u> As of the Effective Date, your annual base salary will be \$363,000 (<u>'Salary</u>"), which will be payable, less any applicable withholdings, in accordance with the Company's normal payroll practices. Your annual base salary will be subject to review and adjustment from time to time by our Board or its Compensation Committee, as applicable, in its sole discretion.
- 4. <u>Annual Bonus</u>. Commencing in the Company's 2022 fiscal year, you will have the opportunity to earn a target annual cash bonus equal to fifty percent (50%) of your annual base salary earned during the fiscal year, based on achieving performance objectives established by the Board or its Compensation Committee, as applicable, in its sole discretion and payable upon achievement of those objectives, and subject to such terms and conditions, as determined by the Board or its Compensation Committee. Unless determined otherwise by the Board or its Compensation Committee, as applicable, any such bonus will be subject to your continued employment through and until the date of payment, and any such bonus amounts paid will be subject to any applicable withholdings. Your annual bonus opportunity and the applicable terms and conditions may be adjusted from time to time by the Board or its Compensation Committee, as applicable, in its sole discretion.
- 5. Equity Awards. You will be eligible to receive awards of stock options or other equity awards pursuant to any plans or arrangements the Company may have in effect from time to time. The Board or its Compensation Committee, as applicable, will determine in its sole discretion whether you will be granted any such equity awards and the terms of any such award in accordance with the terms of any applicable plan or arrangement that may be in effect from time to time.
- 6. Employee Benefits. You will continue to be eligible to participate in the benefit plans and programs established by the Company for its employees from time to time, subject to their applicable terms and conditions, including without limitation any eligibility requirements. The Company will reimburse you for reasonable travel or other expenses incurred by you in the furtherance of or in connection with the performance of your duties under this Agreement, pursuant to the terms of the Company's expense reimbursement policy as may be in effect from time to time. The Company reserves the right to modify, amend, suspend or terminate the benefit plans, programs, and arrangements it offers to its employees at any time.
- 7. <u>Severance</u>. You will be eligible to enter into a Change in Control and Severance Agreement (the "Severance Agreement") applicable to you based on your position within the Company. The Severance Agreement will specify the severance payments and benefits you may become entitled to receive in connection with certain qualifying terminations of your employment with the Company.
- 8. <u>Confidentiality Agreement/Arbitration</u>. As an employee of the Company, you will continue to have access to certain confidential information of the Company and you may, during the course of your employment, develop certain information or inventions that will be the property of the Company. To protect the interests of the Company, your acceptance of this Agreement confirms that the terms of the Company's Employee Confidential Information and Invention Assignment Agreement you previously signed with the Company (the "Confidentiality Agreement") still apply. Further, in the event of any dispute or claim relating to or arising out of our employment relationship, you and the Company agree that any and all disputes between you and the Company shall be fully and finally resolved by binding arbitration, subject to the terms and conditions of the arbitration agreement set forth in Appendix B.
- 9. <u>At-Will Employment</u>. This Agreement does not imply any right to your continued employment for any period with the Company or any parent, subsidiary, or affiliate of the Company. Your employment with the Company is and will continue to be at-will. Accordingly, you are free to resign at any time, for any reason or for no reason. Similarly, the Company is free to conclude its employment relationship with you at any time, with or without cause, and with or without notice. We request that, in the event of resignation, you give the Company at least two weeks' notice.
- 10. Protected Activity Not Prohibited. The Company and you acknowledge and agree that nothing in this Agreement (or any other Company agreement or policy)limits or prohibits you from filing and/or pursuing a charge or complaint with, or otherwise communicating or cooperating with or participating in any investigation or proceeding that may be conducted by, any federal, state or local government agency or commission, including the Securities and Exchange Commission, the Equal Employment Opportunity Commission, the Occupational Safety and Health Administration, and the National Labor Relations Board ("Government Agencies"), including disclosing documents or other information as permitted by law, without giving notice to, or receiving authorization from, the Company. In addition, nothing in this Agreement (or any other Company agreement or policy) is intended to limit your rights to discuss the terms, wages, and working conditions of your employment, nor prevent you from discussing or disclosing information about unlawful acts in the workplace, such as harassment or discrimination or any other conduct that you have reason to believe is unlawful. You further understand that you are not permitted to disclose the Company's attorney-client privileged communications or attorney work product. In addition, you acknowledge that the Company has provided you with notice in compliance with the Defend Trade Secrets Act of 2016 regarding immunity from liability for limited disclosures of trade secrets. The full text of the notice is attached in Appendix A.
- 11. Miscellaneous. This Agreement, together with the Confidentiality Agreement, the Severance Agreement and the stock options granted to you by the Company under its Amended and Restated 2013 Equity Incentive Plan and 2021 Omnibus Equity Incentive Plan and the applicable award agreements thereunder, constitute the entire agreement between you and the Company regarding the terms and conditions of your employment, and they supersede and replace all prior negotiations, representations or agreements between you and the Company. This Agreement will be governed by the laws of the State of California but without regard to the conflict of law provision, except that any

dispute regarding the enforceability of the arbitration section set forth in Appendix B to this Agreement shall be governed by the Federal Arbitration Act. This Agreement may be modified only by a written agreement signed by a duly authorized officer of the Company (other than yourself) and you.

[Signature page follows]

To confirm the current terms and conditions of your employment, please sign and date in the spaces indicated and return this Agreement to me.

Sincerely,

RenovoRx, Inc.

By: /s/ Ramtin Agah

Dr. Ramtin Agah

Chairman of the Board of Directors

Agreed to and accepted:

/s/ Shaun R. Bagai

Shaun R. Bagai

Dated: November 11, 2021

Appendix A

Section 7 of the Defend Trade Secrets Act of 2016

".... An individual shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of a trade secret that—(A) is made—(i) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney; and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. ... An individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the attorney of the individual and use the trade secret information in the court proceeding, if the individual—(A) files any document containing the trade secret under seal; and (B) does not disclose the trade secret, except pursuant to court order."

4

Appendix B

Arbitration and Equitable Relief

A . Arbitration. IN CONSIDERATION OF MY CONTINUED EMPLOYMENT WITH THE COMPANY, ITS PROMISE TO ARBITRATE ALL EMPLOYMENT-RELATED DISPUTES WITH ME, AND MY RECEIPT OF COMPENSATION AND OTHER COMPANY BENEFITS, AT PRESENT AND IN THE FUTURE, I AGREE THAT ANY AND ALL CONTROVERSIES, CLAIMS, OR DISPUTES THAT I MAY HAVE WITH THE COMPANY (INCLUDING ANY COMPANY EMPLOYEE, OFFICER, DIRECTOR, TRUSTEE, OR BENEFIT PLAN OF THE COMPANY, IN THEIR CAPACITY AS SUCH OR OTHERWISE), ARISING OUT OF, RELATING TO, OR RESULTING FROM MY EMPLOYMENT OR RELATIONSHIP WITH THE COMPANY OR THE TERMINATION OF MY EMPLOYMENT OR RELATIONSHIP WITH THE COMPANY, INCLUDING ANY BREACH OF THIS AGREEMENT, SHALL BE SUBJECT TO BINDING ARBITRATION PURSUANT TO THE FEDERAL ARBITRATION ACT (9 U.S.C. SEC. 1 ET SEQ.) (THE "FAA"). THE FAA'S SUBSTANTIVE AND PROCEDURAL PROVISIONS SHALL EXCLUSIVELY GOVERN AND APPLY WITH FULL FORCE AND EFFECT TO THIS ARBITRATION AGREEMENT, INCLUDING ITS ENFORCEMENT, AND ANY STATE COURT OF COMPETENT JURISDICTION SHALL COMPEL ARBITRATION IN THE SAME MANNER AS A FEDERAL COURT UNDER THE FAA. I FURTHER AGREE THAT, TO THE FULLEST EXTENT PERMITTED BY LAW, I MAY BRING ANY ARBITRATION PROCEEDING ONLY IN MY INDIVIDUAL CAPACITY, AND NOT AS A PLAINTIFF, REPRESENTATIVE, OR CLASS MEMBER IN ANY PURPORTED CLASS, COLLECTIVE, OR REPRESENTATIVE LAWSUIT OR PROCEEDING. I UNDERSTAND, HOWEVER, THAT NOTHING IN THIS AGREEMENT PREVENTS ME FROM BRINGING A REPRESENTATIVE LAWSUIT OR PROCEEDING AS PERMITTED BY THE CALIFORNIA LABOR CODE'S PRIVATE ATTORNEYS GENERAL ACT OF 2004. THE FULLEST EXTENT PERMITTED BY LAW, I AGREE TO ARBITRATE ANY AND ALL COMMON LAW AND/OR STATUTORY CLAIMS UNDER LOCAL, STATE, OR FEDERAL LAW, INCLUDING, BUT NOT LIMITED TO, CLAIMS UNDER TITLE VII OF THE CIVIL RIGHTS ACT OF 1964, THE AMERICANS WITH DISABILITIES ACT OF 1990, THE AGE DISCRIMINATION IN EMPLOYMENT ACT OF 1967, THE OLDER WORKERS BENEFIT PROTECTION ACT, THE WORKER ADJUSTMENT AND RETRAINING NOTIFICATION ACT, THE FAIR LABOR STANDARDS ACT, THE CALIFORNIA FAIR EMPLOYMENT AND HOUSING ACT, THE FAMILY AND MEDICAL LEAVE ACT, THE CALIFORNIA FAMILY RIGHTS ACT, THE CALIFORNIA LABOR CODE, CLAIMS RELATING TO EMPLOYMENT STATUS, CLAIMS RELATING TO COMPENSATION (CASH, EQUITY, BONUS, OR OTHERWISE), CLAIMS RELATING TO CLASSIFICATION, AND CLAIMS OF HARASSMENT, DISCRIMINATION, WRONGFUL TERMINATION, AND BREACH OF CONTRACT. TO THE FULLEST EXTENT PERMITTED BY LAW, I ALSO AGREE TO ARBITRATE ANY AND ALL DISPUTES ARISING OUT OF OR RELATING TO THE INTERPRETATION OR APPLICATION OF THIS AGREEMENT TO ARBITRATE, BUT NOT DISPUTES ABOUT THE ENFORCEABILITY, REVOCABILITY, OR VALIDITY OF THIS AGREEMENT TO ARBITRATE OR THE CLASS, COLLECTIVE, AND REPRESENTATIVE PROCEEDING WAIVER HEREIN. WITH RESPECT TO ALL SUCH CLAIMS AND DISPUTES THAT I AGREE TO ARBITRATE, I HEREBY EXPRESSLY AGREE TO WAIVE, AND DO WAIVE, ANY RIGHT TO A TRIAL BY JURY. I FURTHER UNDERSTAND THAT THIS AGREEMENT TO ARBITRATE ALSO APPLIES TO ANY DISPUTES THAT THE COMPANY MAY HAVE WITH ME. I UNDERSTAND THAT NOTHING IN THIS AGREEMENT REQUIRES ME TO ARBITRATE CLAIMS THAT CANNOT BE ARBITRATED UNDER THE SARBANES-OXLEY ACT OR OTHER LAW THAT EXPRESSLY PROHIBITS ARBITRATION OF A CLAIM NOTWITHSTANDING THE APPLICATION OF THE FAA.

5

B. Administration of Arbitration. I AGREE THAT ANY ARBITRATION WILL BE ADMINISTERED BY JAMS PURSUANT TO ITS EMPLOYMENT ARBITRATION RULES & PROCEDURES (THE "JAMS RULES"), WHICH ARE AVAILABLE AT http://www.jamsadr.com/rules-employment-arbitration. IF THE JAMS RULES CANNOT BE ENFORCED AS TO THE ARBITRATION, THEN THE PARTIES AGREE THAT THEY WILL ARBITRATE THIS DISPUTE UTILIZING JAMS COMPREHENSIVE ARBITRATION RULES AND PROCEDURES OR SUCH RULES AS THE ARBITRATOR MAY DEEM MOST APPROPRIATE FOR THE DISPUTE. I AGREE THAT THE ARBITRATOR SHALL HAVE THE POWER TO DECIDE ANY MOTIONS BROUGHT BY ANY PARTY TO THE ARBITRATION, INCLUDING MOTIONS FOR SUMMARY JUDGMENT AND/OR ADJUDICATION, AND MOTIONS TO DISMISS AND DEMURRERS, APPLYING THE

STANDARDS SET FORTH FOR SUCH MOTIONS UNDER THE CALIFORNIA CODE OF CIVIL PROCEDURE. I AGREE THAT THE ARBITRATOR SHALL ISSUE A WRITTEN DECISION ON THE MERITS. I ALSO AGREE THAT THE ARBITRATOR SHALL HAVE THE POWER TO AWARD ANY REMEDIES AVAILABLE UNDER APPLICABLE LAW, AND THAT THE ARBITRATOR MAY AWARD ATTORNEYS' FEES AND COSTS TO THE PREVAILING PARTY, WHERE PERMITTED BY APPLICABLE LAW. I AGREE THAT THE DECREE OR AWARD RENDERED BY THE ARBITRATOR MAY BE ENTERED AS A FINAL AND BINDING JUDGMENT IN ANY COURT HAVING JURISDICTION THEREOF. I UNDERSTAND THAT THE COMPANY WILL PAY FOR ANY ADMINISTRATIVE OR HEARING FEES CHARGED BY THE ARBITRATOR OR JAMS EXCEPT THAT I SHALL PAY ANY FILING FEES ASSOCIATED WITH ANY ARBITRATION THAT I INITIATE, BUT ONLY SO MUCH OF THE FILING FEES AS I WOULD HAVE INSTEAD PAID HAD I FILED A COMPLAINT IN A COURT OF LAW THAT WOULD HAVE HAD JURISDICTION OVER SUCH COMPLAINT. SUBJECT TO THE FAA'S EXCLUSIVE APPLICABILITY TO THE ENFORCEMENT OF THIS AGREEMENT TO ARBITRATE, I AGREE THAT THE ARBITRATOR SHALL ADMINISTER AND CONDUCT ANY ARBITRATION HEARING OR PROCEEDING APPLYING CALIFORNIA SUBSTANTIVE AND DECISIONAL LAW AND THE CALIFORNIA CODE OF CIVIL PROCEDURE, INCLUDING THE CALIFORNIA CIVIL DISCOVERY ACT. I AGREE THAT ANY ARBITRATION UNDER THIS AGREEMENT SHALL BE CONDUCTED IN SANTA CLARA COUNTY, CALIFORNIA.

C. Remedy. FOR PURPOSES OF SEEKING PROVISIONAL REMEDIES ONLY, I AGREE THAT THE COMPANY AND I SHALL BE ENTITLED TO PURSUE ANY PROVISIONAL REMEDY PERMITTED BY THE CALIFORNIA ARBITRATION ACT (CALIFORNIA CODE CIV. PROC. § 1281.8), OR OTHERWISE PROVIDED BY THIS AGREEMENT. EXCEPT FOR SUCH PROVISIONAL RELIEF, I AGREE THAT ANY RELIEF OTHERWISE AVAILABLE TO THE COMPANY OR ME UNDER APPLICABLE LAW SHALL BE PURSUED SOLELY AND EXCLUSIVELY IN ARBITRATION PURSUANT TO THE TERMS OF THIS AGREEMENT.

D. Administrative Relief: I UNDERSTAND THAT THIS AGREEMENT DOES NOT PROHIBIT ME FROM PURSUING AN ADMINISTRATIVE CLAIM WITH A LOCAL, STATE, OR FEDERAL ADMINISTRATIVE BODY OR GOVERNMENT AGENCY THAT IS AUTHORIZED TO ENFORCE OR ADMINISTER LAWS RELATED TO EMPLOYMENT, INCLUDING, BUT NOT LIMITED TO, THE DEPARTMENT OF FAIR EMPLOYMENT AND HOUSING, THE EQUAL EMPLOYMENT OPPORTUNITY COMMISSION, THE NATIONAL LABOR RELATIONS BOARD, THE SECURITIES AND EXCHANGE COMMISSION, OR THE WORKERS' COMPENSATION BOARD. THIS AGREEMENT DOES, HOWEVER, PRECLUDE ME FROM PURSUING A COURT ACTION REGARDING ANY SUCH CLAIM, EXCEPT AS PERMITTED BY LAW.

6

E . Voluntary Nature of Agreement. I ACKNOWLEDGE AND AGREE THAT I AM EXECUTING THIS AGREEMENT TO ARBITRATE VOLUNTARILY AND WITHOUT ANY DURESS OR UNDUE INFLUENCE BY THE COMPANY OR ANYONE ELSE. I FURTHER ACKNOWLEDGE AND AGREE THAT I HAVE CAREFULLY READ THIS AGREEMENT TO ARTBIRATE AND THAT I HAVE ASKED ANY QUESTIONS NEEDED FOR ME TO UNDERSTAND THE TERMS, CONSEQUENCES, AND BINDING EFFECT OF THIS AGREEMENT TO ARBIBTRATE AND FULLY UNDERSTAND IT, INCLUDING THAT I AM WAIVING MY RIGHT TO A JURY TRIAL. I AGREE THAT I HAVE BEEN PROVIDED AN OPPORTUNITY TO SEEK THE ADVICE OF AN ATTORNEY OF MY CHOICE BEFORE SIGNING THIS AGREEMENT.

Agreed to and accepted:

Shaun R. Bagai Dated: November 11, 2021

RENOVORX, INC.

CONSULTING AGREEMENT

THIS CONSULTING AGREEMENT (the "Agreement") is entered on this \(\ldots \) day of January 2018 ("Effective Date") by and among RENOVORX, INC., a Delaware corporation, and its successors or assignees ("Company"), and Ramtin Agah (referred to herein as "Consultant") for the purpose of setting forth the terms and conditions by which the Company will acquire Consultant's services.

- ENGAGEMENT OF SERVICES. Consultant
 will, to the best of his or her ability, render
 the services set forth in Exhibit A attached
 hereto. Consultant shall perform the actions
 necessary to complete such services in
 a timely and professional manner consistent
 with industry standards, and at a location,
 place and time which the Consultant deems
 appropriate. Consultant may not subcontract
 or otherwise delegate his obligations under
 this Agreement without Company's prior
- COMPENSATION. Company will compensate Consultant for services rendered under this Agreement as set forth in Exhibit A attached hereto. Unless otherwise agreed to by the Company in writing, Consultant shall be responsible for all expenses incurred in performing services under this Agreement.
- INDEPENDENT CONTRACTOR RELATIONSHIP. Consultant's relationship with Company will be that of an independent contractor and nothing in this Agreement should be construed to create a partnership, joint venture, or employer-employee relationship. Consultant will be solely responsible for all tax returns and payments required to be filed with or made to any federal, state or local tax authority with respect to Consultant's performance of services and receipt of fees under this Agreement. Because Consultant is an independent Consultant, Company will not withhold or make payments for social security; make unemployment insurance or disability insurance contributions; or obtain worker's compensation insurance on Consultant's behalf. Consultant hereby agrees to indemnify and defend Company against any and all such taxes or contributions, including penalties and interest incurred by Company as a result of Consultant's failure to file or pay any such taxes or payments.

4. PROPRIETARY INFORMATION.

4.1 **Proprietary Information**. Consultant agrees during the term of this Agreement and thereafter that it will take all steps reasonably necessary to hold Company's Proprietary Information

- (as defined below) in trust and confidence, will not use Proprietary Information in any manner or for any purpose not expressly set forth in this Agreement, and will not disclose any such Proprietary Information to any third party without first obtaining the express written consent of the Company. By way of illustration but not limitation "Proprietary Information" includes (a) information relating to products, processes, know-how, designs, techniques, drawings, clinical data, test data, formulas, methods, samples, development or experimental work, improvements, discoveries, trade secrets, inventions, ideas, other works of authorship, (hereinafter collectively referred to as "Inventions"); collectively referred to as "Inventions"); (b) information regarding plans for research, development, new products, marketing and selling, business plans, budgets and non-public financial statements, licenses, contracts, prices and costs, suppliers and customers; (c) information regarding the skills and compensation of the Company's employees, consultants and any other service providers of the Company; and (d) the existence of any business discussions, negotiations, or agreements between the Company and any third party. Notwithstanding the other provisions of this Agreement, nothing received by Consultant will be considered to be Company Proprietary Information if (1) it has been published or is otherwise readily available to the public other than by a breach of this Agreement; (2) it has been rightfully received by Consultant from a third party without any confidentiality limitations; or (3) it was known by the Consultant, as evidenced by his records, prior to its disclosure by the Company.
- 4.2 Third Party Information. Consultant understands that Company has received and will in the future receive from third parties confidential or proprietary information ("Third Party Information") subject to a duty on Company's part to maintain the confidentiality of such information and use it only for certain limited purposes. Consultant agrees to hold Third Party Information in confidence and not to disclose to anyone (other than Company personnel who need to know such information in connection with their work for Company) or to use, except in connection with Consultant's work for Company, Third Party Information unless expressly authorized in

writing by an officer of Company.

5. OWNERSHIP OF WORK PRODUCT.

- 5.1 Disclosure of Work Product. As used in this Agreement, the term "Work Product" means any Invention, whether or not patentable, and all related know-how, designs, trademarks, formulae, processes, techniques, trade secrets, ideas, artwork, software, or any other copyrightable or patentable works. Consultant agrees to disclose promptly in writing to Company, or any person designated by Company, all Work Product which is solely or jointly conceived, made, reduced to practice, authored, or learned by Consultant in the course of any work performed for the Company under this Agreement ("Company Work Product"). Consultant agrees that any and all Company Work Product shall be the sole and exclusive property of Company. For clarification purposes, Company Work Product shall not include, and Consultant shall have no obligation to disclose to Company, any Work Product resulting from Consultant's pre-existing obligations as described in section 8.2; provided, however, Consultant shall notify the Company in advance of any situation arising out of any of these pre-existing obligations or any other obligation that could impair or diminish the Company's full rights to any Work Product developed or created pursuant to this Agreement.
- 5.2 Background Technology. Consultant shall specifically describe and identify in Exhibit B any and all works of authorship and Inventions which (a) Consultant intends to use in performing under this Agreement, (b) is either owned by Consultant or licensed to Consultant with a right to sublicense and (c) is made, conceived, reduced to practice, or is in existence in the form of a writing or fixed in any medium prior to the Effective Date ("Background Technology"). If disclosure of any Background Technology would cause Consultant to violate any prior confidentiality agreement, Consultant understands that it is not to list such Background Technology in Exhibit B but it will disclose a cursory name for each such invention, a listing of the party(ies) to whom it belongs, and the fact that full disclosure as to such Background Technology has not been made for that reason. A space is provided in Exhibit B for such purpose. Consultant further represents that any Work Product which Consultant has made, conceived or reduced to practice prior to signing this Agreement

in the Field (as such term is defined in Section 8.2 below) has been disclosed in writing to Company and attached to this Agreement as Exhibit B ("Prior Technology"). If no such disclosure is attached in Exhibit B, Consultant represents that there is no Prior Technology. If, in the course of performance of this Agreement, Consultant incorporates any Background Technology or Prior Technology into an Invention of the Company, product, process or machine, the Company is hereby granted and shall have a nonexclusive, royalty-free, irrevocable, perpetual, worldwide license (with rights to sublicense through multiple tiers of sublicensees) to make, have made, modify, use and sell such Invention, product, process or machine. Notwithstanding the foregoing, Consultant agrees that he will not incorporate, or permit to be incorporated, any Background Technology or Prior Technology in any Company Inventions, product, process or machine without the Company's prior written consent.

- 5.3 Assignment of Company Work Product. Consultant irrevocably assigns to Company all right, title and interest worldwide in and to the Company Work Product and all applicable intellectual property rights related to the Company Work Product, including without limitation, patents, copyrights, trademarks, trade secrets, contract and licensing rights (the "Proprietary Rights"). Consultant retains no rights to use the Company Work Product.
- 5.4 Assistance. At the expense of Company, Consultant agrees to cooperate with Company or its designee(s), both during and after the term of this Agreement, in the procurement and maintenance of Company's rights in Company Work Product and to execute, when requested, any other documents deemed necessary by Company to carry out the purpose of this Agreement.
- 5.5 Enforcement of Proprietary Rights. At the expense of Company, Consultant will assist Company in every proper way to obtain, and from time to time enforce, United States and foreign Proprietary Rights relating to Company Work Product. To that end Consultant will execute, verify and deliver such documents and perform such other acts (including appearances as a witness) as Company may reasonably request for use in applying for, obtaining, perfecting, evidencing, sustaining and enforcing such Proprietary Rights and the assignment thereof, including any

applicable filings with the U.S. Patent and Trademark Office and the U.S. Food and Drug Administration and the respective foreign counterparts to such government offices or agencies. In addition, Consultant will execute, verify and deliver assignments of such Proprietary Rights to Company or its designee. Consultant's obligation to assist Company with respect to Proprietary Rights relating to such Company Work Product in any and all countries shall continue beyond the termination of this Agreement, but Company shall compensate Consultant at a reasonable rate after such termination for the time actually spent by Consultant at Company's request on such assistance.

- Execution of Documents. In the event Company is unable for any reason, after reasonable effort, to secure Consultant's signature on any document needed in connection with the actions specified in the preceding Sections 5.4 and 5.5, Consultant hereby irrevocably designates and appoints Company and its duly authorized officers and agents as its agent and attorney-in-fact, which appointment is coupled with an interest, to act for and in its behalf to execute, verify and file any such documents and to do all other lawfully permitted acts to further the purposes of the preceding paragraph with the same legal force and effect as if executed by Consultant. Consultant hereby waives and quitclaims to Company any and all claims, of any nature whatsoever, which Consultant now or may hereafter have for infringement of any Proprietary Rights assigned hereunder to Company.
- 6. OBLIGATION TO KEEP COMPANY INFORMED. Subject to the pre-existing obligations as described in Section 8.2, during the period of this Agreement and for twelve (12) months after termination of this Agreement, Consultant shall promptly disclose to the Company fully and in writing all Inventions in the Field authored, conceived or reduced to practice by Consultant, either alone or jointly with others. In addition, subject to the pre-existing obligations as described in Section 8.2, Consultant shall promptly disclose to the Company all patent applications relating to the Field filed by him or on his behalf within a year after termination of this Agreement.
- 7. CONSULTANT REPRESENTATIONS AND WARRANTIES. Consultant hereby represents and warrants that (a) to the best of Consultant's

- knowledge, neither the Company Work Product, nor any element thereof will infringe the Proprietary Rights of any third party; (b) neither the Company Work Product, nor any element thereof will be subject to any restrictions or to any mortgages, liens, pledges, security interests, encumbrances or encroachments: (c) Consultant will not grant, directly or indirectly, any rights or interest whatsoever in the Company Work Product to third parties; (d) Consultant has full right and power to enter into and perform this Agreement without the consent of any third party; (e) Consultant will comply with all laws and regulations applicable to Consultant's obligations under this Agreement; (f) Consultant is not subject to any contract or duty that would be breached by Consultant's entering into or performing Consultant's obligations under this Agreement or that is otherwise inconsistent with this Agreement; and (g) should the Company permit Consultant to use any of the Company's equipment or facilities during the term of this Agreement, such permission shall be gratuitous and Consultant shall be responsible for any injury to any person (including death) or damage to property arising out of use of such equipment or facilities.
- 8. RESTRICTIVE COVENANTS. Consultant acknowledges that: (a) the business of the Company is intensely competitive and that Consultant's relationship with the Company requires that Consultant have access to and knowledge of Proprietary Information; (b) the direct and indirect disclosure of any such Proprietary Information would place the Company at a competitive disadvantage and would do damage, monetary or otherwise, to the Company's business; (c) the Proprietary Information constitutes a trade secret of the Company; and (d) the engaging by Consultant in any of the activities prohibited by this Section 8 may constitute improper misappropriation and/or use of such information and trade secrets.
- 8.1 Nondisclosure of Proprietary Information. Consultant agrees that at all times during and after the termination of his relationship with the Company, Consultant shall not, directly or indirectly, whether individually, as a director, stockholder, owner, partner, employee, principal or agent of any business, or in any other capacity, make known, disclose, furnish, make available or utilize any of the Proprietary Information. This confidentiality covenant has no temporal, geographical or territorial

restriction. Consultant agrees to immediately return all Proprietary Information, Company documents (and all copies thereof) and other Company property and materials in his possession or control, including, but not limited to, Company reports, notes, files, memoranda, records, drawings, business plans and forecasts, financial information, specifications, computer-recorded information, software, tangible property (including, but not limited to, computers and cellular phones), credit cards, travel cards, entry cards, identification badges and keys, and any materials of any kind that contain or embody any Proprietary Information of the Company (and all reproductions thereof).

- 8.2 No Conflict of Interest. Consultant agrees that he will not, at any time during the term of this Agreement and for a period of six (6) months thereafter, without the prior written consent of the Company, engage in any business or activity, accept work or enter into a contract or agreement, or otherwise become associated with any business (i) relating to the research, design, development, transfer of intellectual property rights, commercializing and/or marketing of any product or technology relating to the attempted treatment or enhanced treatment of pancreatic cancer by any endovascular approach, including by means of delivery of any therapeutic materials to the pancreas, and all related devices, accessories, products, kits or services (collectively, the "Field"), or (ii) that is in conflict or incompatible with Consultant's obligations under this Agreement or the scope of services rendered for Company. Consultant represents and warrants that except for this Agreement, he has not entered into any contract or agreement relating to the Field. Consultant further represents that he is not a party to any existing agreement or obligation inconsistent or in conflict with
- 8.3 Non-solicitation of Company Employees and Customers. Consultant hereby agrees that at any time during the term of this Agreement and for a period of six (6) months thereafter, Consultant will not, without first obtaining the Company's prior written permission, (a) directly or indirectly solicit, entice, induce, or encourage employees or consultants of the Company to leave the Company to accept work with a competing business, or (b) directly or indirectly solicit any customer or prospective customer of the Company on Consultant's

own behalf or on behalf of any competitor of the Company, for which Consultant rendered services during his relationship with the Company.

9. TERM AND TERMINATION.

- 9.1 **Term.** The term of this Agreement will be for fifteen (15) years beginning on the Effective Date, unless terminated earlier pursuant to this Section 9; *thereafter*, this Agreement shall automatically be renewed for successive one (1) year periods until terminated.
- 9.2 **Termination.** Either the Company or Consultant may terminate this Agreement at its convenience by providing at least thirty (30) days prior written notice to the other.
- 9.3 Return of Company Property. Upon termination of the Agreement or earlier as requested by Company, Consultant will deliver to Company any and all samples, drawings, notes, memoranda, specifications, devices, formulas, and documents, together with all copies thereof, and any other material containing or disclosing any Company Work Product, Third Party Information or Proprietary Information of Company.

GENERAL PROVISIONS.

- 10.1 Governing Law. This Agreement will be governed by and construed according to the laws of the State of California, as such laws are applied to agreements entered into and to be performed entirely within California between California residents. The Consultant hereby expressly consent to the personal jurisdiction of the state and federal courts located in the county where Company's principle place of business is located for any lawsuit filed there against Consultant by Company arising from or related to this Agreement.
- 10.2 Severability. In case any one or more of the provisions contained in this Agreement shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect the other provisions of this Agreement, and this Agreement shall be construed as if such invalid, illegal or unenforceable provision had never been contained herein. If moreover, any one or more of the provisions contained

in this Agreement shall for any reason be held to be excessively broad as to duration, geographical scope, activity or subject, it shall be construed by limiting and reducing it, so as to be enforceable to the extent compatible with the applicable law as it shall then appear.

- 10.3 No Assignment. This Agreement, and Consultant's rights and obligations herein, may not be assigned, subcontracted, delegated, or otherwise transferred by Consultant without the Company's prior written consent, and any attempted assignment, subcontract, delegation, or transfer in violation of the foregoing will be null and void. The terms of this Agreement shall be binding upon assignees.
- 10.4 Notices. Any notice required or permitted by this Agreement shall be in writing and shall be delivered as follows with the notice deemed given as indicated: (i) by personal delivery when delivered personally; (ii) by overnight courier upon written verification of receipt; (iii) by telecopy or facsimile transmission upon acknowledgment of receipt of electronic transmission; (iv) by certified or registered mail, return receipt requested, upon verification of receipt. Notice shall be sent to the addresses set forth below or such other address as either party may specify in writing.
- of the promises or agreements contained in this Agreement may result in irreparable and continuing damage to Company for which there may be no adequate remedy at law, and Company is therefore entitled to seek injunctive relief as well as such other and further relief as may be appropriate.
- 10.6 **Survival**. Sections 2, 4, 5, 6, 7, 8 and 10 shall survive termination of this Agreement.
- 10.7 **Waiver.** No waiver by Company of any breach of this Agreement shall be a waiver of any preceding or succeeding breach. No waiver by Company of any right under this Agreement shall be construed as a waiver of any other right.
- 10.8 Entire Agreement. This Agreement is the final, complete and exclusive agreement of the parties with respect to the subject matter hereof and supersedes and merges all prior discussions between us. No modification of or amendment to this Agreement, nor any waiver of any rights under this Agreement, will be effective unless in writing and signed by the party to be charged.

IN WITNESS WHEREOF, that parties have caused this Consulting Agreement to be executed as of the date first written above.

CA 94022

COMPANY:

RENOVORX, INC.

Title:

4546 EL CANDON MENT STE 203 LOS ALTOS, CA 94022 Address:

CONSULTANT:

EXHIBIT A

STATEMENT OF WORK

This Statement of Work ("SOW") is incorporated into the Consulting Agreement by and between Company and Consultant (the "Agreement") to which this SOW is an Exhibit. This SOW describes services to be performed and provided by Consultant pursuant to the Agreement. If any item in this SOW is inconsistent with the Agreement prior to such incorporation, the terms of this SOW will control, but only with respect to the services to be performed under this SOW.

1. Scope of Services:

Consultant shall serve as the Company's Chief Medical Officer, and shall provide such services as are consistent with the Company's charter and as may be reasonably requested by the Company's Board of Directors from time to time.

2. Compensation:

Except with respect to reimbursable expenses as described in Section 3 below, the sole compensation payable to Consultant shall be continuation of vesting of shares of common stock of the Company held by Consultant pursuant to the Founder Restricted Stock Purchase Agreement dated on or about the same date hereof ("FRSPA"), and the grant of Stock Options (300,00 shares) in the May 19, 2017 BOD Meeting. Consultant acknowledges and agrees that such grant constitutes payment in full for any and all services performed to date. The Restricted Stock shall vest in accordance with the following:

- (1) 25% of the shares subject to the option will vest on the 1-year anniversary of the Vesting Commencement Date (January 1, 2016); and the remaining 75% of the shares subject to the option will vest in 36 equal monthly installments thereafter for so long as the optionee remains in continuous service, such that on the fourth anniversary of the Vesting Commencement Date, the option will be fully vested.
- (2) Upon a Change in Control (as defined below), and provided that the optionee is still a consultant or employee with the Company pursuant to optionee's consulting agreement or offer letter through and including the consummation of such Change in Control, then all outstanding shares subject to the option shall vest in full effective immediately as of the consummation of such Change of Control. Notwithstanding the foregoing, as a pre-condition of the accelerated vesting referenced in the immediately preceding sentence, optionee will be required to timely sign, date and return to the Company (or its successor), and to not subsequently revoke, a general release of all known and unknown claims in the form provided to the optionee by the Company.
- (3) For purposes of the above paragraph, a "Change in Control" shall mean the following: (A) any consolidation or merger of the Company with or into any other corporation or other entity or person, or any other corporate reorganization, other than any such consolidation, merger or reorganization in which the stockholders of the Company immediately prior to such consolidation, merger or reorganization, continue to hold a majority of the voting power of the surviving entity (or, if the surviving entity is a wholly owned subsidiary, its parent) immediately after such consolidation, merger or reorganization; (B) any transaction or series of related transactions to which the Company is a party in which in excess of fifty percent (50%) of the Company's voting power is transferred; provided that the foregoing shall not include any transaction or series of transactions principally for bona fide equity financing purposes in which cash is received by the Company or indebtedness of the Company is cancelled or converted or a combination thereof; or (C) a sale, lease, exclusive license or other disposition of all or substantially all of the assets of the Company. For clarification

purposes, any dissolution, bankruptcy, liquidation, or cessation of the Company's operations or similar event shall not be deemed a Change in Control.

3. Expenses.

Company will reimburse Consultant for reasonable out-of-pocket business expenses incurred in connection with the Services, provided that such expenses are approved in advance by the Company and fully documented to Company's satisfaction. As of the date hereof, no amount is due to the Consultant for any such expenses. Consultant shall be reimbursed for approved out-of-pocket expenses as soon as practicable after an invoice is received, on a monthly basis.

Consultant Initial

Company Initial

EXHIBIT B BACKGROUND TECHNOLOGY AND PRIOR TECHNOLOGY					
 Except as listed in Section 2 below, the following is a complete list of all Background Technology (as defined in the Agreement) that have been made or conceived or first reduced to practice by me alone or jointly with others prior to my engagement by Company: 					
☐ List Background Technology here:					
☐ Additional sheets attached.					
2. Due to a prior confidentiality agreement, I cannot complete the disclosure under Section 1 above with respect to inventions or improvements generally listed below, the proprietary rights and duty of confidentiality with respect to which I owe to the following party(ies):					

Invention or Improvement Party(ies)

Relationship

1. 2. 3.

Additional sheets attached.

3. The following is a list of all Prior Technology (as defined in the Agreement):

US8546323 Thrombospondin-1 derived peptides and treatment methods

US60693574P0 Thrombospondin-1 derived peptides and treatment methods

 $\underline{US20070166231A1\ Methods\ and\ probes\ for\ identifying\ vulnerable\ plaque}$

 $\underline{US60459646P0\ Targeted\ delivery\ of\ a\ fluorescent\ probe\ to\ atherosclerotic\ plaque\ as\ a\ means\ of\ identifying\ vulnerable\ plaque\ during\ coronary\ angiography}$

Company Initial

-9-

AMENDMENT TO CONSULTING AGREEMENT

This Third Amendment (the "Third Amendment") by and between RenovoRx, Inc. (the "Company") and Ramtin Agah ("Consultant") (together, the "Parties") amends the Consulting Agreement by and between the Parties dated January 1, 2018 ("Consulting Agreement"), as amended by the Second Amendment to Consulting Agreement, effective August 1, 2019 (the "Second Amendment" and together with the Consulting Agreement, the "Agreement"). This Amendment is entered into as of November 11, 2021 ("Amendment Effective Date").

The Parties desire to amend on the terms of the Agreement on the terms set forth herein.

1. Amendment to Section 6. Section 6 of the Agreement is hereby amended and restated in its entirety as follows:

Obligation to Keep Company Informed. During the period of this Agreement, Consultant shall promptly disclose to the Company fully and in writing all Inventions in the Field authored, conceived or reduced to practice by Consultant, either alone or jointly with others.

2. Amendment to Section 8.2. Section 8.2 of the Agreement is hereby amended and restated in its entirety as follows:

Conflicting Obligations. Consultant represents and warrants that Consultant has no agreements, relationships, or commitments to any other person or entity that conflict with the provisions of this Agreement, Consultant's obligations to the Company under this Agreement, and/or Consultant's ability to perform Consultant's services to the Company. Consultant will not enter into any such conflicting agreement during the term of this Agreement. Consultant agrees to notify Company if he enters into any agreement with a third party that relates to the research, design, development, transfer of intellectual property rights, commercializing and/or marketing of any product or technology relating to the attempted treatment or enhanced treatment of solid tumors by any endovascular or catheter-based approach, including by means of delivery of any therapeutic materials to solid tumors, and all related devices, accessories, products, kits or services (collectively, the "Field"). Consultant shall not use the funding, resources and facilities of any other third party, without the prior written consent of the Company, to perform services hereunder and shall not perform the services hereunder in any manner that would give any third party rights or access to the product of such services. The services performed hereunder will not be conducted on time that is required to be devoted to any other third party. Without limiting the foregoing, Consultant agrees to use his or her best efforts (A) to segregate Consultant's services performed under this Agreement from Consultant's work done for any third party so as to minimize any questions of rights under any inventions, (B) to notify the Company if at any time the Consultant believes that such questions may result from Consultant's performance under this Agreement and (C) to assist the Company in fairly resolving any questions in this regard which may arise.

- 2. Deletion of Section 8.3. Section 8.3 of the Agreement is hereby entirely deleted from the Agreement.
- 3. Amendment to 9.1. Section 9.1 of the Agreement is amended as restated in its entirety as follows:

Term. The term of this Agreement began on the Effective Date of this Agreement and will continue until the earlier of (i) final completion of Consultant's services to the Company under this Agreement or (ii) termination as provided in Section 9.2 of this Agreement.

4. Amendment of Section 10.5. Section 10.5 of the Agreement is hereby amended and restated in its entirety as follows:

Arbitration and Equitable Relief.

A . Arbitration. IN CONSIDERATION OF CONSULTANT'S CONSULTING RELATIONSHIP WITH THE COMPANY, ITS PROMISE TO ARBITRATE ALL DISPUTES RELATED TO CONSULTANT'S CONSULTING RELATIONSHIP WITH THE COMPANY AND CONSULTANT'S RECEIPT OF COMPENSATION AND OTHER CONSIDERATION PROVIDED TO CONSULTANT BY THE COMPANY, AT PRESENT AND IN THE FUTURE, CONSULTANT AGREES THAT ANY AND ALL CONTROVERSIES, CLAIMS, OR DISPUTES THAT CONSULTANT MAY HAVE WITH THE COMPANY (INCLUDING ANY COMPANY EMPLOYEE, OFFICER, DIRECTOR, TRUSTEE, OR BENEFIT PLAN OF THE COMPANY, IN THEIR CAPACITY AS SUCH OR OTHERWISE), ARISING OUT OF, RELATING TO, OR RESULTING FROM CONSULTANT'S CONSULTING OR OTHER RELATIONSHIP WITH THE COMPANY OR THE TERMINATION OF CONSULTANT'S CONSULTING OR OTHER RELATIONSHIP WITH THE COMPANY, INCLUDING ANY BREACH OF THIS AGREEMENT, SHALL BE SUBJECT TO BINDING ARBITRATION PURSUANT TO THE FEDERAL ARBITRATION ACT (9 U.S.C. SEC. 1 ET SEQ.) (THE " $\it FAA$ "). THE FAA'S SUBSTANTIVE AND PROCEDURAL PROVISIONS SHALL EXCLUSIVELY GOVERN AND APPLY WITH FULL FORCE AND EFFECT TO THIS ARBITRATION AGREEMENT, INCLUDING ITS ENFORCEMENT, AND ANY STATE COURT OF COMPETENT JURISDICTION SHALL COMPEL ARBITRATION IN THE SAME MANNER AS A FEDERAL COURT UNDER THE FAA. CONSULTANT FURTHER AGREES THAT, TO THE FULLEST EXTENT PERMITTED BY LAW, CONSULTANT MAY BRING ANY ARBITRATION PROCEEDING ONLY IN CONSULTANT'S INDIVIDUAL CAPACITY, AND NOT AS A PLAINTIFF, REPRESENTATIVE, OR CLASS MEMBER IN ANY PURPORTED CLASS, COLLECTIVE, OR REPRESENTATIVE LAWSUIT OR PROCEEDING, CONSULTANT UNDERSTANDS, HOWEVER, THAT NOTHING IN THIS AGREEMENT PREVENTS CONSULTANT FROM BRINGING A REPRESENTATIVE LAWSUIT OR PROCEEDING AS PERMITTED BY THE CALIFORNIA LABOR CODE'S PRIVATE ATTORNEYS GENERAL ACT OF 2004. TO THE FULLEST EXTENT PERMITTED BY LAW, CONSULTANT AGREES TO ARBITRATE ANY AND ALL COMMON LAW AND/OR STATUTORY CLAIMS UNDER LOCAL, STATE, OR FEDERAL LAW, INCLUDING, BUT NOT LIMITED TO, CLAIMS UNDER THE CALIFORNIA LABOR CODE, CLAIMS RELATING TO EMPLOYMENT OR INDEPENDENT CONTRACTOR STATUS, CLAIMS RELATING TO COMPENSATION (CASH, EQUITY, OR OTHERWISE), CLAIMS RELATING TO CLASSIFICATION, AND RELATIONSHIP WITH THE COMPANY, AND CLAIMS OF BREACH OF CONTRACT, TO THE FULLEST EXTENT PERMITTED BY LAW. CONSULTANT ALSO AGREES TO ARBITRATE ANY AND ALL DISPUTES ARISING OUT OF OR RELATING TO THE INTERPRETATION OR APPLICATION OF THIS AGREEMENT TO ARBITRATE, BUT NOT DISPUTES ABOUT THE ENFORCEABILITY, REVOCABILITY OR VALIDITY OF THIS AGREEMENT TO ARBITRATE OR THE CLASS, COLLECTIVE AND REPRESENTATIVE PROCEEDING WAIVER HEREIN. WITH RESPECT TO ALL SUCH CLAIMS AND DISPUTES THAT CONSULTANT AGREES TO ARBITRATE, CONSULTANT HEREBY EXPRESSLY AGREES TO WAIVE, AND DOES WAIVE, ANY RIGHT TO A TRIAL BY JURY. CONSULTANT FURTHER UNDERSTANDS THAT THIS AGREEMENT TO ARBITRATE ALSO APPLIES TO ANY DISPUTES THAT THE COMPANY MAY HAVE WITH CONSULTANT. CONSULTANT UNDERSTANDS THAT NOTHING IN THIS AGREEMENT REQUIRES CONSULTANT TO ARBITRATE CLAIMS THAT CANNOT BE ARBITRATED UNDER THE SARBANES-OXLEY ACT OR OTHER LAW THAT EXPRESSLY PROHIBITS ARBITRATION OF A CLAIM NOTWITHSTANDING THE APPLICATION OF THE FAA.

B. Administration of Arbitration. Consultant agrees that any arbitration will be administered by Jams Pursuant to its employment arbitration rules & procedures (the "Jams Rules"), which are available at http://www.jamsadr.com/rules-employment-arbitration. If the Jams rules cannot be enforced as to the arbitration, then the parties agree that they will arbitrate this dispute utilizing Jams comprehensive arbitration rules and procedures or such rules as the arbitrator may deem most appropriate for the dispute. Consultant agrees that the use of the Jams rules does not change consultant's classification to that of an employee. To the contrary, consultant reaffirms that consultant is an independent contractor. Consultant agrees that the arbitrator shall have the power to decide any motions brought by any party to the arbitration, including motions for summary judgment and/or adjudication and motions to dismiss and demurrers applying the standards for such motions set forth under the california code of civil procedure. Consultant agrees that the arbitrator shall issue a written decision on the merits. Consultant also agrees that the arbitrator shall have the

POWER TO AWARD ANY REMEDIES AVAILABLE UNDER APPLICABLE LAW, AND THAT THE ARBITRATOR MAY AWARD ATTORNEYS' FEES AND COSTS TO THE PREVAILING PARTY, WHERE PERMITTED BY APPLICABLE LAW. CONSULTANT AGREES THAT THE DECREE OR AWARD RENDERED BY THE ARBITRATOR MAY BE ENTERED AS A FINAL AND BINDING JUDGMENT IN ANY COURT HAVING JURISDICTION THEREOF. CONSULTANT UNDERSTANDS THAT THE COMPANY WILL PAY FOR ANY ADMINISTRATIVE OR HEARING FEES CHARGED BY THE ARBITRATOR OR JAMS, EXCEPT THAT CONSULTANT SHALL PAY ANY FILING FEES ASSOCIATED WITH ANY ARBITRATION THAT CONSULTANT INITIATES, BUT ONLY SO MUCH OF THE FILING FEES AS CONSULTANT WOULD HAVE INSTEAD PAID HAD CONSULTANT FILED A COMPLAINT IN A COURT OF LAW THAT WOULD HAVE HAD JURISDICTION OVER SUCH COMPLAINT. SUBJECT TO THE FAA'S EXCLUSIVE APPLICABILITY TO THE ENFORCEMENT OF THIS AGREEMENT TO ARBITRATE, CONSULTANT AGREES THAT THE ARBITRATOR SHALL ADMINISTER AND CONDUCT ANY ARBITRATION HEARING OR PROCEEDING APPLYING CALIFORNIA SUBSTANTIVE AND DECISIONAL LAW AND THE CALIFORNIA CODE OF CIVIL PROCEDURE, INCLUDING THE CALIFORNIA CIVIL DISCOVERY ACT. CONSULTANT AGREES THAT ANY ARBITRATION UNDER THIS AGREEMENT SHALL BE CONDUCTED IN SANTA CLARA COUNTY, CALIFORNIA.

Page 2 of 5

- C . Remedy. FOR PURPOSES OF SEEKING PROVISIONAL REMEDIES ONLY, CONSULTANT AGREES THAT THE COMPANY AND CONSULTANT SHALL BE ENTITLED TO PURSUE ANY PROVISIONAL REMEDY PERMITTED BY THE CALIFORNIA ARBITRATION ACT (CALIFORNIA CODE CIV. PROC. § 1281.8), OR OTHERWISE PROVIDED BY THIS AGREEMENT. EXCEPT FOR SUCH PROVISIONAL RELIEF, CONSULTANT AGREES THAT ANY RELIEF OTHERWISE AVAILABLE TO THE COMPANY OR CONSULTANT UNDER APPLICABLE LAW SHALL BE PURSUED SOLELY AND EXCLUSIVELY IN ARBITRATION PURSUANT TO THE TERMS OF THIS AGREEMENT.
- D. Administrative Relief. CONSULTANT UNDERSTANDS THAT THIS AGREEMENT DOES NOT PROHIBIT CONSULTANT FROM PURSUING AN ADMINISTRATIVE CLAIM WITH A LOCAL, STATE OR FEDERAL ADMINISTRATIVE BODY OR GOVERNMENT AGENCY SUCH AS THE DEPARTMENT OF FAIR EMPLOYMENT AND HOUSING, THE EQUAL EMPLOYMENT OPPORTUNITY COMMISSION, THE NATIONAL LABOR RELATIONS BOARD, THE SECURITIES AND EXCHANGE COMMISSION, OR THE WORKERS' COMPENSATION BOARD. THIS AGREEMENT DOES, HOWEVER, PRECLUDE CONSULTANT FROM BRINGING ANY ALLEGED WAGE CLAIMS WITH THE DEPARTMENT OF LABOR STANDARDS ENFORCEMENT. LIKEWISE, THIS AGREEMENT DOES PRECLUDE CONSULTANT FROM PURSUING A COURT ACTION REGARDING ANY SUCH CLAIM, EXCEPT AS PERMITTED BY LAW.
- E. Voluntary Nature of Agreement. Consultant acknowledges and agrees that consultant is executing this agreement voluntarily and without any duress or undue influence by the company or anyone else. Consultant further acknowledges and agrees that consultant has carefully read this agreement and that consultant has asked any questions needed for consultant to understand the terms, consequences and binding effect of this agreement and fully understand it, including that consultant is waiving consultant's right to a jury trial. Consultant agrees that consultant has been provided an opportunity to seek the advice of an attorney of consultant's choice before signing this agreement.
 - 5. <u>Insertion of Section 10.9</u>. The following is added to the Agreement, as Section 10.9:

Protected Activity Not Prohibited Consultant understands that nothing in this Agreement limits or prohibits Consultant from filing and/or pursuing a charge or complaint with, or otherwise communicating or cooperating with, or participating in any investigation or proceeding that may be conducted by, any federal, state or local government agency or commission, including the Securities and Exchange Commission ("Government Agencies"), including disclosing documents or other information as permitted by law, without giving notice to, or receiving authorization from, the Company. In addition, nothing in this Agreement is intended to prevent Consultant from discussing or disclosing information about unlawful acts in the workplace, such as harassment or discrimination or any other conduct that Consultant has reason to believe is unlawful. Consultant further understands that Consultant is not permitted to disclose the Company's attorney-client privileged communications or attorney work product. Pursuant to the Defend Trade Secrets Act of 2016, Consultant is notified that an individual will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that (i) is made in confidence to a federal, state, or local government official (directly or indirectly) or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, or (ii) is made in a complaint or other document filed in a lawsuit or other proceeding, if (and only if) such filing is made under seal. In addition, an individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the individual's attorney and use the trade secret information in the court proceeding, if the individual files any document containing the trade secret under seal and does not disclose the trade secret, except pursuant to court order.

Page 3 of 5

6. Amendment to Second Amendment (which had replaced and superseded the First Amendment to the Consulting Agreement (which had previously replaced and superseded Exhibit A of the Consulting Agreement)). The Second Amendment (as amended herein) will be known as the Statement of Work ("SOW"). The following replaces and supersedes Section 2.1 of the Second Amendment, and will be considered Section 2 of the SOW.

Section 2: Compensation

- 2.1 <u>Base Consulting Fee</u>. The Company will pay Consultant a monthly consulting fee of \$21,667.67 ("Base Consulting Fee"), based on Consultant spending no less than 24 hours per week on Company matters (the "Allocated Time"). If Consultant's Allocated Time decreases, the Company may, in its discretion, proportionally adjust the Base Consulting Fee. Consultant agrees that any such reduction in the Base Consulting Fee will not constitute "Good Reason" or any similar definition or concept in any agreement, contract, or arrangement between Consultant and the Company notwithstanding any language to the contrary in any such agreement, contract, or arrangement, nor serves as a trigger for any related benefits.
- 2.2 <u>Bonus</u>. Commencing in the Company's 2022 fiscal year, Consultant will have the opportunity to earn a target annual cash bonus equal to thirty-five percent (35%) of Consultant's annualized Base Consulting Fee, based on achieving performance objectives established by the Company's Board of Directors ("*Board*") or its Compensation Committee, as applicable, in its sole discretion and payable upon achievement of those objectives, and subject to such terms and conditions, as determined by the Board or its Compensation Committee. Unless determined otherwise by the Board or its Compensation Committee, as applicable, any such bonus will be subject to Consultant's continued consulting or employment relationship with the Company through and until the date of payment, provided that service as a non-employee member of the Board will not by itself constitute a continued consulting relationship for this purpose. Consultant's annual bonus opportunity and the applicable terms and conditions may be adjusted from time to time by the Board or its Compensation Committee, as applicable, in its sole discretion.
- 2.3 Section 409A. In order to help prevent adverse tax consequences to Consultant under Section 409A (as defined below) should Consultant be or become subject to U.S. taxes, in no event will any payment under Section 2.1 or section 2.2 of this Statement of Work be made later than the later of (1) March 15th of the calendar year following the calendar year in which such payment was earned, or (2) the 15th day of the third (3rd) month following the end of the Company's fiscal year in which such payment was earned. All payments and benefits provided for under this Agreement are intended to be exempt from or otherwise comply with the requirements of Section 409A of the Internal Revenue Code of 1986, as amended, and the regulations and guidance thereunder (together, "Section 409A") so that none of the payments and benefits to be provided hereunder will be subject to the additional tax imposed under Section 409A, and any ambiguities or ambiguous terms herein will be interpreted to be exempt or so comply. Each payment and benefit payable under this Agreement is intended to constitute a separate payment for purposes of Section 1.409A-2(b)(2) of the Treasury Regulations. In no event will the Company have any responsibility, liability or obligation to reimburse, indemnify or hold harmless Consultant or any other person for any taxes penalties or interest that may be imposed, or other costs that may be incurred, by Consultant or any other person as a result of Section 409A.

7. Insertion to Second Amendment. The following is added as Section 3 of the Second Amendment (the SOW).

Section 3: Reimbursement

The Company will reimburse Consultant, in accordance with Company policy, for all reasonable expenses incurred by Consultant in performing the services pursuant to this Agreement, if Consultant receives written consent from an authorized agent of the Company prior to incurring such expenses and submits receipts for such expenses to the Company in accordance with Company policy.

Page 4 of 5

8. Miscellaneous

- A. Full Force and Effect. To the extent not expressly amended hereby, the Agreement shall remain in full force and effect.
- B. Entire Agreement. This Third Amendment, together with the Agreement, constitutes the full and entire understanding and agreement between Company and Consultant with respect to the subjects hereof and thereof.
- C. No Oral Modification No modification of or amendment to this Third Amendment (or the Agreement) will be effective unless in a writing signed by Consultant and an authorized signatory of the Company.

Company Consultant: By: /s/ Shaun R. Bagai By: /s/ Ramtin Agah Shaun R. Bagai Ramtin Agah, MD Name: Name: Title: **Chief Executive Officer** Title: **CMO** November 11, 2021 November 11, 2021 Date: Date: Page 5 of 5

RENOVORX, INC.

CHANGE IN CONTROL AND SEVERANCE AGREEMENT

This Change in Control and Severance Agreement (the "<u>Agreement</u>") is made by and between RenovoRx, Inc., a Delaware corporation (the '<u>Company</u>"), and Shaun R. Bagai ("<u>Executive</u>"), effective as of the Effective Date, as defined in Section 7 below.

This Agreement provides certain protections to Executive in connection with an involuntary termination of Executive's employment with the Company under the circumstances described in this Agreement, including in connection with a change in control of the Company. Certain capitalized terms used in this Agreement are defined in Section 7 below.

The Company and Executive agree as follows:

- 1. Term of Agreement. This Agreement will continue indefinitely until terminated by written consent of the parties hereto, or if earlier, upon the date that all of the obligations of the parties hereto with respect to this Agreement have been satisfied.
- 2. <u>At-Will Employment</u>. The Company and Executive acknowledge that Executive's employment is and will continue to be at-will, as defined under applicable law. No payments, benefits, or provisions under this Agreement will confer upon Executive any right to continue Executive's employment with the Company, nor will they interfere with or limit in any way the right of the Company or Executive to terminate such relationship at any time, with or without cause, to the extent permitted by applicable laws.

3. Severance Benefits.

- (a) <u>Qualifying Termination Outside of the Change in Control Period</u>. In the event of a Qualifying Termination that occurs other than during the Change in Control Period, Executive will receive the following payments and benefits from the Company, subject to the requirements of this Agreement:
- (i) <u>Base Compensation Severance</u>. A single, lump sum, cash payment equal to one-hundred percent (100%) of Executive's Annual Base Compensation.
- (ii) <u>Bonus Severance.</u> A single, lump sum, cash payment equal to the product of (i) the Executive's Target Bonus, if any, that the Executive would have earned for the entire fiscal year in which the Qualifying Termination occurs; and (ii) a fraction, the numerator of which is the number of days the Executive was employed by the Company during the fiscal year in which the Qualifying Termination occurs and the denominator of which is the number of days in such fiscal year.
- (iii) COBRA Severance. Subject to Executive timely electing continuation coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA") and further subject to Section 5(c), Executive will receive Company-paid group health, dental and vision coverage for Executive and any of Executive's eligible dependents, as applicable (the "COBRA Severance"), following the Qualifying Termination until the earliest of: (A) twelve (12) months following the date of the Qualifying Termination, (B) the date on which Executive and Executive's eligible dependents (as applicable) become covered under similar plans, or (C) the expiration of Executive's eligible dependents', as applicable) eligibility for continuation coverage under COBRA.
- (b) <u>Qualifying Termination During the Change in Control Period</u>. In the event of a Qualifying Termination that occurs during the Change in Control Period, Executive will receive the following payments and benefits from the Company, subject to the requirements of this Agreement:
- (i) Base Compensation Severance. A single, lump sum, cash payment equal to one-hundred and fifty percent (150%) of Executive's Annual Base Compensation.
- (ii) <u>COBRA Severance</u>. Subject to Executive timely electing continuation coverage under COBRA and further subject to Section 5(c), Executive will receive COBRA Severance until the earliest of: (A) eighteen (18) months following the date of the Qualifying Termination, (B) the date on which Executive and Executive's eligible dependents (as applicable) become covered under similar plans, or (C) the expiration of Executive's (and any of Executive's eligible dependents, as applicable) eligibility for continuation coverage under COBRA.
- (iii) <u>Vesting Acceleration of Service-based Equity Awards</u>. Notwithstanding the terms of the Company equity plan or plans under which the Executive's Awards are granted or any applicable award agreements, vesting acceleration of one hundred percent (100%) of any Equity Awards that are outstanding and unvested as of the date of the Qualifying Termination.
- (c) <u>Termination Other Than a Qualifying Termination</u>. If the termination of Executive's employment does not constitute a Qualifying Termination, then Executive will not be entitled to receive any severance or other benefits in connection with such termination except for those, if any, as may then be established under the Company's then existing severance and benefits plans or programs.
- (d) Non-duplication of Payment or Benefits Notwithstanding any provision of this Agreement to the contrary, if Executive is entitled to any cash severance, continued health coverage benefits, vesting acceleration of any Awards, or other severance or separation benefits similar to those provided under this Agreement, by operation of applicable law or under a plan, policy, contract, or arrangement sponsored by or to which the Company is a party other than this Agreement ("Other Benefits"), then the corresponding severance payments and benefits under this Agreement will be reduced by the amount of Other Benefits paid or provided to Executive.
- (e) <u>Death of Executive</u>. In the event of Executive's death before all payments or benefits Executive is entitled to receive under this Agreement have been provided, the unpaid amounts will be provided to Executive's designated beneficiary, if living, or otherwise to Executive's personal representative in accordance with the terms of this Agreement.

-2

4. <u>Accrued Compensation</u>. On any termination of Executive's employment with the Company, Executive will be entitled to receive all accrued but unpaid vacation, expense reimbursements, wages, and other benefits due to Executive under any Company-provided plans, policies, and arrangements.

5. Conditions to Receipt of Severance.

(a) <u>Separation Agreement and Release of Claims</u>. Executive's receipt of any severance payments or benefits upon a Qualifying Termination under Section 3 is subject to Executive signing and not revoking the Company's then standard separation agreement and release of claims with the Company (the "<u>Release</u>"), which must become effective and irrevocable no later than the sixtieth (60th) day following the date of the Qualifying Termination (the "<u>Release Deadline Date</u>"). If the Release does not

become effective and irrevocable by the Release Deadline Date, Executive will forfeit any right to severance payments or benefits under Section 3.

(b) <u>Payment Timing</u>. Any lump sum cash severance payments under Section 3 relating to base compensation severance and any bonus severance will be provided to Executive on the first regularly scheduled payroll date of the Company following the date the Release becomes effective and irrevocable, subject to any delay required by Section 5(d) below. Any Equity Awards that are restricted stock units, performance shares, performance units, and/or similar full value awards ("<u>Full Value Awards</u>") that accelerate vesting under Section 3(b)(iii) will be settled, subject to any delay required by Section 5(d) below (or the terms of the Full Value Award agreement or other Company plan, policy, or arrangement governing the settlement timing of the Full Value Award to the extent such terms specifically require any such delay in order to comply with the requirements of Section 409A, as applicable), on a date within ten (10) days following the date the Release becomes effective and irrevocable.

(c) COBRA Severance Limitations. If the Company determines in its sole discretion that it cannot provide the COBRA Severance without potentially violating, or being subject to an excise tax under, applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then in lieu of such COBRA Severance, subject to any delay required by Section 5(d) below and except as provided by the last sentence of this Section 5(c), the Company will provide to Executive a taxable monthly payment payable on the last day of a given month (provided that no such payments will be made prior to the effectiveness of the Release, and any such payments delayed as a result will be paid, subject to any delay required by Section 5(d) below, in a lump sum on the first regularly scheduled payroll date of the Company following the date the Release becomes effective and irrevocable), in an amount equal to the monthly COBRA premium that Executive would be required to pay to continue Executive's group health coverage in effect on the date of the Qualifying Termination (which amount will be based on the premium rates applicable for the first month of COBRA Severance for Executive and any eligible dependents of Executive) (each, a "COBRA Replacement Payment"), which COBRA Replacement Payments will be made regardless of whether Executive elects COBRA continuation coverage and will end on the earlier of (i) the date upon which Executive obtains other employment, or (ii) the date the Company has paid an amount totaling the number of COBRA Replacement Payments equal to the number of months in the applicable COBRA Severance period set forth in clause (A) of Section 3(a)(iii) or Section 3(b)(ii), as applicable. For the avoidance of doubt, the COBRA Replacement Payments may be used for any purpose, including, but not limited to continuation coverage under COBRA, and will be subject to any applicable withholdings. Notwithstanding anything to the contrary under this Agreement, if the Company determines in its sole discretion at

-3-

(d) Section 409A. The Company intends that all payments and benefits provided under this Agreement or otherwise are exempt from, or comply with, the requirements of Section 409A so that none of the payments or benefits will be subject to the additional tax imposed under Section 409A, and any ambiguities and ambiguous terms in this Agreement will be interpreted in accordance with this intent. No payments or benefits to be provided to Executive, if any, under this Agreement or otherwise, when considered together with any other severance payments or separation benefits that are considered deferred compensation under Section 409A (together, the "Deferred Payments") will be paid or otherwise provided until Executive has a "separation from service" within the meaning of Section 409A. To the extent required to be exempt from or comply with Section 409A, references to the termination of Executive's employment or similar phrases used in this Agreement will mean Executive's "separation from service" within the meaning of Section 409A.

(i) Any payments or benefits paid or provided under this Agreement that satisfy the requirements of the "short-term deferral" rule under Treasury Regulations Section 1.409A-1(b)(4), or that qualify as payments made as a result of an involuntary separation from service under Treasury Regulations Section 1.409A-1(b)(9)(iii) that is within the limit set forth thereunder, will not constitute Deferred Payments for purposes of this Section 5(d).

(ii) Notwithstanding any provisions to the contrary in this Agreement, if Executive is a "specified employee" within the meaning of Section 409A at the time of Executive's separation from service (other than due to death), then any payments or benefits under this Agreement that constitute Deferred Payments payable within the first six (6) months after Executive's separation from service; instead will be payable on the date six (6) months and one (1) day after Executive's separation from service; provided that in the event of Executive's death within such six (6) month period, any payments delayed by this subsection (ii) will be paid to Executive in a lump sum as soon as administratively practicable after the date of Executive's death. To the extent that Executive is not a specified employee but Executive's Qualifying Termination occurs at a time during the year whereby the Release Deadline Date will occur in the year immediately following the year in which the Qualifying Termination occurs, then any payments or benefits under this Agreement that constitute Deferred Payments that otherwise would be payable prior to the Release Deadline Date instead will be paid on the first regularly scheduled payroll date of the Company following the Release Deadline Date.

(iii) The Company reserves the right to amend this Agreement as it considers necessary or advisable, in its sole discretion and without the consent of Executive or any other individual, to comply with any provision required to avoid the imposition of the additional tax imposed under Section 409A or to otherwise avoid income recognition under Section 409A prior to the actual payment of any benefits or imposition of any additional tax. Each payment, installment, and benefit payable under this Agreement is intended to constitute a separate payment for purposes of Treasury Regulations Section 1.409A-2(b)(2). In no event will Executive have any discretion to choose Executive's taxable year in which any payments or benefits are provided under this Agreement. In no event will the Company or any parent, subsidiary or other affiliate of the Company have any responsibility, liability or obligation to reimburse, indemnify or hold harmless Executive for any taxes, penalties or interest that may be imposed, or other costs that may be incurred, as a result of Section 409A.

-4-

6. Limitation on Payments.

(a) Reduction of Severance Benefits. If any payment or benefit that Executive would receive from the Company or any other party whether in connection with the provisions in this Agreement or otherwise (the "Payments") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then the Payments will be either delivered in full, or delivered as to such lesser extent that would result in no portion of the Payments being subject to the Excise Tax, whichever of the foregoing amounts, taking into account the applicable federal, state and local income taxes and the Excise Tax, results in Executive's receipt, on an after-tax basis, of the greatest amount of Payments, notwithstanding that all or some of the Payments may be subject to the Excise Tax. If a reduction in Payments is made in accordance with the immediately preceding sentence, the reduction will occur, with respect to the Payments considered parachute payments within the meaning of Code Section 280G, in the following order: (A) reduction of cash payments in reverse chronological order (that is, the cash payment owed on the latest date following the occurrence of the event triggering the Excise Tax will be the first cash payment to be reduced); (B) cancellation of equity awards that were granted "contingent on a change in ownership or control" within the meaning of Section 280G of the Code in the reverse order of date of grant of the equity awards (that is, the most recently granted equity awards will be cancelled first); (C) reduction of the accelerated vesting of equity awards in the reverse order of date of grant of the equity awards (that is, the vesting of the most recently granted equity awards will be cancelled first); and (D) reduction of employee benefits in reverse chronological order (that is, the benefit owed on the latest date following the occurrence of the event triggering the Excise Tax will be the

(b) <u>Determination of Excise Tax Liability</u>. Unless the Company and Executive otherwise agree in writing, any determinations required under this Section 6 will be made in writing by a nationally recognized accounting or valuation firm (the "<u>Firm</u>") selected by the Company, whose determinations will be conclusive and binding upon Executive and the Company for all purposes. For purposes of making the calculations required by this Section 6, the Firm may make reasonable assumptions and approximations concerning applicable taxes and may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. The Company and Executive will furnish to the Firm such information and documents as the Firm reasonably may request in order to make determinations under this Section 6. The Company will bear the costs and make all payments required to be made to the Firm for the Firm's services that are rendered in connection with any calculations contemplated

- 7. <u>Definitions</u>. The following terms referred to in this Agreement will have the following meanings:
- (a) "Annual Base Compensation" means Executive's annual base salary in effect immediately prior to Executive's Qualifying Termination (or, if the termination is due to a resignation for Good Reason based on a material reduction in Executive's annual base salary, then Executive's annual base salary in effect immediately prior to the reduction) or, if Executive's Qualifying Termination occurs during the Change in Control Period and the amount is greater, Executive's annual base salary in effect immediately prior to the Change in Control.
 - (b) "Award" means stock options and other equity awards covering shares of Company common stock granted to Executive.
 - (c) "Board" means the Company's Board of Directors.
- (d) "Cause" means Executive's: (i) dishonesty of a material nature; (ii) theft or embezzlement of Company funds or assets; (iii) being convicted of, or guilty plea or no contest plea to, a felony charge or any misdemeanor involving moral turpitude, or the entry of a consent decree with any governmental body; (iv) noncompliance in any material respect with any U.S. or non-U.S. laws or regulations; (v) violation of any express direction or any rule, regulation or policy established by the Company or the Board; (vi) material breach of this Agreement or the Confidentiality Agreement; (vii) breach of any fiduciary duty to the Company; (viii) gross incompetence, neglect, or misconduct in the performance of Executive's duties; or (ix) repeated failure to perform Executive's duties and responsibilities for the Company or follow the reasonable and lawful instructions of the Company.
 - (e) "Change in Control" means the first occurrence of any of the following events on or after the Effective Date:
- (i) Change in Ownership of the Company. A change in the ownership of the Company which occurs on the date that any one person, or more than one person acting as a group ("Person"), acquires ownership of the stock of the Company that, together with the stock held by such Person, constitutes more than fifty percent (50%) of the total voting power of the stock of the Company; provided, however, that for purposes of this subsection, the acquisition of additional stock by any one Person, who is considered to own more than fifty percent (50%) of the total voting power of the stock of the Company will not be considered a Change in Control; provided, further, that any change in the ownership of the stock of the Company as a result of a private financing of the Company that is approved by the Board also will not be considered a Change in Control. Further, if the stockholders of the Company immediately before such change in ownership continue to retain immediately after the change in ownership, in substantially the same proportions as their ownership of shares of the Company's voting stock immediately prior to the change in ownership, direct or indirect beneficial ownership of fifty percent (50%) or more of the total voting power of the stock of the Company or of the ultimate parent entity of the Company, such event shall not be considered a Change in Control under this subsection (i). For this purpose, indirect beneficial ownership shall include, without limitation, an interest resulting from ownership of the voting securities of one or more corporations or other business entities which own the Company, as the case may be, either directly or through one or more subsidiary corporations or other business entities; or

-6-

(ii) Change in Effective Control of the Company. If the Company has a class of securities registered pursuant to Section 12 of the U.S. Securities Exchange Act of 1934, as amended, a change in the effective control of the Company which occurs on the date that a majority of members of the Board is replaced during any twelve (12) month period by Directors whose appointment or election is not endorsed by a majority of the members of the Board prior to the date of the appointment or election. For purposes of this subsection (ii), if any Person is considered to be in effective control of the Company, the acquisition of additional control of the Company by the same Person will not be considered a Change in Control; or

(iii) Change in Ownership of a Substantial Portion of the Company's Assets. A change in the ownership of a substantial portion of the Company's assets which occurs on the date that any Person acquires (or has acquired during the twelve (12) month period ending on the date of the most recent acquisition by such Person or Persons) assets from the Company that have a total gross fair market value equal to or more than fifty percent (50%) of the total gross fair market value of all of the assets of the Company immediately prior to such acquisition or acquisitions; provided, however, that for purposes of this subsection (iii), the following will not constitute a change in the ownership of a substantial portion of the Company's assets: (A) a transfer to an entity that is controlled by the Company's stockholders immediately after the transfer, or (B) a transfer of assets by the Company to: (1) a stockholder of the Company (immediately before the asset transfer) in exchange for or with respect to the Company's stock, (2) an entity, fifty percent (50%) or more of the total value or voting power of which is owned, directly or indirectly, by the Company, (3) a Person, that owns, directly or indirectly, fifty percent (50%) or more of the total value or voting power of all the outstanding stock of the Company, or (4) an entity, at least fifty percent (50%) of the total value or voting power of which is owned, directly or indirectly, by a Person described in this subsection (iii)(B)(3). For purposes of this subsection (iii), gross fair market value means the value of the assets of the Company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets.

For purposes of this definition, persons will be considered to be acting as a group if they are owners of a corporation that enters into a merger, consolidation, purchase or acquisition of stock, or similar business transaction with the Company.

Notwithstanding the foregoing, a transaction will not be deemed a Change in Control unless the transaction qualifies as a change in control event within the meaning of Section 409A. Further and for the avoidance of doubt, a transaction will not constitute a Change in Control if: (x) its sole purpose is to change the jurisdiction of the Company's incorporation, or (y) its sole purpose is to create a holding company that will be owned in substantially the same proportions by the persons who held the Company's securities immediately before such transaction.

-7-

- (f) "Change in Control Period" means the period beginning on the date of a Change in Control and ending on (and inclusive of) the date that is the one (1) year anniversary of a Change in Control.
- (g) "Code" means the Internal Revenue Code of 1986, as amended. Reference to a specific section of the Code or regulation thereunder will include such section or regulation, any valid regulation promulgated under such section, and any comparable provision of any future legislation or regulation amending, supplementing or superseding such section or regulation.
- (h) "Confidentiality Agreement" means Executive's Employee Confidentiality, Inventions and Non-Interference Agreement entered into with the Company dated January 1, 2016.
 - (i) "Director" means a member of the Board.
 - (j) "Disability" means total and permanent disability as defined in Code Section 22(e)(3).

- (k) "Effective Date" means November 11, 2021.
- (1) "Equity Awards" means Awards that, as of the date of the Qualifying Termination, are held by Executive and subject to continued service-based vesting criteria, but not subject to the achievement of any performance-based or other similar vesting criteria.
- (m) "Good Reason" means Executive's termination of Executive's employment with the Company within thirty (30) days following the end of the Company's Cure Period (as defined below) as a result of the occurrence of any of the following without Executive's written consent: (i) a material diminution in Executive's annual base salary; (ii) the assignment to Executive of duties that are materially inconsistent with Executive's duties that results in a material diminution of Executive's duties with the Company in effect immediately prior to such assignment; (iii) a material diminution in Executive's authority, responsibilities, or job title; (iv) a material change in the location of Executive's primary place of work to a location more than thirty (30) miles from Executive's primary place of work immediately prior to such change and that is further from Executive's residence; provided, however, that Executive must provide written notice to the Company of the condition that could constitute a "Good Reason" event within sixty (60) days following the initial existence of such condition and such condition must not have been remedied by the Company within thirty (30) days (the "Cure Period") of such written notice. To the extent Executive's primary work location is Executive's residence due to a shelter-in-place order or similar work-from-home arrangement that applies to Executive, Executive's primary place of work, from which a change in location under the foregoing clause (iv) will be measured, will be considered the Company's office location where Executive's employment with the Company primarily was based immediately prior to the commencement of such shelter-in-place order or similar work-from-home arrangement.
- (n) "Qualifying Termination" means a termination of Executive's employment with the Company either (i) by the Company without Cause and other than due to Executive's death or Disability, or (ii) by Executive for Good Reason.

-8-

- (o) "Section 409A" means Code Section 409A and the Treasury Regulations and guidance thereunder, and any applicable state law equivalent, as each may be promulgated, amended or modified from time to time.
- (p) "Target Bonus" means Executive's annual (or annualized, if applicable) target bonus in effect immediately prior to Executive's Qualifying Termination or, if Executive's Qualifying Termination occurs during the Change in Control Period and the amount is greater, Executive's annual (or annualized, if applicable) target bonus in effect immediately prior to the Change in Control.
- 8. Successors. This Agreement will be binding upon and inure to the benefit of (a) the heirs, executors, and legal representatives of Executive upon Executive's death, and (b) any successor of the Company. Any such successor of the Company will be deemed substituted for the Company under the terms of this Agreement for all purposes. For this purpose, "successor" means any person, firm, corporation, or other business entity which at any time, whether by purchase, merger, or otherwise, directly or indirectly acquires all or substantially all of the assets or business of the Company. None of the rights of Executive to receive any form of compensation payable pursuant to this Agreement may be assigned or transferred except by will or the laws of descent and distribution. Any other attempted assignment, transfer, conveyance, or other disposition of Executive's right to compensation or other benefits will be null and void.

9. Notice.

(a) <u>General</u>. All notices and other communications required or permitted under this Agreement will be in writing and will be effectively given (i) upon actual delivery to the party to be notified, (ii) upon transmission by email, (iii) twenty-four (24) hours after confirmed facsimile transmission, (iv) one (1) business day after deposit with a recognized overnight courier, or (v) three (3) business days after deposit with the U.S. Postal Service by first class certified or registered mail, return receipt requested, postage prepaid, addressed: (A) if to Executive, at the address Executive will have most recently furnished to the Company in writing, (B) if to the Company, at the following address:

RenovoRx, Inc. 4546 El Camino Real, Suite B1 Los Altos, California 94022 Attention: Chairman of the Board

(b) Notice of Termination. Any termination of Executive's employment by the Company for Cause will be communicated by a notice of termination of Executive's employment to Executive, and any termination by Executive for Good Reason will be communicated by a notice of termination to the Company, in each case given in accordance with Section 9(a). The notice will indicate the specific termination provision in this Agreement relied upon, will set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination under the provision so indicated, and will specify the termination date (which will be not more than thirty (30) days after the later of (i) the giving of the notice or (ii) the end of any applicable cure period).

-9-

10. Resignation. The termination of Executive's employment for any reason also will constitute, without any further required action by Executive, Executive's voluntary resignation from all officer and/or director positions held at the Company or any of its subsidiaries or affiliates, and at the Board's request, Executive will execute any documents reasonably necessary to reflect the resignations.

11. Miscellaneous Provisions.

- (a) No Duty to Mitigate. Executive will not be required to mitigate the amount of any payment contemplated by this Agreement, nor will any payment be reduced by any earnings that Executive may receive from any other source except as specified in Sections 3(d), 5(d) and 6.
- (b) Waiver; Amendment. No provision of this Agreement will be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by an authorized officer of the Company (other than Executive) and by Executive. No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other party will be considered a waiver of any other condition or provision or of the same condition or provision at another time.
 - (c) Headings. Headings are provided herein for convenience only, and will not serve as a basis for interpretation or construction of this Agreement.
- (d) Entire Agreement. This Agreement, together with the Confidentiality Agreement, Executive's offer letter with the Company dated November 11, 2021, and the Company's 2021 Omnibus Equity Incentive Plan and award agreements thereunder governing Executive's Awards, constitutes the entire agreement of the parties and supersedes in their entirety all prior representations, understandings, undertakings or agreements (whether oral or written and whether expressed or implied) of the parties with respect to the subject matter of this Agreement, including without limitation, Executive's offer letter, as amended, with the Company originally dated December 29, 2015.
- (e) Governing Law. This Agreement will be governed by the laws of the State of California but without regard to the conflict of law provision. To the extent that any lawsuit is permitted with respect to any provisions under this Agreement, Executive hereby expressly consents to the personal and exclusive jurisdiction and venue of the state and federal courts located in the State of California for any lawsuit filed against Executive by the Company.

(f) Severability. If any provision of this Agreement is or becomes or is deemed to be invalid, illegal, or unenforceable for any reason, such invalid	dity,
illegality, or unenforceability will not affect the remaining parts of this Agreement, and this Agreement will be construed and enforced as if the invalid, illegal, or unenforces	able
provision had not been included.	

(g) Withholding. The Company (and any parent, subsidiary or other affiliate of the Company, as applicable) will have the right and authority to deduct from any payments or benefits all applicable federal, state, local, and/or non-U.S. taxes or other required withholdings and payroll deductions ("Withholdings"). Prior to the payment of any amounts or provision of any benefits under this Agreement, the Company (and any parent, subsidiary or other affiliate of the Company, as applicable) is permitted to deduct or withhold, or require Executive to remit to the Company, an amount sufficient to satisfy any applicable Withholdings with respect to such payments and benefits. Neither the Company nor any parent, subsidiary or other affiliate of the Company will have any responsibility, liability or obligation to pay Executive's taxes arising from or relating to any payments or benefits under this Agreement.

(h) Counterparts. This Agreement may be executed in counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

[Signature Page Follows]

-10-

By its signature below, each of the parties signifies its acceptance of the terms of this Agreement, in the case of the Company by its duly authorized officer.

COMPANY

By: /s/ K. Angela Macfarlane
K. Angela Macfarlane
Title: Director

Date: November 11, 2021

Shaun R. Bagai

Date: November 11, 2021

-11-

RENOVORX, INC.

CHANGE IN CONTROL AND SEVERANCE AGREEMENT

This Change in Control and Severance Agreement (the "<u>Agreement</u>") is made by and between RenovoRx, Inc., a Delaware corporation (the "<u>Company</u>"), and Ramtin Agah ("<u>Service Provider</u>"), effective as of the Effective Date, as defined in Section 7 below.

This Agreement provides certain protections to Service Provider in connection with an involuntary termination of Service Provider's consulting services with the Company under the circumstances described in this Agreement, including in connection with a change in control of the Company. Certain capitalized terms used in this Agreement are defined in Section 7 below.

The Company and Service Provider agree as follows:

- 1. <u>Term of Agreement</u>. This Agreement will continue indefinitely until terminated by written consent of the parties hereto, or if earlier, upon the date that all of the obligations of the parties hereto with respect to this Agreement have been satisfied.
- 2. Service. The Company and Service Provider acknowledge that Service Provider's consulting services to the Company are and will continue to be governed by the Consulting Agreement originally entered into between Service Provider and the Company as of January 1, 2018, as such agreement has been or may be amended from time to time (the "Consulting Agreement"). The Company and Service Provider acknowledge that if Service Provider becomes an employee of the Company, such employment services will be at-will, as defined under applicable law. No payments, benefits, or provisions under this Agreement will confer upon Service Provider any right to continue Service Provider's services with the Company, nor will they interfere with or limit in any way the right of the Company or Service Provider to terminate such relationship at any time, with or without cause, to the extent permitted by applicable laws and the terms of the Consulting Agreement, if then in place.

3. Severance Benefits.

- (a) Qualifying Termination Outside of the Change in Control Period. In the event of a Qualifying Termination that occurs other than during the Change in Control Period, Service Provider will receive the following payments and benefits from the Company, subject to the requirements of this Agreement:
- (i) <u>Base Compensation Severance</u>. A single, lump sum, cash payment equal to fifty percent (50%) of Service Provider's Annual Base Compensation.
- (ii) <u>Bonus Severance</u>. A single, lump sum, cash payment equal to the product of (i) the Service Provider's Target Bonus, if any, that the Service Provider would have earned for the entire fiscal year in which the Qualifying Termination occurs; and (ii) a fraction, the numerator of which is the number of days the Service Provider was in Service during the fiscal year in which the Qualifying Termination occurs and the denominator of which is the number of days in such fiscal year.
- (iii) COBRA Severance. If Service Provider's Service as of immediately prior to the Qualifying Termination was as an employee, then, subject to Service Provider timely electing continuation coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA") and further subject to Section 5(c), Service Provider will receive Company-paid group health, dental and vision coverage for Service Provider and any of Service Provider's eligible dependents, as applicable (the "COBRA Severance"), following the Qualifying Termination until the earliest of: (A) six (6) months following the date of the Qualifying Termination, (B) the date on which Service Provider and Service Provider's eligible dependents (as applicable) become covered under similar plans, or (C) the expiration of Service Provider's (and any of Service Provider's eligible dependents', as applicable) eligibility for continuation coverage under COBRA.
- (b) <u>Qualifying Termination During the Change in Control Period</u>. In the event of a Qualifying Termination that occurs during the Change in Control Period, Service Provider will receive the following payments and benefits from the Company, subject to the requirements of this Agreement:
- (i) <u>Base Compensation Severance</u>. A single, lump sum, cash payment equal to one-hundred percent (100%) of Service Provider's Annual Base Compensation.
- (ii) <u>COBRA Severance</u>. If Service Provider's Service as of immediately prior to the Qualifying Termination was as an employee, then, subject to Service Provider timely electing continuation coverage under COBRA and further subject to Section 5(c), Service Provider will receive COBRA Severance until the earliest of: (A) twelve (12) months following the date of the Qualifying Termination, (B) the date on which Service Provider and Service Provider's eligible dependents (as applicable) become covered under similar plans, or (C) the expiration of Service Provider's (and any of Service Provider's eligible dependents, as applicable) eligibility for continuation coverage under COBRA.
- (iii) <u>Vesting Acceleration of Service-based Equity Awards</u>. Notwithstanding the terms of the Company equity plan or plans under which the Service Provider's Awards are granted or any applicable award agreements, vesting acceleration of one hundred percent (100%) of any Equity Awards that are outstanding and unvested as of the date of the Qualifying Termination.
- (c) <u>Termination Other Than a Qualifying Termination</u>. If the termination of Service Provider's Service does not constitute a Qualifying Termination, then Service Provider will not be entitled to receive any severance or other benefits in connection with such termination except for those, if any, as may then be established under the Company's then existing severance and benefits plans or programs.
- (d) Non-duplication of Payment or Benefits Notwithstanding any provision of this Agreement to the contrary, if Service Provider is entitled to any cash severance, continued health coverage benefits, vesting acceleration of any Awards, or other severance or separation benefits similar to those provided under this Agreement, by operation of applicable law or under a plan, policy, contract, or arrangement sponsored by or to which the Company is a party other than this Agreement ("Other Benefits"), then the corresponding severance payments and benefits under this Agreement will be reduced by the amount of Other Benefits paid or provided to Service Provider.

_1.

- (e) <u>Death of Service Provider</u>. In the event of Service Provider's death before all payments or benefits Service Provider is entitled to receive under this Agreement have been provided, the unpaid amounts will be provided to Service Provider's designated beneficiary, if living, or otherwise to Service Provider's personal representative in accordance with the terms of this Agreement.
- 4. <u>Accrued Compensation</u>. On any termination of Service Provider's Service, Service Provider will be entitled to receive all accrued but unpaid consulting fee and expense reimbursements, and, if Service Provider's Service as of immediately prior to the Qualifying Termination was as an employee, all accrued but unpaid wages, and other benefits due to Service Provider under any Company-provided plans, policies, and arrangements.

5. Conditions to Receipt of Severance.

(a) <u>Separation Agreement and Release of Claims</u>. Service Provider's receipt of any severance payments or benefits upon a Qualifying Termination under Section 3 is subject to Service Provider signing and not revoking the Company's then standard separation agreement and release of claims with the Company (the "<u>Release</u>"), which must become effective and irrevocable no later than the sixtieth (60th) day following the date of the Qualifying Termination (the "<u>Release Deadline Date</u>"). If the Release does not become effective and irrevocable by the Release Deadline Date, Service Provider will forfeit any right to severance payments or benefits under Section 3.

(b) <u>Payment Timing</u>. Any lump sum cash severance payments under Section 3 relating to base compensation severance and any bonus severance will be provided to Service Provider on the first regularly scheduled payroll date of the Company following the date the Release becomes effective and irrevocable, subject to any delay required by Section 5(d) below. Any Equity Awards that are restricted stock units, performance shares, performance units, and/or similar full value awards ("<u>Full Value Awards</u>") that accelerate vesting under Section 3(b)(iii) will be settled, subject to any delay required by Section 5(d) below (or the terms of the Full Value Award agreement or other Company plan, policy, or arrangement governing the settlement timing of the Full Value Award to the extent such terms specifically require any such delay in order to comply with the requirements of Section 409A, as applicable), on a date within ten (10) days following the date the Release becomes effective and irrevocable.

-2

(c) COBRA Severance Limitations. If the Company determines in its sole discretion that it cannot provide the COBRA Severance, if any is otherwise due under the terms of Section 3, without potentially violating, or being subject to an excise tax under, applicable law (including, without limitation, Section 2716 of the Public Health Service Act), then in lieu of such COBRA Severance, subject to any delay required by Section 5(d) below and except as provided by the last sentence of this Section 5(c), the Company will provide to Service Provider a taxable monthly payment payable on the last day of a given month (provided that no such payments will be made prior to the effectiveness of the Release, and any such payments delayed as a result will be paid, subject to any delay required by Section 5(d) below, in a lump sum on the first regularly scheduled payroll date of the Company following the date the Release becomes effective and irrevocable), in an amount equal to the monthly COBRA premium that Service Provider would be required to pay to continue Service Provider's group health coverage in effect on the date of the Qualifying Termination (which amount will be based on the premium rates applicable for the first month of COBRA Severance for Service Provider and any eligible dependents of Service Provider) (each, a "COBRA Replacement Payment"), which COBRA Replacement Payments will be made regardless of whether Service Provider elects COBRA continuation coverage and will end on the earlier of (i) the date upon which Service Provider obtains other employment (excluding any employment outside of the Company that is in effect as of the Effective Date), or (ii) the date the Company has paid an amount totaling the number of COBRA Replacement Payments equal to the number of months in the applicable COBRA Severance period set forth in clause (A) of Section 3(a)(iii) or Section 3(b)(ii), as applicable. For the avoidance of doubt, the COBRA Replacement Payments may be used for any purpose, including, but not limited to conti

(d) Section 409A. The Company intends that all payments and benefits provided under this Agreement or otherwise are exempt from, or comply with, the requirements of Section 409A so that none of the payments or benefits will be subject to the additional tax imposed under Section 409A, and any ambiguities and ambiguous terms in this Agreement will be interpreted in accordance with this intent. No payments or benefits to be provided to Service Provider, if any, under this Agreement or otherwise, when considered together with any other severance payments or separation benefits that are considered deferred compensation under Section 409A (together, the "Deferred Payments") will be paid or otherwise provided until Service Provider has a "separation from service" within the meaning of Section 409A. To the extent required to be exempt from or comply with Section 409A, references to the termination of Service Provider's Service, termination of Service Provider's consulting services, termination of Service Provider's employment or similar phrases used in this Agreement will mean Service Provider's "separation from service" within the meaning of Section 409A.

(i) Any payments or benefits paid or provided under this Agreement that satisfy the requirements of the "short-term deferral" rule under Treasury Regulations Section 1.409A-1(b)(4), or that qualify as payments made as a result of an involuntary separation from service under Treasury Regulations Section 1.409A-1(b)(9)(iii) that is within the limit set forth thereunder, will not constitute Deferred Payments for purposes of this Section 5(d).

-3

(ii) Notwithstanding any provisions to the contrary in this Agreement, if Service Provider is a "specified employee" within the meaning of Section 409A at the time of Service Provider's separation from service (other than due to death), then any payments or benefits under this Agreement that constitute Deferred Payments payable within the first six (6) months after Service Provider's separation from service instead will be payable on the date six (6) months and one (1) day after Service Provider's separation from service; provided that in the event of Service Provider's death within such six (6) month period, any payments delayed by this subsection (ii) will be paid to Service Provider in a lump sum as soon as administratively practicable after the date of Service Provider's death. To the extent that Service Provider is one a specified employee but Service Provider's Qualifying Termination occurs at a time during the year whereby the Release Deadline Date will occur in the year immediately following the year in which the Qualifying Termination occurs, then any payments or benefits under this Agreement that constitute Deferred Payments that otherwise would be payable prior to the Release Deadline Date instead will be paid on the first regularly scheduled payroll date of the Company following the Release Deadline Date.

(iii) The Company reserves the right to amend this Agreement as it considers necessary or advisable, in its sole discretion and without the consent of Service Provider or any other individual, to comply with any provision required to avoid the imposition of the additional tax imposed under Section 409A or to otherwise avoid income recognition under Section 409A prior to the actual payment of any benefits or imposition of any additional tax. Each payment, installment, and benefit payable under this Agreement is intended to constitute a separate payment for purposes of Treasury Regulations Section 1.409A-2(b)(2). In no event will Service Provider have any discretion to choose Service Provider's taxable year in which any payments or benefits are provided under this Agreement. In no event will the Company or any parent, subsidiary or other affiliate of the Company have any responsibility, liability or obligation to reimburse, indemnify or hold harmless Service Provider for any taxes, penalties or interest that may be imposed, or other costs that may be incurred, as a result of Section 409A.

6. Limitation on Payments.

(a) <u>Reduction of Severance Benefits</u> If any payment or benefit that Service Provider would receive from the Company or any other party whether in connection with the provisions in this Agreement or otherwise (the "<u>Payments</u>") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "<u>Excise Tax</u>"), then the Payments will be either delivered in full, or delivered as to such lesser extent that would result in no portion of the Payments being subject to the Excise Tax, whichever of the foregoing amounts, taking into account the applicable federal, state and local income taxes and the Excise Tax, results in Service Provider's receipt, on an after-tax basis, of the greatest amount of Payments, notwithstanding that all or some of the Payments may be subject to the Excise Tax. If a reduction in Payments is made in accordance with the immediately preceding sentence, the reduction will occur, with respect to the Payments considered parachute payments within the meaning of Code Section 280G, in the following order: (A) reduction of cash payment to be reduced); (B) cancellation of equity awards that were granted "contingent on a change in ownership or control" within the meaning of Section 280G of the Code in the reverse order of date of grant of the equity awards (that is, the most recently granted equity awards will be cancelled first); (C) reduction of the accelerated vesting of equity awards in the reverse order of date of grant of the equity awards (that is, the vesting of the most recently granted equity awards will be cancelled first); and (D) reduction of employee benefits in reverse chronological order (that is, the benefit owed on the latest date following the occurrence of the event triggering the Excise Tax will be the first benefit to be reduced). In no event will Service Provider have any discretion with respect to the ordering of Payment reductions. Service Provider will b

(b) <u>Determination of Excise Tax Liability.</u> Unless the Company and Service Provider otherwise agree in writing, any determinations required under this Section 6 will be made in writing by a nationally recognized accounting or valuation firm (the "<u>Firm</u>") selected by the Company, whose determinations will be conclusive and binding upon Service Provider and the Company for all purposes. For purposes of making the calculations required by this Section 6, the Firm may make reasonable assumptions and approximations concerning applicable taxes and may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. The Company and Service Provider will furnish to the Firm such information and documents as the Firm reasonably may request in order to make determinations under this Section 6. The Company will bear the costs and make all payments required to be made to the Firm for the Firm's services that are rendered in connection with any calculations contemplated by this Section 6. The Company will have no liability to Service Provider for the determinations of the Firm.

7. <u>Definitions</u>. The following terms referred to in this Agreement will have the following meanings:

(a) "Annual Base Compensation" means Service Provider's (x) monthly base consulting fee multiplied by twelve (referred to as the "annual base consulting fee") or, (y) if at the applicable time, Service Provider is an employee of the Company annual base salary, in effect immediately prior to Service Provider's Qualifying Termination (or, if the termination is due to a resignation for Good Reason based on a material reduction in Service Provider's annual base consulting fee or annual base salary, as applicable, then Service Provider's annual base consulting fee or annual base salary, as applicable, in effect immediately prior to the reduction) or, if Service Provider's Qualifying Termination occurs during the Change in Control Period and the amount is greater, Service Provider's annual base consulting fee or annual base salary, as applicable, in effect immediately prior to the Change in Control.

- (b) "Award" means stock options and other equity awards covering shares of Company common stock granted to Service Provider.
- (c) "Board" means the Company's Board of Directors.

(d) "Cause" means Service Provider's: (i) dishonesty of a material nature; (ii) theft or embezzlement of Company funds or assets; (iii) being convicted of, or guilty plea or no contest plea to, a felony charge or any misdemeanor involving moral turpitude, or the entry of a consent decree with any governmental body; (iv) noncompliance in any material respect with any U.S. or non-U.S. laws or regulations; (v) violation of any express direction or any rule, regulation or policy established by the Company or the Board; (vi) material breach of this Agreement or the Consulting Agreement (or any other confidentiality agreement or other agreement with the Company concerning the terms and condition of Service Provider's relationship with the Company); (vii) breach of any fiduciary duty to the Company; (viii) gross incompetence, neglect, or misconduct in the performance of Service Provider's duties; or (ix) repeated failure to perform Service Provider's duties and responsibilities for the Company or follow the reasonable and lawful instructions of the Company.

-5.

(e) "Change in Control" means the first occurrence of any of the following events on or after the Effective Date:

(i) Change in Ownership of the Company. A change in the ownership of the Company which occurs on the date that any one person, or more than one person acting as a group ("Person"), acquires ownership of the stock of the Company that, together with the stock held by such Person, constitutes more than fifty percent (50%) of the total voting power of the stock of the Company; provided, however, that for purposes of this subsection, the acquisition of additional stock by any one Person, who is considered to own more than fifty percent (50%) of the total voting power of the stock of the Company will not be considered a Change in the ownership of the stock of the Company as a result of a private financing of the Company that is approved by the Board also will not be considered a Change in Control. Further, if the stockholders of the Company immediately before such change in ownership continue to retain immediately after the change in ownership, in substantially the same proportions as their ownership of shares of the Company's voting stock immediately prior to the change in ownership, direct or indirect beneficial ownership of fifty percent (50%) or more of the total voting power of the stock of the Company or of the ultimate parent entity of the Company, such event shall not be considered a Change in Control under this subsection (i). For this purpose, indirect beneficial ownership shall include, without limitation, an interest resulting from ownership of the voting securities of one or more corporations or other business entities which own the Company, as the case may be, either directly or through one or more subsidiary corporations or other business entities; or

(ii) Change in Effective Control of the Company. If the Company has a class of securities registered pursuant to Section 12 of the U.S. Securities Exchange Act of 1934, as amended, a change in the effective control of the Company which occurs on the date that a majority of members of the Board is replaced during any twelve (12) month period by Directors whose appointment or election is not endorsed by a majority of the members of the Board prior to the date of the appointment or election. For purposes of this subsection (ii), if any Person is considered to be in effective control of the Company, the acquisition of additional control of the Company by the same Person will not be considered a Change in Control; or

(iii) Change in Ownership of a Substantial Portion of the Company's Assets. A change in the ownership of a substantial portion of the Company's assets which occurs on the date that any Person acquires (or has acquired during the twelve (12) month period ending on the date of the most recent acquisition by such Person or Persons) assets from the Company that have a total gross fair market value equal to or more than fifty percent (50%) of the total gross fair market value of all of the assets of the Company immediately prior to such acquisition or acquisitions; provided, however, that for purposes of this subsection (iii), the following will not constitute a change in the ownership of a substantial portion of the Company's assets: (A) a transfer to an entity that is controlled by the Company's stockholders immediately after the transfer, or (B) a transfer of assets by the Company to: (1) a stockholder of the Company (immediately before the asset transfer) in exchange for or with respect to the Company's stock, (2) an entity, fifty percent (50%) or more of the total value or voting power of which is owned, directly or indirectly, by the Company, (3) a Person, that owns, directly or indirectly, fifty percent (50%) or more of the total value or voting power of all the outstanding stock of the Company, or (4) an entity, at least fifty percent (50%) of the total value or voting power of market value or voting power of this subsection (iii)(B)(3). For purposes of this subsection (iii), gross fair market value means the value of the assets of the Company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets.

-6-

For purposes of this definition, persons will be considered to be acting as a group if they are owners of a corporation that enters into a merger, consolidation, purchase or acquisition of stock, or similar business transaction with the Company.

Notwithstanding the foregoing, a transaction will not be deemed a Change in Control unless the transaction qualifies as a change in control event within the meaning of Section 409A. Further and for the avoidance of doubt, a transaction will not constitute a Change in Control if: (x) its sole purpose is to change the jurisdiction of the Company's incorporation, or (y) its sole purpose is to create a holding company that will be owned in substantially the same proportions by the persons who held the Company's securities immediately before such transaction.

(f) "Change in Control Period" means the period beginning on the date of a Change in Control and ending on (and inclusive of) the date that is the one (1) year anniversary of a Change in Control.

- (g) "Code" means the Internal Revenue Code of 1986, as amended. Reference to a specific section of the Code or regulation thereunder will include such section or regulation, any valid regulation promulgated under such section, and any comparable provision of any future legislation or regulation amending, supplementing or superseding such section or regulation.
 - (h) "Director" means a member of the Board.
 - (i) "Disability" means total and permanent disability as defined in Code Section 22(e)(3).
 - (j) "Effective Date" means November 11, 2021.
- (k) "Equity Awards" means Awards that, as of the date of the Qualifying Termination, are held by Service Provider and subject to continued service-based vesting criteria, but not subject to the achievement of any performance-based or other similar vesting criteria.

-7-

- (l) "Good Reason" means Service Provider's termination of Service Provider's Services within thirty (30) days following the end of the Company's Cure Period (as defined below) as a result of the occurrence of any of the following without Service Provider's written consent: (i) a material diminution in Service Provider's annual base salary or annual base consulting fee, as applicable; (ii) the assignment to Service Provider of duties that are materially inconsistent with Service Provider's duties that results in a material diminution of Service Provider's duties with the Company in effect immediately prior to such assignment; (iii) a material diminution in Service Provider's authority, responsibilities, or job title; (iv) a material change in the location of Service Provider's primary place of work (if Service Provider has been assigned a primary place of work) to a location more than thirty (30) miles from Service Provider's primary place of work immediately prior to such change and that is further from Service Provider's residence; provided, however, that Service Provider must provide written notice to the Company of the condition that could constitute a "Good Reason" event within sixty (60) days following the initial existence of such condition and such condition must not have been remedied by the Company within thirty (30) days (the "Cure Period") of such written notice. To the extent Service Provider's primary work location is Service Provider's residence due to a shelter-in-place order or similar work-from-home arrangement that applies to Service Provider, Service Provider's primary place of work, from which a change in location under the foregoing clause (iv) will be measured, will be considered the Company's office location where Service Provider's Service primarily was based immediately prior to the commencement of such shelter-in-place order or similar work-from-home arrangement.
- (m) "Qualifying Termination" means a termination of Service Provider's Service with the Company either (i) by the Company without Cause and other than due to Service Provider's death or Disability, or (ii) by Service Provider for Good Reason.
- (n) "Section 409A" means Code Section 409A and the Treasury Regulations and guidance thereunder, and any applicable state law equivalent, as each may be promulgated, amended or modified from time to time.
- (o) "Service" means Service Provider's service to the Company or any of its subsidiaries, whether as a consultant or an employee. For the avoidance of doubt, Service will be deemed to continue upon any transfer directly from status as a consultant to the Company or any of its subsidiaries to status as an employee of the Company or any of its subsidiaries, or vice versa.
- (p) "Target Bonus" means Service Provider's annual (or annualized, if applicable) target bonus in effect immediately prior to Service Provider's Qualifying Termination or, if Service Provider's Qualifying Termination occurs during the Change in Control Period and the amount is greater, Service Provider's annual (or annualized, if applicable) target bonus in effect immediately prior to the Change in Control.
- 8. Successors. This Agreement will be binding upon and inure to the benefit of (a) the heirs, executors, and legal representatives of Service Provider upon Service Provider's death, and (b) any successor of the Company. Any such successor of the Company will be deemed substituted for the Company under the terms of this Agreement for all purposes. For this purpose, "successor" means any person, firm, corporation, or other business entity which at any time, whether by purchase, merger, or otherwise, directly or indirectly acquires all or substantially all of the assets or business of the Company. None of the rights of Service Provider to receive any form of compensation payable pursuant to this Agreement may be assigned or transferred except by will or the laws of descent and distribution. Any other attempted assignment, transfer, conveyance, or other disposition of Service Provider's right to compensation or other benefits will be null and void.

-8-

9. Notice.

(a) <u>General</u>. All notices and other communications required or permitted under this Agreement will be in writing and will be effectively given (i) upon actual delivery to the party to be notified, (ii) upon transmission by email, (iii) twenty-four (24) hours after confirmed facsimile transmission, (iv) one (1) business day after deposit with a recognized overnight courier, or (v) three (3) business days after deposit with the U.S. Postal Service by first class certified or registered mail, return receipt requested, postage prepaid, addressed: (A) if to Service Provider, at the address Service Provider will have most recently furnished to the Company in writing, (B) if to the Company, at the following address:

RenovoRx, Inc. 4546 El Camino Real, Suite B1 Los Altos, California 94022 Attention: Chief Executive Officer

- (b) Notice of Termination. Any termination of Service Provider's Service by the Company for Cause will be communicated by a notice of termination of Service Provider's Service to Service to Service Provider, and any termination by Service Provider for Good Reason will be communicated by a notice of termination to the Company, in each case given in accordance with Section 9(a). The notice will indicate the specific termination provision in this Agreement relied upon, will set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination under the provision so indicated, and will specify the termination date (which will be not more than thirty (30) days after the later of (i) the giving of the notice or (ii) the end of any applicable cure period).
- 10. Resignation. The termination of Service Provider's Service for any reason also will constitute, without any further required action by Service Provider, Service Provider's voluntary resignation from all officer and/or director positions held at the Company or any of its subsidiaries or affiliates, and at the Board's request, Service Provider will execute any documents reasonably necessary to reflect the resignations.

11. Miscellaneous Provisions.

- (a) No Duty to Mitigate. Service Provider will not be required to mitigate the amount of any payment contemplated by this Agreement, nor will any payment be reduced by any earnings that Service Provider may receive from any other source except as specified in Sections 3(d), 5(d) and 6.
- (b) Waiver; Amendment. No provision of this Agreement will be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by an authorized officer of the Company (other than Service Provider) and by Service Provider. No waiver by either party of any breach of, or of compliance

with, any condition or provision of this Agreement by the other party will be considered a waiver of any other condition or provision or of the same condition or provision at another time.

- (c) Headings. Headings are provided herein for convenience only, and will not serve as a basis for interpretation or construction of this Agreement.
- (d) Entire Agreement. This Agreement, the Consulting Agreement, and the Company's 2021 Omnibus Equity Incentive Plan and award agreements thereunder governing Service Provider's Awards, constitutes the entire agreement of the parties and supersedes in their entirety all prior representations, understandings, undertakings or agreements (whether oral or written and whether expressed or implied) of the parties with respect to the subject matter of this Agreement.
- (e) Governing Law. This Agreement will be governed by the laws of the State of California but without regard to the conflict of law provision. To the extent that any lawsuit is permitted with respect to any provisions under this Agreement, Service Provider hereby expressly consents to the personal and exclusive jurisdiction and venue of the state and federal courts located in the State of California for any lawsuit filed against Service Provider by the Company.
- (f) Severability. If any provision of this Agreement is or becomes or is deemed to be invalid, illegal, or unenforceable for any reason, such invalidity, illegality, or unenforceability will not affect the remaining parts of this Agreement, and this Agreement will be construed and enforced as if the invalid, illegal, or unenforceable provision had not been included.
- (g) Withholding. The Company (and any parent, subsidiary or other affiliate of the Company, as applicable) will have the right and authority to deduct from any payments or benefits all applicable federal, state, local, and/or non-U.S. taxes or other required withholdings and payroll deductions ("Withholdings"), if any. Prior to the payment of any amounts or provision of any benefits under this Agreement, the Company (and any parent, subsidiary or other affiliate of the Company, as applicable) is permitted to deduct or withhold, or require Service Provider to remit to the Company, an amount sufficient to satisfy any applicable Withholdings with respect to such payments and benefits. Neither the Company nor any parent, subsidiary or other affiliate of the Company will have any responsibility, liability or obligation to pay Service Provider's taxes arising from or relating to any payments or benefits under this Agreement.
- (h) Counterparts. This Agreement may be executed in counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

[Signature Page Follows]

-10-

By its signature below, each of the parties signifies its acceptance of the terms of this Agreement, in the case of the Company by its duly authorized officer.

COMPANY RENOVORX, INC.

SERVICE PROVIDER

/s/ Shaun R. Bagai By:

Shaun R. Bagai

Title: Chief Executive Officer

Date: November 11, 2021

/s/ Ramtin Agah

Ramtin Agah, MD November 11, 2021

Date:

-11-

RENOVORX, INC.

KEY SERVICE PROVIDER INCENTIVE COMPENSATION PLAN

1. <u>Purposes of the Plan</u>. The Plan is intended to increase stockholder value and the success of the Company by motivating Service Providers to (a) perform to the best of their abilities and (b) achieve the Company's objectives.

2. Definitions.

- (a) "Actual Award" means as to any Performance Period, the actual award (if any) payable to a Participant for the Performance Period, subject to the authority of the Administrator (as defined in Section 3) under Section 4(d) to modify the award.
- (b) "Affiliate" means any corporation or other entity (including, but not limited to, partnerships and joint ventures) controlled by the Company.
- (c) "Board" means the Board of Directors of the Company.
- (d) "Bonus Pool" means the pool of funds available for distribution to Participants. Subject to the terms of the Plan, the Administrator establishes the Bonus Pool for each Performance Period.
- (e) "Code" means the U.S. Internal Revenue Code of 1986, as amended. Reference to a specific section of the Code or regulation thereunder will include such section or regulation, any valid regulation promulgated under such section, and any comparable provision of any future legislation or regulation amending, supplementing or superseding such section or regulation.
- (f) "Committee" means a committee appointed by the Board (pursuant to Section 3) to administer the Plan.
- (g) "Company" means RenovoRx, Inc., a Delaware corporation, or any successor thereto.
- (h) "Company Group" means the Company and any Parents, Subsidiaries, and Affiliates.
- (i) "Disability" means a permanent and total disability determined in accordance with uniform and nondiscriminatory standards adopted by the Administrator from time to time.
- (j) "Fiscal Year" means the fiscal year of the Company.
- (k) "Parent" means a "parent corporation," whether now or hereafter existing, as defined in Code Section 424(e).
- (1) "Participant" means as to any Performance Period, a Service Provider who has been selected by the Administrator for participation in the Plan for that Performance Period.
- (m) "<u>Performance Period</u>" means the period of time for the measurement of the performance criteria that must be met to receive an Actual Award, as determined by the Administrator. A Performance Period may be divided into one or more shorter periods if, for example, but not by way of limitation, the Administrator desires to measure some performance criteria over twelve (12) months and other criteria over three (3) months.
- (n) "Plan" means this Key Service Provider Incentive Compensation Plan (including any appendix attached hereto), as may be amended from time to time.
- (o) "Section 409A" means Section 409A of the Code and any proposed or final Treasury Regulations and Internal Revenue Service guidance, compliance programs and other interpretive authority promulgated thereunder, or any state law equivalent, as each may be amended or promulgated, as applicable, from time to time.
- (p) "Service Provider" means any (i) executive or other employee of the Company Group, or (ii) other service provider of the Company Group, in each case whether such individual is so employed or engaged at the time the Plan is adopted or becomes so employed or engaged subsequent to the adoption of the Plan. For clarity, an officer of the Company Group may either be an employee, an independent contractor or other service provider. For avoidance of doubt, for purposes of the Plan, engagement as a service provider to the Company Group may be through a third-party entity (for example, through an agency).
- (q) "Subsidiary" means a "subsidiary corporation," whether now or hereafter existing, as defined in Code Section 424(f), in relation to the Company.
- (r) "Target Award" means the target award, at one hundred percent (100%) of target level performance achievement, payable under the Plan to a Participant for a Performance Period, as determined by the Administrator in accordance with Section 4(b).
- (s) "Tax Withholdings" means any applicable tax, social insurance and social security liability or premium obligations in connection with the awards under the Plan, including without limitation: (i) all federal, state, and local income, employment and any other taxes (including the Participant's U.S. Federal Insurance Contributions Act (FICA) obligation) that are required to be withheld by the Company Group, (ii) the Participant's and, to the extent required by the Company Group, the applicable fringe benefit tax liability of the Company Group associated with an award under the Plan, and (iii) any other applicable taxes or social insurance or social security liabilities or premium the responsibility for which the Participant has, or has agreed to bear, with respect to such award under the Plan.
- (t) "Termination of Service" means a cessation of the employee-employer or independent contractor or other service provider relationship between a Service Provider and the Company Group, including without limitation a termination by resignation, discharge, death, Disability, retirement, or the disaffiliation of a Parent, Subsidiary or Affiliate. For purposes of the Plan, transfer of a Service Provider between any members of the Company Group (for example, between the Company and a Subsidiary) will not be deemed a Termination of Service.

3. Administration of the Plan.

- (a) Administrator. The Plan will be administered by the Board or a Committee (the "Administrator"). To the extent necessary or desirable to satisfy applicable laws, the Committee acting as the Administrator will consist of not less than two (2) members of the Board. The members of any Committee will be appointed from time to time by, and serve at the pleasure of, the Board. The Board may retain the authority to administer the Plan concurrently with a Committee and may revoke the delegation of some or all authority previously delegated. Different Administrators may administer the Plan with respect to different groups of Service Providers. Unless and until the Board otherwise determines, the Board's Compensation Committee will administer the Plan.
- (b) Administrator Authority. It will be the duty of the Administrator to administer the Plan in accordance with the Plan's provisions. The Administrator will have all powers and discretion necessary or appropriate to administer the Plan and to control its operation, including, but not limited to, the power to (i) determine which Service Providers will be granted awards, (ii) prescribe the terms and conditions of awards, (iii) interpret the Plan and the awards, (iv) adopt such procedures and sub-plans as are necessary or appropriate

to permit participation in the Plan by Service Providers who are non-U.S. nationals or employed or engaged outside of the U.S. or to qualify awards for special tax treatment under the laws of jurisdictions other than the U.S., (v) adopt rules for the administration, interpretation and application of the Plan as are consistent therewith, and (vi) interpret, amend or revoke any such rules. Any determinations and decisions made or to be made by the Administrator pursuant to the provisions of the Plan, unless specified otherwise by the Administrator, will be in the Administrator's sole discretion.

- (c) <u>Decisions Binding</u>. All determinations and decisions made by the Administrator and/or any delegate of the Administrator pursuant to the provisions of the Plan will be final, conclusive, and binding on all persons, and will be given the maximum deference permitted by law.
- (d) <u>Delegation by Administrator</u>. The Administrator, on such terms and conditions as it may provide, may delegate all or part of its authority and powers under the Plan to one or more directors and/or officers of the Company. Such delegation may be revoked at any time.
- (e) Indemnification. Each person who is or will have been a member of the Administrator will be indemnified and held harmless by the Company against and from (i) any loss, cost, liability, or expense that may be imposed upon or reasonably incurred by him or her in connection with or resulting from any claim, action, suit, or proceeding to which he or she may be a party or in which he or she may be involved by reason of any action taken or failure to act under the Plan or any award, and (ii) from any and all amounts paid by him or her in settlement thereof, with the Company's approval, or paid by him or her in satisfaction of any judgment in any such claim, action, suit, or proceeding against him or her, provided he or she will give the Company an opportunity, at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf. The foregoing right of indemnification will not be exclusive of any other rights of indemnification to which such persons may be entitled under the Company's Certificate of Incorporation or Bylaws, by contract, as a matter of law, or otherwise, or under any power that the Company may have to indemnify them or hold them harmless.

4. Selection of Participants and Determination of Awards

- (a) <u>Selection of Participants</u>. The Administrator will select the Service Providers who will be Participants for any Performance Period. Participation in the Plan will be on a Performance Period by Performance Period basis. Accordingly, a Service Provider who is a Participant for a given Performance Period in no way is guaranteed or assured of being selected for participation in any subsequent Performance Period or Performance Periods. No Service Provider will have the right to be selected to receive an award under this Plan or, if so selected, to be selected to receive a future award.
- (b) <u>Determination of Target Awards</u>. The Administrator may, in its sole discretion, establish a Target Award for each Participant (which may be expressed as a percentage of a Participant's average annual base salary or annual consulting fee, as applicable, for the Performance Period or a fixed dollar amount or such other amount or based on such other formula or factors as the Administrator determines).
- (c) Bonus Pool. Each Performance Period, the Administrator may, in its discretion, establish a Bonus Pool, which pool may be established before, during or after the applicable Performance Period. Actual Awards will be paid from the Bonus Pool (if a Bonus Pool has been established).
- (d) <u>Discretion to Modify Awards</u>. Notwithstanding any contrary provision of the Plan, the Administrator, at any time, may: (i) increase, reduce or eliminate a Participant's Actual Award, and/or (ii) increase, reduce or eliminate the amount allocated to the Bonus Pool. The Actual Award may be below, at or above the Target Award, as determined by the Administrator. The Administrator may determine the amount of any increase, reduction, or elimination based on such factors as it deems relevant, and will not be required to establish any allocation or weighting with respect to the factors it considers.
- (e) Discretion to Determine Criteria. Notwithstanding any contrary provision of the Plan, the Administrator will determine the performance goals, if any, applicable to any Target Award (or portion thereof) which may include, without limitation, goals related to: attainment of research and development milestones; sales bookings; business divestitures and acquisitions; capital raising; cash flow; cash position; contract awards or backlog; corporate transactions; customer renewals; customer retention rates from an acquired company, subsidiary, business unit or division; earnings (which may include any calculation of earnings, including but not limited to earnings before interest and taxes, earnings before taxes, earnings before interest, taxes, depreciation and amortization and net taxes); earnings per share; expenses; financial milestones; gross margin; growth in stockholder value relative to the moving average of the S&P 500 Index or another index; internal rate of return; leadership development or succession planning; license or research collaboration arrangements; market share; net income; net profit; net sales; new product or business development; new product invention or innovation; number of customers; operating cash flow; operating expenses; operating income; operating margin; overhead or other expense reduction; patents; procurement; product defect measures; product release timelines; productivity; profit; regulatory milestones or regularly-related goals; retained earnings; return on assets; return on capital; return on equity; return on investment; return on sales; revenue; revenue growth; sales results; sales growth; savings; stock price; time to market; total stockholder return; working capital; unadjusted or adjusted actual contract value; unadjusted or adjusted total contract value; and individual objectives such as peer reviews or other subjective or objective criteria. As determined by the Administrator, the performance goals may be based on U.S. generally accepted accounting principles ("GAAP") or non-GAAP results and any actual results may be adjusted by the Administrator for one-time items or unbudgeted or unexpected items and/or payments of Actual Awards under the Plan when determining whether the performance goals have been met. The performance goals may be based on any factors the Administrator determines relevant, including without limitation on an individual, divisional, portfolio, project, business unit, segment or Company-wide basis. Any criteria used may be measured on such basis as the Administrator determines, including without limitation: (i) in absolute terms, (ii) in combination with another performance goal or goals (for example, but not by way of limitation, as a ratio or matrix), (iii) in relative terms (including, but not limited to, results for other periods, passage of time and/or against another company or companies or an index or indices), (iv) on a per-share basis, (v) against the performance of the Company as a whole or a segment of the Company and/or (vi) on a pre-tax or after-tax basis. The performance goals may differ from Participant to Participant and from award to award. Failure to meet the applicable performance goals will result in a failure to earn the Target Award, except as provided in Section 4(d). The Administrator also may determine that a Target Award (or portion thereof) will not have a performance goal associated with it but instead will be granted (if at all) as determined by the Administrator.

5. Payment of Awards.

- (a) Right to Receive Payment. Each Actual Award will be paid solely from the general assets of the Company Group. Nothing in this Plan will be construed to create a trust or to establish or evidence any Participant's claim of any right other than as an unsecured general creditor with respect to any payment to which the Participant may be entitled.
- (b) <u>Timing of Payment.</u> Payment of each Actual Award will be made as soon as practicable after the end of the Performance Period to which the Actual Award relates and after the Actual Award is approved by the Administrator, but in no event after the later of (i) the fifteenth (15th) day of the third (3rd) month of the Company's taxable year immediately following the Company's taxable year in which the Participant's Actual Award first becomes no longer subject to a substantial risk of forfeiture, and (ii) March 15 of the calendar year immediately following the calendar year in which the Participant's Actual Award first becomes no longer subject to a substantial risk of forfeiture. Unless otherwise determined by the Administrator, to earn an Actual Award a Participant must be employed or engaged by the Company Group on the date the Actual Award is paid.
- (c) Form of Payment. Each Actual Award generally will be paid in cash (or its equivalent) in a single lump sum. The Administrator reserves the right to settle an Actual Award with a grant of an equity award with such terms and conditions, including any vesting requirements, as determined by the Administrator.
- (d) Payment in the Event of Death or Disability. If a Termination of Service occurs due to a Participant's death or Disability prior to payment of an Actual Award that the Administrator has determined will be paid for a prior Performance Period, then the Actual Award will be paid to the Participant or the Participant's estate, as the case may be,

subject to the Administrator's discretion to reduce or eliminate any Actual Award otherwise payable.

6. General Provisions.

(a) Tax Matters.

- (i) Section 409A. It is the intent that this Plan be exempt from or comply with the requirements of Section 409A so that none of the payments to be provided hereunder will be subject to the additional tax imposed under Section 409A, and any ambiguities or ambiguous terms will be interpreted to be so exempt or so comply. Each payment under this Plan is intended to constitute a separate payment for purposes of Treasury Regulations Section 1.409A-2(b)(2). In no event will the Company Group have any liability, obligation, or responsibility to reimburse, indemnify or hold harmless any Participant or other Service Provider for any taxes, penalties or interest imposed, or other costs incurred, as a result of Section 409A.
- (ii) Withholding Obligations. The Company Group will have the right and authority to deduct from any Actual Award all applicable Tax Withholdings. Prior to the payment of an Actual Award or such earlier time as any applicable Tax Withholdings are due, the Company Group is permitted to deduct or withhold, or require a Participant to remit to the Company Group, an amount sufficient to satisfy any applicable Tax Withholdings with respect to such Actual Award.
- (b) No Effect on Employment or Service. Neither the Plan nor any award under the Plan will confer upon a Participant any right regarding continuing the Participant's relationship as a Service Provider or other service provider to the Company Group, nor will they interfere with or limit in any way the right of the Company Group or the Participant to terminate such relationship at any time, with or without cause, to the extent permitted by applicable laws.

(c) Forfeiture Events.

- (i) <u>Clawback Policy</u>; <u>Applicable Laws</u>. All awards under the Plan will be subject to reduction, cancellation, forfeiture, or recoupment in accordance with any applicable clawback policy that the Company Group is required to adopt pursuant to the listing standards of any national securities exchange or association on which the Company's securities are listed or as is otherwise required by the Dodd-Frank Wall Street Reform and Consumer Protection Act or other applicable laws. In addition, the Administrator may impose such other clawback, recovery or recoupment provisions with respect to an award under the Plan as the Administrator determines necessary or appropriate, including without limitation a reacquisition right in respect of previously acquired cash, stock, or other property provided with respect to an award. Unless this Section 6(c)(i) is specifically mentioned and waived in a written agreement between a Participant and a member of the Company Group or other document, no recovery of compensation under a clawback policy will give the Participant the right to resign for "good reason" or "constructive termination" (or similar term) under any agreement with a member of the Company Group.
- (ii) Additional Forfeiture Terms. The Administrator may specify when providing for an award under the Plan that the Participant's rights, payments, and benefits with respect to the award will be subject to reduction, cancellation, forfeiture, or recoupment upon the occurrence of specified events, in addition to any otherwise applicable vesting or performance conditions of the award. Such events may include, without limitation, termination of the Participant's status as a Service Provider for "cause" or any act by a Participant, whether before or after the Participant's status as a Service Provider terminates, that would constitute "cause."
- (iii) <u>Accounting Restatements</u>. If the Company is required to prepare an accounting restatement due to the material noncompliance of the Company, as a result of misconduct, with any financial reporting requirement under the securities laws, then any Participant who knowingly or through gross negligence engaged in the misconduct, or who knowingly or through gross negligence failed to prevent the misconduct, and any Participant who is one of the individuals subject to automatic forfeiture under Section 304 of the Sarbanes-Oxley Act of 2002, will reimburse the Company Group the amount of any payment with respect to an award earned or accrued under the Plan during the twelve (12) month period following the first public issuance or filing with the U.S. Securities and Exchange Commission (whichever first occurred) of the financial document embodying such financial reporting requirement.
- (d) <u>Successors</u>. All obligations of the Company under the Plan, with respect to awards under the Plan, will be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation, or otherwise, of all or substantially all of the business or assets of the Company.
- (e) Nontransferability of Awards. No award under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution, and except as provided in Section 5(d). All rights with respect to an award granted to a Participant will be available during his or her lifetime only to the Participant.

7. Amendment, Termination, and Duration

- (a) <u>Amendment, Suspension, or Termination</u>. The Administrator may amend or terminate the Plan, or any part thereof, at any time and for any reason. The amendment, suspension or termination of the Plan will not, without the consent of the Participant, alter or impair any rights or obligations under any Actual Award earned by such Participant. No award may be granted during any period of suspension or after termination of the Plan.
- (b) <u>Duration of Plan</u>. The Plan will commence on the date first adopted by the Board or the Compensation Committee of the Board, and subject to Section 7(a) (regarding the Administrator's right to amend or terminate the Plan), will remain in effect thereafter until terminated.

8. Legal Construction.

- (a) <u>Gender and Number</u>. Unless otherwise indicated by the context, any feminine term used herein also will include the masculine and any masculine term used herein also will include the feminine; the plural will include the singular and the singular will include the plural.
- (b) <u>Severability</u>. If any provision of the Plan is or becomes or is deemed to be invalid, illegal, or unenforceable for any reason in any jurisdiction or as to any Participant, such invalidity, illegality, or unenforceability will not affect the remaining parts of the Plan, and the Plan will be construed and enforced as if the invalid, illegal, or unenforceable provision had not been included.
- (c) Governing Law. The Plan and all awards will be construed in accordance with and governed by the laws of the State of California, but without regard to its conflict of law provisions.
- (d) Bonus Plan. The Plan is intended to be a "bonus program" as defined under U.S. Department of Labor regulations section 2510.3-2(c) and will be construed and administered in accordance with such intention.
- (e) Headings. Headings are provided herein for convenience only, and will not serve as a basis for interpretation or construction of the Plan.
- 9. Compliance with Applicable Laws. Awards under the Plan (including without limitation the granting of such awards) will be subject to all applicable laws, rules and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.

CERTIFICATION PURSUANT TO RULES 13a-14(a) AND 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Shaun R. Bagai, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of RenovoRx, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
- a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- b. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- c. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 15, 2021

By: \(\frac{ss \ Shaun \ R. Bagai}{Chief Executive \ Officer \)

CERTIFICATION PURSUANT TO

RULES 13a-14(a) AND 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934,

AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Christopher J. Lehman, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of RenovoRx, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
- a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- b. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- c. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 15, 2021 By: /s/ Christopher J. Lehman

Chief Financial Officer

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of RenovoRx, Inc. (the "Company") on Form 10-Q for the period ended September 30, 2021 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Shaun R. Bagai, hereby certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- 1) The Report of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

Date: November 15, 2021 By: /s/ Shaun R. Bagai

Chief Executive Officer

CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO

18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of RenovoRx, Inc. (the "Company") on Form 10-Q for the period ended September 30, 2021 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Christopher J. Lehman, hereby certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- 1) The Report of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

Date: November 15, 2021 By: /s/ Christopher J. Lehman

Chief Financial Officer